

# Objectively measured sedentary behaviour, physical activity and health in adolescent females

Kieran P. Dowd

#### Publication date

01-01-2012

#### Licence

This work is made available under the CC BY-NC-SA 1.0 licence and should only be used in accordance with that licence. For more information on the specific terms, consult the repository record for this item.

#### **Document Version**

1

#### Citation for this work (HarvardUL)

Dowd, K.P. (2012) 'Objectively measured sedentary behaviour, physical activity and health in adolescent females', available: https://hdl.handle.net/10344/2865 [accessed 9 Sep 2022].

This work was downloaded from the University of Limerick research repository.

For more information on this work, the University of Limerick research repository or to report an issue, you can contact the repository administrators at ir@ul.ie. If you feel that this work breaches copyright, please provide details and we will remove access to the work immediately while we investigate your claim.



## **UNIVERSITY** of LIMERICK

OLLSCOIL LUIMNIGH

# Objectively Measured Sedentary Behaviour, Physical Activity and Health in Adolescent Females

### Kieran Dowd

A thesis submitted to the University of Limerick in fulfilment of the requirement for the degree of Doctor of Philosophy

University of Limerick

Supervised by: Professor Alan Donnelly

Submitted to the University of Limerick, November 2012

#### Abstract

Title: Objectively Measured Physical Activity, Sedentary Behaviours and Health in Adolescent Females

#### Author: Kieran Dowd

The primary purpose of this thesis was to determine the associations between low intensity physical activities and indices of health in a particularly inactive population, adolescent females. To achieve this, methodologies which enabled the accurate, reliable and objective measurement of low intensity physical activities were examined and developed.

The inclinometer-based activPAL activity monitor was selected as the objective measure of physical activity due to its ability to distinguish between sitting/lying and standing. The validity of the activPAL as a measure of free-living physical activity was investigated, and a count to activity threshold was developed to quantify moderate to vigorous physical activity. Protocols which allowed the examination of both sedentary and physical activity variables were developed and implemented based on activity monitoring best practice.

Three sets of cross-sectional data were pooled to examine whether relations existed between low intensity physical activity variables and body composition in Irish adolescent females (n=195). No associations were evident between body composition and sedentary variables. However, negative associations were observed between light intensity physical activity and body composition. Furthermore, in this population, light intensity physical activity appears to protect adolescent females against sedentary time to a greater extent than moderate to vigorous physical activity. The associations between low intensity physical activity variables and more powerful indices of cardiovascular health, including blood lipids, carotid intima-media thickness and arterial stiffness, were also examined in a smaller sample of adolescent females (n=64). Similarly, no significant relationships were evident between sedentary variables and cardio-metabolic risk factors, while light intensity physical activity was negatively associated with body mass index.

In summary, the activPAL has been identified as a valid measure of physical activity in an adolescent female population. The novel examination of light intensity physical activity, which distinguishes between sitting and standing activities, has been described and presented. Negative associations have been observed between activPAL determined light intensity physical activity and body composition in this population. To conclude, the evidence presented in this thesis suggests that a worthwhile public health initiative may be to encourage Irish adolescent females to increase the amount of time spent in light intensity physical activity by reducing the amount of time spent in sedentary behaviours.

#### **Authors Declaration**

I hereby declare that the work contained within this thesis is my own work other than the counsel of my supervisor, Professor Alan Donnelly of the Department of Physical Education and Sport Sciences, University of Limerick and of collaborative work completed with Dr. Deirdre Harrington, Pennington Biomedical Research Center, Baton Rouge, LA, USA. Forty four sets of data, which have been analysed in Chapter 4 and Chapter 5, were shared data which was collected in collaborative research between Dr. Deirdre Harrington and I during her time as a Ph.D. research student within the Department of Physical Education and Sport Sciences, University of Limerick. All shared data has been processed and analysed using updated techniques and all data has been presented in a different manner to that which has previously been submitted for academic award. This work has not been submitted for any academic award, at this or any other educational establishment. Where the use has been made of the work of other people, it has been fully acknowledged and referenced (Bibliography).

Kieran Dowd, September 2012

#### Acknowledgements

To my supervisor, Professor Alan Donnelly, thank you for everything you have done for me over the last four years. Your patience, leadership, encouragement and support were greatly appreciated throughout this time.

To Dr. Deirdre Harrington, who has been an important colleague, mentor and friend over the past four years, thank you. You have always encouraged and supported me, and I hope our collaboration can continue over the years to come.

To the Irish Research Council of Science, Engineering and Technology (IRCSET), thank you for giving me this opportunity through your scholarship funding. This would not have been possible without it.

To all the staff in the Department of Physical Education and Sport Sciences, thank you for all your assistance over the last four years.

To all the postgraduate students in the Department of Physical Education and Sport Sciences, both past and present, who have made the last four years so memorable and enjoyable. Thank you. A special thanks to Will McCormack, Niamh Ní Chealleachair, Dr. Siobhan Leahy, Dr. Sarah Breen and Dr. Joseph Costello who have helped me throughout my research.

To the technical staff, Steve Clothier and Dr. D.J. Collins, thank you for the hours you both spent assisting with all my technical needs. There were many!

To the principals, physical education teachers and co-operating teachers in the participating schools, thank you for all of your help and assistance.

To all the students who took part in this research. I thoroughly enjoyed working with you all, and will be forever grateful for your help.

To Dr. Alan Bourke and Dr. Phil Kearney, your programs saved me months, maybe years of data processing. Sincere thanks for your patience and help.

To Dr. Helen Purtill, you have guided me through all my statistical nightmares. Thank you.

A special thanks to Phelim Macken and Kathleen Kennedy from the County Limerick Local Sports Partnership, who were key collaborators throughout my research. I will never be able to thank you enough. We had some laugh along the way!

To all my friends, who never understood what I was doing "back in school again", I'm finished. Thanks for making me laugh when I needed it. A special thanks to Angela. You have put up with a lot over the last four years. Thank you for everything.

To my nieces, Lauren and Erin, that have "been neglected" by their uncle over the past few months due to the completion of this thesis. I can bring you for that day out now!

Above all else, thank you to my Mum and family, Michael, Patricia and John. It's been tough, but your constant strength, support and encouragement have always helped me through. Your help over the last four years and beyond will never be forgotten.

Table of (	Contents
------------	----------

Abstract	i			
Authors Declaration	ii			
Acknowledgements	111			
List of Figures	X11			
List of Tables	xiii			
Chapter 1. Introduction	1			
1. 1. Background:	2			
1. 1. 1. The Importance of Physical Activity Type and Intensity	2			
1. 1. 2. Adolescence as a Study Population	2			
1. 1. 3. The Assessment of Subcomponents of Physical Activity	3			
1. 2. Thesis Aims:	4			
1. 3. Thesis Objectives:	4			
1. 4. Thesis Structure:	5			
Chapter 2. Literature Review				
2. 1. Introduction	8			
2. 2. Section A: Physical Activity	8			
2. 2. 1. Early Scientific Observations	9			
2. 2. 2. The Epidemiological Evidence in Adults	9			
2. 2. 3. Non-Exercise Activity Thermogenesis (NEAT)	11			
2. 3. The Physical Inactivity Paradigm: A Modern Phenomenon	12			
2. 3. 1. Sedentariness: A Distinct Activity Behaviour	13			
2. 3. 2. Studies Underpinning the Physical Inactivity Paradigm	14			
2. 3. 3. Sedentariness: The Epidemiological Evidence in Adults	15			
2. 4. Physical Activity and Health in Childhood and Adolescence	16			
2. 4. 1. Physical Activity and Indices of Health in Childhood and Adolescence	17			
2. 4. 1. 1. Physical Activity and Blood Lipids	17			
2. 4. 1. 1. 1 . Blood Lipids	18			
2. 4. 1. 1. 2 . Pathology of Atherosclerosis	19			
2. 4. 1. 1. 3 . Blood Lipids and Cardiovascular Risk in Childhood and				
Adolescence	19			
2. 4. 1. 1. 4 . Physical Activity and Blood Lipids	20			

2. 4. 1. 2. Physical Activity and Blood Pressure	. 21
2. 4. 1. 3. Physical Activity and Adiposity	. 22
2. 4. 2. Associations between Physical Activity in Childhood and Adolescence and	
Indices of Health in Adulthood	. 23
2. 4. 3. Tracking of Physical Activity from Childhood and Adolescence into	
Adulthood	. 25
2. 4. 4. Sedentary Behaviour and Health in Childhood and Adolescence	. 26
2. 5. Current Physical Activity Recommendations	. 28
2. 5. 1. Dose-Response Relationship	. 28
2. 5. 2. Intensity	. 29
2. 5. 3. Duration	. 30
2. 5. 4. Accumulation:	. 30
2. 6. Summary: Section A	. 32
2. 7. Section B: Physical Activity Assessment in Childhood and Adolescence	. 32
2. 7. 1. Calorimetry	. 33
2. 7. 1. 1. Indirect Calorimetry	. 34
2. 7. 1. 2. Doubly labelled water	. 35
2. 7. 1. 3. Behavioural Observation	. 37
2. 7. 1. 4. Reliability and validity:	. 37
2. 7. 1. 5. Practicality	. 38
2. 7. 2. Self-Report	. 39
2. 7. 2. 1. Validity and reliability:	. 39
2. 7. 3. Heart Rate Monitoring	. 41
2. 7. 4. Motion Sensors	. 42
2. 7. 4. 1. Pedometers	. 42
2. 7. 4. 2. Accelerometers	. 44
2. 8. Measuring Physical Activity and Sedentary Behaviours using Accelerometry	. 46
2. 8. 1. Calibration and Validation of Accelerometers	. 46
2. 8. 1. 1. Unit Calibration	. 46
2. 8. 1. 2. Value Calibration	. 47
2. 9. The Measurement of Sedentary Patterns and Behaviours	. 48
2. 9. 1. Self-Reported TV Viewing Time and Sedentary Time	. 48
2. 9. 2. Accelerometer Determined Sedentary Time	. 49
2. 10. The activPAL <sup>TM</sup> Professional Physical Activity Monitor	. 50

2. 11. Summary: Section B	. 51
2. 12. Section C: Levels of Physical Activity and Sedentary Behaviour in Childhood and	
Adolescence	. 52
2. 12. 1. National Levels and Trends of Childhood and Adolescent Physical Activity	
and Sedentary Behaviour	. 52
2. 12. 1. 1. Irish Physical Activity and Sedentary Levels Examined using	
Subjective Measures	. 53
2. 12. 1. 2. Irish Physical Activity and Sedentary Levels Examined using	
Objective Measures	. 56
2. 12. 2. International Levels and Trends of Childhood and Adolescent Physical	
Activity and Sedentary Behaviour	. 57
2. 12. 2. 1. International Levels of Self-Reported Physical Activity and Sedentary	
Behaviours	. 58
2. 12. 2. 2. International Levels of Objectively Determined Physical Activity and	
Sedentary Behaviours	. 59
2. 13. Summary: Section C	. 62
2. 14. General Summary	.63
Chapter 3. Criterion and Concurrent Validity of the activPAL <sup>TM</sup> Professional Physical	
Activity Monitor in Adolescent Females.	. 65
3. 1. Introduction:	. 66
3. 1. 1. Prevalence and Effect of Moderate to Vigorous Physical Activity	.66
3. 1. 2. Methods of Assessing Physical Activity	.66
3. 1. 3. The Sedentary Behaviour Paradigm	.67
3. 1. 4. Methods of Assessing Sedentary Behaviour	.67
3. 2. Methods	. 69
3. 2. 1. Participants	. 69
3. 2. 2. Physical Activity Measurement Devices	. 69
3. 2. 3. Metabolic Unit:	.70
3. 2. 4. Testing Protocol	.70
3. 2. 5. Calibration Activities	.72
3. 2. 6. Data Processing	.73
3. 2. 7. Statistical Analysis	.74
3. 3. Results	.76
3. 3. 1. Descriptive Information	.76

3. 3. 2. Criterion Validity of the activPAL Activity Monitor	. 76
3. 3. 3. Mixed Regression Analysis	. 77
3. 3. 4. ROC Analysis	. 78
3. 3. 5. Cross-Validation of Developed Thresholds	. 78
3. 3. 6. Concurrent Validity	. 79
3. 3. 7. activPAL and ActiGraph GT3X Sitting/Standing/Slow Walking Vs. Direct	
Observation	. 80
3. 4. Discussion	. 81
3. 4. 1. The Identification of the activPAL as a Valid Measure of Physical Activity	. 82
3. 4. 2. The use of both Mixed Regression Analysis and Receiver Operating	
Statistics	. 83
3. 4. 3. The Concurrent Validity of the activPAL Activity Monitor	. 84
3. 4. 4. Differences in activPAL and ActiGraph Determined Sitting and Standing	
Time	. 84
3. 4. 5. The Advantages of using an Inclinometer-Based Activity Monitor as a	
Measure of Physical Activity	. 85
3. 4. 6. Strengths and Limitations	. 86
3. 5. Conclusion:	. 87
Chapter 4. The Measurement of Sedentary Patterns and Behaviours using the activPAL <sup>TM</sup>	
Professional Physical Activity Monitor.	. 89
4. 1. Introduction	. 90
4. 1. 1. Observed Associations between Sedentary Behaviours and Health	. 90
4. 1. 2. The Objective Assessment of Sedentary Behaviours	. 90
4. 1. 3. Fundamental Limitations of Assessment of Sedentary Behaviours using an	
Accelerometer-Based Activity Monitor	. 91
4. 1. 4. The Use of Inclinometer-Based Activity Monitors	. 92
4. 1. 5. Existing Examinations of Inclinometer-Determined Sedentary Time	. 92
4. 2. Methods	. 93
4. 2. 1. The activPAL Professional Physical Activity Monitor	. 93
4. 2. 2. Data from activPAL Output	. 94
4. 2. 3. Examination of Sedentary Periods	. 94
4. 2. 4. Examination of Sedentary Variables	. 95
4. 2. 5. Study Outline and Subject Selection	. 96

4. 3. Results	97
4. 3. 1. Descriptive Information	97
4. 3. 2. Sitting/Lying, Standing and Stepping Output	98
4. 3. 3. Bed Hours, Non-Bed Hours and Percentage of Waking Day Spent in	
Different Sedentary Categories.	98
4. 3. 4. Sedentary Bouts, Sedentary Breaks and Mean Time Spent Daily in Specific	
Sedentary Categories	99
4. 3. 5. Practical Application: A Comparison of Sedentary Patterns and Behaviours	
during the Hour Immediately Before and Immediately After School	101
4. 4. Discussion	102
4. 4. 1. The Examination of Sedentary Time	103
4. 4. 2. The Frequency of Sedentary Periods	104
4. 4. 3. The Importance of a Direct Measure of Sedentary Time	104
4. 4. 4. The Identification of Sedentary Time	105
4. 4. 5. Summary	106
4. 5. Conclusion	106
Chapter 5. Associations between Sedentary Time, Light Intensity Physical Activity and	
Body Composition in Adolescent Females.	109
5. 1. Introduction:	110
5. 1. 1. The Significance of Excess Adiposity	110
5. 1. 2. Associations between Light Intensity Physical Activity and Adiposity	110
5. 1. 3. The Sedentary Behaviour Paradigm and Adiposity	111
5. 1. 4. The Measurement of Sedentary Time and Light Intensity Physical Activity	112
5. 2. Methods:	113
5. 2. 1. Activity Measurement:	114
5. 2. 2. Measurement Protocol:	114
5. 2. 3. Data Processing:	115
5. 2. 3. 1. Identification of Waking Hours:	115
5. 2. 3. 2. Identification of Non-Wear Time:	116
5. 2. 4. Activity Variables:	116
5. 2. 5. Statistical Analysis:	117
5. 3. Results:	119
5. 3. 1. Partial Correlations between Activity Variables and Measures of Body	
Composition and between Activity Variables and Themselves	120

5. 3. 2. Partial Correlations between Activity Variables and Themselves	121
5. 3. 3. Linear Regression Analysis	121
5. 3. 4. Binary Logistic Regression Results	122
5. 4. Discussion:	131
5. 4. 1. Association between Sedentary Time and Sum of Skinfold Thickness	132
5. 4. 2. Association between Light Intensity Physical Activity, Body Mass Index and	
Sum of Skinfold Thickness	133
5. 4. 3. Strengths and Limitations:	135
5. 5. Conclusions:	135
Chapter 6. Objectively Determined Light Intensity Physical Activity is Negatively	
Associated with Skinfold Thickness and Carotid Intima-Media Thickness in Adolescent	
Females	137
6. 1. Introduction	138
6. 1. 1. Cardio-Metabolic Risk Factors in Childhood and Adolescence	138
6. 1. 2. Associations between Moderate to Vigorous Physical Activity and Cardio-	
Metabolic Risk Factors	138
6. 1. 3. Light Intensity Physical Activity and Cardio-Metabolic Risk Factors	139
6. 1. 4. Sedentary Behaviours and Cardio-Metabolic Risk Factors	139
6. 1. 5. The Assessment of Free-Living Sedentary Time and Light Intensity Physical	
Activity	140
6. 2. Methods	141
6. 2. 1. Physical Examination	142
6. 2. 1. 1. Body Composition Measures	142
6. 2. 1. 2. Carotid Intima-Media Thickness Measures	142
6. 2. 1. 3. Large Artery Stiffness	143
6. 2. 1. 4. Stage of Pubertal Development	144
6. 2. 1. 5. Blood Pressure	144
6. 2. 1. 6. Blood Sampling and Analysis	144
6. 2. 2. Clustered Metabolic Risk Score	145
6. 2. 3. Physical Activity and Sedentary Levels and Patterns	146
6. 2. 3. 1. Data Processing	146
6. 2. 4. Statistical Analysis	147
6. 3. Results:	148
6. 3. 1. Descriptive Statistics	148

6. 3. 2. Partial Correlations between Activity Variables and Cardiovascular Risk	
Factors	150
6. 3. 3. Partial Correlations between Activity Variables	150
6. 3. 4. Results of Multiple Linear Regression between Activity Variables and	
Cardio-Metabolic Risk Factors	152
6. 3. 4. 1. Sedentary Time and Cardio-Metabolic Risk Factors	152
6. 3. 4. 2. Light Intensity Physical Activity and Cardio-Metabolic Risk Factors	153
6. 4. Discussion	156
6. 4. 1. Associations between activPAL Determined Light Intensity Physical Activity	
and Cardio-Metabolic Risk Factors	156
6. 4. 2. Potential Areas of Further Research: Associations between Activity	
Variables and Cardio-Metabolic Risk Factors Approaching Significance	.158
6. 4. 3. Associations between Activity Variables Assessed using the activPAL	
Activity Monitor	.159
6. 4. 4. Strengths and Limitations:	.161
6. 5. Conclusions	.161
Chapter 7. General Discussion	.163
7. 1. Introduction	.164
7. 2. The Importance of examining all Intensities of the Activity Intensity Spectrum	.165
7. 2. 1. Limitations of Existing Measures of Sedentary Behaviours	.165
7. 2. 2. Inclinometer-Based Activity Monitors as a Measure of Physical Activity	.166
7. 2. 3. Limitations of the activPAL Activity Monitor	.167
7. 3. The Examination of Light Intensity Physical Activities	.168
7. 4. The Associations between Subcomponents of Habitual Physical Activity and	
Indices of Health in an Adolescent Female Population	.169
7. 4. 1. Sedentary Time and Cardio-Metabolic Risk Factors	.170
7. 4. 2. Light Intensity Physical Activity and Cardio-Metabolic Risk Factors	.171
7. 5. Potential Translation of Research Findings	.172
7. 6. Limitations	.172
7. 7. Future Research	.174
7. 8. Conclusion	.176
Bibliography	.177
Appendices	.211

## List of Figures

Figure 2.1 A comparison of exercise activity thermogenesis and non-exercise activity
thermogenesis
Figure 2.2 The relationship between level of physical activity or exercise capacity and
Coronary Heart Disease Mortality
Figure 2.3: The elimination rates of $^2\mathrm{H}_2\mathrm{O}$ and $\mathrm{H}_2^{\ 18}\mathrm{O}$ over a 14 day period36
Figure 2.4. The number of days participants achieved the recommended amount (>60
minutes) of physical activity55
Figure 3.1 Relationship between the activPAL activity monitor count function and the
ActiGraph GT3X accelerometer across all activities80
Figure 3.2 ActiGraph counts min <sup>-1</sup> directly observed sitting, standing and slow walking
compared with the $\leq 100$ counts min <sup>-1</sup> sedentary threshold81
Figure 4.1 Mean number of sedentary periods per day and mean total amount of time
spent in each sedentary period per day 100
Figure 4.2 Patterns of sedentary bout categories during the hour immediately before
school (8 – 9 am) and the hour immediately after school (3 – 4 pm) 101
Figure 4.3 Sedentary periods of a shorter (< 20 minutes) and longer (> 20 minutes)
duration before and after school
Figure 5.1 Estimated marginal means of high body mass index percentiles stratified by
quartiles of light intensity physical activity
Figure 5.2 Scatterplot of the percentage time spent in light intensity physical activity
and body mass index percentiles
Figure 5.3 Estimated marginal means of sum of skinfolds stratified by quartiles of
sedentary time
Figure 5.4 Scatterplot of the percentage time spent sedentary and the sum of skinfold
thickness
Figure 5.5 Estimated marginal means of sum of skinfolds stratified by quartiles of light
intensity physical activity (excl. standing)
Figure 5.6 Scatterplot of the percentage time spent in light intensity physical activity
(excl. standing) and the sum of skinfold thickness
Figure 5.7 Scatterplot of the percentage of time spent in prolonged sedentary periods
and the sum of four skinfold

### List of Tables

Table 2.1 Blood Lipid Reference Values in Adolescent Females	18
Table 2.2. The percentage of both males and females (all) and females only that	
reported participating in exercise four or more times per week	53
Table 2.3. The percentage of males and females reporting participation in exercise less	
than once a week	54
Table 2.4. Percentage of children and adolescents achieving 60 minutes or more of	
moderate to vigorous physical activity on all days of the week	56
Table 2.5. Percentage of children and adolescents that achieved 60 minutes or more of	
moderate to vigorous physical activity daily	58
Table 2.6. Percentage of children and adolescents spending 2 hours or more watching	
TV daily.	58
Table 2.7. Percentage of males and females that achieved the recommended daily	
amount of daily physical activity on 0 days, 5 days and on all 7 days across grades	59
Table 2.8. Mean total minutes per day in moderate physical activity (MPA) to vigorous	
physical activity (VPA) and mean total minutes per day in MPA and VPA in bouts of	
10 minutes or more. MPA and VPA were examined using age-specific count to	
activity thresholds (Freedson et al., 2005).	60
Table 2.9. Mean hours spent in sedentary behaviours for all participants, males and	
females across the ages of 6 - 19 years	61
Table 3.1 Mean ( $\pm$ SDs) of VO <sub>2</sub> ml·kg <sup>-1</sup> min <sup>-1</sup> , MET scores, speed and both activPAL	
and ActiGraph activity counts	76
Table 3.2 Concordance correlation results comparing actual MET values with	
activPAL counts 15 s <sup>-1</sup> based predicted MET values for all activities, non-locomotor	
activities and locomotor activities in the cross-validation group	77
Table 3.3 Cross-validation results for the activPAL for sensitivity and specificity values	
for activity intensity thresholds developed using both mixed regression analysis and	
receiver operating characteristic (ROC) analysis.	79
Table 3.4 Comparison of activPAL and ActiGraph determined sitting, standing and	
stepping with observed activity category	81
Table 4.1 Mean, standard deviation and range for subject's age, height, weight and	
body mass index.	98

Table 4.2 Mean ( $\pm$ SD) and range of hours spent sedentary, standing and stepping per
day and number of steps achieved per day98
Table 4.3 The percentage of non-bed hours and percentage of waking sedentary hours
spent in sedentary categories99
Table 4.4 Summary of sedentary periods and amount of waking time spent in each
specific sedentary period for each sedentary duration presented as Mean (Standard
Deviation)
Table 5.1 Descriptive characteristics of the population. Values presented as Mean
(±SD) and Range
Table 5.2 The distribution of the number of valid days of accelerometry data (No. of
participants and percentage of participants)
Table 5.3 Partial correlations between physical activity and sedentary variables, body
mass index and sum of skinfolds, controlled for age
Table 5.4 Partial correlations between physical activity and sedentary variables,
controlled for both age and body composition ( $\Sigma$ Skinfolds)
Table 5.5 Relationship between sedentary time, standing time, light intensity physical
activity (both including and excluding standing time) and prolonged sedentary periods
and body composition measures in adolescent females
Table 5.6 Odds ratios for high body mass index across quartiles of sedentary time,
standing time, light intensity physical activity and prolonged sedentary periods125
Table 5.7 Odds ratios for high sum of skinfold thickness across quartiles of sedentary
time, standing time, light intensity physical activity and prolonged sedentary periods 127
Table 5.8 Cross tabulation table comparing Prolonged Sedentary Periods by quartile
with low and high risk sum of skinfold values
Table 6.1 Descriptive characteristics of participants. Values are presented as mean
(range)
Table 6.2 Partial correlations between sedentary time, standing time, light intensity
physical activity (incl. and excl. standing time), moderate to vigorous intensity physical
activity, prolonged sedentary periods and cardio-metabolic risk factors, corrected for
age
Table 6.3 Partial correlations between sedentary variables and physical activity
variables, corrected for age152
Table 6.4 Associations between sedentary time and cardio-metabolic risk factors in
adolescent females

Table	e 6.5	Associations	between	light	intensity	physical	activity	(excluding	standing	
time)	and	cardio-metabo	olic risk fa	ctors	in adolese	cent fema	les			155

Chapter 1. Introduction

#### 1. 1. Background:

#### 1. 1. 1. The Importance of Physical Activity Type and Intensity

The health benefits of moderate to vigorous physical activity are now firmly established (Paffenbarger Jr et al., 1986, Blair et al., 1989b, Lee et al., 2012). Increased levels of moderate to vigorous physical activity have positive effects on human health at all stages of development, while moderate to vigorous physical activity guidelines are widely incorporated into national and international recommendations (Physical Activity Guidelines Advisory Committee, 2008). However, a large proportion of the population achieve little daily moderate to vigorous physical activity (Troiano et al., 2008, Colley et al., 2011a), while activities at the lower end of the activity intensity continuum, such as sitting/lying time and light intensity physical activity, account for the majority of an individual's active time (Matthews et al., 2008, Colley et al., 2011a, Matthews et al., 2012). It is therefore unlikely that moderate to vigorous physical activity is the dominant determinant of variability in daily energy expenditure (Donahoo et al., 2004, Levine et al., 2006, Tremblay et al., 2007). It is now postulated that activity intensities at the lower end of the continuum are more likely the dominant determinant of variability in daily energy expenditure rather than moderate to vigorous physical activity (Donahoo et al., 2004, Levine et al., 2006). Furthermore, physiological evidence on the damaging effects of excess sitting time have encouraged the promotion of reduced sitting time as an added protective mechanism against non-communicable diseases, including obesity, type 2 diabetes and site specific cancers (Hamilton et al., 2007, Hamilton et al., 2008, Owen et al., 2010, Tremblay et al., 2010). Consequently, it has been proposed that the promotion of moderate to vigorous physical activity alone may not be sufficient, and that guidelines on the whole spectrum of activity intensities may be necessary (Owen et al., 2010).

#### 1. 1. 2. Adolescence as a Study Population

Adolescence has been identified as a particularly inactive stage of human development. The prevalence of achieving physical activity recommendations significantly decreases throughout adolescence (Riddoch *et al.*, 2004, Troiano *et al.*, 2008). A recent longitudinal

study has also highlighted that sedentary time increases throughout adolescence at the expense of light intensity physical activity (Mitchell *et al.*, 2012). Furthermore, adolescent females are significantly less active and more sedentary during this period than their male counterparts (Troiano *et al.*, 2008, Mitchell *et al.*, 2012). This has a public health significance, as adolescence is a critical period in the development of excess adiposity (Dietz, 2004). Overweight and obese adolescents are also at increased risk of remaining overweight or obese in adulthood (Must, 2003, Reilly *et al.*, 2003). Similarly, patterns and trends of physical activity are relatively stable from adolescence through to adulthood, meaning less active and more sedentary adolescents are more likely to maintain this behaviour into adulthood (Malina, 1996, Telama, 2009).

#### 1. 1. 3. The Assessment of Subcomponents of Physical Activity

Free-living sedentary behaviours have predominantly been measured using subjective measures such as self-reported time spent in screen-based activities (e.g. TV viewing). Unfortunately, the validity and reliability of self-reported screen-based measures is relatively low (Bryant et al., 2007, Lubans et al., 2011). Self-reported screen-based activities in children and adolescents also may suffer from similar biases observed in self-reported physical activity, while such measures do not examine the subcomponents of sedentary behaviour (Lubans et al., 2011). Accelerometry has also been employed as a measure of sedentary and light intensity activities, whereby sedentary time is estimated using a sedentary threshold. Although this measure of sedentary time is more accurate and reliable than self-reported screen-based activities, it also has fundamental limitations. Accelerometers measure the accelerations of an object relative to free-fall, and provide an estimate of ambulation (Esliger et al., 2005, Warren et al., 2010). Consequently, sedentary time determined using an accelerometer is a measure of the lack of ambulation (Ridgers et al., 2012), and cannot distinguish between sitting time and standing time (Hart et al., 2011, Kozey-Keadle et al., 2011). Technological developments have incorporated inclinometers within accelerometer devices, which now enable the examination of posture while also examining ambulation or non-stationary activities. With this, inclinometer-based activity monitors have been encouraged for use in studies which aim to examine the associations between sedentary behaviours and health (Bassett Jr et al., 2010, Owen et al., 2010).

This thesis presents data from 4 different research studies that took place in the midwestern region of Ireland between September 2008 and December 2011. The first of these studies aimed to validate and perform a unit calibration of an inclinometer-based activity monitor (activPAL<sup>TM</sup> Professional Physical Activity Monitor) to enable the valid and reliable measurement of physical activity and sedentary behaviours in an adolescent female population. The remaining three studies each involved the examination of indices of health and patterns and trends of free-living physical activity and sedentary behaviours in adolescent female populations.

#### 1. 2. Thesis Aims:

- 1. To develop accurate methods of examining free-living activity behaviours using a valid and reliable inclinometer-based activity monitor in an adolescent female population;
- 2. To examine the associations between free-living activity behaviours, specifically low intensity activity behaviours, and indices of health in this population

#### 1. 3. Thesis Objectives:

- To examine the criterion and concurrent validity of the activPAL<sup>TM</sup> Professional Physical Activity Monitor
- To perform a value-calibration of the activPAL for moderate and vigorous physical activity in an adolescent female population
- To develop methodologies to quantify free-living sedentary time, prolonged sedentary periods, standing time, light intensity physical activity and moderate to vigorous physical activity examined using the activPAL
- To examine and describe physical activity patterns and sedentary behaviours in a cohort of adolescent females
- To determine the relationship between activity variables of different intensity in this population

- To examine the associations between activPAL-determined activity variables and adiposity in a cohort of adolescent females
- To examine the associations between activPAL determined activity variables and cardio-metabolic risk factors in a cohort of adolescent females

#### 1. 4. Thesis Structure:

The thesis examines the methods used to assess physical activity and sedentary behaviour, while also considering the potential associations between specific activity behaviours and health. Chapter 1 of this thesis provides a brief introduction to the research topic while also outlining the primary aims of the thesis. Chapter 2 presents a review of the literature on a broad range of research areas, including the associations between activity variables and health, methods of assessment of activity variables and the national and international prevalence of physical activity and inactivity in child and adolescent populations. Chapter 3 examines the potential of the activPAL<sup>TM</sup> Professional Physical Activity Monitor as a measure of physical activity intensities in an adolescent female population. This chapter describes the validation of the activPAL activity monitor, and performs a value calibration and cross-validation of count to activity thresholds for moderate and vigorous physical activity in a group of adolescent females. Chapter 3 has been accepted for publication in the journal PLoS ONE. Chapter 4 describes, employs and discusses the methodologies used to examine and analyse free-living activity variables measured using an inclinometerbased activity monitor. This chapter is an adapted version of a paper that is currently under review in Physiological Measurement. Chapter 5 examines the relationships between activity variables at the lower end of the intensity continuum and health in a group of 194 adolescent females. Activity variables, including sedentary time, light intensity physical activity and prolonged sedentary periods, are quantified using methodologies outlined in Chapters 3 and 4 and associations of these variables with measures of adiposity are examined. Chapter 6 advances on the analysis in Chapter 5, introducing the examination of relationships between low intensity activity variables and cardio-metabolic biomarkers (blood lipids, insulin, glucose, high molecular weight adiponectin) and measures of arterial health (carotid intima-media thickness and large arterial stiffness) in a smaller cohort of adolescent females (n=64). The final chapter, Chapter 7, critically analyses and discusses the results observed and the research completed throughout this thesis. The implications of these results are also discussed in this chapter, while potential areas of future research are highlighted.

Chapter 2. Literature Review

#### 2.1. Introduction

This literature review will describe the origination of physical activity research, and provide evidence of the early epidemiological findings which are the foundation for modern physical activity and health-related research. This review will examine the relationship between physical activity and indices of health in both children and adolescents and adult populations. The review will also examine the evidence to date of the deleterious effect of sedentary patterns and behaviours on indices of health in both adults and youths, and will discuss the contribution of non-exercise activity thermogenesis to total daily energy expenditure. The relationship between activity patterns and behaviours in childhood/adolescence and adulthood will be examined and discussed. The examination of physical activity and sedentary behaviours using a range of assessment methods will be discussed. This review will focus on the use of accelerometry and on the use of inclinometer-based accelerometers when examining free-living physical activity and sedentary behaviours. Methodological issues with the assessment of health indices will be considered, and the validity, reproducibility and reliability of these measures will be examined. Additionally, the suitability of using different health measures in different populations and research settings will be discussed. Irish and international prevalence of physical activity levels and sedentary behaviours in adolescents and adults will be reviewed. Finally, this literature review will examine the effectiveness of interventions in free-living populations which aim to increasing levels of physical activity and/or decrease the amount of time spent in sedentary behaviours.

#### 2. 2. Section A: Physical Activity

Physical activity is defined as any bodily movement produced by skeletal muscles that results in the expenditure of energy (Caspersen *et al.*, 1985). Academics and scholars have highlighted the relationship between physical activity and inactivity and indices of health for centuries (Pate *et al.*, 2008). Many of the fundamental concepts and theories of modern physical activity and health-related research were advocated by Hippocrates and Galen centuries ago (U.S. Dept. of Health and Human Services, 1996). Hippocrates and Galen were among the first in Western history to highlight the importance of the balance between

energy intake and expenditure, the medicinal benefits that specific exercises may have on the human body and the effect of exercise across the lifespan (Jones, 1967, Sallis *et al.*, 2000, Hardman and Stensel, 2009). Their work has been the basis of medical training and teaching in Western civilisation over the centuries, and has been the foundation of modern preventive medicine (U.S. Dept. of Health and Human Services, 1996, Hardman and Stensel, 2009).

#### 2. 2. 1. Early Scientific Observations

Interest in the relationships between non-communicable disease and individual behaviours has increased dramatically since the 1950's (Hardman and Stensel, 2009). One of the first and potentially most significant research studies in the area of physical activity and health-related research was conducted by Dr. Jeremy Morris on London bus conductors. The findings of Morris et al. (1953) identified that active bus conductors (climbing approximately 600 stairs per working day) were at a significantly lower risk of onset of coronary heart disease when compared to their more sedentary bus driving colleagues (Morris *et al.*, 1953). A similar study conducted by Taylor and colleagues (1962) on US railroad employees highlighted lower mortality rates among the most active section men compared with the least active clerks and switchmen (Taylor *et al.*, 1962). Although these studies had limitations, such as robust measures of physical activity and not accounting for additional risk factors (family history, smoking or diet), the results highlighted the positive association between physical activity and health, and consequently have provided the rationale for the investigation of the effects of physical activity on health since (Sallis *et al.*, 2000, Hardman and Stensel, 2009).

#### 2. 2. 2. The Epidemiological Evidence in Adults

These initial observations were later supported by studies of longevity, which examined the effect that physical activity had on mortality rates in adults. The Harvard alumni study, one of the first large-scale longevity studies to examine self-reported physical activity and risk of mortality, highlighted that adults who achieved three hours of sports per week were 53%

less likely to die from all causes compared with those who achieved less than one hour of sport per week (Paffenbarger Jr *et al.*, 1986). Research by Blair and colleagues (1989), who were the first to examine the relationship between physical fitness and mortality in a large population, identified strong inverse associations between increased levels of physical fitness and all-cause mortality in both male and female adults (Blair *et al.*, 1989b). Furthermore, these researchers highlighted that physical activity and physical fitness did not only reduce risk of coronary heart disease in healthy populations, but could also reduce risk of coronary heart disease and mortality in all risk groups, including smokers and those with poor family history and risk factor status (Paffenbarger Jr *et al.*, 1986, Blair *et al.*, 1989b, Lee *et al.*, 2012).

Many factors contribute to the development of coronary heart disease, with the most recognised of these being dyslipidemia (Després *et al.*, 2000), hypertension (Kannel *et al.*, 1969, Chobanian *et al.*, 2003), physical inactivity (Paffenbarger *et al.*, 1983, Haskell, 1994, Blair *et al.*, 1996, Lee *et al.*, 2012), obesity (Hubert *et al.*, 1983, Nielsen and Jensen, 1997, Grundy, 2004), and smoking (Wang *et al.*, 1996, Yusuf *et al.*, 2004). Each of these factors significantly increases the chance of developing coronary heart disease. However, obesity and inactivity are considered particularly important, as they directly affect other risk factors, while also exerting an independent effect of their own (Powell *et al.*, 1987, U.S. Dept. of Health and Human Services, 1996).

Incontrovertible evidence now supports the hypothesis that moderate to vigorous physical activity assists in the prevention of coronary heart disease (Paffenbarger *et al.*, 1983, Ekelund *et al.*, 1988, Blair *et al.*, 1989b, Slattery *et al.*, 1989, Berlin and Colditz, 1990, Helmrich *et al.*, 1991, Sesso *et al.*, 2000, Lee *et al.*, 2012), associated comorbidities such as type II diabetes (Haffner *et al.*, 1998, Fox *et al.*, 2006, Lee *et al.*, 2012) as well as colon and breast cancer (Lee, 2003, Lee *et al.*, 2012). Physical activity induces caloric/energy expenditure, which positively affects weight loss and weight management (Bar-Or *et al.*, 1998, Moran, 1999, Dietz, 2004). It increases the transportation of glucose, and can reduce the production of insulin in the pancreas, aiding the prevention or delay of the onset of type II diabetes mellitus (Helmrich *et al.*, 1991, Manson *et al.*, 1991, Manson *et al.*, 2003, Katzmarzyk and Janssen, 2004) and improves blood lipid profiles, hence reducing the build-up of atherosclerotic plaques in the coronary arteries (Armstrong and Simons-Morton, 1994). This 'prescribed dose' of regular physical activity is essential in reducing

risk of coronary heart disease (Sisson and Katzmarzyk, 2008, Lee *et al.*, 2012), and will be discussed in more detail in Section 2. 5.

#### 2. 2. 3. Non-Exercise Activity Thermogenesis (NEAT)

Caspersen and colleagues have stated that physical activity is a complex behaviour, which can be categorised into mutually exclusive intensities, such as light, moderate and vigorous activity (Caspersen *et al.*, 1985). Researchers have focused on the examination of moderate and vigorous physical activity intensities, while activities of lower intensities of the spectrum have not been studied in great detail (Pate *et al.*, 2008). More recently, researchers have begun to place more importance on the lower intensities of physical activity, and have suggested an alternative categorisation of activity energy expenditure, whereby three components of total daily energy expenditure exist; 1) basal metabolic rate (energy expended due to ingestion, absorption and storage of food) and 3) activity thermogenesis (Levine *et al.*, 2006). Activity thermogenesis is the energy expended by an individual from all other daily activities. The sum of energy expended through basal metabolic rate, thermic effect of food and activity thermogenesis should amount to an individual's total daily energy expenditure (Levine *et al.*, 2006). It has been suggested that the variance in total daily energy expenditure can be best explained through activity thermogenesis (Levine, 2002).

Activity thermogenesis is divided into two sub-categories comprising of energy expenditure of exercise and non-exercise activity thermogenesis (NEAT; all daily activity that is non-volitional) (Levine *et al.*, 1999). Non-exercise activity thermogenesis is defined as the energy expended through physical activity other than volitional exercise (Levine *et al.*, 2006) and is predominantly the result of light intensity physical activity (Hamilton *et al.*, 2004). Evidence from epidemiological studies indicates that the majority of adolescents and adults do not participate in exercise of a moderate to vigorous intensity, with negligible exercise-related energy expenditure (Riddoch *et al.*, 2004, Healy *et al.*, 2008c, Troiano *et al.*, 2008). As a result, it has been postulated that differences in total daily energy expenditure can be explained through variances in NEAT, or low levels of light intensity physical activity, rather than through variance in energy expenditure of exercise (Ravussin *et al.*, 1986, Levine, 2002, Levine *et al.*, 2006). Figure 2.1 provides a graphical representation of the

importance of non-exercise activity thermogenesis, whereby participating in the recommended amount of moderate to vigorous physical activity produces less energy expenditure and fewer muscle contractions than normal amounts of NEAT (Hamilton *et al.*, 2007). Considering the significant effect that energy expenditure has on weight loss/gain in both adolescent and adult populations, it is essential that NEAT is examined when assessing the relationship between activity and health (Levine *et al.*, 2006).



(Hamilton et al. 2007, page 2661)

Figure 2.1 A comparison of exercise activity thermogenesis and non-exercise activity thermogenesis.

#### 2. 3. The Physical Inactivity Paradigm: A Modern Phenomenon

As highlighted in Section 2. 2. 1, early epidemiological evidence from Morris et al. (1953) and Taylor et al. (1962) contrasted the disease outcomes and mortality rates of co-workers who were sedentary for the majority of their working day with those who were exposed to relatively high levels of work related physical activity (Morris *et al.*, 1953, Taylor *et al.*, 1962). Since these initial scientific observations, researchers have focused on the effect of physical activity on health, primarily the frequency (Pate *et al.*, 1995, Physical Activity Guidelines Advisory Committee, 2008, Garber *et al.*, 2011), intensity (Bouchard *et al.*, 1993, Després *et al.*, 2000, Warburton *et al.*, 2000) and duration (DeBusk *et al.*, 1990, Duncan *et al.*, 1991, Jakicic *et al.*, 1995, Murphy *et al.*, 2000) of physical activity. However, evidence suggests that the least active (Morris *et al.*, 1953, Paffenbarger Jr *et al.*, 1986, Leon *et al.*, 1987, Lee *et al.*,

2012) and least fit (Ekelund *et al.*, 1988, Blair *et al.*, 1989b) populations are at the greatest risk of disease and mortality (Figure 2.2)(Haskell, 1994). This evidence has led to the emergence of the inactivity physiology paradigm, whereby researchers have emphasised the necessity to identify the causal links explaining why inactivity significantly raises the risk of disease and mortality (Hamilton *et al.*, 2004).



**Figure 2.2** The relationship between level of physical activity or exercise capacity and Coronary Heart Disease Mortality.

(Haskell 1994, page 653)

#### 2. 3. 1. Sedentariness: A Distinct Activity Behaviour

In population-based research, the definition of the term sedentary determines the prevalence of sedentary lifestyle, and differing definitions can provide dramatically different findings in population activity and behaviour (Caspersen *et al.*, 1985, Bennett *et al.*, 2006). The terms inactive and sedentary have previously been used interchangeably, whereby the term physically inactive or sedentary have described populations that have 1) not achieved any moderate to vigorous physical activity or 2) levels of moderate to vigorous physical activity below a specific threshold (Pate *et al.*, 1995, Bernstein *et al.*, 1999, Booth and Chakravarthy, 2002, Varo *et al.*, 2003, Tremblay *et al.*, 2012). However, research over the

last decade (discussed in more detail in Section 2. 3. 2) has suggested that sedentariness (or the behaviour of sitting and lying) is a distinct risk factor to disease and premature mortality independent of levels of physical activity (Owen *et al.*, 2010). To differentiate between sedentary behaviours and inactivity, clear and distinct definitions are necessary. At present, physical inactivity is described as engaging in no leisure time physical activity, or an individual's/populations failure to achieve sufficient daily physical activity (Pate *et al.*, 2008). In contrast, sedentariness is described as a discrete behaviour separate to physical activity (Schofield *et al.*, 2009), and is distinct to physical inactivity (Owen *et al.*, 2010). The term sedentary refers to a specific behaviour which describes activities that do not substantially increase energy expenditure above resting level (Pate *et al.*, 2008, Tremblay *et al.*, 2012). Resting level is defined as 1.0 metabolic equivalent of task (MET) for both adults and youths, and includes a range of activities such as sleeping, sitting and lying (Ainsworth *et al.*, 2011, Tremblay *et al.*, 2012).

#### 2. 3. 2. Studies Underpinning the Physical Inactivity Paradigm

The basic premise of the inactivity physiology paradigm is that responses to sitting and lying may have a deleterious effect on specific molecular and cellular mechanisms which are important for disease-related proteins. These mechanisms have their own unique metabolic consequences that are different to those of intense exercise training (Hamilton *et al.*, 2004).

The metabolic risks associated with sedentary behaviours are reliant on the activity of the skeletal muscle enzyme lipoprotein lipase (LPL)(Hamilton *et al.*, 1998). Lipoprotein lipase is a soluble enzyme that is attached to the luminal surface of endothelial cells. The enzyme hydrolyses triglycerides into two free fatty acids and one monoacylglycerol molecule in adipose tissue, while LPL also promotes the uptake of free fatty acids, high-density lipoprotein cholesterol and complex lipoprotein remnants from the blood (Hamilton *et al.*, 2004). Insulin stimulates the activity of LPL, and consequently stimulates glucose uptake in adipose tissues.

Local contractile activity or local inactivity of large skeletal muscles has been identified as one of the major physiological variables regulating LPL (Hamilton *et al.*, 1998). The loss of contractile stimulation of large skeletal muscles may lead to the suppression of LPL activity and reduced glucose uptake (Bey and Hamilton, 2003, Hamilton *et al.*, 2004). Additionally, strong and inverse muscle LPL activity has been documented with aging (Bey *et al.*, 2001, Hamilton *et al.*, 2001), while significant increases in LPL activity have been identified during moderate walking (Bey and Hamilton, 2003) and more vigorous activity like running (Hamilton *et al.*, 1998).

These findings are of great significance for physical activity epidemiology and healthrelated research. The suggestion that isometric contraction of antigravity muscles elicit changes to skeletal muscle LPL (Zderic and Hamilton, 2006, Hamilton *et al.*, 2007, Hamilton *et al.*, 2008) implies that activities such as standing, which would previously have been considered sedentary, should now be considered as a distinct activity behaviour (Owen *et al.*, 2010). Furthermore, these findings support the classification of sedentary behaviours as activities that do not involve the contraction of antigravity skeletal muscles (Zderic and Hamilton, 2006).

#### 2. 3. 3. Sedentariness: The Epidemiological Evidence in Adults

Research groups and international bodies have begun to examine the specific effect of sedentary patterns and behaviours on indices of health in large-scale epidemiological studies. To date, large-scale studies have identified strong associations between self-reported TV viewing time and metabolic risk factors in adult populations (Jakes *et al.*, 2003, Dunstan *et al.*, 2004, Dunstan *et al.*, 2005, Dunstan *et al.*, 2007, Healy *et al.*, 2008b, Dunstan *et al.*, 2010, Thorp *et al.*, 2010). Furthermore, it has been highlighted that when individuals achieve the recommended amount of weekly moderate to vigorous physical activity (150 minutes), there remained a significant dose response relationship between TV viewing time and women. Additional risks, including fasting plasma glucose, triglycerides and high-density lipoproteins, were evident in women (Healy *et al.*, 2008b). These initial epidemiological observations supported the hypothesis that sedentariness is a risk factor for disease and mortality, independent of physical activity.

Although TV viewing time has been identified as significantly associated with sedentary time, the objective examination of physical activity and sedentary behaviours has been

recommended, as more detailed patterns and behaviours can be observed (Bassett Jr et al., 2010, Clark et al., 2011). Large-scale epidemiology studies, which have examined the relationship between sedentary time and health using more accurate and objective measures of physical activity and sedentary time, have been conducted (Section 2. 7. 4. 2). Initial objective findings by Healy and colleagues (2007) highlighted that sedentary time was unfavourably associated with blood glucose (Healy et al., 2007). This group provided additional epidemiological support to the inactivity physiology paradigm with their findings on a relatively small sample (AusDIAB study), noting that breaks in sedentary time were beneficially associated with indices of health (Healy et al., 2008a). Results from the National Health and Nutrition Examination Survey (NHANES), have provided perhaps the most significant epidemiologic findings in the examination of sedentary behaviours due to the large sample size ( $n\approx 4750$ ). Sedentary time was detrimentally associated with triglycerides, C-reactive protein and markers of insulin resistance, while further analysis of results highlighted that breaks in sedentary time (i.e. less prolonged sedentary periods), independent of total sedentary time, was beneficially associated with waist circumference (Healy et al., 2011b).

Results from large-scale epidemiological studies support the inactivity physiology paradigm, presenting evidence on the damaging effect of sedentary time on indices of health in adults, independent of physical activity, while additionally emphasising that prolonged sedentary periods have a negative effect on metabolic risk factors (Hamilton *et al.*, 2007, Hamilton *et al.*, 2008, Owen *et al.*, 2010). However, these epidemiological findings should initially be interpreted with some caution, primarily due to issues regarding the measurement of sedentariness, which will be discussed in more detail in Section 2. 7. 2 and Section 2. 7. 4. 2.

#### 2. 4. Physical Activity and Health in Childhood and Adolescence

Increased activity thermogenesis and reduced sedentary time in childhood and adolescence has primarily been associated with physiologic and lifestyle benefits in three specific domains; 1) directly affects indices of health in youth, 2) directly affects indices of health in adulthood and 3) indirectly affects adult health status by increasing the probability of becoming more active in adulthood (Blair *et al.*, 1989a, Hallal *et al.*, 2006).

# 2. 4. 1. Physical Activity and Indices of Health in Childhood and Adolescence

There is much less scientific evidence available in children and adolescents to support the relationship between physical activity and health than that available for adults. This is predominantly due to the fact that children and adolescents to not suffer from diseases associated with physical inactivity, such as coronary heart disease, type II diabetes, hypertension and osteoporosis, as these diseases manifest in adult years (Blair *et al.*, 1989b, Hallal *et al.*, 2006, Hardman and Stensel, 2009, Andersen *et al.*, 2011). Despite this, physiological and epidemiological evidence supports the positive association between increased levels of physical activity and health in children and adolescents. The most effective method of examining the current and future effect of physical activity on the health of children and adolescents is through the examination of surrogate markers of future cardiac risk, including serum lipid profiles, blood pressure and body fat content (Rowlands, 2007b). The relationship between surrogate markers of cardiac risk and physical activity are discussed below.

#### 2. 4. 1. 1. Physical Activity and Blood Lipids

According to the Irish Central Statistics Office, cardiovascular disease is the primary cause of death in Ireland for both men and women (Central Statistics Office, 2004). A key risk factor for cardiovascular disease is dyslipidemia. Dyslipidaemia is defined as an abnormal amount of lipids and lipoproteins in the blood. The most prevalent form of dyslipidemia is hyperlipidaemia, which is an elevated amount of lipids in the blood. General population blood lipid reference values specifically for adolescent females have previously been described (Ruiz *et al.*, 2006) and are presented in Table 2.1.
	Desirable Values in adolescent females
Total Cholesterol	<5.2 mmol/L
HDL Cholesterol	> 2.0  mmol/L
LDL Cholesterol	< 3.3  mmol/L
Triglyceride	< 1.1 mmol/L

Table 2.1 Blood Lipid Reference Values in Adolescent Females

Desirable values describe the values below the 90<sup>th</sup> percentile (i.e. a HDL value of 5.2 mmol/L was defined as in the 90<sup>th</sup> percentile, therefore values below this were deemed desirable).

### 2.4.1.1.1.Blood Lipids

Cholesterol is a steroid that serves as the basis to steroid-based hormones and is not very soluble in blood. It is most commonly transported around the bloodstream while attached to chylomicrons. Different cholesterols are named for their density of protein as compared to lipids. The most commonly referred to lipoproteins are high-density lipoproteins (HDL-C), which have the most protein and least amount of cholesterol and low-density lipoproteins (LDL-C), which have lower protein levels but more cholesterol. The amount of cholesterol which is bound to different plasma protein carriers in the blood is significantly associated with an increased incidence of atherosclerosis and coronary heart disease (Antonopoulos, 2002). Low-density lipoproteins transports cholesterol from the intestinal endothelium around the body, and deposit them at specific sites, including arterials walls, throughout the body. In contrast, HDL-C transports cholesterols from these cells and deposits them in the liver, where they are partially removed from the body.

Additionally, high levels of triglycerides in the blood have also been associated with increased risk of atherosclerosis and coronary heart disease. Triglyceride is a lipid which is composed of one glycerol and three free fatty acids, and is a mechanism for storing unused energy. When required, triglycerides are broken down by the special enzyme lipoprotein lipase (LPL) to release the free fatty acids and glycerol components. Free fatty acids are an energy source for many tissues throughout the body. Glycerol is synthesised to glycogen, which is stored or used as fuel in skeletal and brain muscle or stored as a reserve in the

liver. Unused free fatty acids and glycerol in the bloodstream are catalysed and stored in adipose cells as triacylglycerol (Brouns, 2002, Silverthorn, 2012).

#### 2. 4. 1. 1. 2 . Pathology of Atherosclerosis

High Density Lipoprotein-cholesterol concentration is inversely related to the risk of atherosclerosis in the body (Assmann et al., 1996). High Density Lipoprotein-cholesterol protects the body by inhibiting oxidation of LDL-C (Silverthorn, 2012). The presence of high levels of oxidised LDL-C is particularly important in the process that leads to atherosclerosis and coronary heart disease. Low-density lipoproteins transport cholesterol to specific cell sites, including blood vessel walls. The presence of LDL-C on vessel walls is a trigger for the inflammatory process that leads to the development of atherosclerotic plaques. Essentially, LDL-C is deposited, and cholesterol accumulates beneath the endothelium cells, between the endothelium and the connective tissues, where it is oxidised. Smooth muscle cells beneath the endothelium and connective tissue are attracted towards the deposits, and fatty streaks are formed. Over time, as cholesterol accumulates within the fatty streak, fibrous scar tissue forms at the surface, while smooth muscle cells divide within the plaque build-up, causing a narrowing of the lumen of the artery, placing more pressure on the arterial walls. If this build-up is damaged, platelets stick to the damaged area, and a blood clot is formed. If the flow of blood in the vessel is stopped, a stroke or myocardial infarction is likely to occur.

# 2. 4. 1. 1. 3. Blood Lipids and Cardiovascular Risk in Childhood and Adolescence

It has been widely accepted that risk factors for coronary heart disease, including unfavourable blood lipids, are applicable to children and adolescents (Armstrong and Simons-Morton, 1994, Anding *et al.*, 1996). Early observational evidence examining the relationship between physical activity and blood lipids reported that children who participated in above normal levels of physical activity had lower levels of total cholesterol and higher levels of HDL-C when compared with less active (Armstrong and Simons-Morton, 1994). Findings from the European youth heart study, a large-scale study using objective measures of physical activity, reported significant associations between physical activity levels and both total cholesterol and triglycerides, but not HDL-C (Andersen *et al.*, 2006, Ruiz et al., 2007). Recent reviews have examined the literature relating physical activity to blood lipids (Twisk, 2001, Janssen and LeBlanc, 2010, Andersen et al., 2011). However, these reviews have reported inconsistent findings, stating that evidence from observational studies supporting the positive effect of physical activity on blood lipids are generally weak.

#### 2. 4. 1. 1. 4 . Physical Activity and Blood Lipids

The association between physical activity and blood lipids from school based randomised and non-randomised control trials have also been reviewed. These reviews have concluded that weak improvements in blood lipids were observed with increases in levels of physical activity (Janssen and LeBlanc, 2010, Andersen *et al.*, 2011). In contrast to the findings of the European youth heart study, the most commonly observed improvements across school-based interventions were triglycerides and HDL-C, with limited or no improvements observed in LDL-C or total cholesterol (Strong *et al.*, 2005, Janssen and LeBlanc, 2010, Andersen *et al.*, 2011). Of the reviewed non-school based randomised and non-randomised control trials, it was highlighted that aerobic interventions produced the greatest likelihood of blood lipid improvement. While strength and circuit training interventions were beneficial, no significant improvements were observed (Janssen and LeBlanc, 2010).

Limited results have been published linking increases in physical activity to improvements in blood lipids in children and adolescents. Andersen and colleagues have highlighted that significant improvements in blood lipids are usually observed when significant improvements in physical activity and physical fitness are observed, yet a large number of interventions were unsuccessful in achieving these improvements (Andersen *et al.*, 2011). It has also been suggested that studies that have identified significant associations to date have had the greatest power, while the majority of studies that have reported limited or no associations have tended to be quite small, and possibly underpowered (Janssen and LeBlanc, 2010). Consequently, although it is recommended that further research is required to more accurately determine the association between physical activity and blood lipids in children and adolescence, it has been proposed that a sustained amount of daily physical activity ( $\approx$ 40 minutes) has the potential to improve blood lipid profiles in this population (Andersen *et al.*, 2011).

#### 2. 4. 1. 2. Physical Activity and Blood Pressure

Blood flow through the cardiovascular system stems from ventricular contraction. When the ventricles of the heart contract, the semilunar valves open and release blood into the arteries. To accommodate the pressure, the arteries expand. Upon the relaxation of the ventricles, the semilunar valves close, preventing flow back from the arteries. The elastic arteries recoil due to the stored pressure in the elastic walls, and propel the blood forward throughout the circulatory system. The sustained driving pressure maintains continuous blood flow through the blood vessels (Silverthorn, 2012). The arterial pressure of the circulatory system is measured using blood pressure. Blood pressure is defined as the pressure exerted by blood on the walls of the blood vessels, and usually measured in the systemic arteries. During ventricular systole, pressure on the walls of blood vessels reaches an average high of 120 mmHg, while during ventricular diastole it reaches an average low of 80 mmHg. Criteria for normal blood pressure in child and adolescent populations is defined as less than the 90<sup>th</sup> percentile for age, sex and height, while hypertension (or elevated blood pressure) is defined as systolic or diastolic blood pressure persistently greater than the 95th percentile measured on three separate occasions (National High Blood Pressure Education Program, 2005).

Elevated blood pressure in childhood and adolescence increases the risk of hypertension in adulthood (Bao *et al.*, 1995, Sun *et al.*, 2007). Reviews have highlighted that intervention studies, which aimed to reduce systolic and diastolic blood pressure in obese and hypertensive youths, have reported significant reductions in systolic blood pressure (Strong *et al.*, 2005, Janssen and LeBlanc, 2010). The majority of results from intervention studies in normotensive youths have not reported any significant relationships between blood pressure and physical activity (Alpert and Wilmore, 1994, Kelley *et al.*, 2003, Rowlands, 2007b). However, it has been suggested that these insignificant differences may be due to inadequate intensity and duration of intervention in this specific age population (Kelley *et al.*, 2003, Janssen and LeBlanc, 2010). Only one observational study has examined the dose-response relationship between blood pressure and objectively measured physical activity in youth (Mark and Janssen, 2008). Findings from this study identified that a dose-response relationship exists between physical activity and blood pressure. Furthermore, this study highlighted that children and adolescents who achieved the recommended amount of

daily moderate to vigorous physical activity significantly reduced their risk of hypertension (Mark and Janssen, 2008).

This evidence suggests that increases in levels of physical activity improve blood pressure in hypertensive or obese children and adolescents, and that a dose response relation exists between levels of physical activity and blood pressure. Although the required dose or amount of physical activity necessary to reduce resting systolic and diastolic blood pressure remains unclear (Alpert and Wilmore, 1994), it is likely that increases in physical activity of a moderate to vigorous intensity over a prolonged period will have positive effects on blood pressure in children and adolescents (Kelley *et al.*, 2003).

#### 2. 4. 1. 3. Physical Activity and Adiposity

Obesity is associated with unfavourable blood lipid profiles and elevated blood pressure in childhood and adolescence (McMurray *et al.*, 1995, Dietz, 1998, Sorof and Daniels, 2002, Reich *et al.*, 2003, Weiss *et al.*, 2004). However, complications in relation to childhood overweight and obesity normally do not become apparent until a later stage of life (Must and Strauss, 1999). The predominant risk factor associated with childhood and adolescent obesity is obesity in adulthood (World Health Organisation, 2000, Rowlands, 2007b, Daniels *et al.*, 2009). This is primarily due to the tracking of obesity levels from childhood and adolescence into adulthood (Serdula *et al.*, 1993, Guo and Chumlea, 1999, Freedman *et al.*, 2005, Thompson *et al.*, 2007, Franks *et al.*, 2010a, Nakano *et al.*, 2010).

The fundamental cause of overweight and obesity is an imbalance of energy intake and energy expenditure (Reilly, 2005, Daniels *et al.*, 2009). Physical activity provides the potential to increase energy expenditure (Harrell *et al.*, 2005, Ainsworth *et al.*, 2011), and hence assist in maintaining an adequate energy balance. However, to date, limited evidence is available to support the claim that increased levels of physical activity can reduce childhood overweight and obesity. An early review, which critically appraised paediatric obesity treatment and prevention interventions, suggested that "circumstantial" evidence existed to support the role that levels of physical activity and energy expenditure play the causation of overweight and obesity (Reilly and McDowell, 2003). More recent reviews of the literature concur with this, noting that the majority of studies, which examined both

subjective and objective measure of physical activity and weight gain did not find any association of measures or that associations were small (Wareham *et al.*, 2005, Janssen and LeBlanc, 2010, Wilks *et al.*, 2011). In contrast, Strong and colleagues noted that intervention programs of a moderate intensity ( $\approx$ 30-60 minutes on 3-7 days per week) were effective in reducing total body and visceral adiposity in overweight and obese children and adolescents, but had no significant effect on populations with normal weight profiles (Strong *et al.*, 2005). Strong et al. (2005) added that limited evidence has been provided to suggest that activities of a more vigorous intensity may be more effective in reducing percentage body fat in normal weight children and adolescents. Differing inclusion and exclusion criteria across reviews makes interpretation of the collective evidence of the effect of physical activity on obesity difficult. Additionally, a complex relationship exists between physical activity, diet and obesity, whereby the contribution of energy expenditure and energy intake to obesity remains unclear (Rowlands, 2007b, Wilks *et al.*, 2011).

Regardless of difficulties in determining the causation of childhood and adolescent obesity and of the inconsistent findings on the relation between physical activity and obesity, each review discussed above has drawn similar conclusions. Although limited evidence supports the hypothesis that increased physical activity reduces childhood overweight and obesity, physical activity should continue to be promoted across the lifespan (Must, 2003, Wareham *et al.*, 2005, Wilks *et al.*, 2011).

# 2. 4. 2. Associations between Physical Activity in Childhood and Adolescence and Indices of Health in Adulthood

There is incontrovertible evidence of the associations between physical inactivity and premature mortality and comorbidities including obesity, hypertension, type II diabetes and dyslipidemia in adult populations (Powell *et al.*, 2011). However, it has often been assumed that high levels of physical activity and physical fitness in childhood and adolescence is beneficial for indices of health in adulthood (Boreham *et al.*, 2002). The Harvard Alumni Study was among the first to examine the associations between physical activity in an adolescent and young adult population and indices of health in adulthood (Twisk, 2001). Published findings highlighted that no associations were observed between physical activity

levels in adolescence and the incidence of cardiovascular disease in adulthood (Paffenbarger Jr et al., 1986).

In recent years, research has begun to examine the relationships between levels of physical activity in childhood and adolescence and indices of health in adulthood in more detail (Hallal et al., 2006). Interestingly, Hallal and colleagues have noted that the majority of the literature is in agreement with initial observation from the Harvard Alumni Study, stating that little or no positive long-term effects of adolescent physical activity were associated with risk factors for cardiovascular morbidity in adults (Hallal et al., 2006). However, research studies have identified significant associations between adolescent physical fitness, rather than physical activity, and risk factors for cardiovascular morbidity including total cholesterol, HDL-C, both systolic and diastolic blood pressure and percentage body fat in adulthood (Boreham et al., 2002, Hernelahti et al., 2004). As physical activity is significantly associated (albeit a weak to moderate association) with physical fitness (Kemper and Koppes, 2006, Martínez-Vizcaíno and Sánchez-López, 2008), it has been hypothesised that the lack of association between adolescent physical activity and cardiovascular morbidity in adulthood may be due to measurement issues (Hallal et al., 2006). Although no associations have been observed between increased levels of physical activity in childhood and adolescents and cardiovascular morbidity in adulthood, positive associations have consistently been observed between physical activity in childhood and adolescence and reduced risk of breast cancer (Okasha et al., 2003) and improved bone health in adulthood (Karlsson, 2004, Kohrt et al., 2004).

The evidence associating physical activity in childhood and adolescence and cardiovascular disease in adulthood remains unclear. Regardless, the long term protective effect of participation in physical activity in childhood and adolescents should continue to be promoted. It is plausible that increased levels of physical activity in adolescence may reduce the risk of cardiovascular disease in adulthood. The majority of findings support the argument that physical activity in childhood and adolescence increases bone health, reduces the risk of breast cancer (Hallal *et al.*, 2006) and is positively associated with physical growth and behavioural development (Warburton *et al.*, 2006).

# 2. 4. 3. Tracking of Physical Activity from Childhood and Adolescence into Adulthood

Increased levels of physical activity in adulthood are associated with reduced risk of mortality and chronic diseases (Powell *et al.*, 2011). It has been hypothesised that increasing physical activity levels in childhood and adolescence can indirectly reduce the risk of mortality and chronic disease in adulthood through increasing the likelihood that individuals will become more active in adulthood (Blair *et al.*, 1989a). This idea is known as long-term stability or tracking of physical activity (Twisk, 2001). The concept of tracking refers to an individual's tendency to maintain their relative rank within a group over a period of time (Malina, 1996, Malina, 2001a). In terms of physical activity, tracking examines the tendencies of individuals to maintain their levels of physical activity across specific age ranges, normally from childhood and/or adolescence through to adulthood (Malina, 2001b).

Detailed reviews have examined the literature on the stability of physical activity levels across childhood and adolescence and into adulthood (Malina, 2001b, Telama, 2009). Malina and colleagues (2001) highlighted that a low to moderate inter-age correlation exists between early to middle childhood ( $\approx$ 4-8 years), while limited information was available on the tracking of physical activity between middle childhood and adolescence. Furthermore, this review identified that tracking varied across adolescence depending on the duration of follow up, with shorter studies ( $\approx$  3 years) noting moderate inter-age correlations and longer studies ( $\approx$  5-6 years) noting lower inter-age correlations. However, when examined together, a consistent low to moderate tracking of physical activity was evident across adolescence (Malina, 2001b). In a more recent review of the literature, Telama and colleagues (2009) have acknowledged that tracking studies have significantly low or moderate stability across all ages (Telama, 2009). Both reviews conclude that physical activity tracks reasonably well from childhood and adolescence to adulthood (Malina, 2001b, Telama, 2009).

The stability of sedentary behaviours has also been examined across the lifespan. Biddle and colleagues concluded that sedentary behaviours tracked moderately well (regardless of measurement type) during childhood and adolescence and into adulthood, with TV viewing time producing the strongest inter-age correlations (Biddle *et al.*, 2010). Malina has also

highlighted that long-term stability of inactivity is higher than that of activity (Malina, 1996). However, it is critical to understand that although stability for a specific variable, such as inactivity, is high over time, stability refers to the relative rank within a group. This does not imply that levels of inactivity remain consistent, rather it is more likely that the measured population are becoming less/more inactive together (Twisk, 2001). It is also important to acknowledge that all forms of tracking are highly influenced by measurement error (Twisk, 2001). The majority of studies referred to by Malina, Biddle and Telama were self-reported studies, which have significant validity and reliability limitations (as discussed in more detail in Section 2. 7. 2. 1), particularly in children and young adolescents (Welk et al., 2000, Trost, 2007). This may further contribute to the low to moderate correlations that have been observed when tracking physical activity across childhood, adolescence and adulthood (Twisk, 2001, Trudeau et al., 2004). Regardless of these low to moderate findings, researchers emphasise the importance of participating in regular physical activity and reducing sedentary time during childhood and adolescence. This may be effective in increasing physical activity and reducing sedentary time in adulthood, hence improving indices of health and reducing the risk of chronic disease (Malina, 2001b, Biddle et al., 2010).

## 2. 4. 4. Sedentary Behaviour and Health in Childhood and Adolescence

Technological developments, including the dependence on motor transport and the development of labour saving devices in the home, work and school environments have increased sedentary patterns and behaviours in recent years across all ages, including children and adolescents (Salmon *et al.*, 2011). Although a relatively new area of interest, epidemiological studies have examined the effect of subjectively measured sedentary patterns and behaviours on indices of health in childhood and adolescence. A recent review, which examined the relationship between self-reported TV viewing and sedentary time and health indicators, highlighted that there is moderate evidence of an inverse longitudinal relationship between sedentary behaviour and aerobic fitness in childhood (Chinapaw *et al.*, 2011). However, publications which have identified significant relationships between self-report measures of sedentary time and indices of health should be viewed with caution (Marshall *et al.*, 2004), as TV viewing is associated with unhealthy

snacking (Utter *et al.*, 2006) and sleeping patterns (Johnson *et al.*, 2004), which may distort identified associations. Furthermore, self-reported physical activity and sedentary behaviours in childhood and adolescence have well-documented limitations, as discussed in Section 2. 7. 2.

Researchers have examined the effects of sedentary patterns and behaviours on health parameters in children and adolescents using more accurate objective measures. Martinez-Gomez and colleagues have identified that total daily sedentary time was associated with higher cardiovascular risk in two hundred and ten 13-17 year olds, adding that those with a less favourable adiposity were at a greater cardiovascular risk (Martinez-Gomez et al., 2010). Findings from the European youth heart study identified that an inverse correlation existed between objectively measured total sedentary time and metabolic risk factors in one thousand seven hundred and nine 9-10 and 15-16 year olds (Ekelund et al., 2007a). However, it should be noted that neither of these studies accounted for moderate to vigorous physical activity, which is an independent predictor of cardio-metabolic risk (Carson and Janssen, 2011). Findings from "The HELENA Study" have also highlighted significant associations between total sedentary time and low levels of cardiorespiratory fitness in 12-17 year old adolescents, while also highlighting that adolescent females that spend greater than 2/3 or their waking time sedentary are more likely to have low levels of cardiorespiratory fitness, independent of physical activity (Martinez-Gomez et al., 2011b). In contrast with these findings, research has demonstrated that although a positive association was observed between sedentary behaviour and obesity in a large cohort of 12 year olds, this relationship was not apparent when moderate to vigorous physical activity was taken into consideration (Mitchell et al., 2009). Additionally, Carson and Janssen examined the associations of total sedentary time and patterns of sedentary behaviour with cardio-metabolic risk factors in 6-19 year olds and found no association independent of moderate to vigorous physical activity, while TV viewing time and low levels of moderate to vigorous physical activity were both associated with cardio-metabolic risk (Carson and Janssen, 2011).

The contrasting results identified here are not surprising. Differences in sedentary cutpoints employed and differences in data processing, cleaning and presentation are evident across each article (Lubans *et al.*, 2011). Similar to the objective examination of physical activity, the lack of a standardised method of analysing these variables poses significant problems in the objective examination of sedentary patterns and behaviours, as comparability of research findings between studies is extremely difficult (Ward *et al.*, 2005, Rowlands, 2007a). Furthermore, the effectiveness of the use of cut-points in the examination of sedentary behaviours must be taken into consideration (discussed further in Section 2. 9. 2).

### 2. 5. Current Physical Activity Recommendations

Public health experts have produced physical activity guidelines and recommendations to educate general populations on the optimal amount of physical activity necessary to maintain and/or improve indices of health, hence enabling them to make informed decisions on their level of participation (U.S. Dept. of Health and Human Services, 1996, Department of Health and Ageing, 1999, Haskell et al., 2007, Physical Activity Guidelines Advisory Committee, 2008). There is now a consensus that the risk of chronic disease is reduced when adults participate in at least 30 minutes of moderate to vigorous physical activity on at least 5 days of the week. Furthermore, accumulating the activity in intermittent bouts as short as ten minutes was a suitable way of meeting the 30 min goal (DeBusk et al., 1990, Jakicic et al., 1995, Murphy et al., 2000, Warburton et al., 2006, Pate, 2007). For children and adolescents, it has been suggested that 60 minutes of moderate to vigorous physical activity be accumulated on at least 5 days per week for optimal health benefits (World Health Organisation, 2011, Department of Health and Children, 2012). The prescribed recommendation for frequency of physical activity is based on the reexamination of recommendations by the U.S. Dietary Advisory Committee (U.S. Department of Health and Human Services and U.S. Department of Agriculture, 2005, Department of Health and Children, 2012).

#### 2. 5. 1. Dose-Response Relationship

It is important to recognise that the guidelines outline the minimum recommended frequency, intensity and duration of activity that will reduce the risk of onset of chronic illness (Pate, 2007). A dose-response relation of intensity, duration and frequency exists, suggesting that greater health benefits can be obtained by participating in physical activity

more vigorously, more frequently and for a longer duration then suggested in guidelines (Masse *et al.*, 1999, Kohl, 2001, Haskell *et al.*, 2007, Powell *et al.*, 2011). However, it is essential that the dose-response needs to consider not only what dose induces the most health benefits, but also the risk profile for that dose, for example in people with underlying diseases or health complications (Pate, 2007).

## 2. 5. 2. Intensity

Intensity is a key component to be examined when analysing the appropriate dose of physical activity required to impact specific health and fitness outcomes (U.S. Dept. of Health and Human Services, 1996, Garber et al., 2011). Intensity is described as the required overload that is exerted on specific systems (i.e. cardiovascular) that is necessary to bring about improvements (Sisson and Katzmarzyk, 2008). Initial recommendations of physical activity indicated that vigorous activity was required to produce health benefits (American College of Sports Medicine, 2007). However, it is now accepted that most health benefits experienced at vigorous intensity physical activity are also experienced at activities of a moderate intensity (American College of Sports Medicine, 2007, Haskell et al., 2007, Garber et al., 2011). These recommendations are based on experimental exercise training studies published during the 1980's and 1990's. These studies found similar health improvements, such as increases in levels of high-density lipoprotein (HDL) and reductions in blood pressure between individuals who participated in moderate physical activity and vigorous physical activity (Duncan et al., 1991, Bouchard et al., 1993, Després et al., 2000). As identified above, a dose-response relationship exists between physical activity and health benefits (Powell et al., 2011). This relationship applies to intensity also, highlighting that health improvements occur at low levels of exercise intensity or total energy expenditure, yet a higher intensity of physical activity produces greater improvements to health (U.S. Dept. of Health and Human Services, 1996, Lee et al., 2000, Kohl, 2001).

#### 2. 5. 3. Duration

Physical activity dose is a combination of frequency, intensity and duration of activity. Duration must be viewed with intensity, as the volume of energy expended throughout the day is essential when comparing physical activity and health benefits (American College of Sports Medicine, 2007). The majority of physical activity recommendations identify 30 minutes of moderate to vigorous physical activity as the minimum duration of daily activity necessary for adults, while 60 minutes of moderate to vigorous activity is recommended children and adolescents (U.S. Dept. of Health and Human Services, 1996, Department of Health and Ageing, 1999, Haskell et al., 2007, Physical Activity Guidelines Advisory Committee, 2008, Department of Health and Children, 2012). These recommendations have been based on findings from large-scale epidemiological research (Morris et al., 1953, Paffenbarger Jr et al., 1986, Powell et al., 1987). It is proposed that regularly participating in 30 minutes of moderate to vigorous physical activity (60 minutes for children and adolescents) helps increase caloric/energy expenditure, which positively affects weight loss (Jakicic et al., 1995). It is also suggested that participating in the recommended amount of activity reduces risk of diseases such as osteoporosis, type II diabetes and hypertension (U.S. Department of Health and Human Services, 2008, Powell et al., 2011).

#### 2. 5. 4. Accumulation:

Accumulation of physical activity is the concept of achieving one's daily recommended activity using multiple short bouts of a specific intensity throughout the day (i.e. moderate to vigorous physical activity). Accumulation of physical activity was first included in public health guidelines in 1995 (Pate *et al.*, 1995). The guidelines state the prescribed 30 minutes of daily physical activity can be accumulated in short bouts throughout the day. These guidelines were developed from perspective observational studies relating amounts of activity to coronary heart disease, along with results of experimental studies that examined the effect of a single, longer bout of activity versus three shorter bouts of activity on health benefits (Paffenbarger Jr *et al.*, 1986, Leon *et al.*, 1987, DeBusk *et al.*, 1990). Since the concept of accumulation was introduced in public health guidelines, research reported that similar health benefits achieved by completing 30 minutes of consecutive moderate to

vigorous physical activity can also be achieved by accumulating 3 ten minute bouts of activity (Jakicic et al., 1995, Murphy and Hardman, 1998, Murphy et al., 2000). Murphy and colleagues (2000) highlighted the similarity in levels of triglyceride concentration after two separate exercise interventions, 1 continuous 30 minute bout of moderate physical activity versus 3 intermittent 10 minute bouts of moderate physical activity (Murphy et al., 2000). Jakicic et al. (1995) noted similar increases in peak oxygen uptake and weight loss between participants completing an intervention involving 1 continuous bout of moderate activity versus participants completing 3 intermittent bouts of moderate activity (Jakicic et al., 1995). It is now widely accepted that the accumulation of shorter bouts of activity lasting 10 minutes shows similar benefits when compared with one longer bout, and these guidelines continue to be an important aspect in physical activity recommendations (Physical Activity Guidelines Advisory Committee, 2008, Department of Health and Children, 2012). The promotion of intermittent bouts of activity is essential in encouraging participation in physical activity, as poor adherence to planned physical activity programs due to perceived lack of available time is a key contributing cause to physical inactivity (Marcus et al., 1996, Murphy et al., 2002). Inactive individuals are more likely to perform multiple short bouts of activity throughout the day, as 10 minute bouts of free time are more available in modern, hectic lifestyle (Marcus et al., 1996, Murphy et al., 2002).

It should be acknowledged that researchers have continued to examine the relationship between the length of physical activity bouts and health indices. Miyashita *et al.* (2006) identified similar reductions in postprandial triacylglycerol concentrations when ten threeminute bouts of moderate to vigorous physical activity were compared to 1 continuous 30 minute bout, further questioning whether the duration of activity recommended in current guidelines is accurate or effective (Miyashita *et al.*, 2006). Other studies have highlighted that lower levels of physical activity, as low as half the current recommendations, may be sufficient to induce health benefits (Warburton *et al.*, 2006). Further studies examining the effect of shorter physical activity bouts (< 10 mins) on daily accumulated activity are required to determine if even shorter bouts of physical activity can provide significant health improvements. Nevertheless, until this research is undertaken, current guidelines on accumulated physical activity bouts should be adhered to.

#### 2. 6. Summary: Section A

Although there is evidence that increased levels of physical activity in childhood and adolescence are beneficial to indices of health (Strong *et al.*, 2005, Hallal *et al.*, 2006, Janssen and LeBlanc, 2010), it is clear from the findings presented throughout this section that further research on the effects of physical activity and sedentary behaviours on indices of health is necessary. A significant limitation of existing research may be the lack of accurate and objective measures of both physical activity and sedentary behaviours (Pate *et al.*, 2008, Owen *et al.*, 2010), which has potentially reduced the associations made between specific activity patterns and behaviours and indices of health across all ages. Until further large-scale epidemiological studies have objectively examined both levels of physical activity and sedentary behaviours with accurate health measures, current guidelines on the recommended amount of physical activity across the lifespan should be adhered to. Furthermore, the target of reducing total sedentary time and prolonged sedentary periods, as in the Canadian guidelines for sedentary behaviour (Tremblay *et al.*, 2011), should be recommended in all future physical activity guidelines (Hamilton *et al.*, 2008).

# 2.7. Section B: Physical Activity Assessment in Childhood and Adolescence

As identified in Section 2. 2. 2, physical activity is inversely associated with improvements in indices of health, whereby individuals with higher levels of physical activity are at a lower risk of disease and all-cause mortality (U.S. Dept. of Health and Human Services, 1996, Physical Activity Guidelines Advisory Committee, 2008, Garber *et al.*, 2011). However, physical activity is a complex construct and is a particularly difficult behaviour to assess (Caspersen *et al.*, 1985, Kohl *et al.*, 2000, Butte *et al.*, 2012). Almost 60 years since the publication of initial physical activity and health-related findings, researchers continue to examine human activity behaviour in an attempt to accurately identify the relationship between different activity patterns and behaviours and indices of health (Sallis *et al.*, 2000). Furthermore, the relatively new research areas of sedentary patterns and behaviours and non-exercise activity thermogenesis have added new dimensions to be considered in physical activity assessment (Levine *et al.*, 2001, Bassett Jr *et al.*, 2010, Bauman *et al.*, 2011, Loprinzi and Cardinal, 2011).

The assessment of physical activity provides independent variables when examining relationships with indices of health and behavioural correlates of physical activity (Welk, 2002). The ability to make strong associations between variables of activity and health is dependent on its reliability and validity (Caspersen *et al.*, 1985, Trost, 2007, Westerterp, 2009, Loprinzi and Cardinal, 2011, Intille *et al.*, 2012). The reliability of an instrument is the consistency with which the instrument measures what it is intended to measure (Kohl *et al.*, 2000, Warren *et al.*, 2010). Reliability is a prerequisite for validity, as reliability must be established before a measure can be valid (Morrow, 2002). Validity is the degree to which an instrument can measure what it is intended to measure (Kohl *et al.*, 2002, Warren *et al.*, 2010). The reliability and validity of a measurement tool must be considered prior to being employed in specific populations, specific settings and when examining specific activity variables of physical activity (Morrow, 2002, Ward *et al.*, 2005, Trost, 2007).

Different methods of examining physical activity can be categorised into calorimetry, behavioural observation, self-report, physiological variables, and motion sensors (Westerterp, 2009). During this section, the validity, reliability and practicality of each methodology will be considered.

#### 2. 7. 1. Calorimetry

The law of thermodynamics state that energy is neither created nor destroyed, but is rather changed from one form to another. Within the human body, chemical energy (energy intake in the form of food) is used to produce mechanical energy (energy expended throughout daily living) (Starling, 2002). The predominant categories by which energy is expended in the human body have been discussed in detail in Section 2. 2. 3 (resting metabolic rate, thermic effect of food and activity thermogenesis). Three methods of accurately examining energy expenditure in humans are direct calorimetry, indirect calorimetry and doubly labelled water. Direct calorimetry measures energy expenditure directly by assessing heat production (Jequier, 1981, Brychta *et al.*, 2010). Although this is a valid and reliable measure for energy expenditure, it is extremely expensive to establish and

run, requires enormous knowledge and expertise of equipment and does not provide more accurate results than indirect measures (Levine, 2005, Brychta *et al.*, 2010). Additionally, the use of indirect calorimetry chambers are limited in physical activity and health-related research due to high costs and their inability to examine realistic free-living activities due to its artificial setting (Starling, 2002, Brychta *et al.*, 2010). For this reason, this review will focus on indirect calorimetry using open circuit calorimetry and doubly labelled water.

#### 2. 7. 1. 1. Indirect Calorimetry

Indirect calorimetry measures heat production indirectly through examining oxygen consumption (Starling, 2002). Initially, closed circuit indirect calorimetry systems that consisted of sealed respiratory gas circuits (which measured levels of expired gas over time) were the predominant method of indirect calorimetry measurement. However, the most common and easiest indirect calorimetry measures currently in use are the open circuit indirect calorimeter systems (Levine, 2005). Levine has provided a detailed description of the processes involved in open circuit indirect calorimetry methodologies (Levine, 2005). Briefly, different forms of open circuit systems exist, including ventilated open circuit systems (ventilated hood technique) and expiratory collection systems (portable metabolic systems tend to be laboratory-based systems which are fixed to mobile carts and are predominantly used for the examination of resting metabolic rate and the thermic effect of food, while expiratory collection systems are mobile metabolic systems which can be employed in both laboratory or field-based settings and are normally used to examine activity thermogenesis (Matarese, 1997, Starling, 2002, Brychta *et al.*, 2010).

When compared with similar measures (such as direct calorimetry and doubly labelled water), indirect calorimetry provides an accurate measure of energy expenditure at a lower cost using equipment that does not require specialist training (Levine, 2005). It has commonly been employed as a criterion measure for energy expenditure (Trost, 2007), and has historically been used to validate heart rate monitors (Treiber *et al.*, 1989, Emons *et al.*, 1992, Bitar *et al.*, 1996, Eston *et al.*, 1998, Troutman *et al.*, 1999), pedometers (Eston *et al.*, 1998) and motion sensors (Rowlands *et al.*, 2004, Pate *et al.*, 2006, Calabró *et al.*, 2009, Esliger *et al.*, 2011). Unfortunately, the use of indirect calorimetry in validation studies is

not without its limitations. Mobile metabolic devices (which are worn over the shoulders or on the back) alter the variable being measured through adding additional weight to the individual being measured (Brychta *et al.*, 2010). However, when validating instruments for measurement of habitual physical activity, it is critical to include activities of daily living (Bassett Jr *et al.*, 2012), and the advantages obtained through using indirect calorimetry outweigh the disadvantages. With this, indirect calorimetry has been recommended as an appropriate criterion for minute by minute energy expenditure to be used during the validation of free-living measures of physical activity (Bassett Jr *et al.*, 2012, Freedson *et al.*, 2012).

#### 2. 7. 1. 2. Doubly labelled water

Doubly labelled water has been highlighted as the most valid and reliable criterion measurement for energy expenditure (Schoeller et al., 1986, Schoeller, 1988). This method of examining energy expenditure is based on the chemical reactions of two stable isotopes of water, <sup>2</sup>H<sub>2</sub>O (deuterium labelled water) and H<sub>2</sub><sup>18</sup>O (Oxygen labelled water) (Trost, 2007). An individual is required to provide a baseline urine sample, ingest a standardised dosage of both stable isotopes, provide an additional two urine sample the morning after ingesting the dosage and return to their normal daily life (Starling, 2002). After a specified period (protocols have ranged between 12 and 21 days), two final urine samples are provided (Welk et al., 2000). The urine samples are then examined using a mass spectrometer to identify the rates of elimination of <sup>2</sup>H<sub>2</sub>O and H<sub>2</sub><sup>18</sup>O (Starling, 2002). The declining level of deuterium labelled water is a function of H<sub>2</sub>O turnover, while the decline in oxygen labelled water is a function of both H<sub>2</sub>O and CO<sub>2</sub>. The difference between the two stable isotopes is then examined over the measurement period to determine the total CO<sub>2</sub> production over this period (Goran, 1994, Shetty, 2002). A graph of the elimination of <sup>2</sup>H<sub>2</sub>O and H<sub>2</sub><sup>18</sup>O is provided in Figure 2.3. Once CO<sub>2</sub> production over the examination period has been determined, a series of calculations involving basal metabolic rate and the thermic effect of food are used to provide information of energy expended through activity thermogenesis (Starling, 2002).



Figure 2.3: The elimination rates of  ${}^{2}H_{2}O$  and  $H_{2}{}^{18}O$  over a 14 day period.

Doubly labelled water has been validated in both children and adults with indirect calorimetry and has been found to be accurate between 5 and 10% (Goran, 1994). Due to its high accuracy, doubly labelled water has been suggested as the gold standard of energy expenditure measurement (Schoeller, 1999, Trost, 2007). It is a relatively unobtrusive and non-invasive method of examining total energy expenditure in free-living youths (Welk et al., 2000, Starling, 2002, Trost, 2007). Unlike objective measures of physical activity, there is no continuous measurement over prolonged periods (Welk et al., 2000). Additionally, participants can be blinded from the experiment, as the measurement can also be used to examine body composition, and should not result in individuals reacting by increasing their levels of physical activity (Starling, 2002). Unfortunately, doubly labelled water is not applicable in large-scale studies due to the high cost of the relevant isotopes and the potential burden for participants to visit testing centres to provide urine samples (Schoeller, 1999, Kohl et al., 2000, Welk et al., 2000). Furthermore, doubly labelled water only provides information on total energy expenditure and total activity energy expenditure, and cannot provide any information on frequency, intensity or duration of activities (Kohl et al., 2000, Welk et al., 2000, Trost, 2001, Trost, 2007). However, the accuracy of doubly labelled water has resulted in its use to examine the validity of measures of physical activity such as selfreport (Philippaerts *et al.*, 1999, Conway *et al.*, 2002), accelerometry (Plasqui and Westerterp, 2007) and heart rate monitoring (Emons *et al.*, 1992, Livingstone *et al.*, 1992, Maffeis *et al.*, 1995), which may in turn be used in epidemiological studies.

#### 2. 7. 1. 3. Behavioural Observation

In a physical activity and health-related research setting, direct observation is a behavioural observation methodology which is particularly effective when examining physical activity in young children (Welk *et al.*, 2000, McKenzie, 2002). This is primarily due to measurement limitations with self-reported and accelerometer determined physical activity in these populations. Direct observation is employed to classify physical activity behaviours into specific categories which can be examined and analysed in detail (McKenzie, 2002). This methodology involves observing a child/group of children for a specific time period while recording activity levels and types of activity on a momentary time-sampling basis (Trost, 2007). Specific activities, types of activities and locations are recorded and coded in a log or handheld computer device (e.g. laptop/tablet) (Kohl *et al.*, 2000, Trost, 2007).

## 2. 7. 1. 4. Reliability and validity:

Direct observation has been identified as both a valid and reliable measure of physical activity in children (Trost, 2007). When examining the validity of direct observation protocols, Kohl and colleagues identified that 9 studies were validated with physiological measures (heart rate = 4; observation = 1; self-report = 1; activity monitor = 1; indirect calorimetry = 2) (Kohl *et al.*, 2000). Each study reported some amount of validity, while one particular study, which employed indirect calorimetry (the gold standard when validating an activity measurement device), reported particularly high levels of validity (Bailey *et al.*, 1995). The reviewed studies which employed direct observation methodologies also highlighted high levels of inter-observer reliability (Klesges, 1984, O'Hara *et al.*, 1989, Puhl *et al.*, 1990, McKenzie *et al.*, 1991a, McKenzie *et al.*, 1991b, Bailey *et al.*, 1993, Rowe *et al.*, 1997). In contrast, one study found inconsistent aerobic activity

between 2 day test-retest reliability (Baranowski *et al.*, 1987), which is not unexpected. This is due to the variance in activity patterns of children from day to day (Klesges, 1984).

#### 2.7.1.5. Practicality

Due to the labour intensive nature of direct observation, its use in large-scale epidemiological studies as a measure of activity patterns and behaviours is unrealistic (Kohl *et al.*, 2000). As evidenced in a review by McKenzie, data collection among children and adolescents is normally only possible in a small sample in specific environments (school, home, organised sports locations) for particular periods of the day (class time, break time etc.) (McKenzie, 1991, McKenzie, 2002). Observers are required to be trained in the specific protocol for data collection and to spend long periods examining the momentary changes in an individual's activity patterns and behaviours (Kohl *et al.*, 2000, McKenzie, 2002, Trost, 2007). Additionally, direct observation can be reactive, whereby the presence of an observer causes the observed to react to the variable being measured, and may result in increased levels of physical activity. This limitation can be overcome by repeated observations (Trost, 2007).

However, despite the described shortcomings, direct observation has specific advantages over other forms of physical activity assessment (McKenzie, 1991). This methodology provides researchers with rich information on the type of activity an individual participates in, the location of the activity and the environment in which the activity takes place (Welk *et al.*, 2000, Trost, 2007), which cannot be accurately gathered using other measures of physical activity. Furthermore, the development and validation of community-based behavioural observation methodologies, such as SOPARC, allow the examination of a range of physical activity variables including type of activity, activity domain, age, sex and a range of additional contextual information in large cohorts of individuals at one time (McKenzie *et al.*, 2000). This type of observation may be particularly informative for community and governmental bodies, as it provides details regarding the influences of the physical and social environments on youth physical activity (Trost, 2007).

#### 2.7.2. Self-Report

Self-report is one of the most valuable and commonly employed measurement instrument in physical activity and health-related research (Kohl *et al.*, 2000, Matthews, 2002, Trost, 2007). It is a cheap and easy method of collecting physical activity information from large cohorts in a short time period (Warren *et al.*, 2010). To date, the majority of significant epidemiological evidence associating physical activity patterns and behaviours with health indices in adults has been observed using self-reported physical activity (Paffenbarger Jr *et al.*, 1986, Wannamethee *et al.*, 1998, Gregg *et al.*, 2003). However, the relationships between physical activity and indices of health in youths using self-reported measures are less well established.

#### 2. 7. 2. 1. Validity and reliability:

Self-reported physical activity includes self-administered recall questionnaires, interview administered recall questionnaires, diaries and proxy reports (Sallis and Saelens, 2000, Trost, 2007). As with all measures of physical activity, the accuracy of the information obtained using self-report is dependent on the validity and reliability of the measurement instrument (Chinapaw *et al.*, 2010). A number of detailed reviews have highlighted levels of validity and reliability of self-reported instruments in the measurement of physical activity (Kohl *et al.*, 2000, Sallis and Saelens, 2000, Trost, 2007, Chinapaw *et al.*, 2010).

Published reviews of self-report measurement instruments in youths have acknowledged that validity varies significantly across validation methodology and measurement protocol (Kohl *et al.*, 2000, Sallis and Saelens, 2000, Trost, 2007). Sallis and Saelens highlighted that validity correlations ranged from 0.4 to 0.77 for proxy reports, 0.07 to 0.88 for self-administered questionnaires and from 0.17 to 0.72 for interviews (Sallis and Saelens, 2000). The authors highlighted that although all measures reported some form of validity, interview-administered measures appeared to consistently correlate better (2/3 of interview administered validity correlations were greater than 0.5) (Sallis and Saelens, 2000). Similar findings for validation of self-report instruments were reported by Kohl and colleagues, who added that the majority of instrument validation correlations fell in the 0.3 to 0.5 range (Kohl *et al.*, 2000). Although reviews have compared the validity of self-reported physical

activity in adult populations using the gold standard doubly labelled water (Prince *et al.*, 2008), only one study validated self-reported physical activity against doubly labelled water in children and adolescence (Craig *et al.*, 1996). A correlation of 0.47 was identified between measures, however methodological issues with this validation have been raised (comparison of 1-year recall with doubly labelled water) (Kohl *et al.*, 2000).

The comparison of the reliability of self-reported measures of physical activity from study to study is extremely difficult. This is primarily due to the nature of self-reported physical activity, whereby inaccurate cognitive processing and recall, coupled with high variability in physical activity levels of children and adolescents significantly contribute to variability between test/re-test results (Shephard, 2003, Chinapaw et al., 2010, Warren et al., 2010). A review of self-reported physical activity conducted by Sallis and Saelens reported acceptable levels of reliability for all reviewed self-reported measures, ranging from 0.6 to 0.98 (Sallis and Saelens, 2000), while a recent review by Chinapaw and colleagues identified that test/re-test correlations for self-report instruments varied from 0.02 to 0.96 (Chinapaw et al., 2010). Kohl and colleagues reviewed test-re-test reliability of self-report measures over specified time periods (ranging from 45 minutes to 8 years), and reliability coefficients ranged from 0.2 to 0.99 (Kohl et al., 2000). Upon further analysis, the reliability of measures which were examined over a shorter period produced the greatest correlations, while the reliability of measurement instruments in children younger than 10 years was limited. Furthermore, reliability generally improved with age from childhood to adolescence (Kohl et al., 2000). With the large body of self-reported physical activity literature, it is not surprising to see wide ranges of variability in reliability across self-report measurement instruments due to the variance in cognitive processes and activity levels across childhood and adolescence (Matthews, 2002).

The benefits and limitations of self-report as a measure of physical activity in epidemiological research have been discussed in great detail in a number of literature reviews (Sallis and Saelens, 2000, Shephard, 2003, Prince *et al.*, 2008, Chinapaw *et al.*, 2010). Self-reported physical activity is a cheap method of examining physical activity as it normally only requires a pencil and a self-report measurement instrument and, as a result, can be implemented in large cohorts of individuals at one time (Sallis and Saelens, 2000, Matthews, 2002, Prince *et al.*, 2008). Additionally, the information obtained is non-reactive as it normally involves recall of activities and does not alter the behaviour under investigation (Sallis and Saelens, 2000). Along with recording activity type, self-report also

has the ability to provide researchers with behavioural, environmental and contextual information of physical activity patterns and behaviours (Kohl *et al.*, 2000, Matthews, 2002, Trost, 2007, Warren *et al.*, 2010). However, levels of recall bias and misinterpretation of the term 'physical activity' are significant constraints in self-reported physical activity in children (Baranowski, 1984, Trost *et al.*, 2000). Furthermore, as discussed earlier, the majority of validation studies have reported only moderate correlations, which has led to concerns regarding the accuracy of self-report information from children and adolescents (Sallis and Saelens, 2000). There also appears to be an enormous lack of consistency when examining youth physical activity via self-report, with investigators feeling obligated to design and validate a self-report instrument of their own rather than using an existing instrument with acceptable validity and reliability (Trost, 2007). Although self-report is a practical measure of physical activity in large populations (Shephard, 2003), substantial research of a high quality is needed in children and adolescents to improve their validity and reliability (Chinapaw *et al.*, 2010).

#### 2. 7. 3. Heart Rate Monitoring

Heart rate monitoring provides investigators with an estimation of the level at which the cardio-respiratory system works during movement (Armstrong, 1998). A linear relationship exists between increases in heart rate and increases in energy expenditure during exercise (Berggren and Christensen, 1950), primarily due to the requirement of the heart to deliver oxygen to these muscles (Welk *et al.*, 2000, Janz, 2002, Warren *et al.*, 2010). As physical activity and energy expenditure are interrelated (Caspersen *et al.*, 1985), heart rate monitoring is used to provide an objective, indirect measure of physical activity based on physiological effects (Welk *et al.*, 2000). Heart rate monitoring (using flex HR method at a group level to determine activity intensity) has been validated for free-living physical activity against total energy expenditure and physical activity energy expenditure using calorimetry and doubly labelled water in children and adults, and has provided valid and acceptable data (Ceesay *et al.*, 1989, Livingstone *et al.*, 1990, Emons *et al.*, 1992).

Compared with other objective field measures, heart rate monitoring is a relatively inexpensive method of estimating levels of physical activity (Janz, 2002). Heart rate monitors record information for several days, and provide relatively accurate information

to investigators. Initially, the use of heart rate monitors in large-scale epidemiological studies was recommended (Livingstone et al., 1990, Emons et al., 1992). Although heart rate is a physiologically determined estimate of physical activity, its limitations must be considered. Many different factors affect heart rate and the resulting relationship with  $VO_2$ including age, body composition and cardio-respiratory fitness, resulting in various sources of error influencing validity (Welk et al., 2000, Rowlands and Eston, 2007, Trost, 2007). Additional fundamental measurement limitations of heart rate in the examination of youth physical activity exist. There is a delay between physical activity and heart rate response. Also, heart rate tends to remain elevated for prolonged periods after physical activity. Due to these limitations, the ability of heart rate monitors to provide detailed activity information in youths, who have more sporadic activity patterns than adults, has been questioned (Welk et al., 2000, Trost, 2007). Furthermore, the loss of data due to signal loss and low compliance due to discomfort are additional disadvantages in using heart rate monitors in youths (Welk et al., 2000). The substantial limitations associated with heart rate monitoring have impacted its use in large-scale epidemiological studies, but its full potential may yet be seen, as some motion devices now incorporate both accelerometry and heart rate as an estimate of physical activity and energy expenditure with significant reliability and validity (Brage et al., 2005, Barreira et al., 2009).

## 2.7.4. Motion Sensors

### 2. 7. 4. 1. Pedometers

There is a need for low cost, accurate and objective devices to examine ambulatory activity (Bassett Jr and Strath, 2002). It has been suggested that pedometers may be the most practical solution (Welk *et al.*, 2000, Tudor-Locke *et al.*, 2002). Pedometers are small, objective devices that measure the number of steps taken and distance walked/run over a specific measurement period (Washburn *et al.*, 1980, Sirard and Pate, 2001, Rowlands and Eston, 2007, Warren *et al.*, 2010). Most electronic pedometers contain a horizontal spring suspended lever arm that moves up and down due to vertical acceleration during ambulation of the hips (Bassett Jr and Strath, 2002, Tudor-Locke *et al.*, 2002). This opens and closes an electrical circuit, which registers a step count on a digital display. These

devices are normally attached to the waist using a specific belt or waistband, and are normally worn in the midline of the thigh to best examine the vertical accelerations (Bassett Jr and Strath, 2002).

A review by Tudor-Locke and colleagues has highlighted the concurrent validity of pedometers when compared against accelerometers, behavioural observation, self-report and energy expenditure (Tudor-Locke et al., 2002). Output from pedometers was highly correlated with that from accelerometers when raw data from both was compared for freeliving conditions (median r = 0.86). Pedometers also perform well when validated against direct observation for ambulatory activity, but poorly when ambulation was slow and in individuals with increased adiposity (median r = 0.82). The validity of pedometers varied dramatically when compared with different measures of self-report, with high correlations between measures that examine distance walked and ambulatory activities, and poor correlations between pedometers and self-reported measures that examine sitting times (median r = 0.33). However, this is to be expected due to the sensitivity of pedometers when examining distance and ambulation (Welk et al., 2000, Tudor-Locke et al., 2002). Pedometers correlated relatively poorly when compared to other measures of physical activity against energy expenditure (median r = 0.68). However, fundamental limitations exist when making such comparisons, while physical activity and energy expenditure are interrelated, they are not synonymous nor are they interchangeable (Tudor-Locke et al., 2002).

Pedometers provide a relatively accurate, objective and low cost measure of walking behaviours (Bassett Jr and Strath, 2002, Tudor-Locke *et al.*, 2002). However, pedometers do have significant limitations, which must be considered. The majority of low cost pedometers are unable to provide information on the intensity of physical activity (Rowlands and Eston, 2007, Trost, 2007). Additionally, pedometers only provide ambulatory information and are insensitive to some forms of movement such as upper body activities and activities of a cyclic nature such as cycling or rowing (Rowlands and Eston, 2007, Trost, 2007). Historically, the majority of pedometers do not use real time data storage nor do they have the ability to download information collected (Trost, 2007). As a result of these limitations, the use of pedometers when examining physical activity in health-related research has been limited. However, it has been suggested that pedometers may be an effective intervention tool when used as a motivational or behavioural trigger (Welk *et al.*, 2000, Rowlands and Eston, 2007).

## 2.7.4.2. Accelerometers

Over the last three decades, accelerometry has become the most commonly used method of assessing free-living physical activity in health-related research (Esliger et al., 2005, Freedson et al., 2005, Rowlands, 2007a, Trost, 2007, Reilly et al., 2008). The basic concept behind accelerometry is that accelerations are directly proportional to muscular forces used, and are therefore related to energy expenditure (Freedson and Miller, 2000, Warren et al., 2010). Accelerometers measure the acceleration of an object relative to free-fall. In most accelerometers, the magnitude of the acceleration is measured using piezo-resistive or piezoelectric technology, which emits voltages proportional to acceleration experienced (Chen and Bassett Jr, 2005, Esliger et al., 2005). Filtering programs remove accelerations which are too high or too low to be generated by human movement (Welk, 2002, Chen and Bassett Jr, 2005). The outputs are then processed to produce what are known as accelerometer/activity counts. Proprietary algorithms sum these results over a specific time interval or epoch (Chen and Bassett Jr, 2005, Rowlands, 2007a, Rowlands and Eston, 2007). These accelerometer counts provide an objective measure of movement intensity, with greater accelerations providing more counts (Esliger et al., 2005, Warren et al., 2010). As time is also recorded during measurement, it is possible for researchers to calculate frequency, intensity and duration of PA (Melanson Jr et al., 1996, Welk et al., 2000, Trost, 2001, Puyau et al., 2002, Rowlands and Eston, 2007).

A broad range of commercially available accelerometers are employed in the assessment of physical activity across all ages. Although each device has the same fundamental function, different procedures (e.g. filtering, data storage etc.) are used to process the information (Welk, 2005). However, it is essential that each measurement instrument is effective at measuring what it is intended to measure (that each device is a valid measure) (Chinapaw *et al.*, 2010). In accelerometry, the most common method of evaluating whether an accelerometer device is a valid measure is through criterion validation, normally using indirect calorimetry or doubly labelled water (De Vries *et al.*, 2009). Several reviews have examined the criterion validity of accelerometers in youths using indirect calorimetry and doubly labelled water (Westerterp, 1999, Sirard and Pate, 2001, Welk *et al.*, 2004, Trost *et al.*, 2005, De Vries *et al.*, 2006, Bassett Jr *et al.*, 2012). The most common method of validating these accelerometers is by comparing accelerometer output with energy expenditure through indirect calorimetry in lab-based settings. Some accelerometers have been validated

in field-based settings using doubly labelled water. Literature reviews have examined the published validity of accelerometers and have shown moderate to high correlations between activity counts and energy expenditure (e.g. ActiGraph  $r \approx 0.69 - 0.94$  (Sirard and Pate, 2001); Tritrac-R3D  $r \approx 0.34 - 0.91$ (De Vries *et al.*, 2006); GENEA  $\approx 0.83 - 0.87$  (Esliger *et al.*, 2011)). An examination of different types of accelerometers and their validation and calibration will be further discussed in Section 2. 8. 1.

The use of accelerometers in the assessment of free-living physical activity has many advantages. Accelerometers are valid, objective (Welk *et al.*, 2000, Janz, 2006, Rowlands and Eston, 2007), can directly measure an individual's movement (Rowlands and Eston, 2007, Warren *et al.*, 2010) and can be worn for prolonged periods by all populations (Westerterp, 2009). However, perhaps the greatest advantage of accelerometry is its time sampling capabilities (Janz, 2006, Westerterp, 2009). Accelerometers have the ability to record physical activity information over specific time intervals, or epochs (Trost *et al.*, 2005). With this, researchers have the capability to examine frequency, intensity and duration of physical activity, which is essential when comparing physical activity behaviour to indices of health (Melanson Jr *et al.*, 1996, Welk *et al.*, 2000, Trost, 2007, Butte *et al.*, 2012).

As with all measures of physical activity, accelerometry has distinct limitations. Accelerometers are not "plug and play" devices, as several variables including validity, positioning, compliance and output analysis must all be considered for each specific population prior to use (Trost *et al.*, 2005). Output analysis is a particularly important characteristic that must be considered, and will be discussed in detail in Chapter 4. Additionally, the output received from an accelerometer is not a measure of an individual's total activity, as most accelerometers cannot examine upper body, water-based, cyclic activities (such as cycling and rowing) or loaded activities (such as weight training) (Westerterp, 1999, Welk *et al.*, 2000, Sirard and Pate, 2001, Warren *et al.*, 2010). Nevertheless, accelerometer output is still recognised as an accurate measure of ambulation, which comprises the bulk of daily activity (Welk, 2002, Welk *et al.*, 2004), and accelerometry remains the most effective and practical method of objectively examining physical activity in field-based research (Welk *et al.*, 2000, Ward *et al.*, 2005, Welk, 2005).

# 2.8. Measuring Physical Activity and Sedentary Behaviours using Accelerometry

Accelerometry is currently the most commonly employed method of objectively assessing physical activity in field-based research (Trost *et al.*, 2005, Rowlands, 2007a, Bassett Jr *et al.*, 2012). As described in Section 2. 7. 4. 2, accelerometers produce raw accelerations which are classified into an arbitrary score, or count, through proprietary algorithms (Esliger *et al.*, 2005, Freedson *et al.*, 2012). Accelerometer or activity counts are a dimensionless unit, and it is the analysis of these counts that predicts physical activity (Freedson *et al.*, 2005, Welk, 2005, Rowlands, 2007a). As proprietary algorithms vary across accelerometers, the output obtained from accelerometers is not comparable across differing devices. Consequently, the most common method of analysing accelerometer output is through calibration, where researchers classify counts into physical activity levels or energy expenditure by using predetermined thresholds (Freedson *et al.*, 2005, Trost *et al.*, 2011).

## 2.8.1. Calibration and Validation of Accelerometers

The calibration and validation of motion sensors is essential when seeking to objectively obtain accurate and reliable measures of physical activity (Bassett Jr *et al.*, 2012). Although the majority of motion sensors and activity monitors are sold "off the shelf" with in-built calibration equations, it is essential that researchers independently examine the effectiveness of the device to measure what it is intended to measure across a range of activity intensities (Welk, 2005). The calibration of accelerometers and activity monitors has been categorised into unit calibration and value calibration.

# 2.8.1.1. Unit Calibration

The reliability of an accelerometer is an essential requirement for validity. Essentially, it is critical that each accelerometer measures and processes raw accelerations in the same way. Through completing a unit calibration of a device, researchers can examine the reliability of

a sensor across multiple accelerometers (Welk *et al.*, 2004, Welk, 2005). For accelerometry, the most commonly employed method of examining unit calibration has been the use of mechanical shakers or spinning devices which provides a standardised acceleration (Welk *et al.*, 2004, Welk, 2005). Devices are normally shaken at specific speeds, and the results obtained from the devices are compared with each other and with these fixed accelerations. It has been suggested that newer devices, including the ActiGraph GT1M, which incorporates a direct compression sensor that is integrated into a solid state microelectromechanical system (MEMS) accelerometer with digital filtering, do not require unit calibration checks, as there is very little variance in sensitivity due to tighter manufacturing tolerances (Kozey *et al.*, 2010b, Bassett Jr *et al.*, 2012). However, experts continue to recommend unit calibration for devices, as a manufacturer determined unit calibration (which normally use one acceleration speed and one frequency) may not be sufficient to determine devices reliability across a range of speeds (Welk, 2005, Bassett Jr *et al.*, 2012).

#### 2. 8. 1. 2. Value Calibration

Accelerometers produce a dimensionless output called accelerometer/activity counts (Welk, 2005, Bassett Jr *et al.*, 2012). The output from different makes and models of accelerometers are not directly comparable due to manufacturer specific proprietary algorithms which filter and process the accelerations (Welk, 2005). As a result, researchers have highlighted the importance of employing methods of standardising accelerometer output analysis, which will enable research to compare findings across studies that employ different devices (Freedson *et al.*, 2005). To date, the most commonly used method to translate and interpret accelerometer output is value calibration (Welk, 2005).

Value calibration refers to the procedure used to convert arbitrary count values from accelerometers into a meaningful value, such as an estimate of energy expenditure, and has been described in great detail by experts in the field (Ward *et al.*, 2005, Welk, 2005, Rowlands, 2007a). Briefly, representative samples of a specific population are required to complete a range of free-living activities, including sedentary, non-ambulatory and ambulatory activities. During this time, a criterion measure of energy expenditure - most commonly portable indirect calorimetry - is obtained, while individuals also wear the

specific device which is to be calibrated. The data from both device's is used to predict energy expenditure or to provide activity intensity thresholds which can be used to distinguish between sedentary, light, moderate and vigorous activity intensities (Rowlands, 2007a). A range of methods have been used to calculate these activity intensity thresholds, including linear regression equations (Montoye *et al.*, 1983, Freedson *et al.*, 1998, Trost *et al.*, 1998), two-regression equations (Crouter *et al.*, 2006), random coefficients models (Treuth *et al.*, 2004, Mattocks *et al.*, 2007) and receiver operating characteristics (Esliger *et al.*, 2011). Accelerometer output is then examined for a specific activity variable using the calculated thresholds, such as total time spent in light and/or moderate to vigorous physical activity or time spent in bouts of a specific duration. A detailed description of the methods used in examining physical activity patterns and behaviours is provided in Chapter 4.

### 2. 9. The Measurement of Sedentary Patterns and Behaviours

To date, the most commonly employed methods of examining sedentary behaviours have been the use of surrogate measures including self-reported TV viewing time and the use of sedentary cut-points or thresholds on accelerometer output (Pate *et al.*, 2008, Reilly *et al.*, 2008, Lubans *et al.*, 2011). However, both of these measures have significant limitations in the examination of sedentary patterns and behaviours (Lubans *et al.*, 2011).

# 2. 9. 1. Self-Reported TV Viewing Time and Sedentary Time

Self-reported sedentary time and TV viewing time has been the predominant measure of sedentariness in epidemiological research (Marshall *et al.*, 2004, Clark *et al.*, 2009). As reported in Section 2. 3. 3, TV viewing time and self-reported sedentary time and patterns have been associated with health indicators (Chinapaw *et al.*, 2011). Unfortunately, self-reported TV viewing time is susceptible to similar inconsistencies and inaccuracies as self-reported physical activity, due to poor recall of a ubiquitous behaviour and poor reliability and validity (Bryant *et al.*, 2007, Lubans *et al.*, 2011). Contrasting evidence has been presented on the associations between TV viewing time and total sedentary time. Researchers have suggests that TV viewing time is not a good marker of sedentary

behaviours, primarily due to the wide range of additional sedentary activities completed by individuals across the normal day (Biddle *et al.*, 2009). However, conflicting evidence has been presented by Clark and colleagues, whereby a statistically significant relationship was observed between TV viewing time and accelerometer determined sedentary time, but correlations were weak and inconsistent across populations (Clark *et al.*, 2011). Furthermore, the reference sedentary measure in this study was determined using a threshold of <100 counts per minute, which itself has significant limitations (Lubans *et al.*, 2011). Additionally, self-reported sedentary time and TV viewing time does not allow the examination of sedentary patterns and behaviours (Bassett Jr *et al.*, 2010). This type of examination is particularly important, as research has highlighted the potential effect of prolonged sedentary periods on indices of health (Dunstan *et al.*, 2011).

#### 2. 9. 2. Accelerometer Determined Sedentary Time

It has been suggested that objective measures of sedentary time provide the greatest potential in the examination of free-living sedentary patterns and behaviours (Pate et al., 2008, Reilly et al., 2008). The majority of published literature, which has examined sedentary patterns and behaviours using accelerometry, has employed the use of thresholds or cutpoints when determining sedentary behaviour (Owen et al., 2010). Unfortunately, the use of sedentary thresholds does not provide researchers with a total description of sedentary activities (Bassett Jr et al., 2010). As highlighted by Hart and colleagues, the use of sedentary thresholds with the most commonly employed accelerometer (the ActiGraph) significantly overestimates the amount of time spent sedentary in a free-living environment when compared with a valid and reliable measure of sitting/lying and standing (Hart et al., 2011). This is primarily due to the inability of the accelerometer cut-points to distinguish between sitting/lying and standing time (Bauman et al., 2011, Lubans et al., 2011). Standing upright is not a sedentary activity (Tremblay et al., 2012), but may produce fewer than 100 counts per minute. Through the use of sedentary thresholds, standing time and light intensity physical activity may be classified as sedentary time (Johnson et al., 2004, Bauman et al., 2011). In addition, the inability of such devices to accurately determine the duration of sedentary periods and/or breaks in sedentary periods limits the detailed examination of sedentary variables (Zderic and Hamilton, 2006, Hamilton et al., 2007). Furthermore, recent research

has highlighted that a sedentary threshold of <150 counts per minute is more accurate at determining sedentary time than the commonly employed <100 counts per minute when compared with direct observation in an adult population (Kozey-Keadle *et al.*, 2011). However, it has recently been highlighted that the use of a sedentary threshold of <100 counts per minute was acceptable when compared to total sitting time determined by a valid objective measure of sitting/lying and standing (Ridgers *et al.*, 2012).

Although total sedentary time is a variable of interest in the study of sedentary behaviours, the accurate examination of the frequency and duration of sedentariness is of great importance (Bassett Jr *et al.*, 2010, Owen *et al.*, 2010, Dunstan *et al.*, 2011). Self-reported sedentary time and TV viewing time are not direct measures of sedentariness (Pate *et al.*, 2008, Lubans *et al.*, 2011). Although more accurate at determining sedentary time, the use of cut-points or thresholds does not allow the direct measurement of posture, but rather relies on estimating sedentary time based on the lack of movement (Lubans *et al.*, 2011). As a result, these measures do not provide researchers with direct and reliable measures of sedentary variables, which are essential in the examination of free-living activity patterns and behaviours (Rowlands, 2007a, Loprinzi and Cardinal, 2011, Intille *et al.*, 2012).

# 2. 10. The activPAL<sup>TM</sup> Professional Physical Activity Monitor

Due to the limitations of objectively measured sedentary time highlighted in Section 2. 9. 2, it has been suggested that valid and reliable measures of posture, which can distinguish between sitting and standing time should be employed when examining sedentary patterns and behaviours (Matthews *et al.*, 2008, Bassett Jr *et al.*, 2010, Kozey-Keadle *et al.*, 2011). One such device is the activPAL<sup>TM</sup> Professional Physical Activity Monitor (PAL technologies Ltd., Glasgow, UK). A detailed description of the activPAL and the procedure used when wearing the device is described in Section 3. 2. 2 and Section 4. 2. 1. Briefly, the activPAL is a uniaxial accelerometer which incorporates Intelligent Activity Classification<sup>TM</sup> based on information obtained from MicroElectroMechanical System (MEMS). The device is small (53 x 35 x 7 mm) and lightweight (20 grams), and is worn on the midpoint of the anterior aspect of the right thigh, and is attached using a hydro-gel adhesive pad called a PAL*stickie<sup>TM</sup>*.

The activPAL has been identified as a valid measure of posture allocation in children (Davies *et al.*, 2011, Ridgers *et al.*, 2012) and static and dynamic activities in adults (Grant *et al.*, 2006, Godfrey *et al.*, 2007, Hart *et al.*, 2011, Kozey-Keadle *et al.*, 2011). Consequently, the use of the activPAL to examine total sedentary time and sedentary patterns and behaviours in free-living environments has been recommended (Matthews *et al.*, 2008, Healy *et al.*, 2011a, Kozey-Keadle *et al.*, 2011). In particular, the use of the activPAL has been encouraged when examining sedentariness for more detailed variables such as breaks in sedentariness and sedentary bout duration (Bassett Jr *et al.*, 2010). The activPAL has also been shown to be a valid measure of stepping and cadence in children (Davies *et al.*, 2001), adolescent females (Harrington *et al.*, 2011b) and in both healthy adults (Grant *et al.*, 2006, Ryan *et al.*, 2006, Maddocks *et al.*, 2010) and older adults (Grant *et al.*, 2008). Although the validity of the step count function of the activPAL has not been calibrated in any free-living population. This will be discussed and examined in Chapter 3.

Due to the validity and reliability of the activPAL Professional Physical Activity Monitor as a measure of both physical activity and as a measure of sitting/lying and standing, the activPAL has been employed as a measure of activity patterns and behaviours in children (Martin *et al.*, 2011), in adolescent females (Harrington *et al.*, 2011a) and in a range of adult settings (Dall and Kerr, 2010, Grant *et al.*, 2010, Chastin *et al.*, 2012).

#### 2. 11. Summary: Section B

The valid and reliable measurement of physical activity and sedentary behaviour is essential when examining the associations between activity variables and indices of health (Westerterp, 2009, Loprinzi and Cardinal, 2011, Intille *et al.*, 2012). Currently, a wide variety of assessment methods are available for examining physical activity in child and adolescent populations (Trost, 2007, Westerterp, 2009, Warren *et al.*, 2010). Historically epidemiological evidence has assessed activity variables using self-report and questionnaire-based measures due to their low cost, acceptable levels of reliability and validity and the ease with which they can be administered to large cohorts (Kohl *et al.*, 2000, Matthews, 2002, Trost, 2007). Limited information is currently available on the validity of self-reported measures of sedentary behaviours (Lubans *et al.*, 2011). Over the last number of

decades, the objective examination of physical activity using accelerometers has become increasingly popular, primarily due to increased validity and reliability compared to selfreported measures and affordability of available activity monitors (Rowlands, 2007a, Warren *et al.*, 2010). However, their ability to accurately examine sedentary time (and consequently estimated light intensity physical activity) has also been questioned (Hart *et al.*, 2011, Kozey-Keadle *et al.*, 2011). As a result, the use of multi-sensor devices, which can both objectively examine physical activity and directly examine sedentary time, has been encouraged when examining physical activity and sedentary behaviours in free-living populations (Hamilton *et al.*, 2008, Bassett Jr *et al.*, 2010, Owen *et al.*, 2010).

# 2. 12. Section C: Levels of Physical Activity and Sedentary Behaviour in Childhood and Adolescence

Despite the recommendations for adequate physical activity in adolescence being well documented, national and international data indicates that children and adolescents participate in low levels of physical activity. Furthermore, it is apparent that youths become less physically active between the onset of puberty and age 18 (Schmitz *et al.*, 2005). Irish and international data on these levels and trends of physical activity will be discussed here.

# 2. 12. 1. National Levels and Trends of Childhood and Adolescent Physical Activity and Sedentary Behaviour

Sedentary patterns and behaviours and physical activity levels have predominantly been examined using subjective measures in the Irish context. However, over the last decade, several examinations of physical activity using objective measures have taken place. The findings from these studies are presented and discussed below.

# 2. 12. 1. 1. Irish Physical Activity and Sedentary Levels Examined using Subjective Measures

A number of studies have examined the levels of physical activity participation in Irish children and adolescents over the last two decades. The National Health and Lifestyle Surveys (Slán) completed in 1998 and 2003 and the Irish Health Behaviour in School-aged Children (HBSC) Study completed in 2006 and 2010 are collectively the most comprehensive examination of physical activity participation in childhood and adolescence in Ireland (Kelleher *et al.*, 1999, Kelleher *et al.*, 2003, Nic Gabhainn *et al.*, 2007, Kelly *et al.*, 2012). Table 2.2 presents the percentage of individuals that took part in exercise on four or more days per week in each of the three examinations. The findings acknowledge moderate participation in physical activity at early and middle adolescence, but highlight severe fallout in participation during late adolescence, particularly in female populations. Caution should be taken when interpreting these results, however, as participation in physical activity was defined as the frequency with which the individual exercised so much that they would get out of breath, and did not consider duration of activity.

**Table 2.2.** The percentage of both males and females (all) and females only that reported participating in exercise four or more times per week.

	9-11		12-14		15-17	
	A11	Girls	All	Girls	All	Girls
Slán 1998 (%)	63	59	58	49	40	26
Slán 2003 (%)	59	55	53	44	35	25
HBSC 2006 (%)	64	58	59	51	42	28
HBSC 2010 (%)	62	60	54	42	41	29

Levels of physical inactivity were also examined and presented in each of these studies. Physical inactivity was defined as participating in exercise on less than one day per week. The findings from each of the four examinations are presented in Table 2.3 (Kelleher *et al.*, 1999, Kelleher *et al.*, 2003, Nic Gabhainn *et al.*, 2007, Kelly *et al.*, 2012). It is evident from these findings that inactivity levels remain consistent across teenage years in Irish males.
However, it is also clear that inactivity levels increase dramatically across Irish females, particularly during the latter stages of adolescence, with an 8% increase in 1998, 13% increase in 2003, 14% increase in 2006 and 9% increase in inactivity in 2006 between the ages 12-14 years and 15-17 years.

**Table 2.3.** The percentage of males and females reporting participation in exercise less than once a week.

	9-11			12-14	15-17	
	Boys	Girls	Boys	Girls	Boys	Girls
Slán 1998 (%)	4	4	3	5	5	13
Slán 2003 (%)	8	8	7	8	8	21
HBSC 2006 (%)	7	8	4	8	9	22
HBSC 2010 (%)	4	7	4	10	8	19

Hussey and colleagues also completed a large data collection of 6-9 year old children in the Dublin region (n=786) (Hussey *et al.*, 2001). This study reported that 39% of children participated in hard exercise for 20 minutes per day on at least 3 days of the week. Similar to the studies mentioned above, the levels of participation in physical activity were significantly higher in males (53%) than females (28%). This study also highlighted that 60% of participants spend greater than 3 hours per day watching television.

Another study completed on the East coast of Ireland (the Take PART study) examined self-reported levels of physical activity in 1,500 adolescents (Woods et al., 2004). For the purpose of this study, regular activity was defined as participating in > 60 minutes of physical activity on 4 or more days of the week. This study classified 56% of 15-17 year old as not regularly active, highlighting that females (62%) were significantly less active than their male counterparts (52%). When the definition of regular activity was increased to 5 or more days of the week, 82% of participants were identified as not regularly active, while only 4% of participants achieved the recommendations on every day of the week. Figure 2.4 presents information on the number of days individuals participated in the recommended amount of daily physical activity. As discussed in Section 2. 7. 2. 1, these

results should also be interpreted with caution, as all data here was obtained using selfreported measures.



Figure 2.4. The number of days participants achieved the recommended amount (>60 minutes) of physical activity.

(Woods et al. 2004. p 33)

The Children's Sports Participation and Physical Activity study (CSPPA study) is one of the most recent research studies to have examined physical activity participation in a large sample (n=1215) of Irish children and adolescents (Woods *et al.*, 2010). Physical activity levels were examined using self-reported measures. Table 2.4 presents the percentage of children and adolescents that achieved the recommended amount of daily physical activity (>60 minutes of moderate to vigorous physical activity) on all days of the week. Few differences were observed in physical activity participation between this examination and that completed by Woods and colleagues in 2004 (Woods *et al.*, 2010).

	Males	Females	All
10-12 years (%)	27	13	19
12-13 years (%)	24	13	18
14-15 years (%)	16	8	12
16-18 years (%)	7	6	6

**Table 2.4.** Percentage of children and adolescents achieving 60 minutes or more of moderate to vigorous physical activity on all days of the week.

(Woods et al. 2010, p30)

# 2. 12. 1. 2. Irish Physical Activity and Sedentary Levels Examined using Objective Measures

Limited objectively measured physical activity information is available in the Irish context. Two studies have examined physical activity levels in child and adolescent populations using heart rate monitoring. Results from a sample of 65 Irish children aged between 7 and 9 years reported that 55% of children accumulated > 60 minutes of light physical activity, 35% achieved > 30 minutes of moderate to vigorous physical activity on every measured day, while 8% achieved > 60 minutes of moderate to vigorous physical activity on all measured days (Belton and MacDonncha, 2007). Findings from heart rate monitoring in 28 Irish adolescents presented low levels of physical activity participation in this population, with no subject meeting the international moderate or vigorous intensity physical activity guidelines (Shiely and MacDonncha, 2009). Interestingly, Shiely and MacDonncha also noted that physical activity levels predicted from heart rate monitoring in this population were significantly lower than those estimated with self-report (Shiely and MacDonncha, 2009).

One Irish study has examined pedometer derived step counts of 7-9 year old Irish children (Belton *et al.*, 2010). Although this study did not present levels of physical activity in terms of physical activity intensities or in terms of reaching the recommendations, this study examined the number of steps accumulated at specific times of the day in both males and females. The findings support the findings observed in much of the other Irish physical

activity research, highlighting that males accumulated significantly more steps than females, while participants were more active on weekend days than on weekdays.

To date, only two studies have employed accelerometry as a measure of physical activity in an Irish cohort (Woods *et al.*, 2010, Harrington *et al.*, 2011a). However, the accelerometer data obtained by Woods and colleagues (2010) was used to validate a self-report measure of physical activity, and accelerometry protocols or detailed accelerometry results have not been reported to date (Woods *et al.*, 2010). As a result, this data cannot be compared to other objective measures of physical activity. One study (Appendix 1) has employed an inclinometer-based accelerometer to examine sedentary patterns and behaviours, and has highlighted that adolescent females spend approximately 19 hours per day sitting/lying, with approximately 79% of their total day (including sleep time) spent in sitting and lying behaviours (Harrington *et al.*, 2011a).

This national information on levels and trends of physical activity in Irish youths has consistently identified that 1) levels of physical activity participation across all ages are low, and are a serious public health concern and 2) levels of physical activity decrease while physical inactivity increases throughout adolescence, with the most severe fallout from participation evident in older adolescent females. Furthermore, the frightening levels and trends that have been observed may be worse than previously thought, with results from objectively assessed physical activity suggesting even lower levels of physical activity participation across children and adolescents (Shiely and MacDonncha, 2009).

# 2. 12. 2. International Levels and Trends of Childhood and Adolescent Physical Activity and Sedentary Behaviour

A wealth of data on physical activity and sedentary behaviours in childhood and adolescence is now available, particularly from the United States, Australia and European research groups. The levels and trends of physical activity and sedentary behaviour measured using both subjective and objective measures are discussed below.

# 2. 12. 2. 1. International Levels of Self-Reported Physical Activity and Sedentary Behaviours

Several large-scale studies have examined the levels and trends of physical activity and sedentary behaviours in childhood and adolescence. One of the most notable of these is the Health Behaviour in School-aged Children (HBSC) Study. The most recent report of the HBSC included 43 countries and excess of 200,000 participants (Currie *et al.*, 2012). The findings from the HBSC Study highlighted that physical activity decreased significantly between ages 11 and 15, while males continued to be more active that their female counterparts across all ages (Table 2.5).

**Table 2.5.** Percentage of children and adolescents that achieved 60 minutes or more of moderate to vigorous physical activity daily.

	11 yrs.			13 yrs.			15 yrs.		
	Boys	Girls	All	Boys	Girls	All	Boys	Girls	All
> 60 mins MVPA daily (%)	28	19	23	24	13	19	19	10	15

(Currie et al., 2012, p 130-131)

The report also highlighted that 15 year olds spent significantly more TV viewing time than 11 year olds, with little or no difference observed between males and females (Table 2.6).

**Table 2.6.** Percentage of children and adolescents spending 2 hours or more watching TV daily.

	11 yrs.			13 yrs.			15 yrs.		
	Boys	Girls	A11	Boys	Girls	A11	Boys	Girls	<b>A11</b>
> 2 hrs TV viewing	58	54	56	65	64	65	64	62	63
daily (%)									

(Currie et al., 2012, p 134-135)

When compared to another recent large-scale surveillance survey from the United States, similar trends of low levels of physical activity and high sedentary levels were observed. The Youth Risk Behaviour Surveillance Survey (YRBSS) identified that 18.4 % of youths examined were physically active for 60 minutes or more daily, with significantly higher rates of participation observed in males (24.8%) than in females (11.4%) (Eaton *et al.*, 2010). Compliance with physical activity recommendations decreased across each class group, with twice as many males as females achieving the recommendation across all class groups. When the information was examined for participation in at least 60 minutes of physical activity on 5 days of the week, the percentage of children and adolescents achieving the recommendations increased to 37%. Finally, 23.1% of participants did not achieve 60 minutes of physical activity on any of the previous 7 days. Physical activity participation results across class group and sex are described for each of these variables in Table 2.7.

**Table 2.7.** Percentage of males and females that achieved the recommended daily amount of daily physical activity on 0 days, 5 days and on all 7 days across grades.

> 60 mins MVPA	9 <sup>th</sup> grade		10 <sup>th</sup> g	10 <sup>th</sup> grade		11 <sup>th</sup> grade		12 <sup>th</sup> grade	
	Boys	Girls	Boys	Girls	Boys	Girls	Boys	Girls	
0 days (%)	17.4	26.9	15.7	30.3	16.4	29.8	18.5	33	
5 days (%)	47.5	30.8	47.4	30.5	46.2	26	40.4	22.4	
7 days (%)	28	13.6	25.3	12.7	23.3	10.3	21.9	8.6	

(Eaton et al. 2010, page 25 & 26)

# 2. 12. 2. 2. International Levels of Objectively Determined Physical Activity and Sedentary Behaviours

Large-scale studies have incorporated the use of objective measures. One of the most significant of these examinations has been the National Health and Nutrition Examination Survey (NHANES). Findings from the survey completed in 2003-2004 have identified similar physical activity trends to those observed through subjective methods (Troiano *et al.*, 2008). In a sample of 1778 participants aged between 6 and 19 years, levels of physical activity were examined. Troiano and colleagues highlighted that mean accelerometry counts

per minute significantly declined from childhood through adolescence, with females achieving fewer accelerometry counts per minute than their male counterparts. Age specific count thresholds, developed and described by Freedson and colleagues, were employed to determine the amount of time spent in moderate to vigorous physical activity (Insert Reference here: Freedson 1997). The mean total amount of moderate physical activity was above 60 minutes in both males and females at age 6-11 years. However, levels of participation in both total and bouts of moderate and vigorous physical activity severely declined across age, with a particular decline in female participation at 16-19 years of age. These results are presented in Table 2.8.

**Table 2.8.** Mean total minutes per day in moderate physical activity (MPA) to vigorous physical activity (VPA) and mean total minutes per day in MPA and VPA in bouts of 10 minutes or more. MPA and VPA were examined using age-specific count to activity thresholds (Freedson *et al.*, 2005).

	Males			Females			
(yrs.)	6-11	12-15	16-19	6-11	12-15	16-19	
(mins)							
Total MPA	79.5	39.2	29.8	65.1	21.7	18.5	
Total VPA	16.0	6.0	3.0	10.1	2.9	1.1	
Bouts of MPA	22.4	11.6	7.7	12.8	4.4	4.2	
Bouts of VPA	4.2	1.4	0.9	1.8	0.9	0.4	

(Troiano et al. 2008, p 185)

Matthews and colleagues examined the 2003-2004 NHANES accelerometer output for sedentary behaviour in a similar manner to that described in Table 2.9 (Matthews *et al.*, 2008). These findings also highlighted high levels of sedentary time in each population, with females spending significantly more time sedentary than their male counterparts at age 6-11, 12-15 and 16-19 years.

	Total	Males	Females
6-11 yrs.	6.07	6.00	6.14
12-15 yrs.	7.53	7.37	7.7
16-19 yrs.	8.03	7.91	8.13

**Table 2.9.** Mean hours spent in sedentary behaviours for all participants, males and females across the ages of 6 - 19 years.

(Matthews et al. 2008, Page 878)

The NHANES data of 2003-2004 and 2005-2006 has also recently been collectively examined for physical activity levels and sedentary patterns and behaviours in children and adolescents (Carson and Janssen, 2011). Sedentary time was examined using a count-to-activity threshold of <100 counts per minute. Moderate to vigorous physical activity was determined using age specific count to activity thresholds (Freedson *et al.*, 2005). The median values for the accelerometry wear time identified children and adolescents as spending approximately 50.8% of their wear time in sedentary behaviours, 43.8% in light intensity activity and 4.1% in moderate to vigorous physical activity. Interestingly, this population spent 24.5% of their day in prolonged sedentary periods of greater than 30 minutes in duration, which may have specific deleterious effects (Section 2. 4. 4). Unfortunately, the focus of this paper was on the relationship between sedentary patterns and behaviours and cardio-metabolic health in this population. As a result, further breakdown of sedentary results are not provided.

The European Youth Heart Study has provided objectively assessed physical activity trends in a large cohort of 9 and 15 year old children and adolescents in the European context. Count to activity thresholds of 1000 and 1500 counts per minute respectively were employed to determine moderate to vigorous physical activity in these populations (Riddoch *et al.*, 2004). Similar trends to those presented in the NHANES examination were observed here, with the authors concluding that males are more active than females, with a significant reduction in activity over the adolescent years in both males and females (Riddoch *et al.*, 2004). The HELENA study, in which a large-scale objective examination of physical activity was completed across 10 European cities, have provided further data of physical activity and sedentary patterns and behaviours (Martinez-Gomez *et al.*, 2011b). Findings from this study have concurred with Riddoch and colleagues, stating that adolescent European males have higher levels of MVPA and achieve the recommended amount of daily MVPA more often than females. Additionally, Martinez-Gomez and colleagues noted that adolescent females spent significantly more time in sedentary activities than males.

A recent publication on the largest sample of children and adolescents, which included findings from NHANES and other large-scale objective investigations, has reported physical activity and sedentary time in 20,871 children and adolescents (Ekelund *et al.*, 2012). Sedentary time was identified as time spent below a sedentary count to activity threshold of 100 counts per minute, while a count to activity threshold of <3000 counts per minute identified time spent in moderate to vigorous physical activity. The findings highlight that males achieve significantly higher mean accelerometer counts per minute ( $642 \pm 226 \text{ cpm/d}^{-1}$ ) than female ( $540 \pm 193 \text{ cpm/d}^{-1}$ ; p<0.001). Additionally, significantly more time was spent in moderate to vigorous physical activity in males ( $37 \pm 23 \text{ min/d}^{-1}$ ) than females ( $24 \pm 17 \text{ min/d}^{-1}$ ; p<0.001), while females accumulated significantly more daily sedentary time ( $363 \pm 96 \text{ min/d}^{-1}$ ) than males ( $345 \pm 96 \text{ min/d}^{-1}$ ; p<0.001).

## 2. 13. Summary: Section C

It is clear from both the objective and subjective findings that physical activity in children and adolescents are lower than expected at baseline. It is also evident that physical activity declines severely across childhood and adolescence. It is also obvious that this fallout is greater in females, with the greatest fallout in physical activity observed in adolescent females aged greater than 15 years. Inverse trends are observed for sedentary time, with all research suggesting that sedentary time increases throughout adolescence, with greater sedentary levels observed in females than males. As highlighted throughout this review, it is important that caution is employed when interpreting sedentary patterns and behaviours, as surrogate measures rather than direct measures have been employed in its examination to date (Kozey-Keadle *et al.*, 2011). Collectively, the information presented here suggests that females adolescents are a particularly at risk population, as tracking of activity behaviours (as discussed in Section 2. 4. 3) would infer that these deleterious behaviours (particularly sedentary behaviour) will continue through adolescence and into adult life, further increasing the risk of cardiovascular disease and related comorbidities.

### 2. 14. General Summary

Evidence suggests that associations are evident between levels of moderate to vigorous physical activity in childhood and adolescence and mortality from cardiovascular disease and related disorders in adulthood. Over the last 2 decades, the importance of lower spectrum activity intensities has been emphasised, and research has begun to examine all activity intensities when examining relationships with health. It seems rational that the interaction between the activity profile, rather than specific activity intensities, is the most effective method of associating habitual physical activity with indices of health. In order to effectively examine relationships between indices of health and the activity profile, valid and reliable measures of both activity patterns and behaviours are necessary. To date, no research has examined the relationships between objectively examined physical activities and directly examined sedentary patterns and behaviours and indices of health in any population.

# Chapter 3. Criterion and Concurrent Validity of the activPAL<sup>™</sup> Professional Physical Activity Monitor in Adolescent Females.

(Published in PLoS ONE on 19/10/12)

#### 3. 1. Introduction:

### 3. 1. 1. Prevalence and Effect of Moderate to Vigorous Physical Activity

Increased levels of moderate to vigorous physical activity have the potential to improve cardio-metabolic risk factors, improve bone health, reduce the risk of depression and reduce the risk of becoming overweight/obese in childhood, in adolescence and in adulthood (Warburton *et al.*, 2006, Janssen and LeBlanc, 2010). Despite the widespread publication of the benefits of physical activity, levels remain low in many countries (Troiano *et al.*, 2008, Colley *et al.*, 2011b, Lee *et al.*, 2012). Furthermore, the most significant decrease in levels of physical activity occur in later adolescence, with greater decreases observed in females (Riddoch *et al.*, 2004, Troiano *et al.*, 2008). This is critical, as the processes associated with long term risk of diseases, such as coronary heart disease, begin in childhood and adolescence (Berghöfer *et al.*, 2008).

# 3. 1. 2. Methods of Assessing Physical Activity

Advancing the field of free-living activity measurement requires the development of methodologies that are practical in habitual settings. These methodologies are crucial when relating levels of physical activity to indices of health (Freedson *et al.*, 1998, Esliger *et al.*, 2011). Over the past two decades, accelerometry has become the preferred method of objectively examining physical activity in free-living populations (Trost *et al.*, 2005, Rowlands, 2007a, Welk *et al.*, 2007). This is primarily due to the rich information obtained from the devices (Rowlands, 2007a, Bassett Jr *et al.*, 2012), coupled with relatively high levels of reliability and validity and the lowering costs of the monitoring devices themselves (Trost *et al.*, 2005). Typically, accelerometers record raw accelerations, and proprietary algorithms calculate arbitrary units known as accelerometer or activity levels (light, moderate, vigorous) using predetermined thresholds (Bassett Jr *et al.*, 2012). Total minutes spent per day at each physical activity intensity, along with the frequency and duration of physical activity can then be calculated (Freedson *et al.*, 1998, Bassett Jr *et al.*, 2012).

## 3. 1. 3. The Sedentary Behaviour Paradigm

In recent years, the quantification of sedentary behaviour, as well as physical activity, has become extremely topical, as the deleterious effects of sedentariness have been emphasised (Owen et al., 2010). Inactivity physiologists have highlighted the negative effect of sedentary behaviours on indices of health in rats, and have suggested the loss of contractile stimulation of large skeletal muscles as one the major physiological variables which regulates muscle enzyme lipoprotein lipase (LPL) (Bey and Hamilton, 2003, Hamilton et al., 2007, Hamilton et al., 2008). The suggestion that isometric contraction of antigravity muscles produce electromyographic and skeletal muscle LPL change (Hamilton et al., 2007, Hamilton et al., 2008) implies that activities such as standing, which would previously have been considered sedentary, should now be considered as distinct activity behaviours (Owen et al., 2010). Consequently, sedentary behaviour is now characterised by energy expenditure below 1.5 metabolic equivalents (METs) while in a sitting or lying position during waking hours (Marshall and Ramirez, 2011, Tremblay et al., 2012). To date, epidemiological evidence has supported the physiological observations, highlighting the negative effect of sedentary patterns and behaviours in both adolescents (Martinez-Gomez et al., 2010) and adults (Healy et al., 2007, Healy et al., 2008a, Healy et al., 2011b). Unfortunately, the methods used to examine sedentariness in these studies have significant limitations. Surrogate measures of sedentary behaviour, such as self-reported TV viewing time, do not accurately quantify sedentariness, and only examine one aspect of sedentary behaviour (Pate et al., 2008). Furthermore, the use of indirect measures of sedentariness, such as the use of sedentary thresholds from accelerometer counts (e.g.  $\leq 100$  counts min<sup>-1</sup>) rely on the lack of ambulation or movement rather than directly measuring body position (Hart et al., 2011, Kozey-Keadle et al., 2011).

## 3. 1. 4. Methods of Assessing Sedentary Behaviour

Due to increased interest in sedentary behaviour and the obvious interest in examining levels of physical activity, a device that is both a valid and reliable measure of both domains would be extremely valuable. While the ActiGraph GT1M and GT3X (Manufacturing Technologies Inc. Health Systems, Shalimar, FL), for example, are valid measures of

physical activity, their measurement of sedentary behaviours are dependent on thresholds. It has been suggested that the use of such thresholds to determine sedentary time could lead to errors, as this analysis may include other activities, such as standing (Owen et al., 2010, Hart et al., 2011, Kozey-Keadle et al., 2011). Recent technological developments have provided researchers with the tools to directly examine sedentary behaviours without the use of thresholds. The use of inclinometer-based activity monitor, such as the activPAL Professional Physical Activity Monitor (PAL Technologies Ltd., Glasgow, UK), has enabled researchers to directly identify periods of sitting/lying, standing and stepping, and have been encouraged for studies examining sedentary behaviours in detail (Bassett Jr et al., 2010, Owen et al., 2010). The activPAL is worn on the midline of the anterior aspect of the thigh. Due to this unique positioning, the inbuilt inclinometer is able to distinguish between sitting/lying and standing, while the activity monitoring function of the device allows the examination of ambulation, as with other existing devices. The device has been validated for the measurement of static and dynamic activities in adults (Godfrey et al., 2007), posture during free-living activities (Grant et al., 2006), step and cadence output in females (Harrington et al., 2011b) and in the examination of sedentary time in children (Ridgers et al., 2012) and adults (Kozey-Keadle et al., 2011). While the activPAL has been used as a measure of habitual locomotion using steps (Chastin et al., 2009) and has had a cadence threshold developed for adolescent female populations (Harrington et al., 2012), the activity count function has not been investigated in the same way as with the ActiGraph, and activity counts have not been utilised to examine free-living physical activity in any population. Furthermore, the sit/lie and stand function has not been validated in an adolescent population.

The primary aims of this chapter are to investigate the validity of activPAL activity counts in estimating energy expenditure, to perform a value-calibration of the activPAL and to validate thresholds for defining moderate physical activity and vigorous physical activity for an adolescent female population. This chapter also aims to examine the concurrent validity of the activPAL by comparing activity counts across different activities with the ActiGraph GT3X accelerometer. Finally, the chapter aims to compare the performance of the activPAL with the ActiGraph for estimating time spent sitting, standing and stepping.

### 3. 2. Methods

### 3. 2. 1. Participants

Participants for the study were recruited from a community youth group in the West of Ireland. To be considered for inclusion in this study, participants had to be female, aged between 15 and 18 years and have no injury or illness that limited their ability to be physically active. This population were selected for investigation as female adolescents have been highlighted as a particularly inactive population, and as a result are of great interest to physical activity and health practitioners (Riddoch et al., 2004, Troiano et al., 2008, Matthews et al., 2012). A local community youth group was approached to participate in this study. All female members of this group were invited to an information evening, where the study protocols and objectives were clearly outlined. All participants were provided with parental and participant information sheets and consent forms, and invited to return the completed consent forms at their next youth group meeting. All participants that returned completed participant and parental written informed consent were selected for participation (n = 40). Three participants from the original sample of 40 withdrew from the study. From the remaining participants, five data sets were excluded from analysis due to a malfunction of indirect calorimetry measurement. A total of 32 valid sets of VO<sub>2</sub> data were obtained from the participants for statistical analysis. Two sets of activPAL data were not included in the analysis due to equipment malfunction, resulting in 30 full sets of activPAL, ActiGraph GT3X and simultaneous VO<sub>2</sub> data for the current analysis. Each participant was allocated a number, and a randomisation table was used to assign each participant to either an equation development group or a cross-validation group. The study was approved by the Faculty of Education and Health Sciences Research Ethics Committee at the University of Limerick.

# 3. 2. 2. Physical Activity Measurement Devices

Physical activity was recorded using the activPAL<sup>TM</sup> Professional Physical Activity Monitor (Firmware: v 0.9.9) which is a single unit uniaxial accelerometer (53 x 35 x 7 mm) that weighs approximately 20 grams. The activPAL responds to gravitational accelerations

resulting from segmental movement (Ryan *et al.*, 2006), and data is recorded at 10 Hz for each 15 second time intervals (epoch). Proprietary algorithms provide outputs including time spent sitting/lying, standing and stepping, step counts and cadence and activity counts. The activPAL communicates with a Windows (Microsoft Corporation, Microsoft Excel 2010, One Microsoft Way, Redmond, WA, USA) compatible PC via a USB interface. The ActiGraph GT3X (Firmware v 4.1.0) (Pensacola, FL 32502, USA) is a small triaxial accelerometer (38 mm  $\times$  37 mm  $\times$  18 mm) that weighs approximately 27 grams. Throughout the remainder of this document, the term ActiGraph will refer to the ActiGraph GT3X unless otherwise stated. The device assesses acceleration in three individual orthogonal planes using a vertical axis, horizontal axis and a perpendicular axis. The ActiGraph samples accelerations at a rate of 30 Hz and the data can be reprocessed into epochs ranging from 1 to 60 seconds. The ActiGraph also communicates with a Windows compatible PC via a USB interface. For the purpose of this study, the ActiGraph monitor was initialised to record vertical accelerations every 15 seconds.

### 3. 2. 3. Metabolic Unit:

Oxygen consumption was measured breath-by-breath using a portable metabolic unit (Cosmed K4B<sup>2</sup>, Rome, Italy). The K4B<sup>2</sup> is a lightweight system with a heart rate receiver. The Cosmed K4B<sup>2</sup> has been deemed an appropriate criterion measure for minute-byminute energy expenditure (Bassett Jr *et al.*, 2012). Over the duration of the study, the K4B<sup>2</sup> was calibrated following standard manufacturer procedures. Before each testing session, the device was calibrated using known gas concentrations, and environmental conditions were updated. The VO<sub>2</sub> data was downloaded and stored on a PC after each testing session.

### 3. 2. 4. Testing Protocol

Participants arrived to the testing facility having fasted for two hours, refrained from smoking and drinking caffeine for 2 hours, and having refrained from moderate and vigorous physical activity for 12 hours. Participants wore light shorts, t-shirt or vest, socks and running shoes. Height was measured without shoes and socks to the nearest 0.25 cm using a portable stadiometer (Seca model 214, Seca Ltd, Birmingham, UK) and weight was measured without shoes and socks to the nearest 0.1 kg using portable electronic scales (Seca model 770, Seca Ltd, Birmingham, UK). Body mass index was then calculated from the height and weight measures (kg/m<sup>2</sup>). After participants arrived to the test facility and had their height and weight obtained, participants were fitted with the activPAL, the ActiGraph and the Cosmed K4B<sup>2</sup> metabolic unit. Both accelerometers were initialised prior to participant's arrival and their internal clocks were automatically synchronised with the main investigator computer. To conform to the sampling time used by the activPAL, a 15 s<sup>-1</sup> epoch was used for the ActiGraph. The activPAL was attached directly on the skin of the midline of the anterior aspect of the right thigh using a PAL*stickie*, (double sided hydrogel adhesive pad) and tube bandages were used as extra security to keep the activity monitor in place. The ActiGraph was worn around the waist on an elasticated band over the right hip bone. The metabolic unit was placed over participants shoulders and a mask fitted over the face.

Participants were then introduced to the protocol of activities. Activities were completed in ascending intensity throughout the testing period. Participants were instructed when to start each activity and when to stop each activity by a single observer. The activity category, the exact start time and the exact finish time of each activity were recorded by the observer. Resting  $VO_2$  was measured for 25 minutes, to ensure the participants were provided an adequate amount of time to return to a rested state. During this time period, participants lay in a reclined position on a physiotherapy plinth in a darkened, quiet room. For the sitting activity, participants sat looking straightforward, placed their feet flat on the floor, placed their hands on their knees and were asked not to speak or take part in any other activity. Participants did not lean against or hold on to any support while completing the seated and standing activities.

Participants then completed the 3 locomotor activities. Participants were asked to complete each activity at a pace that was comfortable to them, but within each speed range: slow walking (2.5-4.5 km.h<sup>-1</sup>), brisk walking (4.5-6.5 km.h<sup>-1</sup>) and light jogging (6.5-8.5 km.h<sup>-1</sup>). Prior to the study beginning, the upper and lower time limits required to complete each section of the track during each speed category were calculated. The time taken to

complete each section of the track was then used to estimate the speed of each participant. During the first minute of measurement, the time it took for each participant to complete each section of the track was recorded. If participants completed each section too slowly or too quickly, they were deemed to be travelling too slow or too fast, and they were asked to adjust their speed accordingly. Once participants were comfortable at travelling within the speed category, the time required to complete each section of the track was recorded, and participants were encouraged to maintain this speed throughout the remainder of the activity. Feedback was provided to each participant throughout the remainder of each speed category in an attempt to maintain a relatively constant speed. This approach was used to simulate real-life activity and to reduce the clustering effect that set speeds may have during statistical analysis. Individual rest periods between each movement activity were used to return participants heart rate below 100 beats min<sup>-1</sup>. Once the protocol was completed, the collected data was downloaded to the main investigator's PC.

#### 3. 2. 5. Calibration Activities

The activities included in the protocol were 1) resting  $VO_2$ , 2) sitting on a chair, 3) standing upright unaided, 4) slow walking (2.5-4.5 km.h<sup>-1</sup>), 5) brisk walking (4.5-6.5 km.h<sup>-1</sup>) and 6) light jogging (6.5-8.5 km.h<sup>-1</sup>). The activities included in the protocol were informed by a number of past studies which recommended sedentary and locomotor activities (Treuth *et al.*, 2004, Welk, 2005, Bassett Jr *et al.*, 2012). Other core activities which have been recommended for inclusion in validation and calibration studies, such as car driving, cycling and stair ascending/descending have not been included in this protocol (Bassett Jr *et al.*, 2012). Car driving was not included for insurance reasons, cycling was omitted as extremely low levels of participation in cycling have been observed in this population (Department of Transport, 2009) and stair ascending/descending was not included for practicality reasons.

For the sitting and standing activities, data was collected for 5 minutes, while 7 minutes of data was collected for ambulatory activities, with the mean value of the final two minutes of each activity used for data analysis. These durations were selected as  $VO_2$  remains stable (steady state) after 3 minutes for light activity and after 3-5 minutes for more intense activities (Poole *et al.*, 1991). The internal time clocks of the ActiGraph, activPAL and metabolic unit were synchronised The mean value of the final two minutes of each activity

(excluding resting  $VO_2$ ) was used for data analysis, as participants were deemed to be at steady state energy expenditure during this period (Treuth *et al.*, 2004).

## 3. 2. 6. Data Processing

Once data from both the activPAL and ActiGraph were downloaded, files were processed using the activPAL (v 5.9.1.1) and ActiGraph (Actilife v4.4.1) software. For the ActiGraph, only accelerations measured in the vertical plane were used for comparative analysis and the low frequency extension was not employed. This created time stamped 15 sec<sup>-1</sup> epochby-epoch information. Using the K4B<sup>2</sup> software, the VO<sub>2</sub> information was averaged for every 15 sec<sup>-1</sup> period, synchronising the start time with the protocol start time. The breathby-breath VO<sub>2</sub> data from the K4B<sup>2</sup> and the resulting epoch-by-epoch data from both the activPAL and the ActiGraph were collated, ensuring that the protocol start time for each individual was synchronised for all three devices. The information recorded by the single observer for the activity categories was synchronised with the VO<sub>2</sub>, activPAL and ActiGraph data. The final 2 minutes of each activity performed were selected for analysis, and all data exported to Predictive Analytic Software (PASW) version 18.0 for Windows (SPSS Inc., Chicago, IL, USA).

When preparing data for examining the levels of agreement between sitting/lying, standing and stepping, the inclinometer function of the activPAL was used to examine sitting/lying time and standing time, while the accelerometer function of the activPAL was used to examine stepping time. A sedentary threshold of < 100 counts·min<sup>-1</sup> has been suggested for use with the ActiGraph in youths from studies which have examined levels of agreement between the activPAL and the ActiGraph (Ridgers *et al.*, 2012). For this reason, a sedentary threshold of < 100 counts·min<sup>-1</sup> was used to quantify sedentary time using the ActiGraph only. This information was compared with the recorded activity categories from the single observer.

#### 3. 2. 7. Statistical Analysis

The use of conventional 1 MET values is discouraged for use in both children and adults (Butte *et al.*, 2012). To normalise energy cost between participants during different tasks, participant's individual resting metabolic rate (RMR) were used to calculate their resting MET values, with energy cost during activity being expressed in calculated METs (MET score = Activity VO<sub>2</sub> mL·kg<sup>-1</sup>min<sup>-1</sup>/Resting VO<sub>2</sub> mL·kg<sup>-1</sup>min<sup>-1</sup>). The accelerometer data was plotted to coincide with the steady state VO<sub>2</sub> for each of the performed activities. Spearman's Rho correlation coefficients were calculated between accelerometer output from both the activPAL and the ActiGraph and VO<sub>2</sub> with an r value of > 0.7 considered highly correlated.

A random coefficients statistical model, which accounts for repeated measures taken from the same participants, was used to examine the relationship between MET values (dependant variable) and the activPAL counts using the equation development group only. This has been the predominant statistical method used to examine the relationship between MET values and accelerometer counts (Puyau et al., 2004, Treuth et al., 2004, Pfeiffer et al., 2006). The concordance correlation coefficient (CCC = the mixed model equivalent of  $R^2$ in linear regression) was used to assess the goodness of fit of the equation (Vonesh et al., 1996), and this was presented with the Standard Error Estimate (SEE). Age, height and weight were not found to contribute to the fit of the model. However, linear regression is highly sensitive to outliers, while it also assumes the prediction error is constant across all values of the predictor variables. It has also been highlighted that the relationship between energy expenditure and physical activity is non-linear. The use of decision boundaries and ROC analysis has been proposed as an alternative statistical procedure in this setting as it provides a method of determining accelerometer intensity thresholds with less misclassification than the previously employed regression formulae (Jago et al., 2007). Receiver operating characteristic (ROC) curves were used to calculate an area under the curve (AUC) and to define a threshold which optimises sensitivity and specificity (Zweig and Campbell, 1993). Sensitivity (true positive fraction) refers to the correct identification of an activity intensity (i.e. >3.0 METs) using a specific intensity threshold, and is calculated as Sensitivity = No. of true positive test results/ (No. of true positive test results + No. of false negative test results). Specificity (false positive fraction) refers to the correct identification of different activity intensity (i.e. <3.0 METs) using a specific intensity

threshold, and is calculated as Specificity = No. of false positive test results/ (No. of true negative test results + No. of false positive test results). The advantage of this method is that the sensitivity and specificity can be calculated for each activity intensity threshold, and the threshold with the optimum sensitivity and specificity can then be selected for use within the population. The intensity thresholds were developed using both mixed regression (MR) and ROC analysis. Both sets of thresholds were then cross-validated using ROC analysis on an independent group. Sensitivity, specificity and area under the curve were examined and interpreted (Swets, 1988), and the optimal value for moderate physical activity and vigorous physical activity were identified. All analyses were undertaken using PASW statistics.

A non-parametric Spearman correlation was performed to examine the concurrent validity of the count 15 s<sup>-1</sup> function of the activPAL with that of the ActiGraph. To examine the validity of the activPAL when classifying sitting, standing and stepping time, data from both the activPAL and the ActiGraph were compared with activity information recorded by the single observer. Agreement between both the activPAL and ActiGraph and direct observation was examined on a minute-by-minute basis. This involved observing the amount of time each participant spent in each activity, and examining how well this time agreed with the observed activity category. Percentage agreement, sensitivity and predictive values were calculated (Grant et al., 2006). Percentage agreement was defined as the agreement between all observed samples and activity monitoring samples ((number of observed samples which were correctly identified by activPAL or ActiGraph \* 100/total number of samples). Sensitivity was defined as the degree to which the activity monitors correctly detected the activity category ((number of observed samples which were correctly identified by activPAL or ActiGraph for each activity category \* 100)/total number of samples for each activity category). Predictive Value was defined as the level to which each activity monitor determined category agreed with the observed activity category ((number of matching samples between observed values and activPAL or ActiGraph for activPAL or ActiGraph activity category \* 100)/total number of samples for activPAL or ActiGraph activity category).

# 3.3. Results

# 3. 3. 1. Descriptive Information

Participants' mean age was 17.2 yrs. ( $\pm$  0.9), mean height was 1.7 m ( $\pm$  0.1), mean weight was 65.4 kg ( $\pm$  9.2) and the mean body mass index of the population was 23.2 kg·m<sup>-2</sup> ( $\pm$  2.8). There were no significant differences observed for age, height, weight or BMI between the development group and the cross-validation group.

## 3. 3. 2. Criterion Validity of the activPAL Activity Monitor

Table 3.1 describes the mean and SD of  $VO_2$ , MET scores, speeds and both activPAL and ActiGraph activity counts expressed in count·15 s<sup>-1</sup> from the development group. The chosen activities provided a wide range of accelerometer counts (activPAL Range: 0 to 14198 counts·15 sec<sup>-1</sup>; ActiGraph Range: 0 to 3378 counts·15 sec<sup>-1</sup>) and MET scores (Range: 0.74 to 14.95 METs).

<b>Table 3.1</b> Mean ( $\pm$ SDs) of VO <sub>2</sub> ml·kg <sup>-1</sup>	min <sup>-1</sup> , MET scores,	speed and both	1 activPAL a	and
ActiGraph activity counts.				

Activity	VO <sub>2</sub>	METs	activPAL	ActiGraph	Speed
	(ml·kg <sup>-1</sup> ·min <sup>-1</sup> )		counts ·15 s <sup>-1</sup>	counts ·15 s <sup>-1</sup>	(km/h)
			(n=15)	(n=15)	
Sitting	4.2 (1.2)	1.1 (0.2)	5 (8)	0 (1)	N/A
Standing	4.3 (1.2)	1.1 (0.2)	15 (22)	1 (2)	N/A
Slow Walk	11.0 (1.9)	3.0 (0.7)	3098 (858)	632 (174)	3.6 (0.4)
Brisk Walk	14.3 (2.1)	3.9 (0.8)	5011 (869)	940 (156)	4.9 (0.4)
Light Jogging	31.1 (4.5)	8.5 (1.9)	11086 (1624)	2368 (406)	7.3 (0.5)

The criterion validity of the activPAL was examined using  $VO_2$  as the criterion measure. The MR equation (N=15) for the activPAL was developed for this population to correspond to activity categories (Moderate = 3-5.99 METs; Vigorous = 6 METs or greater) that have been recommended for examination in previously published literature. The developed equation is:

$$METs = 0.971011 + 0.000677 * (counts \cdot 15 s^{-1})$$

The concordance correlation coefficient (CCC) for counts  $\cdot 15 \text{ s}^{-1}$  was identified as 0.93 (SEE = 1.20). When solving the equation, 2997 counts  $\cdot 15 \text{ s}^{-1}$  was the value for a moderate threshold of 3 METs, while 7428 counts  $\cdot 15 \text{ s}^{-1}$  was the value for vigorous threshold of 6 METs.

#### 3. 3. 3. Mixed Regression Analysis

The ability of the MR to predict MET values from activity counts in an independent group was examined. The correlation values from this analysis are presented in Table 3.2. The mean absolute difference between actual and predicted MET values for non-locomotor activities, locomotor activities and all activities are also included in Table 3.2.

**Table 3.2** Concordance correlation results comparing actual MET values with activPAL counts 15 s<sup>-1</sup> based predicted MET values for all activities, non-locomotor activities and locomotor activities in the cross-validation group.

	Actual	Predicted	Mean	SEE	r value
	Mean	Mean	Absolute		
			Difference		
All activities	4.19 (3.16)	4.02 (3.04)	1.32	0.86	0.93*
Non-locomotor Activities	$1.17 \pm 0.21$	$0.99 \pm 0.02$	0.92	0.27	0.25
Locomotor Activities	$5.92 \pm 2.74$	$5.75 \pm 2.5$	1.55	1.06	$0.87^{*}$

N=15; \* p < 0.01.

# 3. 3. 4. ROC Analysis

The development group data was examined using ROC analysis, and revealed an AUC of 0.98 for moderate physical activity and 0.99 for vigorous physical activity. For moderate physical activity, a threshold of 3329 counts  $15 \text{ s}^{-1}$  optimised sensitivity (0.93) and specificity (0.93), while a threshold of 8229 counts  $15 \text{ s}^{-1}$  optimised sensitivity (0.98) and specificity (0.97) for vigorous physical activity. These values would be considered highly accurate (Swets, 1988).

## 3. 3. 5. Cross-Validation of Developed Thresholds

The moderate physical activity and vigorous physical activity thresholds from both mixed regression analysis and ROC analysis were cross-validated, and the cross-validation results are presented in Table 3.3. Each threshold demonstrated high levels of sensitivity and specificity when cross-validated. As the AUC for both moderate physical activity thresholds and both vigorous physical activity thresholds were the same, sensitivity and specificity were summed and the threshold with the highest value was selected as the optimal threshold. An optimal threshold of 2997 counts  $15 \text{ s}^{-1}$  was identified for moderate physical activity, while an optimal threshold of 8229 counts  $15 \text{ s}^{-1}$  was identified for vigorous physical activity. The moderate physical activity threshold was developed using MR, and optimised sensitivity (95.7%) and specificity (94.5%). The vigorous physical activity threshold was developed using ROC analysis and optimised sensitivity (97.7%) and specificity (100%).

**Table 3.3** Cross-validation results for the activPAL for sensitivity and specificity values for activity intensity thresholds developed using both mixed regression analysis and receiver operating characteristic (ROC) analysis.

	Moderate Phys	sical Activity	Vigorous Phys	sical Activity
	Mixed	ROC	Mixed	ROC
	Regression		Regression	
Counts ·15 s <sup>-1</sup>	2997	3329	7428	8229
Sensitivity (%)	95.7	91.3	97.7	97.7
Specificity (%)	94.5	95.9	99.2	100
AUC	0.99	0.99	1.0	1.0

# 3. 3. 6. Concurrent Validity

There was a strong, positive relationship between the counts function of the activPAL and that of the ActiGraph (r = 0.96, p < 0.01) demonstrating very high concurrent validity between the two devices. Figure 3.1 presents the relationship between the count function of the activPAL and ActiGraph.



**Figure 3.1** Relationship between the activPAL activity monitor count function and the ActiGraph GT3X accelerometer across all activities.

# 3. 3. 7. activPAL and ActiGraph GT3X Sitting/Standing/Slow Walking Vs. Direct Observation

The results of the minute-by-minute analysis for levels of agreement, sensitivity and predictive values of the activPAL and the ActiGraph with direct observation for sitting, standing upright and slow walking are presented in Table 3.4. The overall levels of agreement between direct observation and activPAL for correct classification of sitting, standing and slow walking was 99.1%, while the overall agreement between direct observation and the ActiGraph was 66.7%. Additionally, a graphical representation of the activity counts min<sup>-1</sup> for ActiGraph recorded sitting, standing and light stepping activities is presented and compared with the < 100 counts min<sup>-1</sup> sedentary threshold in Figure 3.2.

 Table 3.4 Comparison of activPAL and ActiGraph determined sitting, standing and stepping with observed activity category.

	All activities	Sitting		Stand	ling	Slow Walking		
	Agreement	S %	PV %	S %	PV %	S %	PV %	
activPAL	99.1 %	100 %	100 %	98.1 %	100 %	99.2 %	100 %	
ActiGraph	66.7%	100%	100%	0%	0%	100%	100%	

S = Sensitivity; PV = Predictive Value.



(N=30)

**Figure 3.2** ActiGraph counts  $\cdot$  min<sup>-1</sup> directly observed sitting, standing and slow walking compared with the < 100 counts  $\cdot$  min<sup>-1</sup> sedentary threshold.

# 3.4. Discussion

The purpose of this chapter is to investigate the validity of the activPAL activity monitor as a measure of physical activity, while also determining a count to activity threshold for moderate to vigorous physical activity. The activPAL demonstrated high levels of criterion validity, identifying a mixed model equivalent to an  $R^2$  of 0.93 (SEE = 1.20) when

compared with METs across 7 different activity intensities. A threshold of 2997 and 8229 counts  $\cdot 15s^{-1}$  for moderate physical activity and vigorous physical activity respectively identified optimum levels of sensitivity and specificity after cross validation in an independent sample. The activity count function of the activPAL demonstrated high levels of concurrent validity with the ActiGraph accelerometer count function (r = 0.96, p < 0.01), while the activPAL was more accurate at distinguishing between sitting, standing and stepping than the sedentary thresholds employed when using the ActiGraph.

# 3. 4. 1. The Identification of the activPAL as a Valid Measure of Physical Activity

This study is the first study to develop count values that correspond to different activity intensities for the activPAL, which are typically used when examining physical activity in both observation and intervention research. Thus far, the literature on the use of the activPAL to examine physical activity is limited to sedentary and step-based measurement in laboratory (Grant et al., 2006, Ryan et al., 2006) and habitual settings (Chastin et al., 2009, Harrington et al., 2012). While the existing activPAL software provides an easy to understand output of steps and cadence, which have been compared to MET estimates, it has been suggested that the use of activity counts in the examination of free-living physical activity may be more applicable than the use of steps (Freedson et al., 1998, Harrington et al., 2011b). To date, only one study has attempted to validate the activPAL with a criterion measure (Harrington et al., 2011b). In this paper, Harrington et al. (2011b) highlighted that the relationship between counts and measured METs was stronger than the relationship between steps and measured METs (Harrington et al., 2011b). Similarly, the results presented in the present chapter indicate that activPAL counts are highly associated with MET scores in an adolescent female population while performing a range of everyday activities (CCC = 0.93; SEE = 1.20). Weak correlations were observed when the MR predicted MET values from activity counts for non-locomotor activities (r = 0.25). However, the activPAL separately and accurately distinguishes between sitting and standing using an inbuilt inclinometer, which is more likely to be employed when examining nonlocomotor activities in free-living investigations. Although the activPAL employs a separate mechanism to determine sitting/lying and standing activities, non-locomotor activities are

included to comply with recommendations for calibration and validation (Welk, 2005, Bassett Jr *et al.*, 2012). The MR predicted MET values from activity counts within 1.32 METs, while a strong and significant correlation existed between actual and predicted MET values across all activities (r = 0.93; p < 0.01). Similar to findings from other validation studies, these results suggest that the MR appears to be comparable when examining physical activity in this specific population (Freedson *et al.*, 1998, Treuth *et al.*, 2004, Welk *et al.*, 2007).

# 3. 4. 2. The use of both Mixed Regression Analysis and Receiver Operating Statistics

Methodological differences in estimating intensity thresholds may have a substantial effect on resulting values (Freedson et al., 2005). It has been suggested that the use of linear and non-linear regression equations when developing activity intensity thresholds have significant limitations (Jago et al., 2007). An alternate method for the development of activity intensity thresholds for use in accelerometer-based research has been developed (Jago et al., 2007) and implemented in studies with children and adolescents across a range of accelerometers (Welk et al., 2007, Esliger et al., 2011). Through attempting to maximise the AUC, the ROC method of determining intensity thresholds places an equal emphasis on the importance of both sensitivity and specificity (Welk et al., 2007). The use of ROC analysis and AUC in the development of intensity thresholds maximises the sensitivity and specificity in classifying MET values correctly and reduces the error of estimating the true intensity (Jago et al., 2007). In this chapter, intensity thresholds were developed using both MR and ROC analysis, and a decision on which threshold was recommended for use was based on the sensitivity, specificity and area under the curve after cross-validation. When cross validated with an independent sample, a MR determined threshold of 2997 counts 15  $s^{-1}$  was identified for moderate physical activity (Sens = 95.7; Spec = 94.5; AUC 0.99), while an ROC determined threshold of 8229 counts 15 s<sup>-1</sup> was identified for vigorous physical activity (Sens = 97.7; Spec = 100; AUC 1.0). The observed high levels of sensitivity, specificity and AUC support the use of the developed threshold values within this specific population.

#### 3. 4. 3. The Concurrent Validity of the activPAL Activity Monitor

To date, only one study has compared the concurrent validity of the step count function of the activPAL and the step count function of the ActiGraph GT1M to video recorded steps (Harrington *et al.*, 2011b), while no study has previously examined the concurrent validity of the activity count function of the activPAL with that of the ActiGraph. The findings of Harrington et al. (2011) identified that the activPAL step function was reasonably accurate at measuring moderate walking speeds, and was more accurate at measuring slow walking speeds when compared to the ActiGraph. The results presented in this chapter have identified that the activPAL is at least as effective in measuring locomotor activities as the uniaxial function of the ActiGraph.

# 3. 4. 4. Differences in activPAL and ActiGraph Determined Sitting and Standing Time

Until now, large-scale studies have utilised a sedentary count threshold (e.g. < 100 counts min<sup>-1</sup>) when using the ActiGraph to define sedentary or sitting time (Matthews *et al.*, 2008, Healy *et al.*, 2011b, Matthews *et al.*, 2012). Researchers have examined the effectiveness of using a range of count thresholds for the ActiGraph compared with the activPAL (criterion measure) for examining sedentary time in youth, and a sedentary threshold of <100 counts min<sup>-1</sup> has been suggested (Ridgers *et al.*, 2012). Although the use of such sedentary thresholds may be appropriate for specific sedentary variables or research questions - whereby standing is considered a sedentary activity - the ability of such devices to examine sedentary patterns and behaviours is limited. It has been highlighted that standing is a distinct behaviour from sitting and can confer its own physiological benefits (Hamilton *et al.*, 2007, Hamilton *et al.*, 2008, Owen *et al.*, 2010). This chapter has highlighted the accuracy of the < 100 counts min<sup>-1</sup> in estimating sitting/lying time (100%) and stepping time (100%). However, the inability of the ActiGraph to determine standing time has also been demonstrated (accuracy of 0%). These findings suggest that the use of a < 100 ActiGraph counts min<sup>-1</sup> in an adolescent female population will include time spent

standing still, and highlights a limitation to the use of the <100 ActiGraph counts min<sup>-1</sup> to determine sitting/lying patterns. This < 100 counts min<sup>-1</sup> threshold was applied to the ActiGraph vertical axis only as this threshold was developed using the vertical plane only from an older ActiGraph model. Findings of this study have also identified high levels of agreement between direct observation and the activPAL for sitting/lying (100%), standing (98.1%) and stepping (99.2%). Although the activPAL also records counts that can be examined using sedentary thresholds (similar to the ActiGraph), the activPAL employs an inbuilt inclinometer, which allows inclination of the thigh to be classified into sitting/lying or standing without the user resorting to using thresholds. The inclinometer function of the activPAL has previously been identified as both a valid and reliable measure of posture (Grant et al., 2006). Devices, such as the activPAL, have provided researchers with the capability of directly examining sitting/lying behaviours and standing behaviour, while also examining levels of physical activity. The use of a single activity monitor to distinguish between sitting/lying and standing time, to examine both of these behaviours and to examine physical activity behaviours in a habitual setting has the ability to provide detailed activity profiles in population-based research. The use of inclinometer-based activity monitors has substantial potential in the area of physical activity and health-related research, as results from these monitors enable researchers to make stronger and more accurate associations between activity variables (including sitting/lying, standing and physical activity behaviours) and health variables (Bassett Jr et al., 2010, Owen et al., 2010).

# 3. 4. 5. The Advantages of using an Inclinometer-Based Activity Monitor as a Measure of Physical Activity

This chapter aims to explore the effectiveness of the commonly employed ActiGraph (used in uniaxial mode) and the newer activPAL in measurement of physical activity and sedentary behaviour. The results provide evidence that the activPAL can be used to examine standing, moderate physical activity and vigorous physical activity, while also as a measure of sedentary behaviour (for which it was originally developed). However, to recommend the activPAL for physical activity measurement alone would be premature, as the present study is only the first to validate the activPAL count function and to create thresholds for moderate physical activity and vigorous physical activity, compared with over 2 decades of validation and calibration work on the ActiGraph. However, the limitations of the ActiGraph for measuring sedentary behaviour cannot be overlooked. Our findings, particularly data presented in Figure 3.2, highlight the inability of the < 100counts min<sup>-1</sup> sedentary threshold for the ActiGraph to differentiate between sitting and standing. The obvious approach would be to use both ActiGraph for measurement of physical activity and activPAL for the examination of sedentary behaviour. However, the potential cost and burden of wearing two devices on participants, especially children, cannot be overlooked as it may affect compliance. To our knowledge, only one study has examined levels of physical activity and sedentary behaviour using both the activPAL and ActiGraph devices (Martin et al., 2011). Three days of valid measurement, which is below the recommended measurement period (Trost et al., 2005), were required from each participant, and insufficient data was provided by 28% of participants (Martin et al., 2011). Published findings on the measurement of free-living physical activity in adolescents using the activPAL only have identified high levels of compliance (8% of participants providing less than 4 valid days of measurement) (Harrington et al., 2011a). Newer devices, such as the ActiGraph GT3X+, have incorporated the inclinometer functions into existing accelerometers. Although these devices may have the potential to measure inclination, further work needs to determine their validity and reliability. The potential of a device, such as the activPAL, which has now been validated for both activity and posture in an adolescent female population, to objectively examine physical activity, coupled with the objective examination of sedentary patterns and behaviours in free-living populations, is substantial and of great benefit in large-scale health-related research.

### 3. 4. 6. Strengths and Limitations

This is the first non-treadmill based validation study of the activity count function of the activPAL, which is becoming increasingly popular in physical activity and health-related research. A significant strength of this study is that we employed individualised RMR to normalise energy cost between participants for each activity, rather than the use of standard RMR values (Butte *et al.*, 2012). The use of over ground walking with self-pacing within a particular speed range, which is more effective in simulating real-life activity and reduced the clustering effect that is normally created by using specific speeds, is a significant

strength of this study (Bassett Jr *et al.*, 2012). This study also employs a criterion measures for energy expenditure (indirect calorimetry) while direct observation is used as a measure of posture and activity. The inclusion of sedentary activities and of a range of locomotor activity intensities (light, moderate and vigorous activity) is another strength of this study (Welk, 2005, Bassett Jr *et al.*, 2010). Another important strength of this study is the crossvalidation of the equation, which is employed to examine the ability of the MR to estimate MET values from accelerometer counts for the activPAL (Bassett Jr *et al.*, 2012). An additional strength of the study is the development and cross-validation of activity intensity thresholds which optimise sensitivity and specificity using ROC analysis and the AUC.

This study targeted a specific population, adolescent females, and results from this population may not be generalised to younger children, adolescent males or a wider adult population. High levels of physical inactivity have been observed in adolescent female populations (Troiano *et al.*, 2008), making them a population of great importance when examining activity behaviour, particularly when aiming to implement activity interventions to increase activity levels and decrease sedentary time (Harrington *et al.*, 2011a). The validation of the activPAL in larger and more variable samples is necessary. The threshold development protocol does not represent the full range of activities undertaken by a population, since it does not include weight-bearing and upper body activities (Welk, 2005, Bassett Jr *et al.*, 2012). Additionally, during cross validation, the use of different walking speeds over a specified period may be of greater benefit. This methodology may have examined the ability of the thresholds to detect small and quick change in activity intensity. Finally, the use of the low frequency extension may have an effect on the results.

# 3. 5. Conclusion:

The accurate and objective examination of physical activity and sedentary behaviours is critical when establishing links between activity behaviours and indices of health. The activPAL has been previously identified as a valid and reliable method of directly examining sedentary behaviours. However, the ability of the activPAL to examine physical activity patterns through the accelerometer count function has not been investigated. This chapter has highlighted high levels of criterion and concurrent validity demonstrated by the activPAL count function in an adolescent female population, and has also presented optimum thresholds for moderate physical activity and vigorous physical activity in this population. The chapter has also highlighted the ability of the activPAL to distinguish between sitting, standing and stepping time, while also identifying the limitation of the  $<100 \text{ counts} \cdot \text{min}^{-1}$  sedentary threshold employed by the ActiGraph. The findings of this chapter support the future use of the activPAL not only as a measure of sedentary behaviours, but also as a measure of physical activity in an adolescent female population.

# Chapter 4. The Measurement of Sedentary Patterns and Behaviours using the activPAL<sup>TM</sup> Professional Physical Activity Monitor.

(Published in Physiological Measurement on 31/10/12)
# 4.1. Introduction

# 4.1.1. Observed Associations between Sedentary Behaviours and Health

Over the last decade, research has highlighted the independent risk that sedentary behaviours have on health (Bey *et al.*, 2003, Hamilton *et al.*, 2004). There is growing evidence that excessive bouts of prolonged sitting have negative effects on indices of health, which is distinct from the lack of exercise (Hamilton *et al.*, 2004, Owen *et al.*, 2010). According to Hamilton et al. (2007), prolonged periods of sitting may result in a reduced activity of the enzyme lipoprotein lipase (Hamilton *et al.*, 2007). This enzyme regulates muscle triglyceride uptake, and its down-regulation could raise blood lipid levels and decrease high-density lipoprotein synthesis, both of which would have negative consequences for health. Additionally, prolonged sitting reduces the effects of insulin action, even when energy intake is similarly reduced (Stephens *et al.*, 2010). Supporting evidence for the negative health consequences of sedentary time in both children/adolescents and adults has been provided by epidemiological studies such as the AUSDIAB study (Healy *et al.*, 2008c), the NHANES study (Martinez-Gomez *et al.*, 2010).

#### 4. 1. 2. The Objective Assessment of Sedentary Behaviours

The relationships between activity and inactivity levels and health have been examined throughout history (Pate *et al.*, 2008), and there exists a wealth of evidence on the health benefits of physical activity and the negative effects of inactivity. Consequently, there is a necessity to examine patterns and levels of activity over the whole intensity spectrum, including sedentary, in free-living populations using accurate and reliable measures (Pate *et al.*, 2008). Historically, the accurate and reliable measurement of lifestyle activities has been difficult, with subjective measures, such as self-report, the most commonly employed method of assessment (Ward *et al.*, 2005). Self-reported sedentary time suffers from similar limitations as self-reported physical activity, as individuals across all ages have difficulty recalling such ubiquitous behaviours, primarily due to their intermittent nature (Slootmaker

et al., 2009). Additionally, self-reported sedentary time has predominantly focused on TV viewing time or other surrogate measures, which do not take into consideration all other sedentary patterns that contribute to daily activity (Bassett Jr et al., 2010). Recent technological developments, particularly in the area of wearable motion sensors, have provided reliable, unobtrusive and reasonably affordable methods of examining free-living activity behaviour (Troiano, 2006, Troiano et al., 2008). Such improvements have enabled large-scale objective examinations of free-living populations, such as The Avon Longitudinal Study of Parents and Children (ALSPAC) and the National Health and Nutrition Examination Survey (NHANES) in the United States (Matthews et al., 2008, Troiano et al., 2008).

# 4. 1. 3. Fundamental Limitations of Assessment of Sedentary Behaviours using an Accelerometer-Based Activity Monitor

Typically, physical activity measurement devices record accelerations, and proprietary algorithms calculate arbitrary units known as accelerometer/activity counts over a specified time period or epoch (e.g. 1 minute). Calibration studies are then completed to determine count-to-activity thresholds. These thresholds enable researchers to categorise specific activity intensities, such as moderate to vigorous physical activity, from the produced accelerometer counts (Welk, 2005). If the number of activity counts produced over a specific epoch exceeds the defined threshold, it is categorised as time spent in that specific activity intensity. Studies have begun to present information on sedentary levels using similar threshold methodologies, whereby time spent below a specified accelerometer count threshold (known as a sedentary threshold) is classified as time spent in a sedentary state. ActiGraph (ActiGraph LCC, Pensacola, FL) is a commonly used measurement device and sedentary behaviour has usually been examined using an accelerometer count thresholds of < 100 counts (Matthews et al., 2008) or < 200 counts (Mitchell et al., 2012), while < 150 counts per minute has recently been proposed as more appropriate threshold (Kozey-Keadle et al., 2011). Conclusions on sedentary time or sedentary patterns from using such thresholds must be interpreted cautiously (Owen et al., 2010, Lubans et al., 2011). The count function of motion sensors estimates sedentary time based on the lack of movement (Ridgers et al., 2012). Consequently, such devices cannot distinguish between

sitting time and standing time (standing time should not be considered a sedentary behaviour, as the term sedentary behaviour refers to activities that do not increase energy expenditure significantly above the resting level while in a sitting or lying position (Owen *et al.*, 2010, Tremblay *et al.*, 2012). Additionally, it has been hypothesised that different sedentary thresholds may be necessary for use in different populations (sex, age, race/ethnicity, body composition) (Owen *et al.*, 2010).

# 4. 1. 4. The Use of Inclinometer-Based Activity Monitors

The fundamental limitations of examining static behaviour using accelerometer-based motion sensors have highlighted the necessity for alternative accurate and reliable measures of sedentary activities (Bassett Jr *et al.*, 2010). The development of inclinometer-based motion sensors, such as the activPAL<sup>TM</sup> Professional Physical Activity Monitor (PAL Technologies Ltd., Glasgow, UK), has provided researchers with methods of objectively examining activity behaviours in free-living populations using postural orientation (Chastin and Granat, 2010). The use of such devices has been encouraged, particularly for studies examining sedentary behaviours in more detail (Bassett Jr *et al.*, 2010, Owen *et al.*, 2010). To date, studies have identified the activPAL as a valid measure of posture and motion in adults (Grant *et al.*, 2006, Godfrey *et al.*, 2007), sedentary behaviours in adults (Kozey-Keadle *et al.*, 2011), and step cadence in females (Harrington *et al.*, 2011b).

# 4. 1. 5. Existing Examinations of Inclinometer-Determined Sedentary Time

The most common method of analysing sedentary behaviour recorded by inclinometerbased activity monitors to date has been the calculation of total sedentary time. An alternative approach is that of Chastin and Granat (2010), who described a generic statistical model to examine sedentary patterns and behaviours (Chastin and Granat, 2010). Though this generic model provides information on total sedentary time and the distribution of the length of sedentary bouts (using the Gini index), it does not aid researchers in examining detailed information on sedentary behaviours (frequency and duration of sedentary bouts). The rich information provided by inclinometer-based motion sensors, can be examined and analysed to present valuable details of sedentary patterns and behaviours. Due to the benefits of using such a device, it is expected that inclinometerbased activity monitors will become increasingly popular in the examination of sedentary variables. However, to date, limited published information is available on methodologies employed when analysing inclinometer-determined sedentary variables in greater detail.

With the significant increase in efforts to examine sedentary patterns and behaviours in recent years, and the mounting evidence of its relationship with indices of health, there is a need to present practical and descriptive methodologies in the examination of inclinometer-based accelerometer output. Without discussions on standardised methodologies for the examination of sedentary patterns and behaviours, results obtained from different studies may not be comparable, as has been observed with objectively determined physical activity levels over the last number of decades (Westerterp, 1999, Ward *et al.*, 2005). The primary purpose of this chapter is to provide a detailed description of the methodologies used to quantify sedentary periods in a free-living environment using an inclinometer-based physical activity monitor. The secondary purpose of this chapter is to describe the methodologies used to process the resulting sedentary periods and to identify important sedentary patterns and behavioural variables across a measured day. Finally, this chapter will describe the habitual sedentary patterns and behaviours of a group of adolescent females using these methodologies.

#### 4.2. Methods

## 4. 2. 1. The activPAL Professional Physical Activity Monitor

A detailed description of the physical characteristics and of the wear protocol of the activPAL has been provided in Section 3. 2. 2. (Additional information only included in this section). The activPAL provides two types of information: 1) steps and activity counts, which can both be used to determine physical activity and 2) inclinometer information, used to determine posture (i.e. whether the wearer's thigh is horizontal, as in sitting or lying, or vertical as in standing or walking). At the end of data collection, activPAL data

was downloaded to a PC via a USB interface. This data was stored in a proprietary data format, and was processed so that sedentary data and accelerometry data was compressed into 15 second epochs.

#### 4. 2. 2. Data from activPAL Output

The activPAL proprietary software (activPAL<sup>TM</sup> Professional V5.9.1.1) was used to access the recorded data, and the epoch data for the entire week of recording was exported to a Microsoft Excel format file (Microsoft Corporation, Microsoft Excel 2010, One Microsoft Way, Redmond, WA, USA). The spreadsheet displayed the number of seconds that the subjects engaged in sitting/lying, standing and stepping for each 15 second epoch. These values were summed over the entire 24 hour day, and averaged to get the mean time spent sitting/lying, standing and stepping formance.

#### 4.2.3. Examination of Sedentary Periods

All recorded activPAL data was examined to ensure that a sufficient number of valid days of accelerometry recording were obtained from each participant. The protocol used for data examination and cleaning has previously been described by the authors (Harrington *et al.*, 2011a). Briefly, only data sets that provided 4 full days of accelerometer recording (including at least one weekend day) were processed for this analysis. For the purpose of this chapter, non-wear time was defined as 60 minutes or more of consecutive zero accelerometer counts between the hours of 7 am and 11 pm. A MATLAB® (The MathsWorks Inc., 3 Apple Drive, Natick, MA, 01760-2098, USA) custom software program was developed to examine the sedentary output of the activPAL Professional physical activity monitor. The developed MATLAB® program examined the sedentary function of the activPAL output file epoch-by-epoch. Firstly, the program binary coded each epoch. A sedentary epoch was categorised as an epoch spent entirely in sedentary activities (i.e. sitting/lying) (code = 1), while a non-sedentary epoch (i.e. an epoch with less than 15 seconds spent in sedentary activities (code = 0). The program then

examined sedentary periods, whereby one sedentary epoch identified the start of a sedentary period and the last consecutive sedentary epoch identified the end of a sedentary period. The MATLAB® computer program sequentially examined each daily activPAL file, and for each of the continuous sedentary periods provided the start time, the end time in hh:mm:ss format and the duration in bouts, minutes and seconds. The mean, standard deviation, maximum, minimum and quantity for both the sedentary activity bouts and the dynamic activity bouts were also calculated and presented. An adjustable time of day window (e.g. between 7 am and 11 pm) was also applied to the data to exclude certain out of hours information. This information was then written to a Microsoft Excel file.

#### 4. 2. 4. Examination of Sedentary Variables

1) The Microsoft Excel file containing the non-processed activPAL output by epoch was first examined to provide the total amount of time spent in sitting/lying, standing and stepping behaviour categories. This was achieved by summing all time spent sitting/lying, standing and stepping throughout a measured day.

2) A sedentary analysis Microsoft Excel template was used to categorise sedentary periods by specific duration. For the purpose of this chapter, the selected duration categories for examination were <1 min, 1-5 mins, 6 - 10 mins, 11 - 20 mins, 21 - 40 mins and > 40 mins. These categories were selected to provide a range of shorter sedentary periods, while also identifying longer uninterrupted sedentary periods (e.g. > 20 minutes in duration) to identify daily sedentary patterns. The information obtained from the MATLAB® generated Excel file was exported to the sedentary analysis Microsoft Excel template, where the number of sedentary periods in each category was calculated. Additionally, the total amount of time spent in sedentary periods in each sedentary category was calculated.

3) The number of sit/lie to stand transitions was classified as the number of times an individual moved from a sitting/lying posture to a standing posture throughout the day. This is equivalent to the total number of sedentary periods throughout the waking day, and was calculated by summing the sedentary periods in each category described in (2) above.

4) The identification of percentage non-bed time spent in sedentary activities of specific duration is necessary, as individuals have differing bed/sleep patterns. For these differing bed/sleep patterns, bed hours and waking hours were calculated. The authors first identified 7 am as the start time for each 24 hour measurement day for every participant. This time point was selected as the majority of subjects in our research group would not rise from bed before this time. The first registered non-sedentary epoch after 7 am was identified as rise time. The last registered non-sedentary epoch of the day which was followed by a long, uninterrupted sedentary period (> 2 hours) was identified as the time the subject went to bed. Bed hours were calculated as [time between 7 am and the first non-sedentary epoch] + [time between last non-sedentary epoch and the next 7 am time point]. Waking hours were then calculated by: 24 hours – Bed hours. Particularly short periods of bed hours were manually examined to identify possible breaks in bed hours (e.g. rise to use the toilet or to get a drink), and bed hours were recalculated to present total bed hours.

5) The output produced by the sedentary analysis Microsoft Excel template was examined between the hours of 8 - 9 am (the hour immediately before school) and 3 - 4 pm (the hour immediately after school) to demonstrate the utility of the output produced by an inclinometer-based motion sensor. The purpose of this analysis was to identify potential differences in sedentary patterns and behaviours between the two time periods. The analytical approaches presented here are similar to those presented by Harrington et al. (2011), and the findings supplement those presented (whereby different time periods are analysed). Sedentary variables were processed for these particular time periods, and were examined for normality. As data was identified as non-parametric, Wilcoxon Signed Rank tests were used to compare pre- and post-school findings.

# 4. 2. 5. Study Outline and Subject Selection

Data for this study was collected from adolescent females who lived in County Limerick in the mid-western region of Ireland. Five urban high schools in were selected to participate in this study. Subjects were eligible to participate in this study if they were female, aged between 15 and 18 years and had no illness or injury that may prevent them from participating in regular physical activity. Consent was obtained from each school principal prior to study commencement. With the school principal's agreement, a list of all eligible subjects within the school was obtained from the school administrators. A randomisation table was generated, and fifteen subjects from each school were randomly selected for participation in this study. All subjects provided written informed parental and participant consent. Approval for the research study was obtained from the University of Limerick research ethics committee.

#### 4. 2. 6. Subject Procedure

All subjects reported to their school gymnasium with completed parental consent and subject consent forms. Height and weight measures were recorded from each subject, and each was provided with an activPAL Professional physical activity monitor and a detailed demonstration of how to wear the device. Subjects were instructed to wear the accelerometer for 7 full days, 24 hours a day (including when sleeping) and were instructed to only remove the device when bathing, swimming or before any other water based activities. Subjects were asked to attach the activPAL on the midpoint of the anterior aspect of the right thigh using the provided hydro-gel adhesive PAL*stickie*. An elasticised tube grip bandage was also provided, and subjects were asked to wear this over the device to form an envelope for further protection.

# 4.3. Results

#### 4. 3. 1. Descriptive Information

From the randomly selected 75 subjects, 51 participated in the study. When accelerometer data was examined for number of valid days of recording, a further 7 subjects were removed due to insufficient data (less than 4 days of accelerometer recording or not including at least 1 weekend day). Data from 44 adolescent females is thus presented in this chapter. The mean age of the subjects that participated in this study was 15.9 ( $\pm$  0.9). Participant descriptive statistics are presented in Table 4.1.

N = 44	Mean (SD)	Range
Age (yrs.)	15.9 (0.9)	15 – 17 yrs.
Height (m)	1.6 (0.1)	1.5 - 1.7
Weight (kg)	58.0 (9.0)	42.4 - 80.7
BMI (kg/m <sup>2</sup> )	21.7 (2.9)	17.2 – 28.9

Table 4.1 Mean, standard deviation and range for subject's age, height, weight and body mass index.

# 4. 3. 2. Sitting/Lying, Standing and Stepping Output

Information for the activPAL output was examined to identify total time spent in specific postural positions and activities. The total number of steps accumulated over the measured day was also summed. The results of each of these variables are presented in Table 4.2.

**Table 4.2** Mean ( $\pm$ SD) and range of hours spent sedentary, standing and stepping per day and number of steps achieved per day.

N= 44	Mean (SD)	Range
Sedentary (hrs.)	18.86 (0.96)	16.62 - 20.97
Standing (hrs.)	3.35 (0.74)	1.74 – 4.94
Stepping (hrs.)	1.79 (0.46)	0.78 – 3.12
Steps	9462 (2674)	4483 - 17841

# 4. 3. 3. Bed Hours, Non-Bed Hours and Percentage of Waking Day Spent in Different Sedentary Categories.

The total sedentary time during waking hours was then examined. Waking hours were defined as "24 hours - Bed Hours". The rise time and bed time (time an individual went to bed) were identified from the MATLAB® computer program output as described in 4. 2. 4 and both bed hours and waking hours identified. As highlighted above, subjects spent an

average of 18.86 hrs. ( $\pm$  0.96) daily in total sedentary time. The average subject waking hours was 14.77 hrs. ( $\pm$  0.79), with an average of 9.23 hrs. ( $\pm$  0.79) in bed. Total waking sedentary time was calculated, identifying that subjects spent 9.63 hrs. ( $\pm$  1.1) in sedentary activities during waking hours. The total amount of time spent in specific bout durations was then calculated as a percentage of sedentary time during waking hours and as a percentage of total daily sedentary time spent in specific sedentary categories. These results are presented in Table 4.3.

Sedentary bout duration	% of non-bed sedentary	% of total daily
Categories (minutes) (N = 44)	time	sedentary time
<1	0.86 %	0.44 %
1-5	6.29 %	3.21 %
6 – 10	8.5 %	4.34 %
11 – 20	15.67 %	8 %
21 – 40	32.88 %	16.79 %
> 40	35.79 %	67.22 %
Total	100 %	100 %

 Table 4.3 The percentage of non-bed hours and percentage of waking sedentary hours

 spent in sedentary categories.

# 4. 3. 4. Sedentary Bouts, Sedentary Breaks and Mean Time Spent Daily in Specific Sedentary Categories

The mean number of sit/lie to stand transitions achieved by subjects per waking day was  $48.8 (\pm 9.80)$ . Table 4.4 describes the number of bouts and amount of time spent daily in sedentary periods of specific duration, while Figure 4.1 graphically represents this data.

**Table 4.4** Summary of sedentary periods and amount of waking time spent in each specific

 sedentary period for each sedentary duration presented as Mean (Standard Deviation).

N = 44	Daily Bouts	Daily Minutes
<1 minute	11 (4)	5 (1.6)
1 – 5 minutes	15 (5)	36.3 (11)
6 – 10 minutes	7 (2)	49.1 (12.7)
11 – 20 minutes	6 (2)	90.5 (22.5)
20 – 40 minutes	7 (2)	190 (49.5)
>40 minutes	3 (1)	207.3 (55.8)
Total	49 (10)	578.3 (9.4)



Data presented as mean + standard deviation.

Figure 4.1 Mean number of sedentary periods per day and mean total amount of time spent in each sedentary period per day.

# 4. 3. 5. Practical Application: A Comparison of Sedentary Patterns and Behaviours during the Hour Immediately Before and Immediately After School

The hours immediately before school and immediately after school were examined for all subjects on school days only. The total amount of time spent sedentary during the hour immediately before school was 30.1 minutes, while total time spent sedentary during the hour immediately after school was 36.47 minutes. The distribution of different sedentary bout durations during the hours immediately before and after school are presented in Figure 4.2.





Figure 4.2 Patterns of sedentary bout categories during the hour immediately before school (8 - 9 am) and the hour immediately after school (3 - 4 pm).

The number of shorter sedentary bouts (< 20 minutes) and the number prolonged sedentary periods (> 20 minutes) were examined for the two time periods. Significantly more bouts of < 20 minutes in duration were accumulated immediately before school (2.75  $\pm$  2.04) than immediately after school (2.3  $\pm$  2.15), while significantly more bouts of > 20 minutes in duration accumulated immediately after school (0.6  $\pm$  0.59) than immediately before school (0.4  $\pm$  0.55). This information is presented in Figure 4.3.



(\* p < 0.05; \*\* p < 0.01; Wilcoxon Signed Rank Test; N = 44

Figure 4.3 Sedentary periods of a shorter (< 20 minutes) and longer (> 20 minutes) duration before and after school.

# 4.4. Discussion

A recently published "Sedentary Behaviour Research Strategy" has identified the precise measurement of prolonged periods of sitting as a key phase when researching the population health science of sedentary behaviour. (Owen *et al.*, 2010). As such, there is a need for researchers to develop methods of analysing sedentary output from inclinometer-based motion sensors. Standardised methods of data analysis have also been identified as one of the greatest difficulties in accelerometry research, as comparability of research findings between studies is extremely difficult (Masse *et al.*, 2005, Ward *et al.*, 2005, Healy *et al.*, 2008c). This chapter has described and presented methodologies that may be employed when examining free-living sedentary patterns and behaviours using an inclinometer-based activity monitor. This information has the potential to enable researchers to examine inclinometer-based activity monitor output in a standardised manner, and to present and compare their findings with similar research studies.

#### 4.4.1. The Examination of Sedentary Time

It has been suggested that the examination of sedentary time during waking hours alone may be considered inconsistent, as sleeping periods make up a large proportion of total daily sedentary time (Chastin and Granat, 2010). The majority of large-scale studies, which have reported sedentary time and patterns in this manner, have used accelerometers that were not worn during night/sleep hours, such as the ActiGraph GT1M/GT3X accelerometer (Healy et al., 2007, Matthews et al., 2008, Healy et al., 2011b). A difficulty with not wearing the accelerometer at night is that subjects may potentially fail to re-apply the device immediately after waking or still be awake and sedentary (using laptop or reading for example) for a period of time once the participant is in bed and has removed the device. The use of a device that is an accurate measure of sitting/lying to standing movements that can also be worn throughout the night reduces the uncertainties associated with the assumption of waking hours identified by Chastin and Granat (2009), and allows for a more accurate estimation of waking time. This chapter has examined sedentary patterns and behaviours recorded across an entire 24 hour period using an inclinometer-based activity monitor. The identification of the first rise time and last movements of each individual enables the examination of sedentary patterns including time spent in bed as suggested (Chastin and Granat, 2010), while also allowing researchers to normalise sedentary patterns by presenting results as a percentage of waking sedentary time. The high compliance rates observed in this study (86% of subjects providing sufficient activPAL data) would suggest that wearing the device throughout the night does not deter subjects from wearing the device for recommended measurement period. Similar levels of compliance have been observed in adult studies which have employed the activPAL (Chastin et al., 2009, Dall and Kerr, 2010). Large scale epidemiological studies in children and adolescents that have employed devices that are removed during night hours appear to have provided consistently lower compliance rates (Sardinha et al., 2008, Troiano et al., 2008, Carson and Janssen, 2011). These findings suggest that the activPAL is an acceptable measurement device in an adolescent female population, and should be considered as an acceptable method of monitoring habitual activity in this population. Further examination of the compliance rates of the activPAL is merited to ensure these high levels are apparent across people of different ages, sex and ethnicity.

# 4. 4. 2. The Frequency of Sedentary Periods

Unsurprisingly, data presented in Figure 4.2 (for waking sedentary patterns and behaviours) resembles a power law distribution (larger number of short periods accounts for a small amount of time, while a small amount of long periods accounts for a large amount of time). Similar observations for the distribution of sedentary patterns and behaviours have been highlighted (Chastin and Granat, 2010), although the distribution included here does not include the sedentary hours spent in bed. Discrepancies with the power law distribution of the < 1 minute category are likely due to the duration of the sedentary category rather than the activity behaviour. Sedentary bouts of less than 1 minute are possibly less likely than sedentary bouts of 1-5 minutes, as few activities of daily living would require a change in postural position from standing to sitting and back to standing within a one minute time period. In future, the authors suggest that this category be pooled with sedentary periods of 1-5 minutes in duration, unless specific shorter bouts are required for analysis.

# 4. 4. 3. The Importance of a Direct Measure of Sedentary Time

The ability for physical activity and health-related researchers to identify strong associations and relationships between health indices and patterns and behaviours of particular types of activity (such as sedentary, light activity or moderate to vigorous activity) is dependent on accurate measurement methodologies (Reilly *et al.*, 2008, Healy *et al.*, 2011b) and an ability to interrogate our resulting information in the most efficient and effective manner (Troiano, 2005, Welk, 2005). To make associations between free-living sedentary behaviours and the inactivity physiology work completed by Hamilton and colleagues, researchers must 1) examine sedentary behaviour using the correct methods and 2) interpret the findings in the correct manner. Thus far, inactivity physiologists have highlighted the potentially deleterious effect that high levels of sedentary time may have on indices of health (Bey *et al.*, 2003, Hamilton *et al.*, 2004). However, epidemiological studies that have confirmed these observations in large-scale free-living examinations have measured the lack of ambulation rather than sedentary time (Ridgers *et al.*, 2012) or used self-report methods. The term sedentary refers to activities that do not significantly increase energy expenditure above a resting rate while in a sitting or lying position (Owen *et* 

*al.*, 2010, Tremblay *et al.*, 2012). Existing objective epidemiological evidence does not distinguish between sedentary time, standing time and low levels of ambulation, and consequently has not directly measured sedentary activities. Furthermore, the presentation of sedentary findings has previously focused on total sedentary time (Healy *et al.*, 2008c, Matthews *et al.*, 2008, Healy *et al.*, 2011b). The presentation of this total sedentary time may allow significant details of sedentary patterns and behaviours to go undetected. Healy *et al.* (2008a) progressed the examination of total sedentary time by providing evidence that more interruptions in sedentary time were positively associated with metabolic risk variables (Healy *et al.*, 2008a), yet the association between prolonged sedentary bouts and health indices has not been examined. The use of a robust statistical method of examination of sedentary behaviours, although adding significantly to the field, may also overlook significant detail of sedentary patterns and behaviours. Through the detailed examination of duration of sedentary periods, such as those presented in this chapter, researchers can examine whether sedentary bouts of differing duration have an effect on health risk factors.

#### 4. 4. 4. The Identification of Sedentary Time

It is important to identify sedentary environments and determinants, which promote uninterrupted sedentary periods, as potential areas for intervention (Owen *et al.*, 2010). The methods of analysis presented in this chapter enable the examination of sedentary patterns and behaviours over the full 24 hour day, which can then be subdivided into particular periods of the day (i.e. work, school or home time). Through applying the methodologies reported here, we highlight that this adolescent female population spent significantly more time sedentary during the hour immediately post school than the hour immediately preschool, and this sedentary time was accumulated in longer uninterrupted bouts. Researchers have subjectively identified how children and adolescent populations have become increasingly sedentary in their commute to and from school (Jago and Baranowski, 2004, Faulkner *et al.*, 2009), and have suggested that the introduction of initiatives to increase active transportation (such as walking buses) may increase physical activity and subsequently decrease time spent in sedentary behaviours (Tudor-Locke *et al.*, 2001). The findings presented here indicate that the post-school hour may be a potential target for active transport interventions to reduce sedentary behaviours in this population. Furthermore, we have also recently highlighted the undesirable sedentary profiles of an adolescent female cohort during school hours, with the school setting appearing to promote prolonged sedentary periods. The findings have provided objectively measured sedentary data, which could be used to inform school-based sedentary interventions to decrease prolonged sedentary periods (Harrington *et al.*, 2011a). The measurement of total sedentary time or of sedentary period distribution would not reveal these specific behaviours. Collectively, these findings highlight the value of examining the duration of sedentary behaviour changes over the course of the day.

## 4. 4. 5. Summary

The development of standardised methodologies for examining inclinometer-based sedentary activity at a relatively early stage of the direct measurement of sedentary patterns and behaviours may encourage and enable researchers to collaborate and compare findings from different research studies. The ability of researchers to collate directly examined sedentary time and health indices may result in the identification of statistically stronger associations between risk factors and/or health outcomes and sedentary patterns and behaviours. The methodologies presented herein are an example of a protocol for measuring and analysing sedentary levels and patterns using a particular device. This article adds to the information available on the analysis of the output from the activPAL inclinometer function and provides detailed descriptions of the methodology used to examine a wide range of sedentary variables in free-living environments. The methodologies employed in this chapter are transferable within populations with which the inclinometer function of the activPAL has been validated.

# 4.5. Conclusion

This chapter has provided a detailed description of methodologies used to describe and present a wide range of objectively measured sedentary patterns and behaviours using an inclinometer-based motion sensor. The techniques described and presented in this chapter enable the detailed examination of an individual's sedentary profile across a 24 hour period, across a specific time period during the day and during non-bed hours. Furthermore, the techniques used in this chapter can be applied to all populations in which the inclinometer function of the activPAL Professional Physical Activity Monitor has been validated. The high compliance rates of successful data collection observed in this analysis (86%) would suggest that this device is an acceptable method of examining habitual physical activity and sedentary behaviour in large-scale epidemiological studies. Before large-scale studies are undertaken, it is important to discuss and develop inclinometer-based activity monitor output analysis methods, such as those described in this chapter, which will enable the comparison of sedentary variables from future population-based examinations with each other.

Chapter 5. Associations between Sedentary Time, Light Intensity Physical Activity and Body Composition in Adolescent Females.

# 5.1. Introduction:

# 5. 1. 1. The Significance of Excess Adiposity

Excess adiposity is a physiological state that occurs through an imbalance in energy metabolism, whereby caloric energy intake is increased, energy expenditure is decreased or both occur simultaneously (Katzmarzyk et al., 2003, Reilly, 2005). There is incontrovertible evidence of the associations between obesity and coronary heart disease, type 2 diabetes, hypertension, dyslipidemia, site-specific cancers and all-cause mortality in adults (Malnick and Knobler, 2006, Flegal et al., 2007). Also, high levels of excess adiposity are apparent in many developed nations (Do Carmo et al., 2008, Flegal et al., 2010, Ng et al., 2011, Dalton et al., 2012). The associations between obesity and non-communicable diseases in childhood and adolescence are less well understood, as diseases such as coronary heart disease, type 2 diabetes and cancers tend not to manifest until adulthood (Hallal et al., 2006, Andersen et al., 2011). However, evidence suggests that excess adiposity tracks from childhood and adolescence throughout the lifespan, hence increasing the risk of onset of noncommunicable diseases in adulthood (Singh et al., 2008). This has enormous public health significance, as high levels of overweight and obesity have been observed in adolescent populations (Olds et al., 2009, Ogden and Carroll, 2010, Stamatakis et al., 2010, Janssen et al., 2011), while adolescents appear to be particularly susceptible to weight gain (Kimm et al., 2002, Rogol et al., 2002, Kimm et al., 2005).

# 5. 1. 2. Associations between Light Intensity Physical Activity and Adiposity

A large and consistent body of evidence has shown that higher levels of physical activity protect children and adolescents from the development of overweight and obesity (Jiménez-Pavón *et al.*, 2010), and that moderate to vigorous physical activity is an independent predictor of adiposity in children and adolescents (Strong *et al.*, 2005). To date, the majority of physical activity research has focused on intensities at the higher end of the activity intensity continuum, predominantly moderate to vigorous physical activity (Pate *et al.*, 2008). However, moderate to vigorous physical activity accounts for a very small

amount of total daily energy expenditure in adolescence (Males: 12-15 yrs.: 45.3 (±3.4) min·d<sup>-1</sup>; 16-19 yrs.: 32.7 (±2.2) min·d<sup>-1</sup>; Females: 12-15 yrs.: 24.6 (±1.8) min·d<sup>-1</sup>; 15-19 yrs.: 19.6 ( $\pm 2.4$ ) min·d<sup>-1</sup>) (Troiano *et al.*, 2008). As a result, it is unlikely that moderate to vigorous physical activity is the dominant determinant of variability in daily energy expenditure (Levine, 2002, Donahoo et al., 2004, Tremblay et al., 2007). This has prompted researchers to examine the relationship between lower intensity activities and indices of health. Non-exercise activity thermogenesis (NEAT) is the energy expended throughout activities of daily living (Levine et al., 2006), and includes all activities of a light intensity. It has been suggested that NEAT is the dominant determinant of variability in total daily energy expenditure, and therefore may have a greater effect on adiposity than moderate to vigorous physical activity (Levine et al., 1999, Levine, 2002, Tremblay et al., 2007). Objective evidence from large-scale adult populations has also identified positive association between increased levels of light intensity physical activity and 2h fasting blood glucose (Healy et al., 2007). Unfortunately, limited objective light intensity physical activity data from epidemiological studies is available due to difficulties in measuring such ubiquitous activities (Shephard, 2003, Bassett Jr et al., 2010).

#### 5. 1. 3. The Sedentary Behaviour Paradigm and Adiposity

The examination of sedentary behaviour has also emerged as a relatively new area in physical activity and health-related research (Owen *et al.*, 2010, Dunstan *et al.*, 2011). Sedentary behaviours refer to waking behaviours spent in a sitting/lying position that require an energy expenditure of <1.5 metabolic equivalents (Dunstan *et al.*, 2011, Marshall and Ramirez, 2011, Tremblay *et al.*, 2012). Recent evidence suggests that sedentary behaviours have a unique negative effect on indices of health including obesity, type 2 diabetes and cardiovascular disease risk factors in adult populations (Hamilton *et al.*, 2007, Owen *et al.*, 2010, Dunstan *et al.*, 2011). Unfortunately, the effect of sedentary behaviour on indices of health in child and adolescent populations is less well understood. Mixed evidence on the effect of sedentary behaviours have been shown to have a negative effect on insulin resistance, blood pressure and cardiovascular risk in child and adolescent populations, but these studies failed to examine whether the effect was independent of

moderate to vigorous physical activity (Sardinha *et al.*, 2008, Martinez-Gomez *et al.*, 2009, Martinez-Gomez *et al.*, 2010). No associations have been identified between sedentary time and indices of health that are independent of moderate to vigorous physical activity (Mitchell *et al.*, 2009, Carson and Janssen, 2011, Martinez-Gomez *et al.*, 2011a).

# 5. 1. 4. The Measurement of Sedentary Time and Light Intensity Physical Activity

When examining activity variables, the use of accurate and reliable measures is critical, particularly when making associations with indices of health (Loprinzi and Cardinal, 2011, Intille et al., 2012). To date, the majority of epidemiological studies have examined sedentary behaviours using surrogate measures, such as self-reported TV viewing time (Pate et al., 2008). Significant limitations have been highlighted with this measure (Pate et al., 2008, Bassett Jr et al., 2010, Lubans et al., 2011). The validity of self-reported measures of sedentary behaviour remains largely untested. Such measures are also susceptible to recall and reporting biases (Lubans et al., 2011). Furthermore, surrogate measures such as TV viewing and computer use do not capture all sedentary activities, and provide no information on the sedentary patterns or the time of day that most sedentariness occurs (Bassett Jr et al., 2010, Owen et al., 2010). More recently, studies have employed accelerometers as a measure of sedentary behaviour (Pate et al., 2008, Bassett Jr et al., 2010, Marshall and Ramirez, 2011). Although the use of accelerometers as a measure of sedentary behaviour has greater reliability and validity than self-report (Owen et al., 2010, Lubans et al., 2011), this measure relies on the lack of ambulation to quantify sedentary levels (Ridgers et al., 2012). Typically, physical activity measurement devices record accelerations, and proprietary algorithms compute arbitrary units known as counts over a specified time period or epoch (e.g. 1 minute). A predetermined sedentary threshold is then used to examine sedentary time, whereby time spent below this threshold is included as sedentary time. Consequently, some non-sedentary activities, such as standing and slow walking, can be misclassified as sedentary activities (Kozey et al., 2010a). This misclassification not only results in discrepancies in the quantification of sedentary time, but also in that of light intensity physical activity, as light intensity activities are being included as sedentary activities (Hart et al., 2011). As a result, the use of inclinometer-based activity monitors, which directly examine the inclination/posture of the body, has been recommended when examining the relationships between specific features of sedentary time (Bassett Jr *et al.*, 2010).

The purpose of this chapter is to examine the associations between low intensity activity variables, such as sedentary time and light intensity physical activity, and body composition in a cohort of adolescent females using an inclinometer-based activity monitor. Any relationships between directly measured sedentary time, standing time, light intensity physical activity, prolonged sedentary periods, and moderate to vigorous physical activity will be explored.

# 5.2. Methods:

The aim of this cross-sectional analysis was to examine free-living physical activity and sedentary behaviours, also known as lifestyle physical activities, using an inclinometerbased activity monitor and to compare these findings with measures of adiposity in a population of adolescent females. The sample used in the present analysis was pooled from three studies undertaken in the University of Limerick. Data was collected from 7 urban and 6 rural high schools between 2009 and 2011. Specifically, data from study 1 was collected as part of a cross-sectional analysis of lifestyle physical activities (n=87), data from study 2 was collected as baseline data from a school-based physical activity intervention programme (n=128) and data from study 3 was collected from another cross-sectional analysis of lifestyle physical activities (n=76). The recruitment strategy for study 1 and study 2 have previously been described (Harrington *et al.*, 2011a), while recruitment for study 3 will be described in detail in Chapter 6. Full ethical approval for each study was provided by the institutional research ethics committee.

A total of 291 students were approached to participate in the three different studies. From the three studies, 222 students provided written informed participant and parental consent and participated in the full test days. Due to incomplete information from the inclinometer-based activity monitor (explained in Section 5. 2. 3), 27 sets of data were excluded from this analysis. This resulted in a total of 195 sets of data included in this analysis.

#### 5. 2. 1. Activity Measurement:

The inclinometer-based activity monitor employed in this research was the activPAL<sup>TM</sup> Professional Physical Activity Monitor (PAL Technologies Ltd., Glasgow, UK). A detailed description of the activPAL and the wear protocol employed in this research has been provided in Section 2. 10 and Section 4. 2. 1.

## 5. 2. 2. Measurement Protocol:

Participants were brought to the school gymnasium, where anthropometric measures were obtained from each participant. Participants wore light clothing (vest top and tracksuit pants) and removed their shoes and socks before height and weight was measured. Height was measured to the nearest 0.25 cm using a portable wall stadiometer (Seca model 214, Seca Ltd., Birmingham, UK). Body weight was measured to the nearest 0.01 Kg using a portable electronic scale (Seca model 77, Seca Ltd., Birmingham, UK). Body Mass Index (BMI) was calculated by Weight/Height<sup>2</sup>. All BMI scores were also converted to BMI percentiles based on their age and sex in accordance with CDC recommendations (Kuczmarski et al., 2002). Once height and weight had been measured, participants were asked to stand in the anatomical position. Skin-fold measurements were then obtained from 4 sites according to the skinfold protocol of the International Society for the Advancement of Kinanthropometry (ISAK) (Marfell-Jones et al., 2006). The 4 sites selected for measurements were the biceps, triceps, subscapular and iliac crest. First, body landmarks were identified and marked using an extensible metric tape and a hypo-allergic pen. Skin-fold thickness was measured to the nearest 0.25cm using a Harpenden skinfold calliper (Cranlea & Co, Birmingham, UK). Each skinfold measure was taken 3 times (but not consecutively) on the right side of the body, and the median value of the three measures was used (Marfell-Jones et al., 2006). The four skinfold measures were then summed.

Participants were then provided with their activPAL activity monitors, and were given a full demonstration of how the activPAL was to be attached, when the device was to be worn and what to do if the device was misplaced. For the purpose of this study, participants were instructed to wear the device on their right thigh only, to ensure uniformity across all data collection. After 7 days, the main investigator returned to participating schools, and the activPAL activity monitors were collected. The activity information was downloaded to the main investigator's PC using the activPAL interface program, and data files were exported to a Microsoft Excel file format (Microsoft Corporation, Microsoft Excel 2010, One Microsoft Way, Redmond, WA, USA) for further analysis.

## 5. 2. 3. Data Processing:

Although the optimal wear duration for activity monitors in free living environments remains contentious (Heil *et al.*, 2012), the current and widely used best practise for activity monitoring of free-living physical activity suggests that a minimum of 4 valid days of activity recording (including one weekend day) are obtained from each participant (Trost *et al.*, 2005). It has been recommended that a wear time of 10 hours is necessary for a day to be considered valid when using accelerometer-based devices (Masse *et al.*, 2005). The optimal wear time for the activPAL activity monitor has not yet been identified. For the purpose of this analysis, a valid day was classified as a measured day with  $\leq$  4 hours non-wear time during waking hours.

#### 5. 2. 3. 1. Identification of Waking Hours:

Daily non-wear time was examined from the produced Excel output files using the following procedures. First, daily waking hours were examined for each participant. The examination of waking hours has been previously described (Section 4. 2. 2). Briefly, the first change in inclination from sitting/lying to standing after 7 am was identified as a participants "rise time". The last change in inclination from standing to sitting/lying which was followed by a prolonged sedentary period (> 2 hours in duration) after 11 pm was identified as "bed time". The time between "rise time" and "bed time" was identified as waking hours. The identified waking hours were then examined for non-wear time.

# 5. 2. 3. 2. Identification of Non-Wear Time:

A non-wear period was defined as a period with  $\geq 60$  minutes of consecutive zero activity counts from the Excel output file. This method for identifying periods of non-wear time is consistent with free-living data reduction methodologies (Matthews *et al.*, 2008, Rowlands *et al.*, 2008, Healy *et al.*, 2011b). The non-wear periods for each day were summed, and all measurement days with greater than 4 hours of non-wear time during waking hours were removed from the analysis. Weekly activity information for each participant was then examined. If a participant did not provide 4 valid days of activity recording which included at least 1 weekend day, they were removed from all further analysis. For all remaining participant information, the daily non-wear time was summed, and the 24 hour measured day was adjusted (24 hours – non-wear time).

# 5. 2. 4. Activity Variables:

The output files provided 1) the time of each epoch, 2) the number of steps taken during each epoch, 3) the number of seconds per epoch spent sitting/lying, the number of seconds per epoch spent standing and the number or seconds per epoch spent stepping and 4) the accrued accelerometer counts per epoch. As highlighted in Section 4. 2. 4, to examine daily activity patterns, a 24 hour measurement period had to be defined. For this analysis, the 24 hour measurement period was defined from 7 am - 7 am of the following day.

The output files were examined to produce daily activity variables which include sedentary time, standing time, light intensity physical activity excluding standing time, light intensity physical activity with standing time included, moderate to vigorous physical activity and prolonged sedentary periods. Sedentary time and standing time was calculated by summing the total number of seconds spent in sitting/lying and standing postures over the 24 hour measurement period. For moderate to vigorous physical activity, a threshold of 2997 counts 15 sec<sup>-1</sup> was used to estimate metabolic equivalents for each 15 second period (Section 3. 3. 5). Moderate to vigorous physical activity was defined as  $\geq$  3 metabolic equivalents. This threshold was developed using mixed model regression based on data from a sample of 15-18 year old females. If an epoch with activity counts greater than 2997

counts  $\cdot$ 15 sec<sup>-1</sup> was observed, this was identified as time spent in moderate to vigorous physical activity. The amount of time spent in moderate to vigorous physical activity was then summed over the 24 hour measurement period. Light intensity physical activity excluding standing time (excl. standing) was then calculated as; light intensity physical activity = [24 hours – (sedentary time + standing time + moderate to vigorous physical activity)]. Light intensity physical activity including standing time (incl. standing) was calculated by summing both standing time and light intensity physical activity variables.

The Excel output file was then examined using a MATLAB® computer program to identify sedentary bout durations. This program examined each sedentary epoch throughout the measured day. Prolonged sedentary periods were defined as sedentary periods lasting longer than 30 minutes in duration (Carson and Janssen, 2011, Mitchell *et al.*, 2012). The methodological procedures for this program are described in detail in Section 4. 2. 3. Briefly, the start of a sedentary period was identified as an epoch that had a full 15 seconds spent in a sitting/lying position. The end of a sedentary period was identified as the next sedentary epoch that did not have a full 15 seconds spent in a sitting/lying position. The duration between the start and end of the sedentary period was then calculated. The time spent in prolonged sedentary periods was then summed for each measured day.

As the identified non-wear time was categorised as sedentary time in this analysis, sedentary time was adjusted by subtracting the non-wear values from the sedentary time value. Finally, total daily wear time was calculated by subtracted non-wear time from the total measurement period (24 hours), and each variable was then divided by the total daily wear time to derive the percentage of time spent in each activity variable.

#### 5. 2. 5. Statistical Analysis:

Descriptive statistics were calculated, and are presented as mean (±SD), frequencies or percentages where appropriate. All variables were examined for normality using skewness and kurtosis statistics (Kolmogorov-Smirnov test) and through visual inspection of histograms before additional statistical analysis were undertaken. The natural logarithm was applied to BMI, sum of skinfolds, standing time, light intensity physical activity, moderate

to vigorous physical activity and prolonged sedentary periods to account for skewness. The level at which the threshold for p was set was p < 0.05 throughout this chapter.

Partial correlations, controlling for age, were used to examine the relationships between sedentary time, standing time, light intensity physical activity (both including and excluding standing time), moderate to vigorous physical activity and prolonged sedentary periods and each of the measures of adiposity (BMI,  $\Sigma$  Skinfolds). Partial correlations, controlling for age and body composition (sum of skinfolds), were also used to examine the relationships between sedentary time, standing time, light intensity physical activity (incl. and excl. standing), moderate to vigorous physical activity and prolonged sedentary periods.

Multiple linear regression models were used to examine the relationships between sedentary time, standing time, light intensity physical activity (excl. standing), light intensity physical activity (incl. standing) and prolonged sedentary periods and body composition measures controlling for age. The two models assessed:

- 1. Whether activity variables were associated with BMI and  $\Sigma$  Skinfolds after correcting for age only.
- 2. Whether activity variables were associated with BMI and  $\Sigma$  Skinfolds after correcting for both age and moderate to vigorous physical activity.

Retrospective power estimation calculations were completed on all statistically significant correlations from the multiple linear regression models. This was completed to examine the probability of committing a type 2 statistical error, or not rejecting the null hypothesis when the null hypothesis is false.

Binary logistic regression models were then employed to predict the highest quartile of BMI and the highest quartile of  $\Sigma$  Skinfolds using quartile groups of the activity variables; sedentary time, standing time, light intensity physical activity (incl. and excl. standing time) and prolonged sedentary periods. The first model corrected for age only, while the second model corrected for both age and moderate to vigorous physical activity. Odds ratios with associated 95% confidence intervals are produced by the models. All analyses were undertaken using IBM SPSS Statistics v. 20 (Armonk, New York, USA).

# 5. 3. Results:

.

The descriptive characteristics of this sample are presented in Table 5.1.

**Table 5.1** Descriptive characteristics of the population. Values presented as Mean ( $\pm$ SD) and Range.

	Mean	Range
	(±SD)	
Age (yrs.)	15.7 (0.9)	13.1 – 18.7
Height (m)	1.64 (0.06)	1.46 – 1.82
Weight (kg)	60.8 (12.0)	40.7 – 115.1
<b>BMI</b> (kg/m <sup>2</sup> )	22.7 (4.2)	15.4 – 41.3
Sum of Skinfolds (mm)	68.6 (30.11)	26.0 - 207.1
Sedentary Time (%)	78.8 (4.4)	67.1 – 89.1
Standing Time (%)	14.0 (3.3)	6.8 – 26.0
Light Intensity Physical Activity incl. Stand (%)	17.5 (3.8)	8.8 – 30.1
Light Intensity Physical Activity excl. Stand (%)	3.4 (0.9)	1.4 – 7.3
Moderate to Vigorous Physical Activity (%)	3.7 (1.4)	1.1 – 8.6
Prolonged Sedentary Periods (%)	22.9 (6.2)	5.2 - 45.6

N = 195: All physical activity variables are presented as a percentage of total waking hours.

When the number of valid days of accelerometer data was examined, 14.4 % of the sample provided 4 valid days, 71.8 % provided 5 valid days, while 13.8 % of participants provided 6 full days of accelerometry data Table 5.2.

No. of Valid Days	<b>No. of Participants</b> ( $N = 195$ )	<b>Percentage</b> $(^{0}/_{0})$
1 (o) o1 ( u10 2 ujo		
4 days	28	14.4
5 days	140	71.8
6 days	27	13.8

**Table 5.2** The distribution of the number of valid days of accelerometry data (No. of participants and percentage of participants).

# 5. 3. 1. Partial Correlations between Activity Variables and Measures of Body Composition and between Activity Variables and Themselves

Partial correlations were also used to initially examine the relationships between activity variables and adiposity measures after correcting for age. A weak but significant negative relationship was observed between light intensity physical activity and BMI and light intensity physical activity and  $\Sigma$  Skinfolds. These correlations are presented in Table 5.3 Table 5.3.

**Table 5.3** Partial correlations between physical activity and sedentary variables, body mass index and sum of skinfolds, controlled for age.

	BMI	$\Sigma$ Skinfolds <sup>+</sup>
	Percentiles	
Sedentary Time	.16 (.03)*	.14 (.049)*
Standing Time <sup>+</sup>	10 (.15)	10 (.18)
Light Intensity Physical Activity incl. Standing <sup>+</sup>	15 (.04)*	14 (.06)
Light Intensity Physical Activity excl. Standing <sup>+</sup>	25 (.0003)**	23 (.001)**
Moderate to Vigorous Physical Activity <sup>+</sup>	09 (.20)	12 (.10)
Prolonged Sedentary Periods <sup>+</sup>	.06 (.42)	01 (.94)

<sup>+</sup>Values were log transformed before analysis.

BMI = Body Mass Index;  $\Sigma$  Skinfolds = Sum of Skinfolds.

Data are presented as r (p value). \*p  $\leq 0.05$ ; \*\*p  $\leq 0.01$ ; N = 194.

#### 5. 3. 2. Partial Correlations between Activity Variables and Themselves

Partial correlations which controlled for age and body composition were used to examine the relationships between different activity patterns and behaviours. All correlations, except for the correlation between light intensity physical activity (excl. standing) and moderate to vigorous intensity physical activity, were significant at a p < 0.001 level. The results of these correlations are presented in Table 5.4.

**Table 5.4** Partial correlations between physical activity and sedentary variables, controlled for both age and body composition ( $\Sigma$  Skinfolds).

	SedT	StT	LIPA incl.	LIPA excl.	MVPA	PSP
			StT	StT		
SedT	-	92**	94**	68**	55**	.47**
StT		-	.98**	.56**	.26**	42**
LIPA incl. StT			-	.71**	.27**	44**
LIPA excl. StT				-	.22*	37**
MVPA					-	24**
PSP						-

SedT = Sedentary Time; StT = Standing Time; LIPA incl. StT = Light Intensity Physical Activity including Standing Time; LIPA excl. StT= Light Intensity Physical Activity excluding Standing Time; MVPA = Moderate to Vigorous Physical Activity; PSP = Prolonged Sedentary Periods. \*p < 0.01 level; \*\*p < 0.001; N = 195.

# 5. 3. 3. Linear Regression Analysis

Linear regression models were employed to examine the association between activity variables and body composition measures. Model 1 was corrected for age, while model 2 was corrected for age and moderate to vigorous physical activity. Retrospective power (1- $\beta$ ) calculations were also completed on all significant associations. No significant relationships were observed between sedentary time, standing time, light intensity physical activity (incl. standing) or prolonged sedentary periods and body composition. Light intensity physical activity (excl. standing) was significantly associated with BMI in model 1 ( $\beta$  = - 0.231; p =

.003) and in model 2 ( $\beta$  = - 0.222; p = .007). Acceptable statistical power was observed for the associations between BMI and light intensity physical activity (N = 194, r = -0.25,  $\alpha$  = 0.01, 1 -  $\beta$  = 0.86). Furthermore, light intensity physical activity (excl. standing) was significantly associated with  $\Sigma$  Skinfold measures in both model 1 ( $\beta$  = - 0.236; p = .005) and model 2 ( $\beta$  = - 0.219; p = .009). Acceptable power was also observed for the associations observed between  $\Sigma$  Skinfold and light intensity physical activity in model 2 (N = 194, r = -0.23,  $\alpha$  = 0.01, 1 -  $\beta$  = 0.79). All linear regression results are presented in Table 5.5.

## 5. 3. 4. Binary Logistic Regression Results

Binary logistic regression models were employed to predict the highest quartile of BMI using quartile groups of the activity variables. The first model corrected for age while the second model corrected for both age and moderate to vigorous physical activity. The results of this analysis identified that the amount of sedentary time, standing time, light intensity physical activity (incl. standing) and prolonged sedentary periods did not predict high BMI after correcting for the selected covariates. However, this analysis did identify that low levels of light intensity physical activity (excl. standing) were a significant predictor of high BMI after correcting for age and moderate to vigorous physical activity. Participants that accumulated the lowest levels of light intensity physical activity (excl. standing) (Quartile 1) were 3.81 (95% CI: 1.31 - 11.11) times more likely to have higher BMI than those who accumulated the highest levels of light intensity physical activity (excl. standing) (Quartile 4). For illustrative purposes, the estimated marginal means for BMI percentiles are presented for each of the quartiles in Figure 5.1, controlling for age and moderate to vigorous physical activity. All data from the binary logistic regression models between all activity variables and BMI are included in Table 5.6. A scatterplot presents the relationship between light intensity physical activity and body mass index in Figure 5.2

Binary logistic regression models were also used to determine whether the different activity variables predicted the highest quartile of skinfold thickness independent of age and moderate to vigorous physical activity. The amount of sedentary time predicted the highest  $\Sigma$  Skinfolds after correction for age and moderate to vigorous physical activity. This data suggests that the highest quartile of sedentary participants were 3.95 (1.25 – 12.47) more

likely to have high  $\Sigma$  Skinfolds than their least sedentary counterparts. The amount of standing time, light intensity physical activity (incl. standing) and prolonged sedentary periods did not predict high  $\Sigma$  Skinfolds after correcting for age and moderate to vigorous physical activity. Low levels of light intensity physical activity (excl. standing) were a significant predictor of high  $\Sigma$  Skinfolds after correcting for age and moderate to vigorous physical activity. Participants that accumulated the lowest levels of this type of light intensity physical activity (Quartile 4) were 2.88 (1.04 – 7.92) times more likely to have high  $\Sigma$  Skinfolds than those who accumulated the highest levels (Quartile 1). All data from the binary logistic regression models between activity variables and  $\Sigma$  Skinfolds are included in Table 5.7. The estimated marginal means for log-transformed sum of skinfold thickness are presented for each of the quartiles of sedentary time and light intensity physical activity in Figure 5.3 and Figure 5.5 respectively. Scatterplots are also provided to identify the relationships between sedentary time, light intensity physical activity and sum of skinfold thickness in Figure 5.4 and Figure 5.6.

		Model 1			Model 2	
Sedentary Time	β	р	Adj. R <sup>2</sup>	β	р	Adj. R <sup>2</sup>
BMI	.16	$.048^{*}$	.02	.15	.11	.01
$\Sigma$ Skinfold	.14	.14	.01	.11	.22	.01
Standing Time						
BMI	10	.19	.01	08	.24	.01
$\Sigma$ Skinfold	10	.39	.00	07	.29	.00
Light Intensity Ph						
BMI	15	.06	.02	13	.11	.02
$\Sigma$ Skinfold	12	.23	.01	10	.22	.01
Light Intensity Physical Activity excl. Standing						
BMI	25	.001**	.06	24	.003**	.05
$\Sigma$ Skinfold	23	.005**	.05	22	.009**	.04
Prolonged Sedent	ary Peri	ods				
BMI	.06	.36	00	.04	.35	.00
$\Sigma$ Skinfold	.01	.92	01	02	.40	.00

**Table 5.5** Relationship between sedentary time, standing time, light intensity physical activity (both including and excluding standing time) and prolonged sedentary periods and body composition measures in adolescent females.

N = 194; Multiple Linear Regression; \*p < 0.05; \*\*p < 0.01;

BMI = Body Mass Index;  $\Sigma$  Skinfold = Sum of four skinfolds;

Model 1 corrected for age; Model 2 corrected for age, and objectively measured MVPA.

	<b>Model 1</b> (95% CI)	<b>Model 2</b> (95% CI)
Sedentary Time		
Quartile 1	1	1
Quartile 2	1.39 (0.52 – 3.73)	1.49 (0.55 – 4.07)
Quartile 3	1.24 (0.46 – 3.37)	1.53 (0.52 – 4.46)
Quartile 4	2.14 (0.83 – 5.56)	2.85 (0.94 - 8.66)
Standing Time		
Quartile 1	1	1
Quartile 2	1.50 (0.60 – 3.72)	1.49 (0.60 – 3.71)
Quartile 3	0.86 (0.32 – 2.29)	0.87 (0.32 – 2.36)
Quartile 4	1.16 (0.45 – 2.98)	1.15 (0.43 – 3.03)
Light Intensity Physical Activity	incl. Standing Time	
Quartile 1	1	1
Quartile 2	0.80 (0.28 – 2.24)	0.77 (0.27 – 2.18)
Quartile 3	2.30 (0.93 – 5.72)	2.51 (0.98 - 6.42)
Quartile 4	1.19 (0.45 – 3.14)	1.22 (0.45 – 3.34)
Light Intensity Physical Activity	excl. Standing Time	
Quartile 1	1	1
Quartile 2	2.61 (0.90 – 7.58)	2.85 (0.95 - 8.57)
Quartile 3	2.77 (0.95 - 8.07)	2.91 (0.99 - 8.56)
Quartile 4	3.52 (1.24 - 10.01)*	3.81 (1.31 – 11.11)*
Prolonged Sedentary Periods		
Quartile 1	1	1
Quartile 2	0.52 (0.20 – 1.38)	0.53 (0.20 – 1.41)
Quartile 3	0.78 (0.32 – 1.92)	0.81 (0.32 – 2.04)
Quartile 4	0.94 (0.38 – 2.29)	0.96 (0.39 – 2.41)

**Table 5.6** Odds ratios for high body mass index across quartiles of sedentary time,

 standing time, light intensity physical activity and prolonged sedentary periods.

Model 1 corrected for age;

Model 2 corrected for both age, and objectively measured MVPA;

N = 194; Binary Logistic Regression; \*p < 0.05.


n = 195; Univariate Analysis of Variance; Mean  $\pm$  95% CI; \*p  $\leq$  0.01; Significant differences observed between Quartile 1 and Quartile 4; Quartile 1 = Lowest levels of light intensity physical activity, Quartile 4 = Highest levels of light intensity physical activity;

Figure 5.1 Estimated marginal means of high body mass index percentiles stratified by quartiles of light intensity physical activity.



N = 195

Figure 5.2 Scatterplot of the percentage time spent in light intensity physical activity and body mass index percentiles.

	<b>Model 1</b> (95% CI)	<b>Model 2</b> (95% CI)
Sedentary Time		
Quartile 1	1	1
Quartile 2	1.81 (0.63 – 5.21)	1.78 (0.61 – 5.20)
Quartile 3	2.02 (0.71 – 5.72)	2.00 (0.66 - 6.08)
Quartile 4	3.79 (1.38 – 10.38)*	3.95 (1.25 – 12.47)*
Standing Time		
Quartile 1	1	1
Quartile 2	1.66 (0.63 – 4.36)	1.68 (0.63 – 4.45)
Quartile 3	1.35 (0.50 – 3.67)	1.22 (0.44 – 3.38)
Quartile 4	2.16 (0.83 - 5.61)	1.97 (0.73 – 5.27)
Light Intensity Physical Activity + S	Standing Time	
Quartile 1	1	1
Quartile 2	1.26 (0.44 – 3.62)	1.26 (0.44 – 3.63)
Quartile 3	2.51 (0.94 – 6.65)	2.40 (0.89 - 6.47)
Quartile 4	2.51 (0.94 – 6.71)	2.36 (0.86 - 6.48)
Light Intensity Physical Activity		
Quartile 1	1	1
Quartile 2	1.34 (0.46 – 3.95)	1.23 (0.40 – 3.71)
Quartile 3	2.79 (1.02 – 7.66)*	2.63 (0.95 – 7.29)
Quartile 4	3.16 (1.17 – 8.54)*	2.88 (1.04 - 7.92)*
Prolonged Sedentary Periods		
Quartile 1	1	1
Quartile 2	0.36 (0.14 – 0.96)*	0.34 (0.13 – 0.93)*
Quartile 3	0.34 (0.13 – 0.90)*	$0.29 (0.11 - 0.79)^*$
Quartile 4	0.85 (0.36 – 1.99)	0.72 (0.30 – 1.75)

**Table 5.7** Odds ratios for high sum of skinfold thickness across quartiles of sedentary

 time, standing time, light intensity physical activity and prolonged sedentary periods.

Model 1 corrected for age;

Model 2 corrected for both age, and objectively measured MVPA;

N = 194; Binary Logistic Regression; \*p < 0.05.



N = 194; Univariate Analysis of Variance; Mean  $\pm 95\%$  CI; \*p  $\leq 0.01$ ; Significant difference observed between Quartile 1 and Quartile 4; Quartile 1 = Lowest levels of sedentary time; Quartile 4 = highest levels of sedentary time.

Figure 5.3 Estimated marginal means of sum of skinfolds stratified by quartiles of sedentary time.



N = 194

Figure 5.4 Scatterplot of the percentage time spent sedentary and the sum of skinfold thickness.



N = 194; Univariate Analysis of Variance; Mean 95% CI; \*p  $\leq 0.01$ ; Significant difference observed between Quartile 1 and Quartile 4; Quartile 1 = Lowest levels of light intensity physical activity; Quartile 4 = Highest levels of light intensity physical activity.

Figure 5.5 Estimated marginal means of sum of skinfolds stratified by quartiles of light intensity physical activity (excl. standing).



N = 194

**Figure 5.6** Scatterplot of the percentage time spent in light intensity physical activity (excl. standing) and the sum of skinfold thickness.

Significant findings were also observed when binary logistic regression was used to examine prolonged sedentary periods. These results unexpectedly suggest that those who spend the lowest amount of time in prolonged sedentary periods have a similar risk for high  $\Sigma$ Skinfolds as those with the highest amount of time in prolonged sedentary periods, while the middle two quartiles have the lowest potential risk for high  $\Sigma$  Skinfolds. A scatterplot was constructed and cross tabulation analysis performed to determine the relationship and explain the unexpected findings. As Figure 5.7 demonstrates, there appears to be no association between the two variables. A large number of participants that have moderate levels of prolonged sedentary periods have relatively low  $\Sigma$  Skinfold values. This causes the quartiles to have conflicting values (Table 5.8). Consequently, the data appears to have spuriously created odds ratios that appear significant.



# N = 194

Figure 5.7 Scatterplot of the percentage of time spent in prolonged sedentary periods and the sum of four skinfold.

n = 194	n = 194 Prolonged Sedentary Periods									
		Quartile 1	Quartile 2	Quartile 3	Quartile 4					
Sum of	Low Risk (%)	64.6	83.7	84	68.1					
Skinfold	High Risk (%)	35.4	16.3	16	31.9					

**Table 5.8** Cross tabulation table comparing Prolonged Sedentary Periods by quartile with low and high risk sum of skinfold values.

# 5. 4. Discussion:

This analysis aimed to examine the associations between sedentary time, standing time, light intensity physical activity (incl. standing), light intensity physical activity (excl. standing) and prolonged sedentary periods and body composition measures in a cohort of adolescent females using an inclinometer-based activity monitor. The majority of activity variables were highly or very highly associated with each other. However, weak associations were found between moderate to vigorous physical activity and sedentary variables. The findings of this study indicate that light intensity physical activity (excl. standing) is negatively associated with both BMI and sum of skinfold thickness, independent of age and moderate to vigorous physical activity (incl. standing) or prolonged sedentary periods and body composition measures. Finally, only light intensity physical activity (excl. standing) was identified as a significant predictor of the highest levels of BMI, while sedentary time and light intensity physical activity (excl. standing) were both significant predictors of the highest levels of sum of skinfold thickness, independent of age and moderate to vigorous physical activity physical activity (excl. standing) were both significant predictors of the highest levels of sum of skinfold thickness, independent of age and moderate to vigorous physical activity.

# 5. 4. 1. Association between Sedentary Time and Sum of Skinfold Thickness

A novel finding of this study was that female adolescents with the highest objectively measured sedentary time were substantially more likely to have high sum of skinfold thickness compared to their least sedentary counterparts (Figure 5.3). These findings are in contrast with existing objective examinations of sedentary time, which have identified that these associations did not exist or disappeared after correction for moderate to vigorous physical activity (Purslow et al., 2008, Carson and Janssen, 2011). The observed differences between the present findings and the previously published literature may be due to the different methods of quantifying sedentary time. Previous objective examinations have employed accelerometer-based activity monitors, such as the ActiGraph (ActiGraph LCC, Pensacola, FL) as measures of physical activity, and have estimated sedentary time using an activity count threshold (e.g. < 100 counts min<sup>-1</sup>). This method estimates sedentary time based on the lack of ambulation (Ridgers et al., 2012). In contrast, the present study has employed the activPAL activity monitor, which directly measures sedentary time using an inbuilt inclinometer. Significant differences have been observed between ActiGraph determined sedentary time and activPAL determined sedentary time, with the activPAL demonstrating increased accuracy at measuring sedentariness (Hart et al., 2011, Kozey-Keadle et al., 2011). Furthermore, the use of the activPAL has been encouraged in studies that aim to examine specific sedentary patterns and behaviours (Bassett Jr et al., 2010), while it has been employed as the reference measure for objective sedentary behaviour when validating accelerometer-based activity monitors (Ridgers et al., 2012).

Our findings are consistent with proposed physiological mechanisms which suggest that the absence or lack of large skeletal muscle contractions may play an important role in the development of obesity, cardiovascular disease and site-specific cancers (Hamilton *et al.*, 2007). These observations have public health significance. It is now widely accepted that sitting/lying is a distinct behaviour to standing with its own set of unique physiologic mechanisms (Hamilton *et al.*, 1998, Bey *et al.*, 2001, Zderic and Hamilton, 2006), while sitting/lying accounts for the majority of an individual's daily activities (Owen *et al.*, 2010). Furthermore, it has been proposed that even though modern humans currently accumulate more sedentary time than at any other period in our history, sedentary levels will continue to increase over the coming years due to further developments in technology (Hamilton *et*  *al.*, 2008, Owen *et al.*, 2010). Collectively, this information should encourage the introduction of sedentary behaviour guidelines in physical activity recommendations, similar to those presented in the Canadian Sedentary Guidelines, which encourage individuals to limit the amount of time spent indoors, in screen-based activities, in prolonged sedentary periods and in sedentary transport (Tremblay *et al.*, 2011). Additionally, the examination of a sedentary time threshold that provides clear and distinct guidelines on the maximum recommended amount of daily sedentary time, as is the case for moderate to vigorous physical activity, should also be investigated. Such guidelines may be clearer, easier to follow and consequently more appropriate as a recommendation for the general population.

# 5. 4. 2. Association between Light Intensity Physical Activity, Body Mass Index and Sum of Skinfold Thickness.

The predominant reason for the lack of information on the associations between light intensity physical activity and body composition variables is due to difficulties in detecting and assessing this specific activity behaviour (Shephard, 2003). The evidence available on the associations between light intensity physical activity and adiposity in adolescent populations is limited and contrasting in nature. Findings from the European Youth Heart Study have identified no associations between objectively assessed light intensity physical activity (light physical activity defined as 500-1999 counts per minute) and body fatness (BMI and sum of 5 skinfold measures) after correction for sex, maturity and a range of activity variables including moderate to vigorous physical activity (Ekelund et al., 2004). More recently, significant associations have been observed between objectively measured light intensity physical activity (light physical activity defined as 101-1999 counts per minute) and BMI in adolescent females, independent of moderate to vigorous physical activity (Steele et al., 2009). Similar to the contrasting associations identified between sedentary time and adiposity in childhood and adolescence, inconsistencies observed across these studies is likely due to the method that was used to examine sedentariness. Both studies employed accelerometer-based activity monitors, while a broad range of activity thresholds were employed across these two studies (< 100 counts  $\cdot$  min<sup>-1</sup> (Steele *et al.*, 2009) Vs. < 500 counts min<sup>-1</sup> (Ekelund *et al.*, 2004)). As identified by Ridgers and colleagues,

significant differences in estimated sedentary time can be observed using sedentary thresholds which differ greatly (Ridgers *et al.*, 2012). Additionally, the accelerometer-based activity monitors (such as the ActiGraph) have been shown to consistently and significantly overestimate sitting and walking activities, primarily due to its inability to examine standing time (Hart *et al.*, 2011). Standing time makes up a large percentage of individuals daily activity, and by definition should be categorised as light intensity physical activity (Owen *et al.*, 2010). Previous publications that have employed non-inclinometer-based accelerometers may have significantly under or overestimated light intensity physical activity through misclassifying standing time as sedentary time or light intensity physical activity (Hart *et al.*, 2011). Through the use of an inclinometer-based activity monitor which accurately and reliably distinguishes between sitting and standing, this study has provided more valid and reliable estimates of sedentary time, standing time and light intensity physical activity compared to previously published literature.

Displacement theory suggests that sedentary time replaces time spent in physical activity (Steele et al., 2009). According to recent objective longitudinal findings, sedentary time increases throughout adolescence at the expense of light intensity physical activity (Mitchell et al., 2012). Findings from this study would appear to agree with these observations, whereby light intensity physical activity (incl. standing) and light intensity physical activity (excl. standing) appears to protect against sedentary time to a greater extent than moderate to vigorous physical activity (Table 5.4). Furthermore, the findings of the current study suggest that light intensity physical activity (excl. standing) is significantly associated with both BMI and sum of skinfold thickness in adolescent females. Lower levels of light intensity physical activity (excl. standing) are a significant predictor of increased adiposity in this population, with those in the lowest quartile of light intensity physical activity (Quartile 1) 2.869 times more likely to have elevated sum of skinfolds than those in the highest light intensity physical activity quartile (Quartile 4). Observations in adult populations have also identified positive associations between participation in light intensity physical activity during breaks in sedentary time and both BMI and waist circumference (Healy et al., 2008a). Collectively, this information would suggest that increasing light intensity physical activity at the expense of sedentary time may be of great benefit in the maintenance of a healthy weight profile. To date, physical activity recommendations have focused on increasing participation in moderate to vigorous physical activity (Physical Activity Guidelines Advisory Committee, 2008). These findings suggest that activity intensities across the

continuum have an influence on health outcomes in both adolescence and adulthood, and that additional recommendations for sedentary time and light intensity physical activity are required. Future physical activity recommendations should encourage the replacement of sedentary time with ambulatory light intensity physical activity alongside existing moderate to vigorous physical activity recommendations.

## 5. 4. 3. Strengths and Limitations:

There are limitations to this study. This study only focused on adolescent females. These findings are not generalisable to male adolescents, and further research is required to examine the associations between directly measured sedentary behaviours, light intensity physical activity and adiposity in adolescent males. Additionally, our cross sectional design represents a group of adolescent females in one geographical area over a two year period, and may not be representative for all populations. The majority of research studies have included additional covariates, such as stage of pubertal development, nutritional information and parental health status. The use of additional covariates would add more strength to our findings. Significant strengths of this study should also be noted. This is the first study to examine the associations between lifestyle physical activities and adiposity using an inclinometer-based activity monitor in any population. Furthermore, we examined the associations between each activity variables and adiposity independently of time spent in moderate to vigorous physical activity.

## 5. 5. Conclusions:

In summary, these population-based observations have identified important associations between physical activities at the low end of the activity intensity continuum and body composition measures in an adolescent female population. Firstly, evidence of the positive associations between total sedentary time and adiposity in adolescent females, independent of moderate to vigorous physical activity, has been presented. Secondly, negative associations between light intensity physical activity (excluding standing time) and adiposity, independent of moderate to vigorous physical activity, have been observed. The results of this study suggest that increasing light intensity physical activity at the expense of sedentary time is a worthwhile initiative that should be promoted along with the wellestablished regular participation in moderate to vigorous physical activity. Additional crosssectional research is required to examine whether these associations are evident across all populations, including children, male adolescents and adults of all ages, while longitudinal and interventional evidence is necessary to determine the effects of reduced sedentary time through increased light intensity physical activity on adiposity and additional cardiovascular risk factors in all populations. Chapter 6. Objectively Determined Light Intensity Physical Activity is Negatively Associated with Skinfold Thickness and Carotid Intima-Media Thickness in Adolescent Females

#### 6.1. Introduction

#### 6.1.1. Cardio-Metabolic Risk Factors in Childhood and Adolescence

Cardiovascular disease continues to be the leading causes of premature death in both industrialised and developing countries (Lloyd-Jones et al., 2009). Risk factors for these pathologies manifest over time, and are predominantly observed in adulthood. Over the last two decades, alarming levels of cardio-metabolic risk factors, which include obesity, hypertension, dyslipidemia, insulin resistance and glucose intolerance have also been detected in children and adolescents (Steinberger et al., 2009). Data from population-based samples have shown that the prevalence of the metabolic syndrome in children and adolescents may range between 3-12% (Steele et al., 2008). A more recent US populationbased examination of 2456 participants has estimated that 50% of US children and adolescents possessed one or more cardio-metabolic risk factors (Johnson et al., 2009). These observations are of particular public health concern, as evidence suggests that cardio-metabolic risk factors track from childhood through adolescents and into adulthood, further increasing the risk of early onset cardiovascular disease (Bao et al., 1994, Raitakari et al., 2003). Consequently, the reduced incidence of cardio-metabolic risk factors in children and adolescents has long term implications to quality of life and longevity (Franks et al., 2010b).

# 6. 1. 2. Associations between Moderate to Vigorous Physical Activity and Cardio-Metabolic Risk Factors

Physical inactivity has been highlighted as a significant contributor to the high prevalence of cardiovascular disease worldwide (Lloyd-Jones *et al.*, 2010). Historically, research has predominantly focused on the effect of moderate to vigorous physical activity, and has emphasised the positive effect that increased levels of moderate to vigorous physical activity has on cardio-metabolic risk factors in children and adolescents. Furthermore, it has been suggested that increased levels of moderate to vigorous physical activity in youth may have a positive effect on cardio-metabolic risk factors in adulthood (Hallal *et al.*, 2006).

# 6. 1. 3. Light Intensity Physical Activity and Cardio-Metabolic Risk Factors

Moderate to vigorous physical activity accounts for a very small proportion of an individual's daily energy expenditure compared with activities at the lower end of the activity intensity continuum (Troiano et al., 2008). Light intensity physical activity accounts for the majority of our daily energy expenditure, yet these lower intensity activities have received relatively little focus (Levine, 2002, Pate et al., 2008, Owen et al., 2010). Few population-based studies have examined the effect of light intensity physical activity on cardio-metabolic risk factors in adults. These studies have highlighted that objectively measured light intensity physical activity was independently associated with 2 hour fasting glucose (Healy et al., 2007), waist circumference and clustered metabolic risk (Healy et al., 2008c) in adult populations. These relationships have not been addressed using objective measures in a population of children or adolescents. Interestingly, a recent longitudinal study has examined the distribution of sedentary time, light intensity physical activity and moderate to vigorous physical activity throughout adolescence, and has identified that increases in sedentary time from childhood through adolescence are largely due to decreases in light intensity physical activity rather than decreases in moderate to vigorous physical activity (Mitchell et al., 2012). These findings further support the significance of accurately examining light intensity physical activity in adolescent populations.

## 6.1.4. Sedentary Behaviours and Cardio-Metabolic Risk Factors

The mechanisms by which prolonged periods of sedentary time are detrimental to indices of health are not yet fully understood. One potential mechanism, proposed by Bey and Hamilton (2003), is that inactivity reduces the activity of the enzyme lipoprotein lipase in skeletal muscle, thereby altering triglyceride metabolism and presumably increasing the risk of cardiovascular disease (Bey and Hamilton, 2003). Several studies have examined the independent relationship between sedentary time and cardio-metabolic risk factors in adults, highlighting that sedentary time has an effect on cardio-metabolic risk factors in adults, independent of moderate to vigorous physical activity (Healy *et al.*, 2008c, Helmerhorst *et al.*, 2009, Healy *et al.*, 2011b). Additionally, evidence has highlighted that

prolonged sedentary periods are negatively associated with cardiovascular and inflammatory markers in adults (Healy *et al.*, 2011b). The associations of sedentary time on cardio-metabolic risk factors in children and adolescents has been examined using objective activity monitors, yet findings have highlighted that the volume of sedentary time does not have an effect on cardio-metabolic risk factors independent of moderate to vigorous physical activity (Carson and Janssen, 2011, Martinez-Gomez *et al.*, 2011a). Likewise, the amount of time spent in prolonged sedentary periods has not been shown to have an effect on cardio-metabolic risk factors in children and adolescents independent of moderate to vigorous physical activity (Carson and Janssen, 2011, Martinez-Gomez *et al.*, 2011a). Likewise, the amount of time spent in prolonged sedentary periods has not been shown to have an effect on cardio-metabolic risk factors in children and adolescents independent of moderate to vigorous physical activity (Carson and Janssen, 2011).

# 6. 1. 5. The Assessment of Free-Living Sedentary Time and Light Intensity Physical Activity

The term sedentary refers to a specific behaviour which describes activities that do not substantially increase energy expenditure above resting level (Marshall and Ramirez, 2011, Tremblay *et al.*, 2012). Objectively-measured sedentary time has previously been determined based on a lack of movement using a measure of ambulation rather than a measure of body position (Ridgers *et al.*, 2012). Unfortunately, cut points are unable to distinguish between sitting and standing and may also count time spent in other light intensity activities as sedentary behaviour, hence inaccurately quantifying sedentary time (Hart *et al.*, 2011, Kozey-Keadle *et al.*, 2011). Consequently, if sedentary time is not accurately quantified as a result of the measurement tool being used, the resulting light intensity physical activity, which is calculated as total measurement time minus [sedentary time + moderate to vigorous physical activity], will also be miscalculated. Alternatively, the use of inclinometer-based activity monitors to distinguish standing from sitting, such as the activPAL<sup>TM</sup> Professional Physical Activity Monitor (PAL technologies Ltd, Glasgow, Scotland, UK), have been encouraged when examining the effects of sedentary behaviour and sedentary time on health outcomes (Bassett Jr *et al.*, 2010, Kozey-Keadle *et al.*, 2011).

To date, no research studies have employed an inclinometer-based activity monitor to examine the relationships between sedentary time and light intensity physical activity and cardio-metabolic risk factors independent of moderate to vigorous physical activity in any population. The purpose of this chapter is to address this issue by examining the associations between the volume of sedentary time, patterns of sedentary time and the volume of light intensity physical activity with cardio-metabolic risk factors in an adolescent female population using an inclinometer-based activity monitor. Specifically, this chapter aims to determine 1) whether an association exists between the sedentary variables, light intensity physical activity and cardio-metabolic risk factors in female adolescents, 2) to explore whether any observed association between the subcomponents of physical activity and 3) to explore whether any observed association between the subcomponents the subcomponents of physical activity and cardio-metabolic risk factors persist after adjustment for moderate to vigorous physical activity and cardio-metabolic risk factors persist after adjustment for both moderate to vigorous physical activity and cardio-metabolic risk factors persist after adjustment for both moderate to vigorous physical activity and cardio-metabolic risk factors persist after adjustment for both moderate to vigorous physical activity and cardio-metabolic risk factors persist after adjustment for both moderate to vigorous physical activity and cardio-metabolic risk factors persist after adjustment for both moderate to vigorous physical activity and cardio-metabolic risk factors persist after adjustment for both moderate to vigorous physical activity and adiposity.

## 6.2. Methods

A list of all second level schools in the Limerick region was obtained. From this list, a sample of schools was selected. The inclusion criteria for a school to be selected for participation was 1) the school must be public schools and 2) the school must have had a large student base (in excess of 500 students). A list of suitable schools was generated, and four schools were selected for participation using a random numbers table. If a school declined the opportunity to take part in the study the next randomly selected school was approached until 4 high schools had consented to participate. A list of all female students between the ages of 13 and 18 years was obtained from each school, and 40 students per school were randomly selected using a random numbers table. A total of 77 students provided written informed parental and written informed participant consent before the study began. Of these participants, 6 (7.8%) were absent on the initial test days, which automatically removed them from participation in the study. This rate of absence is in line with national absence statistics at post primary level, identifying the average rate of nonattendance at 7.9% (Millar, 2012). A further 7 participants were removed from the analysis due to insufficient accelerometer data (explained in Section 4. 2. 2). This provided a final sample of 64 participants. Recruitment and data collection took place in autumn 2011. Ethical approval was obtained for this study from the Research Ethics Committee of the Mid-Western Regional Hospital, Limerick, Ireland.

#### 6.2.1. Physical Examination

All participants arrived to the school gymnasium in groups of three to five. Participants wore light clothing (vest top and tracksuit pants) and removed their shoes and socks before all measures were obtained.

## 6. 2. 1. 1. Body Composition Measures

Height, weight and 4 sites skinfold measurement was assessed in all participants. Height was measured to the nearest 0.25 cm using a portable stadiometer (Seca model 214, Seca Ltd., Birmingham, UK) and weight was measured to the nearest 0.1 kg using a portable electronic scale (Seca model 770, Seca Ltd., Birmingham, UK) following standard procedures. Body mass index (BMI) was calculated as weight (kg) /height (m)<sup>2</sup>. Four sites skinfold measurement was employed as an estimate of general adiposity. All skinfold measures were obtained using a Harpenden skinfold calliper (Assist creative resources Ltd., Wrexham, United Kingdom). The 4 sites included for measurement in this study were the bicep, triceps, subscapular and iliac crest, and the sum of the four skinfold sites were used. All skinfold measures were recorded three times and the median value of the three measures was calculated.

# 6. 2. 1. 2. Carotid Intima-Media Thickness Measures

Carotid intima-media thickness is the most commonly used and best validated ultrasound measure of early and intermediate stages of atherosclerosis (Knoflach *et al.*, 2003). Atherosclerosis refers to the build-up of plaque on the walls of the arteries and is the most common cause of CVD (Skålén *et al.*, 2002). As atherosclerotic plaque accumulates, the vessel wall undergoes compensatory remodelling in order to avoid luminal narrowing, which results in an increase in the intima-media layer (Silverthorn, 2012). Increments in carotid intima-media thickness are associated with an increased risk for myocardial infarction and stroke (Bots *et al.*, 1997, Lorenz *et al.*, 2007). For an absolute carotid intima-

media difference of 0.1 mm, the future risk of myocardial infarction increases by 10% - 15% (Lorenz et al., 2007).

Thickness of the carotid intima-media was assessed using a 12.0 MHz linear-array transducer (SonoSite, MicroMax). Recordings were obtained with the subject resting in a supine position. The common carotid artery, including the carotid bulb, was visualised and 2 longitudinal  $\beta$ -mode images of the left and the right common carotid arteries at end diastole were recorded and electronically stored. Measurements of CIMT was conducted in the 10-mm linear segment proximal to the carotid bulb at 2 plaque-free sites three times at the near wall and three times at the far wall on both sides and combined as mean CIMT. A combination of readings from both the near and far walls were used as they yield the strongest association with cardiovascular disease (Bots *et al.*, 2003).

# 6. 2. 1. 3. Large Artery Stiffness

The build-up of atherosclerotic plaques on the interior walls of arteries causes them to stiffen, which in turn causes the pressure at which blood is pumped through the arteries around the body to increase (Silverthorn, 2012). The build-up of these plaques often results in narrowing of the arteries, which in time may lead to clots or aneurysms. Recent technological developments have enabled researchers to examine arterial stiffness more accurately through examining the velocity at which the blood is pumped around the body (pulse wave velocity). Through examining arterial stiffness, researchers can now more accurately estimate the prevalence of fatty streaks and atherosclerotic lesions in the coronary arteries (Dangardt *et al.*, 2008). Until now, the study of pulse wave velocity and arterial stiffness was an expensive and invasive procedure. However, more cost effective devices have been developed that estimate pulse wave velocity.

In this study, large artery stiffness was estimated using a Pulse Trace PCA 2 (Micro Medical Ltd, Rochester, UK). The Pulse Trace PCA 2 estimates stiffness index by recording and analysing the digital volume pulse using a digital photoplethysmographic probe on the fingertip of the participant (Millasseau *et al.*, 2000). Pulse Trace PCA 2 stiffness index has previously been shown to be positively correlated with pulse wave velocity in paediatric populations (Simonetti *et al.*, 2008).

#### 6. 2. 1. 4. Stage of Pubertal Development

Stage of pubertal development Tanner Scale pictures for self-identification of Tanner pubertal status were employed in this research, which precluded the need to examine the participants for pubertal status assignment (Tanner and Whitehouse, 1976). The self-identification of Tanner pubertal status was completed with a paediatrician present in the Mid-Western regional hospital, Limerick, Ireland.

# 6.2.1.5. Blood Pressure

A detailed description of blood pressure and its relationship with health is discussed in Section 2. 4. 1. 2. Blood Pressure was measured using an automated sphygmomanometer (Omron model M7, Kyoto, Japan), and mean arterial pressure was calculate (Mean Arterial Pressure = Diastolic Pressure + 1/3 \* (Systolic Pressure – Diastolic Pressure)).

#### 6. 2. 1. 6. Blood Sampling and Analysis

The importance of examining blood lipids (including HDL-C, LDL-C, total cholesterol and triglycerides) in relation to cardiovascular disease and health have been discussed in Section 2. 4. 1. 1. 1. Insulin, glucose and high molecular weight (HMW) adiponectin were also examined within this sample. Insulin is a hormone which is secreted by the pancreas. Insulin stimulates increased glucose transport into the cell. However, if skeletal muscle becomes insulin resistant, the pancreas increases its production in an effort to encourage glucose uptake. A high level of insulin in the blood is a precursor of type 2 diabetes. As the insulin resistance grows, the amount of insulin required for glucose homeostasis cannot be produced by the pancreas, resulting in increased blood sugar levels. This is the onset of type 2 diabetes. Adiponectin is also involved in the regulation of glucose levels in the blood, while it is similarly important in the breakdown of fatty acids. Increased levels of HMW adiponectin in the blood is associated with a reduced risk of diabetes (Zhu *et al.*, 2010).

Blood testing was carried out between 9am and midday after each participant was instructed to fast for a minimum of 10 hours. Biochemical analysis was undertaken by the Clinical Chemistry Department at the Mid-Western regional hospital, Limerick, Ireland. Fasting venous blood samples were taken from the cubital vein using a paediatric cannula and were collected in both EDTA and heparin sampling tubes. Cholesterol linked to high-density lipoproteins (HDL-C), cholesterol linked to low-density lipoproteins (LDL-C), total cholesterol (TC), triglycerides (TG) and Glucose (Glu) were measured on a Beckman auto-analyser using standard measurement kits. Insulin and HMW Adiponectin was measured by Enzyme-linked Immunosorbent assay (ELISA) using standard measurement kits. The analytical performance of the clinical laboratory used in this study is continually assessed by the Irish External Quality Assessment Scheme.

#### 6. 2. 2. Clustered Metabolic Risk Score

It has been suggested that a continuous metabolic syndrome score, or clustered metabolic risk score (CMRS), is less prone to error and statistically more sensitive by comparison to standard dichotomous approaches (Brage *et al.*, 2004). Variables included in a CMRS must reflect the adult metabolic syndrome criteria, including obesity, blood lipids, insulin/glucose and hypertension (Eisenmann, 2008). In this study, the CMRS was calculated using sum of skinfolds, HDL-C, TG, Insulin and Mean Arterial Pressure. Each variable was standardised using the linear regression of the results for both age and pubertal developmental stage (Eisenmann, 2008). The standardised results for HDL-C were then multiplied by -1 due to its inverse relationship with cardiovascular risk (Eisenmann, 2008). The results were then summed to produce a clustered metabolic risk score (CMRS) for each participant. Additionally, the standardised scores were summed excluding the adiposity component to examine whether associations between light intensity physical activity, sedentary variables and clustered metabolic risk were mediated by adiposity (CMRS-Obs).

## 6. 2. 3. Physical Activity and Sedentary Levels and Patterns

Physical activity and sedentary levels and patterns were examined using the activPAL<sup>TM</sup> Professional Physical Activity Monitor (PAL Technologies Ltd, Glasgow, UK). A detailed description of the activPAL has been provided in Section 2. 10. The activPAL responds to gravitational accelerations resulting from segmental movement (Ryan *et al.*, 2006), and data is recorded at 10 Hz for each 15 second time intervals (epoch). The activPAL has been identified as a valid and reliable measure of sitting/lying time and standing time in children (Davies *et al.*, 2011) and adults (Kozey-Keadle *et al.*, 2011), and as a measure of light and moderate to vigorous physical activity in an adolescent female population (Section 3. 3. 5).

## 6.2.3.1. Data Processing

The methods employed in processing the activPAL output have previously been described in Section 4. 2. Briefly, for a day to be categorised as valid, a minimum of 10 hours of daily recording time was required. The activity monitor output was then examined for number of valid days of activity data provided. For a dataset to be included in the examination of habitual activity profiles, 4 valid days of accelerometer data (including one weekend day) are required (Trost *et al.*, 2005). All data sets were examined for non-wear time, where nonwear time was defined as periods of 60 minutes of concurrently recorded sedentary time and continuous 0 accelerometer counts from the activPAL output file. Time identified as non-wear time was removed from the data set, and the 24 hour day was adjusted accordingly.

Data were then examined for time spent sedentary, standing, in light intensity physical activity and in moderate to vigorous physical activity. The start time for each measured day was identified at 6 am, and the following 24 hour period was examined. The total amount of sedentary time and standing time was summed over the entire day. The amount of time spent in moderate to vigorous physical activity was examined using count to activity thresholds. A specific count to activity threshold of 2997 per epoch has been derived for use in this population (Section 3. 3. 5). The amount of time spent in moderate to vigorous physical activity amount of time spent in epochs above this threshold. Light intensity physical activity was then calculated as all remaining daily time (light

intensity physical activity = 24 hours – (time spent in sedentary activities + time spent in standing activities + time spent in moderate to vigorous physical activity)). Since participants were encouraged to wear the activPAL device for 24 hours each day, all activities variables are presented as a percentage of waking time to standardise activity scores across all participants. The first change in inclination from sitting/lying to standing after a prolonged sedentary periods (e.g. sleep time) was identified as the "rise time", while the last change in inclination form standing to sitting before a prolonged SP identified the "bed time". From this information, the number of waking hours was the calculated. Each of the activity variables were then presented as percentage of waking time (e.g. (percentage of moderate to vigorous physical activity waking hours = no. of hours spent in moderate to vigorous physical activity waking hours = no. of hours spent in moderate to vigorous physical activity making)\*100).

The percentage of time spent in prolonged sedentary periods (e.g. sedentary periods greater than 30 minutes in duration) was also examined. A customised MATLAB® computer software program examined the activPAL files for all sedentary periods observed across the entire day. This methodology has been described in great detail in Section 4. 2. 3.

#### 6. 2. 4. Statistical Analysis

Descriptive statistics were calculated and are presented as median and interquartile ranges. All variables were examined for normality of distribution before the analysis. Fasting triglycerides, insulin and adiponectin were identified as not normally distributed. Natural logarithm was applied to these non-normal variables.

Partial correlations, adjusted for age, were used to examine the relationships between these activity variables (sedentary time, light intensity physical activity, moderate to vigorous physical activity and time spent in prolonged sedentary periods) and cardio-metabolic risk factors. Partial correlations were also used to identify relationships between physical activity and sedentary variables while adjusting for age and stage of pubertal development.

Multiple linear regression models were used to examine the independent relationships between percentage sedentary time, percentage of time spent in prolonged sedentary period and the percentage of time spent in a light intensity physical activity (both including and excluding standing time) and the metabolic risk factors, adjusting for age and stage of pubertal development. The three models employed assessed:

- 1. Whether activity variables were associated with each cardio-metabolic risk factor after correcting for age and stage of pubertal development;
- 2. Whether activity variables were associated with each risk factor while adjusting for the original confounding variables in model 1 and moderate to vigorous physical activity;
- 3. Whether activity variables were associated with each risk factor after adjusting for all earlier confounding variables plus a measure of body composition (when body composition was not the outcome of interest).

The independent associations between activity variables and clustered metabolic risk score (CMRS) were examined in two ways. Firstly, HDL-C, mean arterial pressure, triglycerides, insulin and sum of skinfolds were included in calculating the CMRS and the model was corrected for all confounders identified in model 1 and 2 above. Then the adiposity component (sum of four skinfolds) was removed from the CMRS (CMRS-Obs). The independent associations between activity variables and CMRS-Obs were then examined using models 1, 2 and 3 described above. Retrospective power calculations were completed on all statistically significant correlations from multiple linear regression models. This was completed to examine the probability of committing a type 2 statistical error, or not rejecting the null hypothesis when the null hypothesis is false. All statistical analysis was completed using Predictive Analysis Software version 18.0 for windows (SPSS Inc., Chicago, IL, USA).

#### 6.3. Results:

#### 6. 3. 1. Descriptive Statistics

Table 6.1 displays the descriptive characteristics for the sample. The median value for percentage of time spent sitting/lying was 70%, percentage of time spent in light intensity physical activity only was 4.8%, the percentage of time spent in light intensity physical

activity including standing time was 24.6% and percentage of time spent in moderate to vigorous physical activity was 5.5%.

Variables	Median (Range)
Age (yrs.)	15.1 (14.4 – 16.0)
Height (m)	1.63 (1.59 – 1.68)
Weight (kg)	59.9 (51.2 - 68.1)
BMI $(kg/m^2)$	22.6 (20.4 – 25.7)
Body Fat (%)	32.3 (28.8 – 34.6)
$\Sigma$ Skinfolds (mm)	69.7 (53.0 - 82.6)
HDL-C (mmol/l)	1.4 (1.2 – 1.5)
LDL-C (mmol/l)	2.3 (1.9 – 2.6)
Total Cholesterol (mmol/l)	3.9 (3.6 – 4.4)
Triglycerides (mmol/l)	0.54 (0.41 – 0.70)
Insulin (pmol/l)	56.3 (44.1 – 65.2)
Glucose (mmol/l)	4.5 (4.3 – 4.7)
Adiponectin (ug)	2.8 (2.1 – 4.1)
Large Artery Stiffness (m/s <sup>-1</sup> )	5.2 (5.0 – 5.3)
Mean Arterial Pressure (mmHg)	86.3 (80.8 – 90.0)
Carotid Intima-Media Thickness (mm)	0.05 (0.04 – 0.06)
Sitting/Lying (%)	70.0 (63.6 – 74.7)
Standing Time (%)	19.8 (17.1 – 23.7)
Light Intensity Physical Activity (%)	4.8 (3.8 – 5.4)
Light Intensity Physical Activity + Standing (%)	24.6 (20.9 – 29.2)
Moderate to Vigorous Physical Activity (%)	5.4 (3.8 – 7.3)
Sedentary Periods > 30 mins (%)	33.8 (27.1 – 40.0)

Table 6.1 Descriptive characteristics of participants. Values are presented as mean (range).

All variables, excluding mean arterial pressure (MAP), had a sample of n=64. MAP had a sample size of 48 participants. All activity variables are presented as a percentage of total waking hours.

# 6. 3. 2. Partial Correlations between Activity Variables and Cardiovascular Risk Factors

The results of the partial correlations between activity variables and cardio-metabolic risk factors, corrected for age, are presented in Table 6.2. The percentage of daily sedentary time demonstrated a moderate but significant positive association with BMI (r = 0.268; p < 0.05). The percentage of daily time spent in light intensity physical activity demonstrated a moderate but significant association with BMI (r = .349; p = 0.034). Percentage of daily light intensity physical activity demonstrated a negative but significant association with BMI (r = ..467; p = 0.006). No other associations were observed between physical activity and sedentary patterns and levels and cardio-metabolic risk factors in this population.

#### 6. 3. 3. Partial Correlations between Activity Variables

The partial correlations between sedentary variables and activity variables adjusted for age and stage of pubertal development are presented in Table 6.3. Moderate to vigorous physical activity was not correlated with standing time (p = .07), light intensity physical activity (excl. standing time) (p = .81), light intensity physical activity (incl. standing time) (p = .11) or with prolonged sedentary periods (p = .14). Significant associations at a p < 0.01level existed between all other measured activity variables in this sample.

**Table 6.2** Partial correlations between sedentary time, standing time, light intensity physical activity (incl. and excl. standing time), moderate to vigorous intensity physical activity, prolonged sedentary periods and cardio-metabolic risk factors, corrected for age.

	ST	StT	LIPA excl. StT	LIPA incl. StT	MVPA	PSP
BMI	.35*	26	47**	31	19	.21
% Body Fat	.25	15	23	16	28	.11
$\Sigma$ Skinfolds	.28	19	27	21	25	.13
HDL-C	07	.02	.08	.03	.02	01
LDL-C	.00	14	.04	14	.04	08
Total Chol.	.01	14	.03	14	.02	05
Triglycerides	.21	21	09	20	06	04
Insulin	.21	19	18	20	02	.05
Glucose	.14	04	05	03	18	06
Adiponectin	29	.22	.34*	.25	.21	25
LAS	.03	00	.10	.02	23	.17
MAP	.12	07	29	12	.13	.14
CIMT	.27	23	34	25	09	.31

ST = Sedentary Time; StT = Standing Time; LIPA incl. StT = Light Intensity Physical Activity including Standing Time; LIPA excl. StT = Light Intensity Physical Activity excluding Standing Time; MVPA = Moderate to Vigorous Physical Activity; PSP = Sedentary Periods greater than 30 minutes in duration; BMI = Body Mass Index; HDL-C = High Density Lipoprotein cholesterol; LDL-C = Low Density Lipoprotein cholesterol; Total Chol. = Total Cholesterol; LAS = Large Artery Stiffness; MAP = Mean Arterial Pressure; CIMT = Carotid Intima-Media Thickness; N = 64 (excluding MAP); \* p < 0.05; \*\*p < 0.01.

	ST	StT	LIPA excl. StT	LIPA incl. StT	MVPA	PSP
SedT	-	94**	66**	96**	48**	.69**
StT		-	.56**	.98**	.24	71*
LIPA excl. StT			-	.71**	.03	45**
LIPA incl. StT				-	.22	71**
MVPA					-	20
<b>SP &gt; 30</b> mins						-

Table 6.3 Partial correlations between sedentary variables and physical activity variables, corrected for age.

ST = Sedentary Time; StT = Standing Time; LIPA excl. StT = Light Intensity Physical Activity excluding Standing Time; LIPA incl. StT = Light Intensity Physical Activity including Standing Time; MVPA = Moderate to Vigorous Physical Activity; PSP = Prolonged Sedentary periods.

All correlations were significant (\*\* $P \le 0.01$ ) except for the correlations between StT, LIPA incl. StT, LIPA excl. StT and PSP with MVPA.

# 6. 3. 4. Results of Multiple Linear Regression between Activity Variables and Cardio-Metabolic Risk Factors

As the partial correlations identified no associations between the percentage of waking time spent standing, percentage of time spent in light intensity physical activity (including standing), moderate to vigorous physical activity, prolonged sedentary periods and cardiometabolic risk factors in Table 6.2, no further results from these variables will be examined using multiple linear regression in this chapter. However, all correlations from these activity variables and cardio-metabolic risk factors are presented in Appendix H.

## 6. 3. 4. 1. Sedentary Time and Cardio-Metabolic Risk Factors

The associations between sedentary time and cardio-metabolic risk factors using multiple linear regression models, which adjusted for age and stage of pubertal development, are displayed in Table 6.4. All outcomes are displayed using standardised beta scores. The results of the regression analysis highlighted that directly measured sedentary time was not associated with cardio-metabolic risk factors.

# 6. 3. 4. 2. Light Intensity Physical Activity and Cardio-Metabolic Risk Factors

The results from the multiple linear regression models, which examined the association between the percentage of waking time spent in light intensity physical activity (excl. standing time) and cardio-metabolic risk factors, are displayed in Table 6.5. A significant association was observed between the percentage of waking time spent in light intensity physical activity and body mass index ( $\beta = -.332$ ; p = .014). After further adjustment for moderate to vigorous physical activity, this association remained ( $\beta = -.333$ ; p = .015). A retrospective power calculation identified power of 0.56 for the association between light intensity physical activity and BMI (N = 64, r = -0.33,  $\alpha = 0.01$ , 1 –  $\beta = 0.56$ ). No additional associations were observed.

	Model 1				Mode	el 2	Model 3			
	β	р	Adj. R <sup>2</sup>	β	р	Adj. R <sup>2</sup>	β	р	Adj. R <sup>2</sup>	
BMI	.23	.08	.04	.06	.29	.03	-	-	-	
% BF	.20	.13	.06	.21	.17	.04	-	-	-	
ΣSF	.22	.10	.05	.25	.11	.04	-	-	-	
HDL-C	10	.47	.01	15	.35	.00	09	.57	.02	
LDL-C	.09	.50	03	.11	.50	05	.03	.83	.02	
T Chol.	.04	.76	.01	.04	.82	01	.02	.92	02	
TG	.14	.32	02	.18	.25	03	.11	.50	.05	
Insulin	.22	.12	00	.29	.07	01	.20	.20	.12	
Glucose	.19	.15	.10	.15	.32	.09	.12	.44	.09	
HMW Apo	15	.27	02	14	.38	04	15	.37	06	
LAS	.06	.75	06	11	.57	01	69	.50	02	
MAP	.06	.72	04	.15	.43	04	.12	.54	05	
CIMT	.26	.06	.04	.28	.08	.03	.30	.07	.01	
CMRS	.26	.11	01	.29	.14	03	-	-	-	
<b>CMRS-Obs</b>	.21	.19	03	.28	.15	04	.19	.28	.17	

**Table 6.4** Associations between sedentary time and cardio-metabolic risk factors in adolescent females.

 $\rm BMI = Body Mass Index; \% BF=$  Percentage Body Fat;  $\Sigma$  SF = Sum of Skinfold Thickness; HDL-C = High Density Lipoprotein cholesterol; LDL-C = Low Density Lipoprotein cholesterol; T Chol. = Total Cholesterol; TG = Triglycerides; HMW Apo = High Molecular Weight Adiponectin; LAS = Large Artery Stiffness; MAP = Mean Arterial Pressure; CIMT = Carotid Intima-Media Thickness; CMRS = cardiometabolic risk score; CMRS-Obs = cardio-metabolic risk score excluding the measure of adiposity. Model 1 adjusted for age and stage of pubertal development; Model 2 adjusted for age, stage of pubertal development and objectively measured MVPA; Model 3 adjusted for age, stage of pubertal development, objectively measured MVPA and adiposity (sum of four skinfolds); N = 64 (excluding MAP); Multiple Linear Regression Analysis; \*p < 0.05.

	Model 1				Model 2			Model 3		
	β	р	Adj. R <sup>2</sup>	β	р	Adj. R <sup>2</sup>	β	р	Adj. R <sup>2</sup>	
BMI	33	.01*	.09	33	.02*	.08	-	-	-	
% BF	25	.06	.08	25	.07	.06	-	-	-	
ΣSF	27	.05	.07	26	.05	.06	-	-	-	
HDL-C	.14	.32	.02	.14	.32	00	.08	.56	.02	
LDL-C	14	.33	03	14	.34	04	06	.69	.02	
T Chol.	08	.59	.01	07	.60	01	05	.72	.02	
TG	19	.18	.00	19	.19	02	10	.46	.06	
Insulin	07	.63	05	07	.65	07	.06	.68	.12	
Glucose	.05	.73	.07	.06	.64	.08	.11	.42	.10	
HMW Apo	.19	.18	01	.19	.19	02	.20	.18	04	
LAS	.06	.75	04	.09	.62	02	.12	.53	03	
MAP	20	.21	01	21	.20	03	18	.29	03	
CIMT	25	.08	.03	24	.09	.02	27	.08	.00	
CMRS	22	.19	03	20	.23	04	-	-	-	
CMRS-Obs	19	.26	03	18	.30	05	08	.60	.20	

**Table 6.5** Associations between light intensity physical activity (excluding standing time)

 and cardio-metabolic risk factors in adolescent females.

BMI = Body Mass Index; % BF= Percentage Body Fat;  $\Sigma$  SF = Sum of Skinfold Thickness; HDL-C = High Density Lipoprotein cholesterol; LDL-C = Low Density Lipoprotein cholesterol; T Chol. = Total Cholesterol; TG = Triglycerides; HMW Apo = High Molecular Weight Adiponectin; LAS = Large Artery Stiffness; MAP = Mean Arterial Pressure; CIMT = Carotid Intima-Media Thickness; CMRS = cardiometabolic risk score; CMRS-Obs = cardio-metabolic risk score excluding the measure of adiposity. Model 1 adjusted for age and stage of pubertal development; Model 2 adjusted for age, stage of pubertal development and objectively measured MVPA; Model 3 adjusted for age, stage of pubertal development, objectively measured MVPA and adiposity (sum of four skinfolds). N = 64 (excluding MAP); Multiple Linear Regression Analysis; \*p < 0.05.

No associations were observed between any of the activity or sedentary variables and the CMRS. The associations between the activity variables independent of adiposity and the CMRS were examined by excluding the sum of four skinfold measurements from the risk score and adjusting for this adiposity measure (CMRS-Obs). No associations were observed between sedentary time, standing time, light intensity physical activity (incl. or excl. standing time) or prolonged sedentary periods and CMRS-Obs.

# 6.4. Discussion

The purpose of this analysis was to determine whether associations exist between levels and patterns of sedentary time and light intensity physical activity and cardio-metabolic risk factors in female adolescents, independent of moderate to vigorous physical activity. The key findings of this chapter suggest that light intensity physical activity which does not include standing time is significantly associated with body mass index, independent of moderate to vigorous physical activity. This chapter has also identified that sedentary time is associated with body mass index in this population when controlling for age and stage of pubertal development, but these findings disappear when correcting for moderate to vigorous physical activity. Finally, an association which approaches significance has been observed between light intensity physical activity and carotid intima-media thickness in this adolescent female population. This chapter also indicates that adolescent females spend the majority of their time in sedentary and standing activities.

# 6. 4. 1. Associations between activPAL Determined Light Intensity Physical Activity and Cardio-Metabolic Risk Factors

The results of this study have highlighted that in this sample, light intensity physical activity is negatively correlated with BMI independent of moderate to vigorous physical activity, while sedentary time, prolonged sedentary periods, standing time and light intensity physical activity including standing time are not significantly associated with any of the cardio-metabolic risk factors when corrected for moderate to vigorous physical activity and adiposity. This suggests that a reduction in the amount of light intensity physical activity is linked to increased body mass index. Furthermore, although sedentary time did not correlate directly with cardio-metabolic risk factors, it may be that sedentary time is a risk when it replaces light intensity physical activity. These findings may have public health and clinical importance. Adverse adiposity is a critical component of metabolic syndrome and an important risk factor for cardiovascular disease (Lloyd-Jones *et al.*, 2010), while large population-based studies in youths have identified alarming trends of increasing levels of obesity, hypertension and type II diabetes mellitus in paediatric populations (Lloyd-Jones *et al.*, 2009).

This is the first study which has examined the independent effect of standing time and light intensity physical activity on cardio-metabolic risk factors using an inclinometer-based activity monitor. Previously, the examination of light intensity physical activity has relied on the use of thresholding for both sedentary time and moderate to vigorous physical activity, which are unable to distinguish between different postural positions such as sitting and standing (Grant et al., 2006), and may also have categorised activities of a lower intensity as sedentary (Davies et al., 2011, Kozey-Keadle et al., 2011). Our findings have identified that light intensity physical activity was inversely associated with BMI independent of moderate to vigorous physical activity, while no associations were observed between standing time or light intensity physical activity including standing time and cardio-metabolic risk factors in this population of adolescent females. Recent physiological evidence has suggested that reduced sedentary time through increased light intensity physical activity (or an increase in the amount of light contractile activity) may have an independent positive effect on an individual's health (Hamilton et al., 2004, Zderic and Hamilton, 2006). It is now accepted that non-exercise activity thermogenesis accounts for the majority of daily energy expenditure, and that increases in the amount of daily light intensity physical activity achieved by an individual can be important in the protection against overweight or obesity (Levine, 2002, Owen et al., 2010). Our findings support these observations, suggesting that light intensity physical activity can protect against sedentary time regardless of participation in moderate to vigorous physical activity, while also having a significant effect on BMI, independent of moderate to vigorous physical activity.

In adults, significant associations have been observed between objectively measured sedentary time and cardio-metabolic risk factors, independent of moderate to vigorous physical activity and other covariates (Ekelund *et al.*, 2007b, Healy *et al.*, 2007, Healy *et al.*, 2008c, Healy *et al.*, 2011b). Researchers have also identified the significant associations of surrogate measures of the duration of sedentary periods, such as breaks in sedentary time and cardio-metabolic risk factors independent of moderate to vigorous physical activity and other covariates in adults (Healy *et al.*, 2008a, Healy *et al.*, 2011b). Studies in youths have observed associations between sedentary time and cardio-metabolic risk factors, yet these associations were not independent of moderate to vigorous physical activity (Ekelund *et al.*, 2007a, Sardinha *et al.*, 2008, Martinez-Gomez *et al.*, 2009). Only two published studies were identified that have examined the associations between sedentary time, prolonged sedentary periods and cardio-metabolic risk factors in child and adolescent populations, independent

of moderate to vigorous physical activity (Carson and Janssen, 2011, Martinez-Gomez et al., 2011a). Similar to the present findings, neither study identified any significant independent associations. As hypothesised by Martinez-Gomez and colleagues (2011), the absence of association between cardio-metabolic risk factors and sedentary variables may be age dependent (Martinez-Gomez et al., 2011a). It is widely acknowledged that weaker associations between activity variables (including moderate to vigorous physical activity) and cardio-metabolic risk factors have been observed in children and adolescents than those observed in adults, as associated diseases do not tend to manifest until adult years (Hallal et al., 2006, Andersen et al., 2011). Additionally, sedentary time has been shown to increase across the lifespan (Janz et al., 2005, Treuth et al., 2005, Matthews et al., 2008), which may further increase the risk of cardiovascular diseases.

# 6. 4. 2. Potential Areas of Further Research: Associations between Activity Variables and Cardio-Metabolic Risk Factors Approaching Significance

Although no further significant relationships were observed during this analysis, relationships which approached significance were observed between sedentary time and carotid intima-media thickness (Table 6.4; Model 3:  $\beta = 0.3$ ; p = 0.07, Adj.  $R^2 = 0.01$ ), independent of moderate to vigorous physical activity. Although insignificant, this relationship may suggest that there is a greater accumulation of atherosclerotic plaques in the large arteries of sedentary adolescents compared to their less sedentary counterparts. This is the first study to examine associations between carotid intima-media thickness and sedentary time, making comparison with existing epidemiological observations difficult. It is well established that initial processes associated with atherosclerosis begin in childhood and develop throughout adolescence and into adulthood, where it is more likely to manifest in the form of myocardial infarction or stroke (Raitakari et al., 2003). If future work identifies a real relationship, it would support the increased use of vascular health measures when examining the relationships between sedentary time, standing time and light intensity physical activity and health in future epidemiological studies. These examinations may present further interesting observations that may provide more evidence on the deleterious effect of sedentary time in childhood and adolescence.

Another relationship which approached significance was observed between standing time and fasting insulin levels (Appendix H; Model 3:  $\beta = -0.19$ ; p = 0.15, Adj.  $R^2 = 0.12$ ), independent of moderate to vigorous physical activity. Epidemiological relationships have been observed between sedentary time and insulin levels in adults, independent of moderate to vigorous physical activity (Healy et al., 2011b). The proposed theory behind these findings supported the work of Hamilton and colleagues, suggesting that fewer skeletal muscle contractions may result in reduced clearance of oral glucose loads from plasma and less glucose-stimulated insulin secretion (Hamilton et al., 2007). The sedentary measure employed by Healy and colleagues (2011) is based on a lack of ambulation rather than postural position (Healy et al., 2011b). Consequently, claims of observing results regarding the contractions of large skeletal muscles in large populations may be premature. Although the independent relationships between standing and insulin presented in this study only approach significance, these findings may be of great importance. The distinction between sitting and standing presented here provides a more accurate measure of large skeletal muscle contraction, resulting in a more accurate insight into the relationship between sedentary time and insulin resistance. Should future studies establish that this trend towards significance for the relationship between sedentary time and insulin is a real effect, then it may be that the increased volume of active skeletal muscle achieved through standing increases plasma glucose clearance and decrease glucose-stimulated insulin secretion (Healy et al., 2008c). Future large-scale epidemiological studies which employ inclinometer-based activity monitors are critical in determining the relationships between activity intensities at the lower end of the spectrum, such as sedentary time, standing time and light intensity physical activity and fasting insulin.

# 6. 4. 3. Associations between Activity Variables Assessed using the activPAL Activity Monitor

One of the key findings of this analysis is that although moderate to vigorous physical activity is negatively associated with sedentary time, the negative association between standing time, light intensity physical activity (excl. standing time) and light intensity physical activity (including standing time) are stronger. Furthermore, moderate to vigorous physical activity did not correlate significantly with standing time, light intensity physical activity excluding standing time and light intensity physical activity including standing time. These findings suggest that light intensity physical activity is reduced more than moderate to vigorous physical activity in female adolescents who are more sedentary, and that sedentary behaviour may be replacing light intensity physical activity rather than moderate to vigorous physical activity. This concurs with recent observations from Mitchell and colleagues (2012), who have highlighted decreases in light intensity physical activity and increases in sedentary time (measured using accelerometer counts below 200 counts per minute) from childhood through to adolescence in both males and females (Mitchell *et al.*, 2012). Furthermore, there is a relatively strong negative correlation between long unbroken periods of sedentary time (greater than 30 minutes) and a reduction in light intensity physical activity. It seems that adolescent girls who sit for long periods during the day have significantly reduced light intensity physical activity, suggesting that prolonged sedentary periods may be displacing other activities that include light intensity physical activity.

No relationship between light intensity physical activity and moderate to vigorous physical activity existed after adjusting for age and stage of pubertal development in this population. This contrasts with similar analysis from other studies in children and adolescent populations, which have documented significant associations between objectively measured light intensity physical activity and moderate to vigorous physical activity (Carson and Janssen, 2011). The observed differences can most likely be attributed to measurement differences. The activPAL is a direct measure of sedentariness, and thus can differentiate between standing time and sitting time, while also differentiating between standing time and light intensity physical activity (Section 2. 10). The ActiGraph (ActiGraph LCC, Pensacola, FL), which relies on the measurement of ambulation, has estimated sedentary time through the use of sedentary thresholds, and is unable to accurately distinguish between sedentary time and standing time (Hart et al., 2011, Kozey-Keadle et al., 2011). Studies which have examined and compared both devices as measures of sedentary time have acknowledged the improved accuracy the activPAL physical activity monitor, while also recommending its use as a measure of sedentary behaviour (Davies et al., 2011, Kozey-Keadle et al., 2011, Ridgers et al., 2012). The activPAL has also been identified as a valid and reliable measure of postural allocation (Grant et al., 2006) and as a measure of physical activity intensity within this population (Section 3. 5). These population-based findings suggest that a device which can examine both activities at the low and high end of the

activity intensity continuum should be employed when making associations between freeliving activity variables and indices of health.

#### 6. 4. 4. Strengths and Limitations:

A particular strength of this study was the use of an objective measure of ambulation with an inbuilt direct measure of postural position, which enabled the accurate and reliable classification of activities including sedentary, standing, light and moderate to vigorous physical activity. We also measured some novel cardio-metabolic risk factors which have not previously been examined in relation to adolescent sedentary time. However, the limitations of this study must be acknowledged. We are unable to make causal inference regarding the identified relationships due to the cross sectional design of this study. Although the sample was randomly selected, the sample size was small and healthy. Significant large-scale epidemiological studies which employ inclinometer-based activity monitors are necessary to make more powerful and substantial statements regarding the associations between sedentary patterns and behaviours, light intensity physical activity and cardio-metabolic risk factors in adolescent females. This study only focused on female adolescents due to their high levels of physical inactivity. Similar examinations are necessary in children and male adolescents, to inform sedentary and physical activity guidelines in young populations. Only two confounders were included in our statistical analysis, age and stage of pubertal development. The use of further confounder variables such as socio-economic status, smoking status and diet would have strengthened the results obtained from our linear regression models, but may not have been appropriate given the relatively low population numbers.

## 6.5. Conclusions

These results are the first to categorise sedentary time, prolonged sedentary periods and light intensity physical activity using an inclinometer-based activity monitor, while also observing an independent association between one of these measured variables and a cardio-metabolic risk factor. A strong negative correlation between prolonged periods of
sitting/lying and light physical activity was observed, suggesting that sitting is displacing light intensity physical activity, irrespective of daily moderate to vigorous physical activity. Additionally, an independent association has been identified between light intensity physical activity (which did not include standing time) and body mass index in an adolescent female population. Taken together, these results suggest that targeting moderate to vigorous physical activity alone may not be a fully effective strategy in changing health risk in this population. Reducing prolonged sitting, which in this population appeared to displace light intensity physical activity, may be an effective target for intervention. These findings add some support to recently published sedentary guidelines, yet further epidemiological evidence from a range of ages, sexes, ethnicities and clinical groups are required to provide more detailed and effective guidelines on the optimum activity profile. **Chapter 7. General Discussion** 

## 7.1. Introduction

The literature review presented in Chapter 2 of this thesis has described existing research on the importance of sedentary patterns and physical activity behaviours to indices of health in all populations. The examination of sedentary patterns and behaviours and physical activities of differing intensity using a range of assessment methodologies was also examined, while the implications of using such measures when making associations between sedentary and light intensity physical activity variables and health were also discussed. The key findings from research studies completed in this thesis, which aimed to address some of the issues raised, have been summarised as:

- The activPAL Professional Physical Activity Monitor (an inclinometer-based activity monitor) is a valid and reliable measure of physical activity intensity in an adolescent female population. As a result, its use within this population as a measure of free-living physical activity is recommended (discussed in Section 7. 2).
- Through determining a threshold for moderate to vigorous physical activity for the activPAL, we have enabled the examination of light intensity physical activity, which can distinguish between sitting time and standing time, for the first time in any population (discussed in Section 7. 3).
- In a cross-sectional examination of physical activity variables and cardio-metabolic risk factors in an adolescent female population, light intensity physical activity (quantified using an inclinometer-based activity monitor) is significantly associated with body composition in an adolescent female population, independent of moderate to vigorous physical activity (discussed in Section 7. 4).

These key findings will be discussed throughout this chapter. Additionally, important areas of future research for physical activity measurement and physical activity and health-related research will be outlined. The limitations of this thesis will be outlined, and potential methods of overcoming these limitations will be suggested. Finally, the conclusion of this thesis will be presented

# 7.2. The Importance of examining all Intensities of the Activity Intensity Spectrum

A wealth of research has examined the relationships between moderate to vigorous physical activity and health in childhood, adolescence and adulthood. It is now widely accepted that increased levels of moderate to vigorous physical activity in childhood and adolescence reduce the risk of mortality from cardiovascular disease and related disorders in adulthood (Paffenbarger Jr *et al.*, 1986, Blair *et al.*, 1989b). However, it is plausible that the interaction between the entire activity profile, rather than specific activity intensities, may be the most effective method of associating habitual physical activity with indices of health (Owen *et al.*, 2000, Hamilton *et al.*, 2004, Hamilton *et al.*, 2007). It has been demonstrated that the dominant determinant of variability in total daily energy expenditure is not moderate to vigorous physical activity, but rather light intensity physical activity (Donahoo *et al.*, 2004, Levine *et al.*, 2006, Tremblay *et al.*, 2007), while evidence also suggests that there is a unique physiology of sedentary time, whereby biological processes differ from those experienced during exercise training (Hamilton *et al.*, 2001, Bey and Hamilton, 2003, Zderic and Hamilton, 2006).

#### 7. 2. 1. Limitations of Existing Measures of Sedentary Behaviours

In order to effectively examine relationships between indices of health and physical activity, valid and reliable measures are necessary (Welk, 2005, Bassett Jr *et al.*, 2012, Freedson *et al.*, 2012). Until now, physical activity and sedentary behaviours have predominantly been examined using subjective measures such as self-report or objective measures using accelerometry. Self-report measures of physical activity and sedentary behaviours in child and adolescent populations have inherent limitations, including self-reporting bias and recall difficulty (Trost, 2007, Chinapaw *et al.*, 2010, Marshall and Ramirez, 2011). Additionally, the validity of a large proportion of self-reported sedentary measures has not been examined (Lubans *et al.*, 2011), while self-reported screen-based activities only examine one type of sedentary behaviour (TV viewing or computer use) and fail to capture the complete sedentary profile (Biddle *et al.*, 2009, Bassett Jr *et al.*, 2010, Owen *et al.*, 2010).

Objective measures of physical activity and sedentary behaviours, such as accelerometry, have demonstrated improved validity and reliability compared to self-report (Westerterp, 1999, Trost, 2001, Trost et al., 2005, Warren et al., 2010). However, it is critical that a measurement instrument is effective at measuring what it is intended to measure (Chinapaw et al., 2010). The majority of existing accelerometers, such as the ActiGraph GT1M and GT3X (ActiGraph LCC, Pensacola, FL), examine accelerations of a body relative to free-fall, and have been identified as a valid measure of physical activity (Esliger et al., 2005, Warren et al., 2010). However, such devices estimate sedentary time using activity thresholds, whereby an epoch with activity counts below a predetermined threshold is quantified as sedentary time (Matthews et al., 2008, Matthews et al., 2012). Sedentary behaviour is now defined as energy expenditure below 1.5 METs while in a sitting or lying position during waking hours (Owen et al., 2010, Marshall and Ramirez, 2011, Tremblay et al., 2012). Accelerometer determined estimates of sedentary time are based on a lack of ambulation, and are unable to distinguish between sitting and lying (Hart et al., 2011, Kozey-Keadle et al., 2011, Marshall and Ramirez, 2011). Consequently, conclusions on the relationship between sedentary behaviours and indices of health from surrogate measures of sedentariness, such as self-reported TV viewing (Pate et al., 2008, Owen et al., 2010) and from measures reliant on the lack of ambulation (Kozey-Keadle et al., 2011, Marshall and Ramirez, 2011, Ridgers et al., 2012), must be interpreted with caution.

# 7. 2. 2. Inclinometer-Based Activity Monitors as a Measure of Physical Activity

Due to the limitations of self-reported and accelerometer determined sedentary time outlined above, the use of valid and reliable inclinometer-based accelerometers has been encouraged (Kozey-Keadle *et al.*, 2011, Bassett Jr *et al.*, 2012). One such device, the activPAL<sup>TM</sup> Professional Physical Activity Monitor (PAL technologies Ltd., Glasgow, UK), has been validated for the examination of sedentary time in both child (Ridgers *et al.*, 2012) and adult populations (Kozey-Keadle *et al.*, 2011). As identified above, evidence would now advocate that researchers should examine the entire activity profile, including both sedentary and ambulatory activity. Preliminary findings from Harrington and colleagues (2011 b) have identified very strong correlations between the activPAL activity count

function and energy expenditure (Harrington *et al.*, 2011b). However, until now, no study has examined the criterion and concurrent validity of the activPAL activity monitor or has completed a value calibration of the device to enable the valid and reliable examination of both physical activity and sedentary behaviours.

Due to the large amount of available literature on the use of the ActiGraph as a measure of physical activity, it may be suggested that two devices be worn to examine the full activity spectrum, with objective measures of ambulation measuring physical activity (e.g. ActiGraph) and with valid inclinometer-based devices measuring sedentary behaviour (activPAL). However, it would be expected that the increased burden of wearing two activity monitoring devices for prolonged periods would affect compliance rates. To date, only one study has simultaneously employed both the activPAL and the Actigraph as measures of habitual physical activity in a sample of children (Martin et al., 2011). Results from this study identified high levels of non-compliance, with 23 out of 32 children providing at least 3 valid days of accelerometry data (which would not be classified as a stringent minimum wear protocol). Additionally, wear protocols differ significantly among both measures, whereby activPAL can be worn throughout sleep hours while ActiGraph is normally removed during sleep hours. This required collating of data from two separate devices may pose further difficulties with data processing and analysis. Meanwhile, studies which have employed activPAL only (including Chapter 6 of this thesis) have identified high levels of compliance in adolescent populations (Harrington et al., 2011a). The findings presented in Chapter 3 of this thesis have now identified the activPAL as a valid estimate of energy expenditure, while the activPAL also appears to be at least as effective at examining ambulation as the ActiGraph GT3X in an adolescent female population. These findings have provided researchers with the potential to examine both sedentary time and physical activity using a single measurement device. Coupled with the high levels of compliance, this thesis supports the use of the activPAL as a measure of physical activity and sedentary behaviour in future population-based studies in adolescent females.

#### 7. 2. 3. Limitations of the activPAL Activity Monitor

Although the activPAL has previously been identified as a valid and reliable measure of posture in child (Davies *et al.*, 2011) and adult populations (Grant *et al.*, 2006), and as a valid

measure of steps in adolescents (Harrington *et al.*, 2011b) and adults (Ryan *et al.*, 2006, Grant *et al.*, 2008) and now as a measure of physical activity (using accelerometer counts) in adolescent females (Chapter 3), it is important to consider the limitations of using an accelerometer-based device as a measure of physical activity. Similar to existing activity monitors, the activPAL is unable to examine energy expenditure from upper body activities. Devices which examine physical activity using accelerometers are also not accurate at measuring cyclic movements like bicycle riding and rowing. Consequently, such devices will underestimate levels of physical activity in individuals that participate in high levels of such activities. Furthermore, although efforts can be made to waterproof the activPAL, enabling the device to be worn during water-based activities, there is no evidence of its effectiveness at examining water-based activities such as swimming. Each of these limitations must be considered when using the activPAL activity monitor in free-living environments. It is also important for additional activPAL validation studies be completed in child, male adolescent and adult populations before the use of the devices as a measure of physical activity is encouraged in large-scale epidemiological research.

#### 7. 3. The Examination of Light Intensity Physical Activities

As discussed in Section 2. 3, the accurate and reliable examination of light intensity physical activity is desirable, as it has been suggested that activities across the activity intensity spectrum should be considered when examining the associations between habitual activity and health (Levine *et al.*, 2006, Hamilton *et al.*, 2007, Hamilton *et al.*, 2008). Unfortunately, light intensity physical activity is potentially the most difficult activity intensity to quantify in free-living epidemiological research (Healy *et al.*, 2007). The examination of a ubiquitous behaviour such as light intensity physical activity using self-report suffers similar bias and recall limitations experienced when examining sedentary and moderate to vigorous physical activity in child and adolescent populations (Shephard, 2003, Trost, 2007, Chinapaw *et al.*, 2010, Marshall and Ramirez, 2011). Furthermore, objectively determined light intensity physical activity using sedentary time, whereby very light intensity physical activities (below the predetermined sedentary threshold) may be determined as sedentary time, while standing activities are also often miscategorised as sedentary time (Hart *et al.*, 2011, Kozey-Keadle *et* 

*al.*, 2011). As a result, the ability to accurately determine associations between light intensity physical activity and health remain difficult.

The identification of the activPAL as a valid measure of physical activity intensities in Chapter 3 has not only enabled the examination of habitual moderate and vigorous physical activity in adolescent females, but also enabled the examination of light intensity physical activity in this population. Light intensity physical activity is characterised as activity behaviours that are not sedentary (< 1.5 METs in a sitting and lying posture) that have an energy expenditure of < 2.9 METs (Owen *et al.*, 2010). The accurate distinction between sitting time, standing time and stepping time of the activPAL allows users to accurately determine sedentary time, while the activity intensity threshold developed in Section 3. 3. 5 enable the determination of time spent in moderate to vigorous physical activity. The remaining waking behaviours can therefore be classified as standing time and light intensity physical activity. This provides a more accurate estimate of light intensity physical activity, which can then be quantified both including and excluding standing time. The resulting variable provides researchers with the opportunity to examine the effect of light intensity physical activity on indices of health in adolescent female populations. The validation of the activPAL as a measure of physical activity intensities in other groups (e.g. adults, adolescent males, children etc.) is necessary prior to the examination of free-living light intensity physical activity in these populations.

# 7. 4. The Associations between Subcomponents of Habitual Physical Activity and Indices of Health in an Adolescent Female Population

The primary purpose of physical activity assessment in epidemiological research is to examine exposures to specific activity behaviours, examine associations between these behaviours and indices of health and to examine whether changes in these behaviours due to intervention have a significant effect on indices of health (Freedson *et al.*, 2012). A wealth of evidence is available on the validity of examining moderate to vigorous physical activity, on the prevalence and trends of this behaviour across all populations and on the association between this behaviour and health. As described in Section 2. 9, fundamental limitations of existing measures of sedentary time and light intensity physical activity have

provided cause for concern when interpreting identified associations between these variables and indices of health. The use of valid and reliable multi-sensor activity monitors, which combine accelerometry with inclinometry, have been encouraged due to increased functionality in examining sedentary time (Esliger and Tremblay, 2007, Bassett Jr *et al.*, 2010, Kozey-Keadle *et al.*, 2011).

Chapter 5 and Chapter 6 of this thesis have examined free-living physical activity variables, including sedentary time and light intensity physical activity, using a valid and reliable inclinometer-based activity monitor (the activPAL<sup>TM</sup> Professional Physical Activity Monitor) in an adolescent female population. These chapters have examined the associations between these activity variables and cardio-metabolic risk factors, independent of moderate to vigorous physical activity. This type of analysis is critical when determining the associations between activity intensities at the lower end of the continuum and cardio-metabolic risk factors, as moderate to vigorous physical activity is an independent predictor of cardio-metabolic risk factors (Ekelund *et al.*, 2006).

#### 7. 4. 1. Sedentary Time and Cardio-Metabolic Risk Factors

The overall findings of Chapter 5 and Chapter 6 suggest that sedentary time is not associated with the majority of cardio-metabolic risk factors in adolescent females. However, Chapter 5 does highlight that the most sedentary adolescent females are significantly more likely to have higher levels of adiposity than their least sedentary counterparts. These results are in contrast with previous observations. Carson and Janssen (2011) identified no associations between accelerometer determined sedentary time and cardio-metabolic risk factors, independent of moderate to vigorous physical activity (Carson and Janssen, 2011). As discussed in Chapter 5 a potential cause for this may have been the use of additional confounders during statistical analysis. Another potential cause for observed differences may be the disparity in participant numbers between the two studies (n = 194 Vs. n = 2527). However, the different methods of examining sedentary time must also be considered as a potential cause for the contrasting results, as the device employed by Carson and Janssen (2011) was unable to distinguish between sitting and standing (Kozey-Keadle *et al.*, 2011) and has been shown to overestimate sedentary time (Hart *et al.*, 2011). Additionally, in Chapter 6, associations that approached significance

were also observed between sedentary time and carotid intima-media thickness, which is a high powered measure of arterial health. Although the association was clearly nonsignificant, it is plausible that the small sample size included in this analysis may be the cause of this "almost" association. These findings are by no means conclusive of the association between sedentary time and indices of health in adolescent female populations. However, they do suggest that this is a worthy area of further research. Further large-scale examinations of adolescent female populations which 1) employ inclinometer-based activity monitors, 2) examine a wide range of high powered health measures and 3) collect information on a broader range of covariates (i.e. smoking status, socio-economic status and dietary information) are required to further examine these associations.

# 7. 4. 2. Light Intensity Physical Activity and Cardio-Metabolic Risk Factors

Chapter 5 and Chapter 6 provide consistent evidence on the positive associations between light intensity physical activity and indices of health in adolescent females. These chapters have both identified significant associations between adiposity and light intensity physical activity in an adolescent female population, independent of moderate to vigorous physical activity. This association is of great importance, as this is the first examination of light intensity physical activity using an assessment method which can distinguish between sitting time, standing time and stepping time. Although associations between objectively measured light intensity physical activity and 2-hr plasma glucose have been identified in an adult population (Healy et al., 2007), these examinations have not distinguished between sitting time, standing time and stepping time. The use of devices that do not detect sitting time, and do not distinguish between sitting and standing, tend to significantly overestimate sedentary time and light intensity physical activity (Hart et al., 2011). To determine whether these associations are evident in adult populations, further large-scale epidemiological studies, which employ a more accurate and effective inclinometer-based activity monitor are necessary. Additionally, further research studies of a larger scale with a greater range of measures are required in both adolescent females and males to provide more support to the associations identified in this thesis.

#### 7. 5. Potential Translation of Research Findings

The findings presented in this research thesis have significant implications to the health of the general population. For the first time, associations have been observed between objectively assessed light intensity physical activity and sedentary behaviours and indices of health. As highlighted in section 7. 4, light intensity physical activity has a negative association with adiposity, while sedentary time is positively associated with adiposity in this sample of adolescent females. The implications of these findings have the potential to inform future national and international physical activity recommendations. The existing physical activity recommendations, which promote increasing participation in moderate to vigorous physical activity, is based on 5 decades of research, and its continued promotion is essential in the area of physical activity and health. However, the inclusion of an additional recommendation that advocates increasing the amount of time spent in light intensity physical activity at the expense of the amount of time spent in sedentary behaviours may be appropriate. Changes in light intensity physical activity behaviour may be a more achievable than changes in moderate to vigorous physical activity. Consequently, the inclusion of this additional recommendation may be perceived as a more realistic and inviting initiative to the most inactive of our populations.

### 7.6. Limitations

• According to best practice and research recommendations for the validation and value calibration of activity monitoring devices, the use of a broad, representative population is necessary. Unfortunately, our validation of the activPAL only included female adolescents between the ages of 15 and 18 years. For this validation study, the use of a broader population, which included females between the age of 13 and 15 and male participants of the same age range, would have made the results of the study more generalisable. Additionally, the activities included in the validation could have included more free-living activities, including household chores, stair ascending/descending and game based activities (e.g. basketball etc.).

- As is the case with all existing devices that measure physical activity using accelerometers, the activPAL was unable to identify non-wear time. As a result, non-wear protocols followed the methodologies employed with other activity monitors (e.g. ActiGraph), whereby 60 minutes of consecutive zero counts were identified as non-wear time. Resulting non-wear time was then deleted. The accuracy of this methodology has not been investigated in the activPAL, and further examination and analysis is required to determine the validity of such estimates of non-wear time in all populations.
- The limitations of the activPAL have already been described in Section 7. 2. 3. However, it is important to reemphasise the inability of this device to examine upper body activities (strength training, carrying loads etc.), water based activity (swimming) and lower body activities which involve a cyclic motion (cycling, rowing). The inability of devices to examine such activities may result in the underestimation of physical activity and energy expenditure, particularly in individuals that perform large amounts of the described activities. The use of an activity log, which encouraged participants to record when they were involved in particular activities may reduce the effect of this limitation. Additionally, the introduction of a physiological variable such as heart rate or heat flux sensor would also eliminate the underestimation of physical activity or energy expenditure when using an accelerometer-based device.
- The small sample size used in Chapter 6 must be acknowledged as a limitation of this work. Due to a relatively low volunteer rate, coupled with absentees on test days, the sample size was reduced to a total of 64 participants. A larger, more representative sample is required to make more statistically powerful associations between activity variables and cardio-metabolic risk factors in an adolescent female population. The results of Chapter 6 should be viewed as pilot work for larger studies, which encourage the further investigation of the effects of sedentary behaviours and physical activity on health in large-scale epidemiological studies.
- The non-examination of additional variables that are significant predictors of noncommunicable diseases, including smoking, dietary information and parental health is a significant limitation when examining the associations of activity variables and indices of health, particularly cardio-metabolic biomarkers. Through adjusting for

additional covariates, a more powerful analysis of the associations between different activity variables and health measures is possible.

## 7.7. Future Research

The results of this thesis have addressed some important research questions outlined by Owen and colleagues (2010), including 1) the identification of a valid measure of sedentary time and physical activity, 2) the development of methodologies to examine the output from this measure, 3) the identification of the type and intensity of activity that is most protective against sedentary time and 4) the examination of relationships between the observed activity variables and indices of health in a specific population (Owen *et al.*, 2010). However, it is important to acknowledge that there remain many important research questions relating to the sedentary behaviour paradigm that must be addressed. Some of the most pressing of these include:

- The validation of the activPAL (or alternative inclinometer-based activity monitor) as a measure of physical activity in broader populations (e.g. represent different ages, gender and ethnicities). This validation work is necessary before the activPAL can be considered a potential measure of free-living activity in epidemiological research.
- The development of valid and reliable methodologies to determine non-wear time is critical to the future use of the activPAL and all other activity monitoring devices. Without such developments, researchers may continue to under/overestimate specific activity variables, particularly sedentary time, standing time and light intensity physical activity.
- The further examination of associations between sedentary time, standing time, light intensity physical activity and moderate to vigorous intensity physical activity and indices of health in populations of all ages, sexes and ethnicities using inclinometer-based devices is essential. This information will strengthen existing sedentary behaviour/disease findings.

- The development and implementation of longitudinal studies, which will examine the stability of activity variables from childhood and adolescence through to adulthood, are critical. Such studies will provide changes in patterns and trends of activity variables over time, while also highlighting how changes in these patterns and trends affect health over time. In particular, the tracking of variables that have not previously been directly examined, including sedentary time, standing time and light intensity physical activity would be particularly informative.
- The quantification of the relative risk associated with different activity variables is of great public health importance (e.g. does the amount of time spent sedentary have the same importance to health as the amount of time spent in moderate to vigorous physical activity, or how much relative weight does sedentary time have compared to moderate to vigorous physical activity). Such examinations will inform whether 1) additional sedentary and light intensity physical activity recommendations should be included in future public health guidelines and 2) what these sedentary and light intensity physical activity guidelines should include.
- The valid and reliable identification of the determinants of sedentary behaviour is critical for the future of sedentary behaviour research. Such examinations are critical in determining why sedentariness varies between individuals and why sedentariness increases or decreases at specific time points throughout the lifespan. Already, researchers are implementing interventions which aim to reduce sedentary time, while little information is available on the determinants of sedentary behaviours. Over the coming years, a plethora of research will aim to reduce sedentary time through increased light, moderate or vigorous intensity physical activity. The early identification of such determinants will inform future sedentary intervention studies.
- The examination of the effectiveness of a range of different sedentary interventions in representative populations for age, sex and ethnicity are necessary. These interventions will seek to reduce sedentary behaviour through increased standing time, light intensity physical activity or moderate to vigorous physical activity, and will provide further insight into the causation and treatment of risk factors associated with physical activity, physical inactivity and sedentary patterns and

behaviours. It is essential that such intervention studies are informed by solid, strong scientific information.

## 7.8. Conclusion

This Ph.D. thesis has identified the activPAL<sup>TM</sup> Professional Physical Activity Monitor as a valid and reliable measure of physical activity in an adolescent female population, and has recommended its use in the examination of free-living physical activity and sedentary behaviour within this population. For the first time, the associations between subcomponents of physical activity, including sedentary time, standing time and light intensity physical activity, have been determined using an inclinometer-based activity monitor. Additionally, for the first time, the quantified values of sedentary time, standing time and light intensity physical activity (both including and excluding standing time) have been associated with indices of health. The findings of this thesis suggest that light intensity physical activity guards against sedentary time to a greater extent than moderate to vigorous physical activity. Furthermore, light intensity physical activity (that does not include standing time) is significantly associated to adiposity and carotid intima media thickness, while limited associations are evident between sedentary patterns or behaviours and cardiometabolic risk factors in an adolescent female population. The results of this thesis encourage the reduction of sedentary time through increased participation in light intensity physical activity as a potential method to improve cardio-metabolic risk factors in an adolescent female population.

#### Bibliography

- Ainsworth, B. E., Haskell, W. L., Herrmann, S. D., Meckes, N., Bassett Jr, D. R., Tudor-Locke, C., Greer, J. L., Vezina, J., Whitt-Glover, M. C. & Leon, A. S. (2011). Compendium of Physical Activities: A Second Update of Codes and MET Values. *Medicine and Science in Sports and Exercise*, 43 (8), 1575-1581.
- Alpert, B. S. & Wilmore, J. H. (1994). Physical Activity and Blood Pressure in Adolescents. *Pediatric Exercise Science*, 6 (4), 361-380.
- American College of Sports Medicine 2007. ACSM's Guidelines for Exercise Testing and Prescription, Baltimore, MD., Lippincott Williams and Wilkins.
- Andersen, L. B., Harro, M., Sardinha, L. B., Froberg, K., Ekelund, U., Brage, S. & Anderssen, S. A. (2006). Physical activity and clustered cardiovascular risk in children: a cross-sectional study (The European Youth Heart Study). *The Lancet*, 368 (9532), 299-304.
- Andersen, L. B., Riddoch, C., Kriemler, S. & Hills, A. (2011). Physical activity and cardiovascular risk factors in children. *British Journal of Sports Medicine*, 45 (11), 871-876.
- Anding, J. D., Kubena, K. S., Mcintosh, W. & O'brien, B. (1996). Blood Lipids, Cardiovascular Fitness, Obesity and Blood Pressure:: The Presence of Potential Coronary Heart Disease Risk Factors in Adolescents. *Journal of the American Dietetic* Association, 96 (3), 238-242.
- Antonopoulos, S. (2002). Third report of the National Cholesterol Education Program (NCEP) expert panel on detection, evaluation, and treatment of high blood cholesterol in adults (Adult Treatment Panel III) final report. *Circulation*, 106, 3143-3421.
- Armstrong, N. (1998). Young people's physical activity patterns as assessed by heart rate monitoring. *Journal of Sports Sciences*, 16 (1), S 9-16.
- Armstrong, N. & Simons-Morton, B. (1994). Physical activity and blood lipids in adolescents. *Pediatric Exercise Science*, 6 (4), 381-405.
- Assmann, G., Schulte, H., Von Eckardstein, A. & Huang, Y. (1996). High-density lipoprotein cholesterol as a predictor of coronary heart disease risk. The PROCAM experience and pathophysiological implications for reverse cholesterol transport. *Atherosclerosis*, 124, S 11-20.
- Bailey, R. C., Olson, J., Pepper, S. L., Porszasz, J., Barstow, T. J. & Cooper, D. (1995). The level and tempo of children's physical activities: an observational study. *Medicine and Science in Sports and Exercise*, 27 (7), 1033-1041.

- Bao, W., Srinivasan, S. R., Wattigney, W. A. & Berenson, G. S. (1994). Persistence of multiple cardiovascular risk clustering related to syndrome X from childhood to young adulthood: the Bogalusa Heart Study. *Archives of Internal Medicine*, 154 (16), 1842-1847.
- Bao, W., Threefoot, S. A., Srinivasan, S. R. & Berenson, G. S. (1995). Essential hypertension predicted by tracking of elevated blood pressure from childhood to adulthood: the Bogalusa Heart Study. *American Journal of Hypertension*, 8 (7), 657-665.
- Bar-Or, O., Foreyt, J., Bouchard, C., Brownell, K. D., Dietz, W. H., Ravussin, E., Salbe, A. D., Schwenger, S., St Jeor, S. & Torun, B. (1998). Physical activity, genetic, and nutritional considerations in childhood weight management. *Medicine and Science in Sports and Exercise*, 30 (1), 2-10.
- Baranowski, T. (1984). Reliability and Validity of Self Report of Aerobic Activity: Family Health Project. *Research Quarterly for Exercise and Sport*, 55 (4), 309-317.
- Baranowski, T., Hooks, P., Tsong, Y., Cieslik, C. & Nader, P. R. (1987). Aerobic physical activity among third-to sixth-grade children. *Journal of Developmental & Behavioral Pediatrics*, 8 (4), 203-206.
- Barreira, T. V., Kang, M., Caputo, J. L., Farley, R. S. & Renfrow, M. S. (2009). Validation of the Actiheart monitor for the measurement of physical activity. *International Journal* of Exercise Science, 2 (1), 60-71.
- Bassett Jr, D. R., Freedson, P. & Kozey, S. (2010). Medical hazards of prolonged sitting. Exercise and Sport Sciences Reviews, 38 (3), 101-102.
- Bassett Jr, D. R., Rowlands, A. & Trost, S. G. (2012). Calibration and validation of wearable monitors. *Medicine and Science in Sports and Exercise*, 44 (1), S 32-38.
- Bassett Jr, D. R. & Strath, S. J. 2002. Use of Pedometers to Assess Physical Activity. In: Welk, G. J. (ed.) Physical Activity Assessments for Health Related Research. Champaign, IL: Human Kinetics Publishers.
- Bauman, A., Ainsworth, B. E., Sallis, J. F., Hagströmer, M., Craig, C. L., Bull, F. C., Pratt, M., Venugopal, K., Chau, J. & Sjöström, M. (2011). The Descriptive Epidemiology of Sitting: A 20-Country Comparison Using the International Physical Activity Questionnaire (IPAQ). *American Journal of Preventive Medicine*, 41 (2), 228-235.
- Belton, S., Brady, P., Meegan, S. & Woods, C. (2010). Pedometer step count and BMI of Irish primary school children aged 6-9 years. *Preventive Medicine*, 50 (4), 189-192.
- Belton, S. & Macdonncha, C. 2007. The physical activity levels of children aged 7–9 years. Available: <u>http://www.irishsportscouncil.ie/Research/Physical Education Physical Activity</u> <u>and Youth Sport PEPAYS Forum/Proceedings of Second PE PAYS Forum</u> <u>/PAYS Forum Proceedings.pdf</u>.

- Bennett, J. A., Winters-Stone, K., Nail, L. & Scherer, J. (2006). Definitions of sedentary in physical-activity-intervention trials: a summary of the literature. *Journal of Aging and Physical Activity*, 14 (4), 456-477.
- Berggren, E. & Christensen, E. (1950). Heart Rate as a Means of Measuring Metabolic Rate in Man. *Arbeitsphysiologie*, 14, 255-260.
- Berghöfer, A., Pischon, T., Reinhold, T., Apovian, C. M., Sharma, A. M. & Willich, S. N. (2008). Obesity prevalence from a European perspective: a systematic review. *BMC Public Health*, 8 (1), 200-210.
- Berlin, J. A. & Colditz, G. A. (1990). A meta-analysis of physical activity in the prevention of coronary heart disease. *American Journal of Epidemiology*, 132 (4), 612-628.
- Bernstein, M. S., Morabia, A. & Sloutskis, D. (1999). Definition and prevalence of sedentarism in an urban population. *American Journal of Public Health*, 89 (6), 862-867.
- Bey, L., Akunuri, N., Zhao, P., Hoffman, E. P., Hamilton, D. G. & Hamilton, M. T. (2003). Patterns of global gene expression in rat skeletal muscle during unloading and lowintensity ambulatory activity. *Physiological Genomics*, 13 (2), 157-167.
- Bey, L., Areiqat, E., Sano, A. & Hamilton, M. T. (2001). Reduced lipoprotein lipase activity in postural skeletal muscle during aging. *Journal of Applied Physiology*, 91 (2), 687-692.
- Bey, L. & Hamilton, M. T. (2003). Suppression of skeletal muscle lipoprotein lipase activity during physical inactivity: a molecular reason to maintain daily low-intensity activity. *The Journal of Physiology*, 551 (2), 673-682.
- Biddle, S. J. H., Gorely, T. & Marshall, S. J. (2009). Is television viewing a suitable marker of sedentary behavior in young people? *Annals of Behavioral Medicine*, 38 (2), 147-153.
- Biddle, S. J. H., Pearson, N., Ross, G. M. & Braithwaite, R. (2010). Tracking of sedentary behaviours of young people: a systematic review. *Preventive Medicine*, 51 (5), 345-351.
- Bitar, A., Vermorel, M., Fellmann, N., Bedu, M., Chamoux, A. & Coudert, J. (1996). Heart rate recording method validated by whole body indirect calorimetry in 10-yr-old children. *Journal of Applied Physiology*, 81 (3), 1169-1173.
- Blair, S. N., Clark, D., Cureton, K. J. & Powell, K. E. (1989a). Exercise and fitness in childhood: implications for a lifetime of health. *Perspectives in Exercise Science and Sports Medicine*, 2, 401-430.
- Blair, S. N., Kampert, J. B., Kohl, H. W., Barlow, C. E., Macera, C. A., Paffenbarger, R. S. & Gibbons, L. W. (1996). Influences of cardiorespiratory fitness and other precursors on cardiovascular disease and all-cause mortality in men and women. *JAMA: the Journal of the American Medical Association*, 276 (3), 205-210.

- Blair, S. N., Kohl, H. W., Paffenbarger, R. S., Clark, D. G., Cooper, K. H. & Gibbons, L. W. (1989b). Physical fitness and all-cause mortality. *JAMA: the Journal of the American Medical Association*, 262 (17), 2395-2401.
- Booth, F. W. & Chakravarthy, M. V. (2002). Cost and consequences of sedentary living: New battleground for an old enemy. *President's Council on Physical Fitness and Sports Research Digest*, 3 (16), 1-8.
- Boreham, C., Twisk, J., Neville, C., Savage, M., Murray, L. & Gallagher, A. (2002). Associations between physical fitness and activity patterns during adolescence and cardiovascular risk factors in young adulthood: the Northern Ireland Young Hearts Project. *International Journal of Sports Medicine*, 23 (1), 22-26.
- Bots, M. L., Evans, G. W., Riley, W. A. & Grobbee, D. E. (2003). Carotid intima-media thickness measurements in intervention studies. *Stroke*, 34 (12), 2985-2994.
- Bots, M. L., Hoes, A. W., Koudstaal, P. J., Hofman, A. & Grobbee, D. E. (1997). Common carotid intima-media thickness and risk of stroke and myocardial infarction: the Rotterdam Study. *Circulation*, 96 (5), 1432-1437.
- Bouchard, C., Depres, J. P. & Tremblay, A. (1993). Exercise and Obesity. *Obesity Reviews*, 1 (2), 133-147.
- Brage, S., Brage, N., Franks, P., Ekelund, U. & Wareham, N. (2005). Reliability and validity of the combined heart rate and movement sensor Actiheart. *European Journal of Clinical Nutrition*, 59 (4), 561-570.
- Brage, S., Wedderkopp, N., Ekelund, U., Franks, P. W., Wareham, N. J., Andersen, L. B. & Froberg, K. (2004). Features of the metabolic syndrome are associated with objectively measured physical activity and fitness in Danish children. *Diabetes Care*, 27 (9), 2141-2148.
- Brouns, F. 2002. Essentials of Sports Nutrition, Chichester, England, Wiley.
- Bryant, M., Lucove, J., Evenson, K. & Marshall, S. (2007). Measurement of television viewing in children and adolescents: a systematic review. *Obesity Reviews*, 8 (3), 197-209.
- Brychta, R., Wohlers, E., Moon, J. & Chen, K. (2010). Energy expenditure: measurement of human metabolism. *Engineering in Medicine and Biology Magazine*, 29 (1), 42-47.
- Butte, N. F., Ekelund, U. & Westerterp, K. R. (2012). Assessing physical activity using wearable monitors: measures of physical activity. *Medicine and Science in Sports and Exercise*, 44 (1), S 5-12.

- Calabró, M. A., Welk, G. J. & Eisenmann, J. C. (2009). Validation of the SenseWear Pro Armband algorithms in children. *Medicine and Science in Sports and Exercise*, 41 (9), 1714-1720.
- Carson, V. & Janssen, I. (2011). Volume, Patterns, and types of sedentary behavior and cardio-metabolic health in children and adolescents: a cross-sectional study. *BMC Public Health*, 11 (1), 274-284.
- Caspersen, C. J., Powell, K. E. & Christenson, G. M. (1985). Physical activity, exercise, and physical fitness: definitions and distinctions for health-related research. *Public Health Reports*, 100 (2), 126-131.
- Ceesay, S. M., Prentice, A. M., Day, K. C., Murgatroyd, P. R., Goldberg, G. R., Scott, W. & Spurr, G. (1989). The use of heart rate monitoring in the estimation of energy expenditure: a validation study using indirect whole-body calorimetry. *British Journal of Nutrition*, 61 (2), 175-186.
- Central Statistics Office. 2004. Vital Statistics, 4th Quarter and Yearly Summary 2004. Dublin: Stationary Office.
- Chastin, S. & Granat, M. H. (2010). Methods for objective measure, quantification and analysis of sedentary behaviour and inactivity. *Gait & posture*, 31 (1), 82-86.
- Chastin, S. F. M., Dall, P. M., Tigbe, W. W., Grant, M. P., Ryan, C. G., Rafferty, D. & Granat, M. H. (2009). Compliance with physical activity guidelines in a group of UK-based postal workers using an objective monitoring technique. *European Journal* of *Applied Physiology*, 106 (6), 893-899.
- Chastin, S. F. M., Ferriolli, E., Stephens, N. A., Fearon, K. C. H. & Greig, C. (2012). Relationship between sedentary behaviour, physical activity, muscle quality and body composition in healthy older adults. *Age and Ageing*, 41 (1), 111-114.
- Chen, K. Y. & Bassett Jr, D. R. (2005). The technology of accelerometry-based activity monitors: current and future. *Medicine and Science in Sports and Exercise*, 37 (11), S 490-500.
- Chinapaw, M., Proper, K., Brug, J., Van Mechelen, W. & Singh, A. (2011). Relationship between young peoples' sedentary behaviour and biomedical health indicators: a systematic review of prospective studies. *Obesity Reviews*, 12 (7), 621-632.
- Chinapaw, M. J. M., Mokkink, L. B., Van Poppel, M. N. M., Van Mechelen, W. & Terwee, C. B. (2010). Physical activity questionnaires for youth: a systematic review of measurement properties. *Sports Medicine*, 40 (7), 539-563.
- Chobanian, A. V., Bakris, G. L., Black, H. R., Cushman, W. C., Green, L. A., Izzo, J. L., Jones, D. W., Materson, B. J., Oparil, S. & Wright, J. T. (2003). Seventh report of the joint national committee on prevention, detection, evaluation, and treatment of high blood pressure. *Hypertension*, 42 (6), 1206-1252.

- Clark, B. K., Healy, G. N., Winkler, E. a. H., Gardiner, P. A., Sugiyama, T., Dunstan, D. W., Matthews, C. E. & Owen, N. (2011). Relationship of television time with accelerometer-derived sedentary time: NHANES. *Medicine and Science in Sports and Exercise*, 43 (5), 822-828.
- Clark, B. K., Sugiyama, T., Healy, G. N., Salmon, J., Dunstan, D. W. & Owen, N. (2009). Validity and reliability of measures of television viewing time and other nonoccupational sedentary behaviour of adults: a review. *Obesity Reviews*, 10 (1), 7-16.
- Colley, R. C., Garriguet, D., Janssen, I., Craig, C. L., Clarke, J. & Tremblay, M. S. (2011a). Physical activity of Canadian adults: Accelerometer results from the 2007 to 2009 Canadian Health Measures Survey. *Health Reports*, 22 (1), 4-11.
- Colley, R. C., Garriguet, D., Janssen, I., Craig, C. L., Clarke, J. & Tremblay, M. S. (2011b). Physical activity of Canadian children and youth: accelerometer results from the 2007 to 2009 Canadian Health Measures Survey. *Health Reports*, 22 (1), 15-23.
- Conway, J. M., Seale, J. L., Jacobs, D. R., Irwin, M. L. & Ainsworth, B. E. (2002). Comparison of energy expenditure estimates from doubly labeled water, a physical activity questionnaire, and physical activity records. *The American Journal of Clinical Nutrition*, 75 (3), 519-525.
- Craig, S. B., Bandini, L. G., Lichtenstein, A. H., Schaefer, E. J. & Dietz, W. H. (1996). The impact of physical activity on lipids, lipoproteins, and blood pressure in preadolescent girls. *Pediatrics*, 98 (3), 389-395.
- Crouter, S. E., Clowers, K. G. & Bassett Jr, D. R. (2006). A novel method for using accelerometer data to predict energy expenditure. *Journal of Applied Physiology*, 100 (4), 1324-1331.
- Currie, C., Zanotti, C., Morgan, A., Currie, D., Delooze, M., Roberts, C., Samdal, O., Smith, O. & Barnekow, V. 2012. Social determinants of health and well-being among young people. Health Behaviour in School-aged Children (HBSC) study: International report from the 2009/2010 survey. Copenhagen: Who Regional Office for Europe.
- Dall, P. M. & Kerr, A. (2010). Frequency of the sit to stand task: An observational study of free-living adults. *Applied Ergonomics*, 41 (1), 58-61.
- Dalton, M., Cameron, A., Zimmet, P., Shaw, J., Jolley, D., Dunstan, D. & Welborn, T. (2012). Waist circumference, waist-hip ratio and BMI and their correlation with cardiovascular risk factors in Australian adults. *Journal of internal medicine*, 254 (6), 555-563.
- Dangardt, F., Osika, W., Volkmann, R., Gan, L. M. & Friberg, P. (2008). Obese children show increased intimal wall thickness and decreased pulse wave velocity. *Clinical Physiology and Functional Imaging*, 28 (5), 287-293.

- Daniels, S. R., Jacobson, M. S., Mccrindle, B. W., Eckel, R. H. & Sanner, B. M. H. (2009). American Heart Association Childhood Obesity Research Summit Report. *Circulation*, 119, E 489-517.
- Davies, G., Reilly, J. J., Mcgowan, A. J., Dall, P. M., Granat, M. H. & Paton, J. Y. (2011). Validity, Practical Utility and Reliability of the Activpal in Pre-School Children. *Medicine and Science in Sports and Exercise*, 44 (4), 761-768.
- De Vries, S. I., Bakker, I., Hopman-Rock, M., Hirasing, R. A. & Van Mechelen, W. (2006). Clinimetric review of motion sensors in children and adolescents. *Journal of Clinical Epidemiology*, 59 (7), 670-680.
- De Vries, S. I., Van Hirtum, H. W., Bakker, I., Hopman-Rock, M., Hirasing, R. A. & Van Mechelen, W. (2009). Validity and reproducibility of motion sensors in youth: a systematic update. *Medicine and Science in Sports and Exercise*, 41 (4), 818-827.
- Debusk, R. F., Stenestrand, U., Sheehan, M. & Haskell, W. L. (1990). Training effects of long versus short bouts of exercise in healthy subjects. *The American Journal of Cardiology*, 65 (15), 1010-1013.
- Department of Health and Ageing. 1999. National Physical Activity Guidelines for Australians. Canberra: Department of Health and Ageing.
- Department of Health and Children. 2012. *Get Ireland Active: The National Guidelines on Physical Activity for Ireland*. Dublin: The Department of Health and Children and the Health Service Executive.
- Department of Transport. 2009. National Cycle Policy Framework. Dublin, Ireland: Department of Transport.
- Després, J. P., Lemieux, I., Dagenais, G. R., Cantin, B. & Lamarche, B. I. (2000). HDLcholesterol as a marker of coronary heart disease risk: the Quebec cardiovascular study. *Atherosclerosis*, 153 (2), 263-272.
- Dietz, W. (2004). The effects of physical activity on obesity. Quest, 56 (1), 1-11.
- Dietz, W. H. (1998). Health consequences of obesity in youth: childhood predictors of adult disease. *Pediatrics*, 101 (2), S 518-525.
- Do Carmo, I., Dos Santos, O., Camolas, J., Vieira, J., Carreira, M., Medina, L., Reis, L., Myatt, J. & Galvão-Teles, A. (2008). Overweight and obesity in Portugal: national prevalence in 2003–2005. *Obesity Reviews*, 9 (1), 11-19.
- Donahoo, W. T., Levine, J. A. & Melanson, E. L. (2004). Variability in energy expenditure and its components. *Current Opinion in Clinical Nutrition & Metabolic Care*, 7 (6), 599-605.

- Duncan, J. J., Gordan, N. F. & Scott, C. B. (1991). Women walking for health and fitness. *JAMA: the Journal of the American Medical Association*, 266 (23), 3295-3299.
- Dunstan, D., Barr, E., Healy, G., Salmon, J., Shaw, J., Balkau, B., Magliano, D., Cameron, A., Zimmet, P. & Owen, N. (2010). Television Viewing Time and Mortality. *Circulation*, 121 (3), 384-391.
- Dunstan, D. W., Salmon, J., Healy, G. N., Shaw, J. E., Jolley, D., Zimmet, P. Z. & Owen, N. (2007). Association of television viewing with fasting and 2-h postchallenge plasma glucose levels in adults without diagnosed diabetes. *Diabetes Care*, 30 (3), 516-522.
- Dunstan, D. W., Salmon, J., Owen, N., Armstrong, T., Zimmet, P. Z., Welborn, T., Cameron, A., Dwyer, T., Jolley, D. & Shaw, J. E. (2005). Associations of TV viewing and physical activity with the metabolic syndrome in Australian adults. *Diabetologia*, 48 (11), 2254-2261.
- Dunstan, D. W., Salmon, J., Owen, N., Armstrong, T., Zimmet, P. Z., Welborn, T. A., Cameron, A. J., Dwyer, T., Jolley, D. & Shaw, J. E. (2004). Physical activity and television viewing in relation to risk of undiagnosed abnormal glucose metabolism in adults. *Diabetes Care*, 27 (11), 2603-2609.
- Dunstan, D. W., Thorp, A. A. & Healy, G. N. (2011). Prolonged sitting: is it a distinct coronary heart disease risk factor? *Current Opinion in Cardiology*, 26 (5), 412-419.
- Durant, R. H., Baranowski, T., Puhl, J. & Rhodes, T. (1993). Evaluation of the Children's Activity Rating Scale (CARS) in young children. *Medicine and Science in Sports and Exercise*, 25 (12), 1415-1421.
- Eaton, D. K., Kann, L., Kinchen, S., Shanklin, S., Ross, J., Hawkins, J., Harris, W. A., Lowry, R., Mcmanus, T., Chyen, D., Lim, C., Whittle, L., Brener, N. D. & Wechsler, H. 2010. *Youth Risk Behavior Surveillance — United States, 2009.* Atlanta, GA: Centre for Disease Control and Prevention.
- Eisenmann, J. C. (2008). On the use of a continuous metabolic syndrome score in pediatric research. *Cardiovascular Diabetology*, 7 (17), 1-6.
- Ekelund, L. G., Haskell, W. L., Johnson, J. L., Whaley, F. S., Criqui, M. H. & Sheps, D. S. (1988). Physical fitness as a predictor of cardiovascular mortality in asymptomatic North American men. *New England Journal of Medicine*, 319 (21), 1379-1384.
- Ekelund, U., Anderssen, S. A., Froberg, K., Sardinha, L. B., Andersen, L. B. & Brage, S. (2007a). Independant associations of physical activity and cardiorespiratory fitness with metabolic risk factors in children: the European youth heart study. *Diabetologia*, 50 (9), 1832-1840.
- Ekelund, U., Brage, S., Froberg, K., Harro, M., Anderssen, S. A., Sardinha, L. B., Riddoch, C. & Andersen, L. B. (2006). TV Viewing Time and Physical Activity are

Independantly Associated with Metabolic Risk in Children: The European Youth Heart Study. *PLoS Medicine*, 3 (12), 2449-2457.

- Ekelund, U., Griffin, S. J. & Wareham, N. J. (2007b). Physical activity and metabolic risk in individuals with a family history of type 2 diabetes. *Diabetes Care*, 30 (2), 337-342.
- Ekelund, U., Luan, J., Sherar, L. B., Esliger, D. W., Griew, P. & Cooper, A. (2012).
   Moderate to Vigorous Physical Activity and Sedentary Time and Cardiometabolic
   Risk Factors in Children and Adolescents. JAMA: the Journal of the American Medical
   Association, 307 (7), 704-712.
- Ekelund, U., Sardinha, L. B., Anderssen, S. A., Harro, M., Franks, P. W., Brage, S., Cooper, A. R., Andersen, L. B., Riddoch, C. & Froberg, K. (2004). Associations between objectively assessed physical activity and indicators of body fatness in 9-to 10-y-old European children: a population-based study from 4 distinct regions in Europe (the European Youth Heart Study). *The American journal of clinical nutrition*, 80 (3), 584-590.
- Emons, H. J. G., Groenenboom, D., Westerterp, K. & Saris, W. H. M. (1992). Comparison of heart rate monitoring combined with indirect calorimetry and the doubly labelled water (2 H 2 18 O) method for the measurement of energy expenditure in children. *European Journal of Applied Physiology and Occupational Physiology*, 65 (2), 99-103.
- Esliger, D., Copeland, J., Barnes, J. & Tremblay, M. (2005). Standardizing and optimizing the use of accelerometer data for free-living physical activity monitoring. *Journal of Physical Activity and Health*, 2 (3), 366-383.
- Esliger, D. W., Rowlands, A. V., Hurst, T. L., Catt, M., Murray, P. & Eston, R. G. (2011). Validation of the GENEA Accelerometer. *Medicine and Science in Sports and Exercise*, 43 (6), 1085-1093.
- Esliger, D. W. & Tremblay, M. (2007). Physical activity and inactivity profiling: the next generation *Applied Physiology, Nutrition, and Metabolism,* 32 (2), S 195-207.
- Eston, R. G., Rowlands, A. V. & Ingledew, D. K. (1998). Validity of heart rate, pedometry, and accelerometry for predicting the energy cost of children's activities. *Journal of Applied Physiology*, 84 (1), 362-371.
- Faulkner, G. E. J., Buliung, R. N., Flora, P. K. & Fusco, C. (2009). Active school transport, physical activity levels and body weight of children and youth: a systematic review. *Preventive Medicine*, 48 (1), 3-8.
- Flegal, K. M., Carroll, M. D., Ogden, C. L. & Curtin, L. R. (2010). Prevalence and trends in obesity among US adults, 1999-2008. JAMA: the journal of the American Medical Association, 303 (3), 235-241.

- Flegal, K. M., Graubard, B. I., Williamson, D. F. & Gail, M. H. (2007). Cause-specific excess deaths associated with underweight, overweight, and obesity. JAMA: the journal of the American Medical Association, 298 (17), 2028-2037.
- Fox, C. S., Pencina, M. J., Meigs, J. B., Vasan, R. S., Levitzky, Y. S. & D'agostino, R. B. (2006). Trends in the Incidence of Type 2 Diabetes Mellitus From the 1970s to the 1990s. *Circulation*, 113 (25), 2914-2918.
- Franks, P. W., Hanson, R. L., Knowler, W. C., Sievers, M. L., Bennett, P. H. & Looker, H. C. (2010a). Childhood Obesity, other cardiovascular risk factors, and premature death. *New England Journal of Medicine*, 362 (2), 485-493.
- Franks, P. W., Hanson, R. L., Knowler, W. C., Sievers, M. L., Bennett, P. H. & Looker, H. C. (2010b). Childhood obesity, other cardiovascular risk factors, and premature death. *New England Journal of Medicine*, 362 (6), 485-493.
- Freedman, D. S., Khan, L. K., Serdula, M. K., Dietz, W. H., Srinivasan, S. R. & Berenson, G. S. (2005). The relation of childhood BMI to adult adiposity: the Bogalusa Heart Study. *Pediatrics*, 115 (1), 22-27.
- Freedson, P., Bowles, H. R., Troiano, R. & Haskell, W. (2012). Assessment of physical activity using wearable monitors: recommendations for monitor calibration and use in the field. *Medicine and Science in Sports and Exercise*, 44 (1), S 1-4.
- Freedson, P. & Miller, K. (2000). Objective monitoring of physical activity using motion sensors and heart rate. *Research Quarterly for Exercise and Sport*, 71 (2), S 21-29.
- Freedson, P., Pober, D. & Janz, K. F. (2005). Calibration of accelerometer output for children. *Medicine and Science in Sports and Exercise*, 37 (11), S 523-530.
- Freedson, P. S., Melanson, E. & Sirard, J. (1998). Calibration of the computer science and applications, inc. accelerometer. *Medicine and Science in Sports and Exercise*, 30 (5), 777-781.
- Garber, C. E., Blissmer, B., Deschenes, M. R., Franklin, B. A., Lamonte, M. J., Lee, I., Nieman, D. C. & Swain, D. P. (2011). American College of Sports Medicine position stand. Quantity and quality of exercise for developing and maintaining cardiorespiratory, musculoskeletal, and neuromotor fitness in apparently healthy adults: guidance for prescribing exercise. *Medicine and Science in Sports and Exercise*, 43 (7), 1334-1359.
- Godfrey, A., Culhane, K. & Lyons, G. (2007). Comparison of the performance of the activPAL<sup>™</sup> Professional physical activity logger to a discrete accelerometer-based activity monitor. *Medical Engineering & Physics,* 29 (8), 930-934.
- Goran, M. I. (1994). Application of the doubly labeled water technique for studying total energy expenditure in young children: a review. *Pediatric Exercise Science*, 6 (1), 11-30.

- Grant, P. M., Dall, P. M., Mitchell, S. L. & Granat, M. H. (2008). Activity-monitor accuracy in measuring step number and cadence in community-dwelling older adults. *Journal* of Aging and Physical Activity, 16 (2), 201-214.
- Grant, P. M., Granat, M. H., Thow, M. K. & Maclaren, W. M. (2010). Analyzing free-living physical activity of older adults in different environments using body-worn activity monitors. *Journal of Aging and Physical Activity*, 18 (2), 171-184.
- Grant, P. M., Ryan, C. G., Tigbe, W. W. & Granat, M. H. (2006). The validation of a novel activity monitor in the measurement of posture and motion during everyday activities. *British Journal of Sports Medicine*, 40 (12), 992-997.
- Gregg, E. W., Cauley, J. A., Stone, K., Thompson, T. J., Bauer, D. C., Cummings, S. R. & Ensrud, K. E. (2003). Relationship of changes in physical activity and mortality among older women. *JAMA: the Journal of the American Medical Association*, 289 (18), 2379-2386.
- Grundy, S. M. (2004). Obesity, metabolic syndrome, and cardiovascular disease. Journal of Clinical Endocrinology & Metabolism, 89 (6), 2595-2600.
- Guo, S. S. & Chumlea, W. C. (1999). Tracking of body mass index in children in relation to overweight in adulthood. *The American Journal of Clinical Nutrition*, 70 (1), S 145-148.
- Haffner, S. M., Lehto, S., Rönnemaa, T., Pyörälä, K. & Laakso, M. (1998). Mortality from coronary heart disease in subjects with type 2 diabetes and in nondiabetic subjects with and without prior myocardial infarction. *New England Journal of Medicine*, 339 (4), 229-234.
- Hallal, P. C., Victora, C. G., Azevedo, M. R. & Wells, J. C. K. (2006). Adolescent Physical Activity and Health: A systematic review. *Sports Medicine*, 36 (12), 1019-1030.
- Hamilton, M. T., Areiqat, E., Hamilton, D. G. & Bey, L. (2001). Plasma triglyceride metabolism in humans and rats during aging and physical inactivity. *International Journal of Sport Nutrition and Exercise Metabolism*, 11, S 97-104.
- Hamilton, M. T., Etienne, J., Mcclure, W. C., Pavey, B. S. & Holloway, A. K. (1998). Role of local contractile activity and muscle fiber type on LPL regulation during exercise. *American Journal of Physiology, Endocrinology And Metabolism,* 275 (6), E 1016-1022.
- Hamilton, M. T., Hamilton, D. G. & Zderic, T. W. (2004). Exercise physiology versus inactivity physiology: an essential concept for understanding lipoprotein lipase regulation. *Exercise and Sport Sciences Reviews*, 32 (4), 161-166.
- Hamilton, M. T., Hamilton, D. G. & Zderic, T. W. (2007). Role of low energy expenditure and sitting in obesity, metabolic syndrome, type 2 diabetes, and cardiovascular disease. *Diabetes*, 56 (11), 2655-2667.

- Hamilton, M. T., Healy, G. N., Dunstan, D. W., Zderic, T. W. & Owen, N. (2008). Too little exercise and too much sitting: inactivity physiology and the need for new recommendations on sedentary behavior. *Current Cardiovascular Risk Reports*, 2 (4), 292-298.
- Hardman, A. E. & Stensel, D. J. 2009. *Physical Activity and health: The Evidence Explained*, New York, NY, Routledge Publishing.
- Harrell, J. S., Mcmurray, R. G., Baggett, C. D., Pennell, M. L., Pearce, P. F. & Bangdiwala, S. I. (2005). Energy costs of physical activities in children and adolescents. *Medicine* and Science in Sports and Exercise, 37 (2), 329-336.
- Harrington, D. M., Dowd, K. P., Bourke, A. K. & Donnelly, A. E. (2011a). Cross-Sectional analysis of levels and patterns of objectively measured sedentary time in adolescent females. *International Journal of Behavioral Nutrition and Physical Activity*, 8 (1), 120-127.
- Harrington, D. M., Dowd, K. P., Tudor-Locke, C. & Donnelly, A. E. (2012). A Steps/Minute Value for Moderate Intensity Physical Activity in Adolescent Females. *Pediatric Exercise Science*, 24 (3), 399-408.
- Harrington, D. M., Welk, G. J. & Donnelly, A. E. (2011b). Validation of MET estimates and step measurement using the activPAL physical activity logger. *Journal of Sports Sciences*, 29 (6), 627-633.
- Hart, T. L., Ainsworth, B. E. & Tudor-Locke, C. (2011). Objective and subjective measures of sedentary behavior and physical activity. *Medicine and Science in Sports and Exercise*, 43 (3), 449-456.
- Haskell, W. L. (1994). Health consequences of physical activity: understanding and challenges regarding dose-response. *Medicine and Science in Sports and Exercise*, 26 (6), 649-660.
- Haskell, W. L., Lee, I. M., Pate, R., Powell, E., Blair, S., Franklin, B., Macera, C., Heath, G., Thompson, P. & Bauman, A. (2007). Physical activity and public health: Updated recommendation for adults from the American college of sports medicine and the American heart association. *Medicine and Science in Sports and Exercise*, 39 (8), 1423-1434.
- Healy, G. N., Clark, B. K., Winkler, E. a. H., Gardiner, P. A., Brown, W. J. & Matthews, C. E. (2011a). Measurement of adults' sedentary time in population-based studies. *American Journal of Preventive Medicine*, 41 (2), 216-227.
- Healy, G. N., Dunstan, D. W., Salmon, J., Cerin, E., Shaw, J. E., Zimmet, P. Z. & Owen, N. (2007). Objectively measured light-intensity physical activity is independently associated with 2-h plasma glucose. *Diabetes Care*, 30 (6), 1384-1389.

- Healy, G. N., Dunstan, D. W., Salmon, J., Cerin, E., Shaw, J. E., Zimmet, P. Z. & Owen, N. (2008a). Breaks in sedentary time: beneficial associations with metabolic risk. *Diabetes Care*, 31 (4), 661-666.
- Healy, G. N., Dunstan, D. W., Salmon, J., Shaw, J. E., Zimmet, P. Z. & Owen, N. (2008b). Television time and continuous metabolic risk in physically active adults. *Medicine* and Science in Sports and Exercise, 40 (4), 639-645.
- Healy, G. N., Matthews, C. E., Dunstan, D. W., Winkler, E. a. H. & Owen, N. (2011b). Sedentary time and cardio-metabolic biomarkers in US adults: NHANES 2003–06. *European Heart Journal*, 32 (5), 590-597.
- Healy, G. N., Wijndaele, K., Dunstan, D. W., Shaw, J. E., Salmon, J., Zimmet, P. Z. & Owen, N. (2008c). Objectively measured sedentary time, physical activity, and metabolic risk. *Diabetes Care*, 31 (2), 369-371.
- Heil, D. P., Brage, S. & Rothney, M. P. (2012). Modeling physical activity outcomes from wearable monitors. *Medicine and Science in Sports and Exercise*, 44 (1), S 50-60.
- Helmerhorst, H. J. F., Wijndaele, K., Brage, S., Wareham, N. J. & Ekelund, U. (2009). Objectively measured sedentary time may predict insulin resistance independent of moderate-and vigorous-intensity physical activity. *Diabetes*, 58 (8), 1776-1779.
- Helmrich, S. P., Ragland, D. R., Leung, R. W. & Paffenbarger Jr, R. S. (1991). Physical activity and reduced occurrence of non-insulin-dependent diabetes mellitus. *New England Journal of Medicine*, 325 (3), 147-152.
- Hernelahti, M., Levälahti, E., Simonen, R. L., Kaprio, J., Kujala, U. M., Uusitalo-Koskinen, A. L. T., Battié, M. C. & Videman, T. (2004). Relative roles of heredity and physical activity in adolescence and adulthood on blood pressure. *Journal of Applied Physiology*, 97 (3), 1046-1052.
- Hubert, H. B., Feinleib, M., Mcnamara, P. M. & Castelli, W. P. (1983). Obesity as an independent risk factor for cardiovascular disease: a 26-year follow-up of participants in the Framingham Heart Study. *Circulation*, 67 (5), 968-977.
- Hussey, J., Gormley, J. & Bell, C. (2001). Physical activity in Dublin children aged 7–9 years. *British Journal of Sports Medicine*, 35 (4), 268-272.
- Intille, S. S., Lester, J., Sallis, J. F. & Duncan, G. (2012). New horizons in sensor development. *Medicine and Science in Sports and Exercise*, 44 (1), S 24-31.
- Jago, R. & Baranowski, T. (2004). Non-curricular approaches for increasing physical activity in youth: a review. *Preventive Medicine*, 39 (1), 157-163.

- Jago, R., Zakeri, I., Baranowski, T. & Watson, K. (2007). Decision boundaries and receiver operating characteristic curves: new methods for determining accelerometer cutpoints. *Journal of Sports Sciences*, 25 (8), 937-944.
- Jakes, R., Day, N., Khaw, K., Luben, R., Oakes, S., Welch, A., Bingham, S. & Wareham, N. (2003). Television viewing and low participation in vigorous recreation are independently associated with obesity and markers of cardiovascular disease risk: EPIC-Norfolk population-based study. *European Journal of Clinical Nutrition*, 57 (9), 1089-1096.
- Jakicic, J. M., Wing, R., Butler, B. & Robertson, R. (1995). Prescribing exercise in multiple short bouts versus one continuous bout: effects on adherence, cardiorespiratory fitness, and weight loss in overweight women. *International Journal of Obesity and Related Metabolic Disorders*, 19 (12), 893-901.
- Janssen, I. & Leblanc, A. G. (2010). Systematic review of the health benefits of physical activity and fitness in school aged children and youth. *International Journal of Behavioral Nutrition and Physical Activity*, 7 (1), 40-56.
- Janssen, I., Shields, M., Craig, C. & Tremblay, M. (2011). Prevalence and secular changes in abdominal obesity in Canadian adolescents and adults, 1981 to 2007–2009. Obesity Reviews, 12 (6), 397-405.
- Janz, K. (2006). Physical activity in epidemiology: moving from questionnaire to objective measurement. *British Journal of Sports Medicine*, 40 (3), 191-192.
- Janz, K. F. 2002. Use of Heart Rate Monitors to Assess Physical Activity. In: Welk, G. J. (ed.) Physical Activity Assessment for Health Related Research. Champaign, IL: Human Kinetics Publishing.
- Janz, K. F., Burns, T. L. & Levy, S. M. (2005). Tracking of activity and sedentary behaviors in childhood: the Iowa Bone Development Study. *American Journal of Preventive Medicine*, 29 (3), 171-178.
- Jequier, E. 1981. Long-term measurement of energy expenditure in man: direct or indirect calorimetry. *In:* Bjorntorp, P., Cairella, M. & Howard, A. N. (eds.) *Recent Advances in Obesity Research III.* London: J. Libbey.
- Jiménez-Pavón, D., Kelly, J. & Reilly, J. J. (2010). Associations between objectively measured habitual physical activity and adiposity in children and adolescents: Systematic review. *International Journal of Pediatric Obesity*, 5 (1), 3-18.
- Johnson, J. G., Cohen, P., Kasen, S., First, M. B. & Brook, J. S. (2004). Association between television viewing and sleep problems during adolescence and early adulthood. *Archives of Pediatrics and Adolescent Medicine*, 158 (6), 562-568.
- Johnson, W. D., Kroon, J. J. M., Greenway, F. L., Bouchard, C., Ryan, D. & Katzmarzyk, P. T. (2009). Prevalence of risk factors for metabolic syndrome in adolescents:

National Health and Nutrition Examination Survey (NHANES), 2001-2006. *Archives of Pediatrics and Adolescent Medicine*, 163 (4), 371-377.

- Jones, W. H. S. 1967. *Hippocrates (trans. W. H. S. Jones)*, Cambridge, Harvard University Press.
- Kannel, W. B., Schwartz, M. J. & Mcnamara, P. M. (1969). Blood pressure and risk of coronary heart disease: the Framingham study. *Diseases of the Chest*, 56 (1), 43-52.
- Karlsson, M. (2004). Physical activity, skeletal health and fractures in a long term perspective. *Journal of Musculoskeletal and Neuronal Interactions*, 4 (1), 12-21.
- Katzmarzyk, P., Janssen, I. & Ardern, C. (2003). Physical inactivity, excess adiposity and premature mortality. *Obesity Reviews*, 4 (4), 257-290.
- Katzmarzyk, P. T. & Janssen, I. (2004). The economic costs associated with physical inactivity and obesity in Canada: an update. *Canadian Journal of Applied Physiology*, 29 (1), 90-115.
- Kelleher, C., Nic Gabhainn, S., Friel, S., Corrigan, H., Nolan, G., Sixsmith, J., Walsh, O. & Cooke, M. 2003. National health and lifestyles survey: Survey of lifestyles, attitudes and nutrition (Slan 2002) and the Irish health behaviour in school-aged children survey (HBSC). Health Promotion Unit Department of Health and Children Dublin.
- Kelleher, C., Nicgabhainn, S. & Friel, S. 1999. The national health and lifestyle surveys: Survey of lifestyle, attitudes and nutrition (Slan 1998) and the Irish health behaviour in school-aged children survey (HSBC). NUI Galway: Centre for Health Promotion Studies.
- Kelley, G. A., Kelley, K. S. & Tran, Z. V. (2003). The Effect of Exercise on Resting Blood Pressure in Children and Adolescents: A Meta-Analysis of Randomized Controlled Trials. *Preventive Cardiology*, 6 (1), 8-16.
- Kelly, C., Gavin, A., Molcho, M. & Nic Gabhainn, S. 2012. *The Irish Health Behaviour in School-aged Children (HBSC) Study 2010.* NUI Galway: Health Promotion Research Centre.
- Kemper, H. C. G. & Koppes, L. L. J. (2006). Linking Physical Activity and Aerobic Fitness: Are We Active Because We Are Fit, or Are We Fit Because We Are Active? *Pediatric Exercise Science*, 18 (2), 173-181.
- Kimm, S., Glynn, N. W., Obarzanek, E., Kriska, A. M., Daniels, S. R., Barton, B. A. & Liu, K. (2005). Relation between the changes in physical activity and body-mass index during adolescence: a multicentre longitudinal study. *The Lancet*, 366 (9482), 301-307.
- Kimm, S. Y. S., Glynn, N. W., Kriska, A. M., Barton, B. A., Kronsberg, S. S., Daniels, S. R., Crawford, P. B., Sabry, Z. I. & Liu, K. (2002). Decline in physical activity in black

girls and white girls during adolescence. New England journal of medicine, 347 (10), 709-715.

- Klesges, R. C. (1984). The FATS: An observational system for assessing physical activity in children and associated parent behavior. *Behavioral Assessment*, 6 (4), 333-345.
- Knoflach, M., Kiechl, S., Kind, M., Said, M., Sief, R., Gisinger, M., Van Der Zee, R., Gaston, H., Jarosch, E. & Willeit, J. (2003). Cardiovascular risk factors and atherosclerosis in young males. *Circulation*, 108 (9), 1064-1069.
- Kohl, H. W. (2001). Physical activity and cardiovascular disease: evidence for a dose response. *Medicine and Science in Sports and Exercise*, 33 (6), S 472-483.
- Kohl, H. W., Fulton, J. E. & Caspersen, C. J. (2000). Assessment of physical activity among children and adolescents: a review and synthesis. *Preventive Medicine*, 31 (2), S 54-76.
- Kohrt, W. M., Bloomfield, S. A., Little, K. D., Nelson, M. E. & Yingling, V. R. (2004). American College of Sports Medicine Position Stand: Physical activity and bone health. *Medicine and Science in Sports and Exercise*, 36 (11), 1985-1996.
- Kozey-Keadle, S., Libertine, A., Lyden, K., Staudenmayer, J. & Freedson, P. S. (2011). Validation of wearable monitors for assessing sedentary behavior. *Medicine and Science in Sports and Exercise*, 43 (8), 1561-1567.
- Kozey, S. L., Lyden, K., Howe, C. A., Staudenmayer, J. W. & Freedson, P. S. (2010a). Accelerometer output and MET values of common physical activities. *Medicine and Science in Sports and Exercise*, 42 (9), 1776.
- Kozey, S. L., Staudenmayer, J. W., Troiano, R. P. & Freedson, P. S. (2010b). A comparison of the ActiGraph 7164 and the ActiGraph GT1M during self-paced locomotion. *Medicine and Science in Sports and Exercise*, 42 (5), 971-976.
- Kuczmarski, R. J., Ogden, C. L., Guo, S. S., Grummer-Strawn, L. M., Flegal, K. M., Mei, Z., Wei, R., Curtin, L. R., Roche, A. F. & Johnson, C. L. (2002). 2000 CDC Growth Charts for the United States: methods and development. *Vital and health statistics 11*, 246 (246), 1-190.
- Lee, I. M. (2003). Physical activity and cancer prevention-data from epidemiologic studies. *Medicine and Science in Sports and Exercise*, 35 (11), 1823-1827.
- Lee, I. M., Sesso, H. D. & Paffenbarger Jr, R. S. (2000). Physical activity and coronary heart disease risk in men: Does the duration of exercise episodes predict risk? . *Circulation*, 102 (9), 981-986.
- Lee, I. M., Shiroma, E. J., Lobelo, F., Puska, P., Blair, S. N. & Katzmarzyk, P. T. (2012). Effect of physical inactivity on major non-communicable diseases worldwide: an analysis of burden of disease and life expectancy. *The Lancet*, 380 (9838), 219-229.

- Leon, A. S., Connett, J., Jacobs, D. R. & Rauramaa, R. (1987). Leisure-time physical activity levels and risk of coronary heart disease and death. JAMA: the Journal of the American Medical Association, 258 (17), 2388-2395.
- Levine, J., Melanson, E. L., Westerterp, K. R. & Hill, J. O. (2001). Measurement of the components of nonexercise activity thermogenesis. *American Journal of Physiology*, *Endocrinology And Metabolism*, 281 (4), 670-675.
- Levine, J. A. (2002). Non-exercise activity thermogenesis (NEAT). Best Practice & Research Clinical Endocrinology & Metabolism, 16 (4), 679-702.
- Levine, J. A. (2005). Measurement of energy expenditure. *Public Health Nutrition*, 8 (7 (a)), 1123-1132.
- Levine, J. A., Eberhardt, N. L. & Jensen, M. D. (1999). Role of nonexercise activity thermogenesis in resistance to fat gain in humans. *Science*, 283 (5399), 212-214.
- Levine, J. A., Vander Weg, M. W., Hill, J. O. & Klesges, R. C. (2006). Non-exercise activity thermogenesis. *Arteriosclerosis, Thrombosis, and Vascular Biology,* 26 (4), 729-736.
- Livingstone, M., Coward, W. A., Prentice, A. M., Davies, P., Strain, J. J., Mckenna, P. G., Mahoney, C. A., White, J. A., Stewart, C. M. & Kerr, M. (1992). Daily energy expenditure in free-living children: comparison of heart-rate monitoring with the doubly labeled water (2H2 (18) O) method. *The American Journal of Clinical Nutrition*, 56 (2), 343-352.
- Livingstone, M., Prentice, A. M., Coward, W. A., Ceesay, S. M., Strain, J. J., Mckenna, P. G., Nevin, G. B., Barker, M. E. & Hickey, R. J. (1990). Simultaneous measurement of free-living energy expenditure by the doubly labeled water method and heart-rate monitoring. *The American Journal of Clinical Nutrition*, 52 (1), 59-65.
- Lloyd-Jones, D., Adams, R., Carnethon, M., De Simone, G., Ferguson, T. B., Flegal, K., Ford, E., Furie, K., Go, A. & Greenlund, K. (2009). Heart Disease and Stroke Statistics—2009 Update: A Report From the American Heart Association Statistics Committee and Stroke Statistics Subcommittee. *Circulation*, 119 (3), 480-486.
- Lloyd-Jones, D. M., Hong, Y., Labarthe, D., Mozaffarian, D., Appel, L. J., Van Horn, L., Greenlund, K., Daniels, S., Nichol, G. & Tomaselli, G. F. (2010). Defining and Setting National Goals for Cardiovascular Health Promotion and Disease Reduction The American Heart Association's Strategic Impact Goal Through 2020 and Beyond. *Circulation*, 121 (4), 586-613.
- Loprinzi, P. D. & Cardinal, B. J. (2011). Measuring Children's Physical Activity and Sedentary Behaviors. *Journal of Exercise Science & Fitness*, 9 (1), 15-23.
- Lorenz, M. W., Markus, H. S., Bots, M. L., Rosvall, M. & Sitzer, M. (2007). Prediction of clinical cardiovascular events with carotid intima-media thickness. *Circulation*, 115 (4), 459-467.

- Lubans, D., Hesketh, K., Cliff, D., Barnett, L., Salmon, J., Dollman, J., Morgan, P., Hills, A. & Hardy, L. (2011). A systematic review of the validity and reliability of sedentary behaviour measures used with children and adolescents. *Obesity Reviews*, 12 (10), 781-799.
- Maddocks, M., Petrou, A., Skipper, L. & Wilcock, A. (2010). Validity of three accelerometers during treadmill walking and motor vehicle travel. *British Journal of Sports Medicine*, 44 (8), 606-608.
- Maffeis, C., Pinelli, L., Zaffanello, M., Schena, F., Iacumin, P. & Schutz, Y. (1995). Daily energy expenditure in free-living conditions in obese and non-obese children: comparison of doubly labelled water (2H2 (18) O) method and heart-rate monitoring. *International Journal of Obesity and Related Metabolic Disorders*, 19 (9), 671-677.
- Malina, R. M. (1996). Tracking of physical activity and physical fitness across the lifespan. *Research Quarterly for Exercise and Sport*, 67 (3), S 48-57.
- Malina, R. M. (2001a). Adherence to Physical Activity from Childhood to Adulthood: A Perspective from Tracking Studies. *Quest*, 53 (3), 346-55.
- Malina, R. M. (2001b). Physical activity and fitness: pathways from childhood to adulthood. *American Journal of Human Biology*, 13 (2), 162-172.
- Malnick, S. & Knobler, H. (2006). The medical complications of obesity. *Qjm*, 99 (9), 565-579.
- Manson, J. a. E., Nathan, D. M., Krolewski, A. S., Stampfer, M. J., Willett, W. C. & Hennekens, C. H. (1992). A prospective study of exercise and incidence of diabetes among US male physicians. *JAMA: the Journal of the American Medical Association*, 268 (1), 63-67.
- Manson, J. E., Stampfer, M. J., Colditz, G. A., Willett, W. C., Rosner, B., Hennekens, C. H., Speizer, F. E., Rimm, E. B. & Krolewski, A. S. (1991). Physical activity and incidence of non-insulin-dependent diabetes mellitus in women. *The Lancet*, 338 (8770), 774-778.
- Marcus, B. H., King, T. K., Clark, M. M., Pinto, B. M. & Bock, B. C. (1996). Theories and techniques for promoting physical activity behaviours. *Sports Medicine*, 22 (5), 321-331.
- Marfell-Jones, T. O. M., Stewart, A. & Carter, L. 2006. *International standards for anthropometric assessment*, International Society for the Advancement of Kinanthropometry.
- Mark, A. E. & Janssen, I. (2008). Dose-Response Relation between Physical Activity and Blood Pressure in Youth. *Medicine and Science in Sports and Exercise*, 40 (6), 1007-1012.

- Marshall, S. J., Biddle, S. J. H., Gorely, T., Cameron, N. & Murdey, I. (2004). Relationship between media use, body fatness and physical activity in children and youth: a meta-analysis. *International Journal of Obesity*, 28 (10), 1238-1246.
- Marshall, S. J. & Ramirez, E. (2011). Reducing Sedentary Behavior: A new paradigm in physical activity promotion. *American Journal of Lifestyle Medicine*, 5 (6), 518-530.
- Martin, A., Mc Neil, M., Penpraze, V., Dall, P., Granat, M., Paton, J. & Reilly, J. (2011). Objective measurement of habitual sedentary behavior in pre-school children: comparison of activPAL with actigraph monitors. *Pediatric Exercise Science*, 23 (4), 468-476.
- Martinez-Gomez, D., Eisenmann, J. C., Gomez-Martinez, S., Veses, A., Marcos, A. & Veiga, O. L. (2010). Sedentary Behavior, Adiposity, and Cardiovascular Risk Factors in Adolescents. The AFINOS Study. *Revista Espanyola de Cardiologia*, 63 (3), 277-285.
- Martinez-Gomez, D., Eisenmann, J. C., Healy, G. N., Gomez-Martinez, S., Esperanza Diaz, L., Dunstan, D. W., Veiga, O. L. & Marcos, A. (2011a). Sedentary Behaviors and Emerging Cardiometabolic Biomarkers in Adolescents. *The Journal of Pediatrics*, 160 (1), E 104-110.
- Martinez-Gomez, D., Ortega, F. B., Ruiz, J. R., Vicente-Rodriquez, G., Veiga, O. L.,
  Widhalm, K., Manios, Y., Beghin, L., Valtuena, J., Kafatos, A., Molnar, D., Moreno,
  L. A., Marcos, A., Castillo, M. J. & Sjostrom, M. (2011b). Excessive Sedentary Time
  and Low Cardiorespiratory Fitness in European Adolescents: The Helena Study. *Archives of Disease in Childhood*, 96 (3), 240-246.
- Martinez-Gomez, D., Tucker, J., Heelan, K. A., Welk, G. J. & Eisenmann, J. C. (2009). Associations between sedentary behavior and blood pressure in young children. *Archives of Pediatrics and Adolescent Medicine*, 163 (8), 724-730.
- Martínez-Vizcaíno, V. & Sánchez-López, M. (2008). Relationship between physical activity and physical fitness in children and adolescents. *Revista Espanola de Cardiologia*, 61 (2), 108-111.
- Masse, L. C., Fuemmeler, B. F., Anderson, C. B., Matthews, C. E., Trost, S. G., Catellier, D. J. & Treuth, M. (2005). Accelerometer data reduction: a comparison of four reduction algorithms on select outcome variables. *Medicine and Science in Sports and Exercise*, 37 (11), S 544-554.
- Masse, L. C., Fulton, J. E., Watson, K. L., Heesch, K. C., Kohl, H. W., Blair, S. N. & Tortolero, S. R. (1999). Detecting bouts of physical activity in a field setting. *Research Quarterly for Exercise and Sport*, 70 (3), 212-219.
- Matarese, L. E. (1997). Indirect calorimetry: technical aspects. *Journal of the American Dietetic Association*, 97 (10), S 154-160.

- Matthews, C. E. 2002. Use of Self-Report Instruments to Assess Physical Activity. In: Welk, G. J. (ed.) Physical Activity Assessment for Health-Related Research. Champaign, IL: Human Kinetics Publishers.
- Matthews, C. E., Chen, K. Y., Freedson, P. S., Buchowski, M. S., Beech, B. M., Pate, R. R. & Troiano, R. P. (2008). Amount of time spent in sedentary behaviors in the United States, 2003–2004. *American Journal of Epidemiology*, 167 (7), 875-881.
- Matthews, C. E., George, S. M., Moore, S. C., Bowles, H. R., Blair, A., Park, Y., Troiano, R. P., Hollenbeck, A. & Schatzkin, A. (2012). Amount of time spent in sedentary behaviors and cause-specific mortality in US adults. *The American journal of clinical nutrition*, 95 (2), 437-445.
- Mattocks, C., Leary, S., Ness, A., Deere, K., Saunders, J., Tilling, K., Kirkby, J., Blair, S. N.
  & Riddoch, C. (2007). Calibration of an accelerometer during free-living activities in children. *International Journal of Pediatric Obesity*, 2 (4), 218-26.
- Mckenzie, T. L. (1991). Observational measures of children's physical activity. *Journal of School Health*, 61 (5), 224-227.
- Mckenzie, T. L. 2002. Use of Direct Observation to Assess Physical Activity. *In:* Welk, G. J. (ed.) *Physical Activity Assessment for Health Related Research*. Champaign, IL: Human Kinetics Publishers.
- Mckenzie, T. L., Marshall, S. J., Sallis, J. F. & Conway, T. L. (2000). Leisure-time physical activity in school environments: an observational study using SOPLAY. *Preventive Medicine*, 30 (1), 70-77.
- Mckenzie, T. L., Sallis, J. F. & Nader, P. R. (1991a). SOFIT: system for observing fitness instruction time. *Journal of Teaching in Physical Education*, 11 (2), 195-205.
- Mckenzie, T. L., Sallis, J. F., Nader, P. R., Patterson, T. L., Elder, J., Berry, C. C., Rupp, J. W., Atkins, C. J., Buono, M. J. & Nelson, J. A. (1991b). BEACHES: an observational system for assessing children's eating and physical activity behaviors and associated events. *Journal of Applied Behavior Analysis*, 24 (1), 141-151.
- Mcmurray, R., Harrell, J., Levine, A. & Gansky, S. (1995). Childhood obesity elevates blood pressure and total cholesterol independent of physical activity. *International Journal of Obesity and Related Metabolic Disorders*, 19 (12), 881-886.
- Melanson Jr, E. L., Freedson, P. S. & Blair, D. S. (1996). Physical activity assessment: a review of methods. *Critical Reviews in Food Science & Nutrition*, 36 (5), 385-396.
- Millar, D. 2012. Analysis of School Attendance Data in Primary and Post-Primary Schools, 2009/10. Dublin: National Educational Welfare Board.

- Millasseau, S. C., Guigui, F. G., Kelly, R. P., Prasad, K., Cockcroft, J. R., Ritter, J. M. & Chowienczyk, P. J. (2000). Noninvasive assessment of the digital volume pulse: comparison with the peripheral pressure pulse. *Hypertension*, 36 (6), 952-956.
- Mitchell, A. M., Mattocks, C., Ness, A. R., Leary, S. D., Pate, R. R., Dowda, M., Blair, S. N. & Riddoch, C. (2009). Sedentary Behaviour and Obesity in a Large Cohort of Children. Obesity, 17 (8), 1596-1602.
- Mitchell, J. A., Pate, R. R., Dowda, M., Mattocks, C., Riddoch, C., Ness, A. R. & Blair, S. N. (2012). A Prospective Study of Sedentary Behavior in a Large Cohort of Youth. *Medicine and Science in Sports and Exercise*, 44 (6), 1081-1087.
- Miyashita, M., Burns, S. F. & Stensel, D. J. (2006). Exercise and postprandial lipemia: effect of continuous compared with intermittent activity patterns. *American Journal of Clinical Nutrition*, 83, 24-29.
- Montoye, H. J., Washburn, R., Servais, S., Ertl, A., Webster, J. G. & Nagle, F. J. (1983). Estimation of energy expenditure by a portable accelerometer. *Medicine and Science in Sports and Exercise*, 15 (5), 403-407.
- Moran, R. (1999). Evaluation and treatment of childhood obesity. *American Family Physician*, 59 (4), 861-877.
- Morris, J., Heady, J., Raffle, P., Roberts, C. & Parks, J. (1953). Coronary heart-disease and physical activity of work. *The Lancet*, 265 (6795), 1053-1057.
- Morrow, J. R. 2002. Measurement Issues for the Assessment of Physical Activity. *In:* Welk, G. J. (ed.) *Physical Activity Assessment for Health Related Research*. Champaign, IL.: Human Kinetics Publishers.
- Murphy, M. H. & Hardman, A. E. (1998). Training effects of short and long bouts of brisk walking in sedentary women. *Medicine and Science in Sports and Exercise*, 30 (1), 152-157.
- Murphy, M. H., Nevill, A. M. & Hardman, A. E. (2000). Different patterns of brisk walking are equally effective in decreasing postprandial lipaemia. *International Journal of Obesity*, 24 (10), 1303-1309.
- Murphy, M. H., Nevill, A. M., Neville, C., Biddle, S. & Hardman, A. E. (2002). Accumulating brisk walking for fitness, cardiovascular risk and psychological health. *Medicine and Science in Sports and Exercise*, 34 (9), 1468-1474.
- Must, A. (2003). Does overweight in childhood have an impact on adult health? *Nutrition Reviews*, 61 (4), 139-142.
- Must, A. & Strauss, R. S. (1999). Risks and consequences of childhood and adolescent obesity. *International Journal of Obesity and Related Metabolic Disorders*, 23 (3), S 2-11.
- Nakano, T., Sei, M., Ewis, A. a. A., Munakata, H., Onishi, C. & Nakahori, Y. (2010). Tracking overweight and obesity in Japanese children; a six years longitudinal study. *Journal of Medical Investigation*, 57 (1-2), 114-123.
- National High Blood Pressure Education Program 2005. The fourth report on the diagnosis, evaluation, and treatment of high blood pressure in children and adolescents, Bethesda, MD, US Department of Health and Human Services, National Institutes of Health, National Heart, Lung, and Blood Institute, National High Blood Pressure Education Program.
- Ng, S., Zaghloul, S., Ali, H., Harrison, G. & Popkin, B. (2011). The prevalence and trends of overweight, obesity and nutrition-related non-communicable diseases in the Arabian Gulf States. *obesity reviews*, 12 (1), 1-13.
- Nic Gabhainn, S., Kelly, C. & Molcho, M. 2007. *HBSC Ireland 2006: National Report of the 2006 Health Behaviour in School-aged Children in Ireland*. Dublin: Department of Health and Children.
- Nielsen, S. & Jensen, M. D. (1997). Obesity and cardiovascular disease: is body structure a factor? *Current Opinion in Lipidology*, 8 (4), 200-204.
- O'hara, N., Baranowski, T., Simons-Morton, B., Wilson, B. & Parcel, G. (1989). Validity of the observation of children's physical activity. *Research Quarterly for Exercise and Sport*, 60 (1), 42-47.
- Ogden, C. & Carroll, M. (2010). Prevalence of obesity among children and adolescents: United States, trends 1963–1965 through 2007–2008. *Atlanta: Centers for Disease Control and Prevention. National Center for Health Statistics.*
- Okasha, M., Mccarron, P., Gunnell, D. & Davey Smith, G. (2003). Exposures in childhood, adolescence and early adulthood and breast cancer risk: a systematic review of the literature. *Breast Cancer Research and Treatment*, 78 (2), 223-276.
- Olds, T. S., Tomkinson, G. R., Ferrar, K. E. & Maher, C. A. (2009). Trends in the prevalence of childhood overweight and obesity in Australia between 1985 and 2008. *International journal of obesity*, 34 (1), 57-66.
- Owen, N., Healy, G. N., Matthews, C. E. & Dunstan, D. W. (2010). Too much sitting: the population health science of sedentary behavior. *Exercise and Sport Sciences Reviews*, 38 (3), 105-113.
- Owen, N., Leslie, E., Salmon, J. & Fotheringham, M. J. (2000). Environmental determinants of physical activity and sedentary behavior. *Exercise and Sport Sciences Reviews*, 28 (4), 153-158.

- Paffenbarger Jr, R. S., Hyde, R., Wing, A. L. & Hsieh, C. (1986). Physical activity, all-cause mortality, and longevity of college alumni. *New England Journal of Medicine*, 314 (10), 605-613.
- Paffenbarger, R. S., Wing, A. L., Hyde, R. T. & Jung, D. L. (1983). Physical activity and incidence of hypertension in college alumni. *American Journal of Epidemiology*, 117 (3), 245-257.
- Pate, R. R. 2007. Historical Perpective on Physical Activity, Fitness, and Health. In: Bouchard, C., Blair, S. N. & Haskell, W. L. (eds.) Physical Activity and Health. Champaign, IL.: Human Kinetics.
- Pate, R. R., Almeida, M. J., Mciver, K. L., Pfeiffer, K. A. & Dowda, M. (2006). Validation and Calibration of an Accelerometer in Preschool Children&ast. Obesity, 14 (11), 2000-2006.
- Pate, R. R., O'neill, J. R. & Lobelo, F. (2008). The Evolving Definition of "Sedentary". Exercise and Sport Sciences Reviews, 36 (4), 173-178.
- Pate, R. R., Pratt, M., Blair, S. N., Haskell, W. L., Macera, C. A., Bouchard, C., Buchner, D., Ettinger, W., Heath, G. W. & King, A. C. (1995). Physical activity and public health. A recommendation from the centres for disease control and prevention and the American college of sports medicine. JAMA: the Journal of the American Medical Association, 273 (5), 402-407.
- Pfeiffer, K. A., Mciver, K. L., Dowda, M., Almeida, M. J. & Pate, R. R. (2006). Validation and calibration of the Actical accelerometer in preschool children. *Medicine and Science in Sports and Exercise*, 38 (1), 152-157.
- Philippaerts, R., Westerterp, K. & Lefevre, J. (1999). Doubly labelled water validation of three physical activity questionnaires. *International Journal of Sports Medicine*, 20 (5), 284-289.
- Physical Activity Guidelines Advisory Committee. 2008. *Physical activity guidelines advisory committee report.* Washington, DC: Us Department of Health and Human Services.
- Plasqui, G. & Westerterp, K. R. (2007). Physical Activity Assessment With Accelerometers: An Evaluation Against Doubly Labeled Water. *Obesity*, 15 (10), 2371-2379.
- Poole, D. C., Schaffartzik, W., Knight, D. R., Derion, T., Kennedy, B., Guy, H. J., Prediletto, R. & Wagner, P. D. (1991). Contribution of excising legs to the slow component of oxygen uptake kinetics in humans. *Journal of Applied Physiology*, 71 (4), 1245-1260.
- Powell, K. E., Paluch, A. E. & Blair, S. N. (2011). Physical Activity for Health: What Kind? How Much? How Intense? On Top of What? *Annual Review of Public Health*, 32, 349-365.

- Powell, K. E., Thompson, P. D., Caspersen, C. J. & Kendrick, J. S. (1987). Physical activity and the incidence of coronary heart disease. *Annual Review of Public Health*, 8, 253-287.
- Prince, S. A., Adamo, K. B., Hamel, M. E., Hardt, J., Gorber, S. C. & Tremblay, M. (2008). A comparison of direct versus self-report measures for assessing physical activity in adults: a systematic review. *International Journal of Behavioral Nutrition and Physical Activity*, 5 (1), 56-80.
- Puhl, J., Greaves, K., Hoyt, M. & Baranowski, T. (1990). Children's Activity Rating Scale (CARS): description and calibration. Research Quarterly for Exercise and Sport, 61 (1), 26-36.
- Purslow, L. R., Hill, C., Saxton, J., Corder, K. & Wardle, J. (2008). Differences in physical activity and sedentary time in relation to weight in 8–9 year old children. *International Journal of Behavioral Nutrition and Physical Activity*, 5 (67), 1-9.
- Puyau, M. R., Adolph, A. L., Vohra, F. A. & Butte, N. F. (2002). Validation and calibration of physical activity monitors in children. *Obesity*, 10 (3), 150-157.
- Puyau, M. R., Adolph, A. L., Vohra, F. A., Zakeri, I. & Butte, N. F. (2004). Prediction of activity energy expenditure using accelerometers in children. *Medicine and Science in Sports and Exercise*, 36 (9), 1625-1631.
- Raitakari, O. T., Juonala, M., Kähönen, M., Taittonen, L., Laitinen, T., Mäki-Torkko, N., Järvisalo, M. J., Uhari, M., Jokinen, E. & Rönnemaa, T. (2003). Cardiovascular risk factors in childhood and carotid artery intima-media thickness in adulthood. JAMA: the Journal of the American Medical Association, 290 (17), 2277-2283.
- Ravussin, E., Lillioja, S., Anderson, T. E., Christin, L. & Bogardus, C. (1986). Determinants of 24-hour energy expenditure in man. Methods and results using a respiratory chamber. *Journal of Clinical Investigation*, 78 (6), 1568-1578.
- Reich, A., Müller, G., Gelbrich, G., Deutscher, K., Gödicke, R. & Kiess, W. (2003). Obesity and blood pressure—results from the examination of 2365 schoolchildren in Germany. *International Journal of Obesity*, 27 (12), 1459-1464.
- Reilly, J. J. (2005). Physical activity and obesity in childhood and adolescence. *The Lancet*, 366 (9482), 268-269.
- Reilly, J. J. & Mcdowell, Z. C. (2003). Physical activity interventions in the prevention and treatment of paediatric obesity: systematic review and critical appraisal. *Proceedings of* the Nutrition Society, 62 (3), 611-620.
- Reilly, J. J., Methven, E., Mcdowell, Z. C., Hacking, B., Alexander, D., Stewart, L. & Kelnar, C. J. H. (2003). Health consequences of obesity. *Archives of Disease in Childhood*, 88 (9), 748-752.

- Reilly, J. J., Penpraze, V., Hislop, J., Davies, G., Grant, S. & Paton, J. Y. (2008). Objective measurement of physical activity and sedentary behaviour: review with new data. *Archives of Disease in Childhood*, 93 (7), 614-619.
- Riddoch, C. J., Bo Andersen, L., Wedderkopp, N., Harro, M., Klasson-Heggebø, L., Sardinha, L. B., Cooper, A. R. & Ekelund, U. (2004). Physical activity levels and patterns of 9 and 15 yr-old European children. *Medicine and Science in Sports and Exercise*, 36 (1), 86-92.
- Ridgers, N. D., Salmon, J., Ridley, K., O'connell, E., Arundell, L., Timperio, A., Linde, J., Nygaard, K., Maclehose, R. & Mitchell, N. (2012). Agreement between activPAL and ActiGraph for assessing children's sedentary time. *International Journal of Behavioral Nutrition and Physical Activity*, 9 (1), 1-15.
- Rogol, A. D., Roemmich, J. N. & Clark, P. A. (2002). Growth at puberty. *Journal of Adolescent Health*, 31 (6), 192-200.
- Rowe, P., Schuldheisz, J. & Van Der Mars, H. (1997). Validation of SOFIT for measuring physical activity of first-to eighth-grade students. *Pediatric Exercise Science*, 9 (2), 136-149.
- Rowlands, A. V. (2007a). Accelerometer assessment of physical activity in children: an update. *Pediatric Exercise Science*, 19 (3), 252-266.
- Rowlands, A. V. & Eston, R. G. (2007). The measurement and interpretation of children's physical activity. *Journal of Sports Science and Medicine*, 6 (3), 270-276.
- Rowlands, A. V., Pilgrim, E. L. & Eston, R. G. (2008). Patterns of habitual activity across weekdays and weekend days in 9-11-year-old children. *Preventive Medicine*, 46 (4), 317-324.
- Rowlands, A. V., Thomas, P. W. M., Eston, R. G. & Topping, R. (2004). Validation of the RT3 triaxial accelerometer for the assessment of physical activity. *Medicine and Science in Sports and Exercise*, 36 (3), 518-524.
- Rowlands, T. W. 2007b. Physical Activity, Fitness, and Children. In: Bouchard, C., Blair, S. N. & Haskell, W. L. (eds.) Physical Activity and Health. Champaign, IL.: Human Kinetics.
- Ruiz, J. R., Ortega, F. B., Moreno, L. A., Wärnberg, J., Gonzalez-Gross, M., Cano, M. D., Gutierrez, A. & Castillo, M. J. (2006). Reference values for serum lipids and lipoproteins in Spanish adolescents: the AVENA study. *Social and Preventive Medicine*, 51 (2), 99-109.
- Ruiz, J. R., Ortega, F. B., Warnberg, J. & Sjostrom, M. (2007). Association of low grade inflammation with physical activity, fitness and fatness in prepubertal children: The European Youth Heart Study. *International Journal of Obesity*, 31 (10), 1545-1551.

- Ryan, C. G., Grant, P. M., Tigbe, W. W. & Granat, M. H. (2006). The validity and reliability of a novel activity monitor as a measure of walking. *British Journal of Sports Medicine*, 40 (9), 779-784.
- Sallis, J. F., Owen, N. & Fotheringham, M. J. (2000). Behavioral epidemiology: a systematic framework to classify phases of research on health promotion and disease prevention. *Annals of Behavioral Medicine*, 22 (4), 294-298.
- Sallis, J. F. & Saelens, B. E. (2000). Assessment of physical activity by self-report: status, limitations, and future directions. Research Quarterly for Exercise and Sport, 71 (2), S 1-14.
- Salmon, J., Tremblay, M. S., Marshall, S. J. & Hume, C. (2011). Health Risks, Correlates, and Interventions to Reduce Sedentary Behavior in Young People. *American Journal* of Preventive Medicine, 41 (2), 197-206.
- Sardinha, L. B., Andersen, L. B., Anderssen, S. A., Quitério, A. L., Ornelas, R., Froberg, K., Riddoch, C. J. & Ekelund, U. (2008). Objectively measured time spent sedentary is associated with insulin resistance independent of overall and central body fat in 9to 10-year-old Portuguese children. *Diabetes Care*, 31 (3), 569-575.
- Schmitz, K. H., Treuth, M., Hannan, P., Mcmurray, R., Ring, K. B., Catellier, D. & Pate, R. (2005). Predicting energy expenditure from accelerometry counts in adolescent girls. *Medicine and Science in Sports and Exercise*, 37 (1), 155-161.
- Schoeller, D. A. (1988). Measurement of energy expenditure in free-living humans by using doubly labeled water. *Journal of Nutrition*, 118 (11), 1278-1289.
- Schoeller, D. A. (1999). Recent advances from application of doubly labeled water to measurement of human energy expenditure. *Journal of Nutrition*, 129 (10), 1765-1768.
- Schoeller, D. A., Ravussin, E., Schutz, Y., Acheson, K. J., Baertschi, P. & Jequier, E. (1986). Energy expenditure by doubly labeled water: validation in humans and proposed calculation. *American Journal of Physiology-Regulatory, Integrative and Comparative Physiology*, 250 (5), 823-830.
- Schofield, G., Quigley, R. & Brown, R. (2009). Does sedentary behaviour contribute to chronic disease or chronic disease risk in adults? *Scientific Committee of Agencies for Nutrition Action*.
- Serdula, M. K., Ivery, D., Coates, R. J., Freedman, D. S., Williamson, D. F. & Byers, T. (1993). Do obese children become obese adults? A review of the literature. *Preventive Medicine*, 22 (2), 167-177.
- Sesso, H. D., Paffenbarger, R. S. & Lee, I. (2000). Physical activity and coronary heart disease in men: The Harvard Alumni Health Study. *Circulation*, 102 (9), 975-980.

- Shephard, R. (2003). Limits to the measurement of habitual physical activity by questionnaires. *British Journal of Sports Medicine*, 37 (3), 197-206.
- Shetty, P. (2002). Human energy requirements: where are we now? Issues emerging from the 2001 Expert Consultation on Energy in Human Nutrition. *Food, Nutrition and Agriculture*, 30, 5-13.
- Shiely, F. & Macdonncha, C. (2009). Meeting the international adolescent physical activity guidelines: a comparison of objectively measured and self-reported physical activity levels. *Irish Medical Journal*, 102 (1), 15-19.
- Silverthorn, D. U. 2012. *Human physiology: An Integrated Approach,* Glenview, IL., Pearson Educations Inc.
- Simonetti, G. D., Eisenberger, U., Bergmann, I. P., Frey, F. J. & Mohaupt, M. G. (2008). Pulse contour analysis: a valid assessment of central arterial stiffness in children? *Pediatric Nephrology*, 23 (3), 439-444.
- Singh, A., Mulder, C., Twisk, J. W. R., Van Mechelen, W. & Chinapaw, M. J. M. (2008). Tracking of childhood overweight into adulthood: a systematic review of the literature. Obesity reviews, 9 (5), 474-488.
- Sirard, J. R. & Pate, R. R. (2001). Physical activity assessment in children and adolescents. *Sports Medicine*, 31 (6), 439-454.
- Sisson, S. & Katzmarzyk, P. (2008). International prevalence of physical activity in youth and adults. *Obesity Reviews*, 9 (6), 606-614.
- Skålén, K., Gustafsson, M., Rydberg, E. K., Hultén, L. M., Wiklund, O., Innerarity, T. L. & Borén, J. (2002). Subendothelial retention of atherogenic lipoproteins in early atherosclerosis. *Nature*, 417 (6890), 750-754.
- Slattery, M., Jacobs, D. & Nichaman, M. (1989). Leisure time physical activity and coronary heart disease death. The US Railroad Study. *Circulation*, 79 (2), 304-311.
- Slootmaker, S. M., Schuit, A. J., Chinapaw, M. J. M., Seidell, J. C. & Van Mechelen, W. (2009). Disagreement in physical activity assessed by accelerometer and self-report in subgroups of age, gender, education and weight status. *International Journal of Behavioral Nutrition and Physical Activity*, 6 (1), 17-26.
- Sorof, J. & Daniels, S. (2002). Obesity hypertension in children. *Hypertension*, 40 (4), 441-447.
- Stamatakis, E., Zaninotto, P., Falaschetti, E., Mindell, J. & Head, J. (2010). Time trends in childhood and adolescent obesity in England from 1995 to 2007 and projections of prevalence to 2015. *Journal of epidemiology and community health*, 64 (2), 167-174.

- Starling, R. D. 2002. Use of Doubly Labeled Water and Indirect Calorimetry to Assess Physical Activity. In: Welk, G. J. (ed.) Physical Activity Assessment for Health Related Research. Champaign, IL: Human Kinetics Publishers.
- Steele, R. M., Brage, S., Corder, K., Wareham, N. J. & Ekelund, U. (2008). Physical activity, cardiorespiratory fitness, and the metabolic syndrome in youth. *Journal of Applied Physiology*, 105 (1), 342-351.
- Steele, R. M., Van Sluijs, E. M. F., Cassidy, A., Griffin, S. J. & Ekelund, U. (2009). Targeting sedentary time or moderate-and vigorous-intensity activity: independent relations with adiposity in a population-based sample of 10-y-old British children. *The American journal of clinical nutrition*, 90 (5), 1185-1192.
- Steinberger, J., Daniels, S. R., Eckel, R. H., Hayman, L., Lustig, R. H., Mccrindle, B. & Mietus-Snyder, M. L. (2009). Progress and challenges in metabolic syndrome in children and adolescents. *Circulation*, 119 (4), 628-647.
- Stephens, B. R., Granados, K., Zderic, T. W., Hamilton, M. T. & Braun, B. (2010). Effects of 1 day of inactivity on insulin action in healthy men and women: interaction with energy intake. *Metabolism*, 60 (7), 941-949.
- Strong, W. B., Malina, R. M., Blimkie, C. J. R., Daniels, S. R., Dishman, R. K., Gutin, B., Hergenroeder, A. C., Must, A., Nixon, P. A., Pivarnik, J. M., Rowlands, T. W., Trost, S. G. & Trudeau, F. (2005). Evidence Based Physical Activity for School-Age Youth. *Journal of Pediatrics*, 146 (6), 732-737.
- Sun, S. S., Grave, G. D., Siervogel, R. M., Pickoff, A. A., Arslanian, S. S. & Daniels, S. R. (2007). Systolic blood pressure in childhood predicts hypertension and metabolic syndrome later in life. *Pediarics*, 119 (2), 237-246.
- Swets, J. A. (1988). Measuring the accuracy of diagnostic systems. *Science*, 240 (4857), 1285-1293.
- Tanner, J. M. & Whitehouse, R. H. (1976). Clinical longitudinal standards for height, weight, height velocity, weight velocity, and stages of puberty. Archives of Disease in Childhood, 51 (3), 170-179.
- Taylor, H. L., Klepetar, E., Keys, A., Parlin, W., Blackburn, H. & Puchner, T. (1962). Death rates among physically active and sedentary employees of the railroad industry. *American Journal of Public Health*, 52 (10), 1697-1707.
- Telama, R. (2009). Tracking of physical activity from childhood to adulthood: a review. *Obesity facts,* 2 (3), 187-195.
- Thompson, D. R., Obarzanek, E., Franko, D. L., Barton, B. A., Morrison, J., Biro, F. M., Daniels, S. R. & Striegel-Moore, R. H. (2007). Childhood overweight and cardiovascular disease risk factors: the National Heart, Lung, and Blood Institute Growth and Health Study. *Journal of Pediatrics*, 150 (1), 18-25.

- Thorp, A. A., Healy, G. N., Owen, N., Salmon, J., Ball, K., Shaw, J. E., Zimmet, P. Z. & Dunstan, D. W. (2010). Deleterious associations of sitting time and television viewing time with cardiometabolic risk biomarkers. *Diabetes Care*, 33 (2), 327-334.
- Treiber, F. A., Musante, L., Hartdagan, S., Davis, H., Levy, M. & Strong, W. B. (1989). Validation of a heart rate monitor with children in laboratory and field settings. *Medicine and Science in Sports and Exercise*, 21 (3), 338-342.
- Tremblay, M., Barnes, J., Behrens, T., Benden, M., Biddle, S., Bond, D., Brassard, P., Brown, H., Carr, L., Chaput, J. & Christian, H. (2012). Standardized use of the term "Sedentary" and "Sedentary Behaviours". *Applied Physiology Nutrition and Metabolism*, 37 (3), 540-542.
- Tremblay, M., Colley, R. C., Saunders, T. J., Healy, G. N. & Owen, N. (2010). Physiological and health implications of a sedentary lifestyle. *Applied Physiology, Nutrition, and Metabolism,* 35 (6), 725-740.
- Tremblay, M. S., Esliger, D. W., Tremblay, A. & Colley, R. (2007). Incidental movement, lifestyle-embedded activity and sleep: new frontiers in physical activity assessment *Applied Physiology, Nutrition and Metabolism,* 32 (2), S 208-217.
- Tremblay, M. S., Leblanc, A. G., Janssen, I., Kho, M. E., Hicks, A., Murumets, K., Colley, R. C. & Duggan, M. (2011). Canadian sedentary behaviour guidelines for children and youth. *Applied Physiology, Nutrition and Metabolism,* 36 (1), 59-64.
- Treuth, M. S., Hou, N., Young, D. R. & Maynard, L. M. (2005). Accelerometry-Measured Activity or Sedentary Time and Overweight in Rural Boys and Girls. *Obesity*, 13 (9), 1606-1614.
- Treuth, M. S., Schmitz, K., Catellier, D. J., Mcmurray, R. G., Murray, D. M., Almeida, M. J., Going, S., Norman, J. E. & Pate, R. (2004). Defining accelerometer thresholds for activity intensities in adolescent girls. *Medicine and Science in Sports and Exercise*, 36 (7), 1259-1266.
- Troiano, R. P. (2005). A timely meeting: objective measurement of physical activity. *Medicine and Science in Sports and Exercise*, 37 (11), S 487-489.
- Troiano, R. P. (2006). Translating accelerometer counts into energy expenditure: advancing the quest. *Journal of Applied Physiology*, 100 (4), 1107-1108.
- Troiano, R. P., Berrigan, D., Dodd, K. W., Mâsse, L. C., Tilert, T. & Mcdowell, M. (2008). Physical activity in the United States measured by accelerometer. *Medicine and Science in Sports and Exercise*, 40 (1), 181-188.
- Trost, S. G. (2001). Objective measurement of physical activity in youth: current issues, future directions. *Exercise and Sport Sciences Reviews*, 29 (1), 32-36.

- Trost, S. G. (2007). State of the art reviews: measurement of physical activity in children and adolescents. *American Journal of Lifestyle Medicine*, 1 (4), 299-314.
- Trost, S. G., Loprinzi, P. D., Moore, R. & Pfeiffer, K. A. (2011). Comparison of accelerometer cut points for predicting activity intensity in youth. *Medicine and Science in Sports and Exercise*, 43 (7), 1360-1368.
- Trost, S. G., Mciver, K. L. & Pate, R. R. (2005). Conducting accelerometer-based activity assessments in field-based research. *Medicine and Science in Sports and Exercise*, 37 (11), S 531-543.
- Trost, S. G., Morgan, A. M., Saunders, R., Felton, G., Ward, D. S. & Pate, R. R. (2000). Children's understanding of the concept of physical activity. *Pediatric Exercise Science*, 12 (3), 293-299.
- Trost, S. G., Ward, D. S., Moorehead, S. M., Watson, P. D., Riner, W. & Burke, J. R. (1998). Validity of the computer science and applications (CSA) activity monitor in children. *Medicine and Science in Sports and Exercise*, 30 (4), 629-633.
- Troutman, S., Allor, K., Hartmann, D. & Pivarnik, J. (1999). MINI-LOGGER--reliability and validity for estimating energy expenditure and heart rate in adolescents. *Research Quarterly for Exercise and Sport,* 70 (1), 70-74.
- Trudeau, F., Laurencelle, L. & Shephard, R. O. Y. J. (2004). Tracking of physical activity from childhood to adulthood. *Medicine and Science in Sports and Exercise*, 36 (11), 1937-1943.
- Tudor-Locke, C., Ainsworth, B. E. & Popkin, B. M. (2001). Active Commuting to School: An Overlooked Source of Childrens Physical Activity? *Sports Medicine*, 31 (5), 309-313.
- Tudor-Locke, C., Williams, J. E., Reis, J. P. & Pluto, D. (2002). Utility of pedometers for assessing physical activity: convergent validity. *Sports Medicine*, 32 (12), 795-808.
- Twisk, J. W. (2001). Physical activity guidelines for children and adolescents: a critical review. *Sports Medicine*, 31 (8), 617-627.
- U.S. Department of Health and Human Services. 2008. *Physical activity guidelines for Americans 2008*. Atlanta, GA.: U.S. Department of Health and Human Services.
- U.S. Department of Health and Human Services and U.S. Department of Agriculture. 2005. *Dietary Guidelines for Americans*. Washington, D.C.: U.S. Government Printing Office.
- U.S. Dept. Of Health and Human Services. 1996. *Physical Activity and Health: A report from the Surgeon General*. Atlanta, GA.

- Utter, J., Scragg, R. & Schaaf, D. (2006). Associations between television viewing and consumption of commonly advertised foods among New Zealand children and young adolescents. *Public Health Nutrition*, 9 (5), 606-612.
- Varo, J. J., Martínez-González, M. A., De Irala-Estévez, J., Kearney, J., Gibney, M. & Martínez, J. A. (2003). Distribution and determinants of sedentary lifestyles in the European Union. *International Journal of Epidemiology*, 32 (1), 138-146.
- Vonesh, E. F., Chinchilli, V. M. & Pu, K. (1996). Goodness-of-fit in generalized nonlinear mixed-effects models. *Biometrics*, 52 (2), 572-587.
- Wang, X. L., Sim, A. S., Badenhop, R. F., Mccredie, R. M. & Wilcken, D. E. L. (1996). A smoking–dependent risk of coronary artery disease associated with a polymorphism of the endothelial nitric oxide synthase gene. *Nature Medicine*, 2 (1), 41-45.
- Wannamethee, S. G., Shaper, A. G. & Walker, M. (1998). Changes in physical activity, mortality, and incidence of coronary heart disease in older men. *The Lancet*, 351 (9116), 1603-1608.
- Warburton, D. E. R., Nicol, C. W. & Bredin, S. S. D. (2006). Health benefits of physical activity: the evidence. *Canadian Medical Association Journal*, 174 (6), 801-809.
- Ward, D. S., Evenson, K. R., Vaughn, A., Rodgers, A. B. & Troiano, R. P. (2005). Accelerometer Use in Physical Activity: Best Practices and Research Recommendations. *Medicine and Science in Sports and Exercise*, 37 (11), S 582-588.
- Wareham, N. J., Van Sluijs, E. M. F. & Ekelund, U. (2005). Physical activity and obesity prevention: a review of the current evidence. *Proceedings of the Nutrition Society*, 64 (2), 229-247.
- Warren, J. M., Ekelund, U., Besson, H., Mezzani, A., Geladas, N. & Vanhees, L. (2010). Assessment of physical activity–a review of methodologies with reference to epidemiological research: a report of the exercise physiology section of the European Association of Cardiovascular Prevention and Rehabilitation. *European Journal of Cardiovascular Prevention & Rehabilitation*, 17 (2), 127-139.
- Washburn, R., Chin, M. & Montoye, H. (1980). Accuracy of pedometer in walking and running. *Research Quarterly for Exercise and Sport*, 51 (4), 695-702.
- Weiss, R., Dziura, J., Burgert, T. S., Tamborlane, W. V., Taksali, S. E., Yeckel, C. W., Allen, K., Lopes, M., Savoye, M. & Morrison, J. (2004). Obesity and the metabolic syndrome in children and adolescents. *New England Journal of Medicine*, 350 (23), 2362-2374.
- Welk, G. J. 2002. Use of Accelerometry-Based Activity Monitors to Assess Physical Activity. In: Welk, G. J. (ed.) Physical Activity Assessments for Health-Related Research. Champaign, IL: Human Kinetics Publishers.

- Welk, G. J. (2005). Principles of design and analyses for the calibration of accelerometrybased activity monitors. *Medicine and Science in Sports and Exercise*, 37 (11), S 501-511.
- Welk, G. J., Corbin, C. B. & Dale, D. (2000). Measurement issues in the assessment of physical activity in children. *Research Quarterly for Exercise and Sport*, 71 (2), S 59-73.
- Welk, G. J., Eisenmann, J. C., Schaben, J., Trost, S. G. & Dale, D. (2007). Calibration of the biotrainer pro activity monitor in children. *Pediatric Exercise Science*, 19 (2), 145-158.
- Welk, G. J., Schaben, J. A. & Morrow Jr, J. R. (2004). Reliability of accelerometry-based activity monitors: a generalizability study. *Medicine and Science in Sports and Exercise*, 36 (9), 1637-1645.
- Westerterp, K. (1999). Physical activity assessment with accelerometers. *International Journal* of Obesity and Related Metabolic Disorders, 23 (4), S 45-49.
- Westerterp, K. R. (2009). Assessment of physical activity: a critical appraisal. *European Journal of Applied Physiology*, 105 (6), 823-828.
- Wilks, D. C., Besson, H., Lindroos, A. K. & Ekelund, U. (2011). Objectively measured physical activity and obesity prevention in children, adolescents and adults: a systematic review of prospective studies. *Obesity Reviews*, 12 (5), E 119-129.
- Woods, C. B., Moyna, N., Quinlan, A., Tannehill, D. & Walsh, J. 2010. *The Children's Sport Participation and Physical Activity Study (CSPPA)*. Dublin: The School of Health and Human Performance and the Irish Sports Council.
- World Health Organisation. 2000. Obesity: Preventing and Managing the Global Epidemic. Geneva: Who Press.
- World Health Organisation. 2011. Physical Activity and Young People. Geneva: Who Press.
- Yusuf, S., Hawken, S., Ôunpuu, S., Dans, T., Avezum, A., Lanas, F., Mcqueen, M., Budaj, A., Pais, P. & Varigos, J. (2004). INTERHEART Study Investigators. Effect of potentially modifiable risk factors associated with myocardial infarction in 52 countries (the INTERHEART study): case-control study. *The Lancet*, 364 (9438), 937-952.
- Zderic, T. W. & Hamilton, M. T. (2006). Physical inactivity amplifies the sensitivity of skeletal muscle to the lipid-induced downregulation of lipoprotein lipase activity. *Journal of Applied Physiology*, 100 (1), 249-257.
- Zhu, N., Pankow, J. S., Ballantyne, C. M., Couper, D., Hoogeveen, R. C., Pereira, M., Duncan, B. B. & Schmidt, M. I. (2010). High-molecular-weight adiponectin and the risk of type 2 diabetes in the ARIC study. *Journal of Clinical Endocrinology and Metabolism*, 95 (11), 5097-5104.

Zweig, M. H. & Campbell, G. (1993). Receiver-operating characteristic (ROC) plots: a fundamental evaluation tool in clinical medicine. *Clinical Chemistry*, 39 (4), 561-577.

# Appendices

Prof. Alan Donnelly, PESS Dept., University of Limerick, Castletroy, Co. Limerick.

Dear Committee,

I am writing to you today on behalf of our research team in the University of Limerick. We are in the final stages of preparation for a physical activity research project. The project title is *"Predicting Energy Expenditure from activPAL accelerometry counts in Adolescent Females"*. We are enquiring about whether members of your group, the No Name Club, may be interested in participating in the research.

The aim of this study is to develop an accelerometer threshold for adolescent females that can be used to examine energy expenditure and physical activity in free-living activities.

It is expected that a total of 40 female adolescents in will participate in this study. We would hope that approximately 30 of your members would participate, with the remainder of participants from a school in the Limerick region. I have attached a project plan of this study which will explain the full involvement required of your members.

I would ask if you could read the attached information sheet. If you have any questions regarding this research study, please feel free to contact my Ph.D. research student, who is in charge of study planning and preparation, Mr. Kieran Dowd on (061) 234715 or via email at <u>kieran.dowd@ul.ie</u>. Mr. Kieran Dowd will contact you within the next couple of days to arrange a meeting, where you can discuss and examine the advantages and disadvantages of this research project. We have also attached a consent form, in case you do not require a meeting and are happy to consent to participation of your school.

Yours sincerely,

Prof. Alan Donnelly PESS Dept., University of Limerick, Castletroy, Co. Limerick.



# OLLSCOIL LUIMNIGH Committee Information Sheet

Study Title: Predicting Energy Expenditure from activPAL accelerometry counts in Adolescent Females.

### Background:

Physical activity has been shown to positively affect a person's current and future health. Currently, all studies that have measured physical activity levels in Ireland have used questionnaires and diaries. These methods of assessing physical activity have limitations, particularly with accuracy. When examining physical activity, it is important to use valid and accurate measures. Currently the most frequently employed measure of physical activity is accelerometers. This study aims to validate a new accelerometer, the activPAL, in female adolescents and to investigate if it is an accurate measure of predicted energy expenditure.

What is required of you, the committee: As adult committee members, we ask for your consent to carry out the proposed research with your members. We ask you to volunteer a member of the committee to be our contact point throughout the research. This will simply involve contacting this person to arrange meetings with the host/hostesses. After this, the only other request we ask is one of support while completing the study.

**Participant Selection:** We hope to select participants from your members with the help of your host and hostesses. They will be asked to contact members of the club, and to recruit the members to participate in the study.

**Parental/Participant consent:** We will meet with the selected students at the earliest possible time. We will discuss the study outline, and explain what is involved in the study. We will then provide the students with informed parental and participant consent forms, which are required prior to the study beginning. We will then ask the students to consider participating in the study, and to complete the required forms and return them to the hosts and hostesses of the no name club.

### **Testing:**

All testing will take place in an appropriately sized space (yet to be confirmed) at times that are suitable with all participants. The first type of testing the girls will take part in a simple physiological test of height and weight. This will take place in a private space, and all measurements obtained will be private and confidential. At <u>no time</u> will the information be available to anyone else, <u>except the researcher's names on this document</u>. At no time will the participants be on their own with any researcher, and all efforts will be made to make them as comfortable as possible.

• Height and Weight: Height and weight will be measured in stocking feet, light shorts and t-shirt. This information will be used to calculate BMI (Body Mass Index).

The members will be given a heart rate monitor for the main testing. This is simply a watch on the wrist, which picks up a signal from a strap fitted around the chest. Participants will be given a picture of where to place the heart rate monitor, and will fit the monitor behind a private screen. They will also be asked to wear an activPAL accelerometer. This is attached to the midline of the thigh, halfway between the knee and hip, using a *PAL stickie*.

The students will wear a Cosmed Mobile metabolic unit, which records both inspired gas (breathed in gas) and expired gas (breathed out gas). This allows us to collect and analyse the different levels of oxygen and carbon dioxide that is used during specific activities. It is a very light device, which is worn on the back and chest. It is attached to a mask, which is placed over the face, to ensure all gases are measured. A picture of the device in use is provided below.



For the main test, the students will be asked to take part in a number of tasks. These will include lying down, sitting down, standing upright, walking and running. The students will complete each of these tasks for between 5 and 10 minutes. They will also be provided with a rest period between each activity. The students will be free to stop the testing whenever they wish.

**Benefits of this research:** Your members will not be informed directly of the overall results of this research study. However, it is hoped that the results of this study will give us a greater understanding of the uses and accuracy of the activPAL physical activity monitor in a female population. The participants will learn about some of the research methodologies used in a University setting, and may learn some beneficial information about the importance of physical activity for health in adolescents.

#### What are the risks?

The EHS Research Ethics Committee has assessed all risks associated with each testing procedure. As with all forms of exercise, there is a slight risk of injury to participants. This includes, but is not restricted to, strains, sprains and overuse injuries. To reduce these risks, your members will have a warm up and cool down.

# What if the members do not want to take part or change their mind during the study?

We will be delighted with your members taking part in the study. However if they do not wish to take part it is entirely up to them. They can withdraw from the study at any time and we will thank them for their time.

## What happens to the information?

The information is completely confidential and will only be used for the purpose of this project. Only the people involved in this project will have access to the results of the testing. Once the project is complete, all personal information will be destroyed. Remaining data will be retained for the recommended period of 7-10 years.

## Who else is taking part?

It is hoped that 39 other females in the 15-18 age bracket will be involved in this study.

## What if something goes wrong?

In the event of an accident or an emergency, the investigator will follow standard no name club protocol in place to deal with such incidents. The researcher will be informed of all accident and emergency protocol in place for the no name club before the study begins.

## Who can I contact for additional information?

If you would like to know more about this study, please do not hesitate to contact any of the investigators listed below.

Name	Title	Contact No.	E-mail
Mr. Kieran Dowd	Postgraduate Student	087-91XXXX	kieran.dowd@ul.ie
Prof. Alan Donnelly	Professor	061-20XXX	alan.donnelly@ul.ie

Also if you have concerns about this study and wish to contact someone independent, you may contact:

Chairman Education and Health Sciences Research Ethics Committee

EHS Faculty Office

University of Limerick

## Tel (061) 23XXXX

### Email: <u>ehsresearchethics@ul.ie</u>



# OLLSCOIL LUIMNIGH Committee Consent

We have read and understand the committee information sheet and have had due time to consider it. We now fully understand what this project: *Predicting Energy Expenditure from activPAL accelerometry counts in Adolescent Females* is all about and what the data will be used for. We are fully aware of the procedures that are involved in this project. We are aware that the members of our organisation can pull out of this project at any time, and that their participation is completely voluntary. The members do not have to give any reasons as to why they wish to leave the study if they so choose to. We are aware that the student's results will be completely confidential. We agree to allow the research team to select their sample population from our members.

I \_\_\_\_\_\_ understand all that is involved in this study. I hereby agree to our members taking part in this research project.

Signed:

Witness:

Date: \_/\_/\_\_\_



# OLLSCOIL LUIMNIGH

Parent/Carer Information Sheet

Study Title: Predicting Energy Expenditure from activPAL accelerometry counts in Adolescent Females.

### What is this study about?

Physical activity has been shown to positively affect a person's current and future health. Currently, all studies that have measured physical activity levels in Ireland have used questionnaires and diaries. These methods of assessing physical activity have limitations, particularly with accuracy. When examining physical activity, it is important to use accurate measures. Currently the most frequently employed measure of physical activity is accelerometry. This study aims to validate a new accelerometer, the activPAL, in female adolescents and to investigate if it is an accurate measure of predicted energy expenditure.

### What will my daughter have to do?

Your daughter will be asked to come to the testing venue in Claremorris (yet to be confirmed). She will be asked to bring with her a pair of light shorts, a t-shirt and gel soled runners as opposed to flat soled runners.

**Test Day:** On arrival at the testing centre, your daughter will be asked to change into the exercise clothing that she will have brought with her. The first type of testing she will take part in is the height and weight measurement. All measurements obtained will be private and confidential, and at <u>no time</u> will information be available to anyone, <u>except the researcher's named on the last page of this document</u>. Height and weight will be measured

Appendix A: Information Sheet and Consent forms: Study 1

in stocking feet, light shorts and t-shirt by the researcher and this will be used to calculate BMI (Body Mass Index).

Your daughter will then be given a Polar heart rate monitor for the main testing. This is simply a watch to be worn on the wrist, which picks up a signal from a strap fitted around the chest. Your daughter will also be asked to wear an activPAL accelerometer. This is attached to the midline of the thigh, halfway between the knee and hip. Finally, your daughter will wear a Cosmed *Mobile* ergospirometry system. This is a device which records inspired (breathed in gas) and expired gas (breathed out gas). This allows us to collect and analyse the different levels of oxygen and carbon dioxide that your daughter uses during specific activities. It is a very light device, which is worn on the shoulder and back. It is attached to a mask, which is placed over the nose and mouth, to ensure all gases are measured (pictured below).



For the main test, your daughter will be asked to take part in a number of tasks. These will include lying down, sitting down, standing upright, walking and a light jog. Your daughter will complete each of these tasks for between 5 and 12 minutes each. A rest period will be provided between each activity.

#### What are the benefits of this Project?

Your daughter will not be informed directly of the overall results of this research study. However, it is hoped that the results of this study will give us a greater understating of the uses and accuracy of the activPAL physical activity monitor in a female population. Your daughter may obtain some interesting information regarding physical activity and health. She may also ask the researcher any questions about the research and physical activity, which will be answered to the best of the researcher's capabilities.

#### What are the risks to my daughter for this study?

The Education and Health Sciences Research Ethics Committee has assessed all risks associated with each testing procedure. As with all forms of exercise there is a risk of injury to participants. This includes, but is not restricted to, strains, sprains and overuse injuries. To reduce these risks you will have a warm up and cool down.

#### What if my daughter does not want to take part or changes her mind?

We will be delighted with your daughter taking part in the study. However if she does not wish to take part it is entirely up to her. Your daughter can withdraw from the study at any time and we will thank her for her time and help.

#### What happens to the information obtained?

The information is completely confidential and will only be used for the purpose of this project. Only the people involved in this project will have access to the questionnaires and results of the testing. Once the project is complete, all personal information will be destroyed. Remaining data will be retained for the recommended period of 7-10 years.

#### Who else is taking part?

It is hoped that 39 other females in the 15-18 age bracket will participate in this study.

### What if something goes wrong?

In the event of an accident or an emergency, the investigator will follow standard no name club protocol in place to deal with such incidents. The researcher will be informed of all accident and emergency protocol in place for the no name club before the study begins.

Appendix A: Information Sheet and Consent forms: Study 1

## Who can I contact for additional information?

If you would like to know more about this study, please do not hesitate to contact any of the investigators listed below.

Name	Title	Contact No.	E-mail
Mr. Kieran Dowd	Postgraduate Student	087-91XXXX	kieran.dowd@ul.ie
Prof. Alan Donnelly	Professor	061-20XXXX	alan.donnelly@ul.ie

Also if you have concerns about this study and wish to contact someone independent, you may contact:

Chairman Education and Health Sciences Research Ethics Committee

EHS Faculty Office University of Limerick Tel (061) 23XXXX Email: <u>ehsresearchethics@ul.ie</u>



# OLLSCOIL LUIMNIGH

# Parent/Carer Informed Consent Form

I have read and understand the Parent/Carer information sheet and have had due time to consider it. I now fully understand the purposes of the project: *Predicting Energy Expenditure from activPAL*<sup>TM</sup> accelerometry counts in Adolescent Females and what the data will be used for. I am fully aware of the procedures that will involve my child, and I am aware that she can withdraw from this project at any time, and that her participation is completely voluntary. She will not have to give any reasons as to why she leaves the study if she so chooses to. I am aware that her results will be completely confidential.

I\_\_\_\_\_, understand all that is involved in this study. I hereby agree to take part in this research project.

Signed:

Date:



# OLLSCOIL LUIMNIGH

Participant Information Sheet

**Study Title:** Predicting Energy Expenditure from activPAL<sup>TM</sup> accelerometry counts in Adolescent Females

## What is this study about?

Physical activity has been shown to positively affect a person's current and future health. Currently, all studies that have measured physical activity levels in Ireland have used questionnaires and diaries. These methods of assessing physical activity have limitations, particularly with accuracy. When examining physical activity, it is important to use accurate measures. Currently the most frequently employed measure of physical activity is accelerometers. This study aims to examine a new accelerometer, the activPAL in female adolescents and to investigate if it is an accurate measure of predicted energy expenditure.

## What will I have to do?

You are asked to come to the testing venue in Claremorris (yet to be confirmed). You are also asked to bring with you a pair of light shorts, a t-shirt and gel soled runners as opposed to flat soled runners. You will be asked to sign an informed consent form. Your parent/guardian will also be asked to sign a consent form, before you will be allowed to participate.

**Test Day:** On arrival at the testing area, you will be asked to change into the exercise clothing that you have brought with you. The first type of testing you will take part in is the measurement of height and weight. All measurements obtained will be private and confidential, and at <u>no time</u> will your information be available to anyone, <u>except the</u>

<u>researcher's</u> names on this document. At no time will you be on your own with the researcher, and all efforts will be made to make you as comfortable as possible. Height and weight will be measured in stocking feet, light shorts and t-shirt by the researcher and this will be used to calculate BMI (Body Mass Index).

You will then be given a heart rate monitor for the main testing. This is simply a watch on your wrist which picks up a signal from a strap fitted around your chest. Your will be shown how to attach the monitor using a diagram. You will also be asked to wear an activPAL accelerometer. This is attached to the midline of your thigh, halfway between your knee and hip, using a *PAL stickie*.

You will also wear a *Cosmed Mobile ergospirometry system*. This is a device which records the inspired gas (breathed in gas) and expired gas (breathed out gas). This allows us to collect and analyse the different levels of oxygen and carbon dioxide that you use during specific activities. It is a very light device, which is worn on the chest and back. It is attached to a mask, which is placed over the face, to ensure all gases are measured. A picture of the device in use is provided below.



For the main test you will be asked to take part in a number of tasks. These will include lying down, sitting down, standing upright, walking and a light jog. You will complete each of these tasks for between 5 and 10 minutes each. You will also be provided with a rest period between each activity. You will be free to stop the testing whenever you want if you feel too tired, feel unwell or if you get very out of breath.

#### What are the Benefits of this Project?

You will not be informed directly of the overall results of this research study. However, it is hoped that the results of this study will give us a greater understating of the uses and accuracy of the activPAL physical activity monitor in a female population. You may learn some valuable information about physical activity, or perhaps you may have questions Appendix A: Information Sheet and Consent forms: Study 1

about physical activity that the researcher may be able to answer. You will also learn about how some things are measured like body composition, energy expenditure or activity.

#### What are the risks?

The EHS Research Ethics Committee has assessed all risks associated with each testing procedure. As with all forms of exercise there is a slight risk of injury to participants. This includes, but is not restricted to, strains, sprains and overuse injuries. To reduce these risks you will have a warm up and cool down.

#### What if I do not want to take part or change my mind during the study?

We will be delighted with you taking part in the study. However if you do not wish to take part it is entirely up to you. You can withdraw from the study at any time and we will thank you for your time and help.

#### What happens to the information?

The information is completely confidential and will only be used for the purpose of this project. Only the people involved in this project will have access to the questionnaires and results of the testing. Once the project is complete, all personal information will be destroyed. Remaining data will be retained for the recommended period of 7-10 years.

#### Who else is taking part?

It is hoped that 39 other females in the 15-18 age will be involved in this study.

#### What if something goes wrong?

In the event of an accident or an emergency, the investigator will follow standard no name club protocol in place to deal with such incidents. The researcher will be informed of all accident and emergency protocol in place for the no name club before the study begins.

Appendix A: Information Sheet and Consent forms: Study 1

## Who can I contact for additional information?

If you would like to know more about this study, please do not hesitate to contact any of the investigators listed below.

Name	Title	Contact No.	E-mail
Mr. Kieran Dowd	Postgraduate Student	087-91XXXX	kieran.dowd@ul.ie
Prof. Alan Donnelly	Professor	061-20 XXXX	alan.donnelly@ul.ie

Also if you have concerns about this study and wish to contact someone independent, you may contact:

Chairman Education and Health Sciences Research Ethics Committee

EHS Faculty Office

University of Limerick

# Tel (061) 23XXXX

# Email: <u>ehsresearchethics@ul.ie</u>



# OLLSCOIL LUIMNIGH

Participant Informed Consent Form

I have read and understand the subject information sheet and have had due time to consider it. I now fully understand what this project: *Predicting Energy Expenditure from activPAL accelerometry counts in Adolescent Females* is all about and what the data will be used for. I am fully aware of the procedures that are involved in this project. I am aware that I can pull out of this project at any time, and that my participation is completely voluntary. I do not have to give any reasons as to why I leave the study if I so choose to. I am aware that my results will be completely confidential.

I\_\_\_\_\_, understand all that is involved in this study. I hereby agree to take part in this research project.

Signed:

Date:

WITNESS SINATURE:

Signed:

Date:



OLLSCOIL LUIMNIGH

Principal [Name],

[Address]

Dear Principal,

I am making contact with you to discuss the possibility of administering a research study in your Secondary School. A key area of research interest in the Department of Physical Education and Sport Sciences at the University of Limerick is physical activity levels of female adolescents. Regular physical activity during childhood and adolescence is associated with improvements in numerous physiological and psychological variables such as reducing risk of obesity, cardiovascular disease, non-insulin-dependant diabetes, osteoporosis, certain cancer and depression. Regular Physical Activity is characterised by taking part in moderate intensity activity for 30 minutes a day 5 times a week but despite recommendations for adequate Physical Activity, data indicates that youths become less physically active between puberty and age 18.

The purpose of the proposed research is to investigate daily Physical Activity levels of adolescent females in Limerick using accelerometry, and comparing the results with prescribed levels of Physical Activity and with a questionnaire .The study also aims to use anthropometric measurements, along with a 20 meter endurance shuttle run test (bleep test) to compare levels of Physical Activity with physiological fitness and body composition.

The purposes of the proposed study are 1) to establish the Physical Activity rates of 15-17 year old adolescent females using accelerometry and a questionnaire; 2) to examine the weight profiles of students, their blood pressure and their cardiovascular fitness and 3) to examine the interrelationships that exist among the measured variables.

#### Appendix B: Information Sheet and Consent form: Study 2

The study will be of great benefit to those who participate as it will:

- Provide information on the importance of being physically active,
- Educate participants on physiological measures for body composition
- Inform participants on their weight profiles and cardiovascular fitness.

Another benefit of participating in the study is, upon their full completion of the study, each participant will receive a 20% discount for Lifestyle Sports shops in Limerick. This discount provides a more attractive incentive for participants to complete all aspects of the study.

A random sample of 12-15 students from your school is required. If you agree to participate in the study, a list of all children who fit the inclusion criteria will be required, and from this list, participants will be selected. Participation in the study will be subject to the consent of parents and participants. The study will last for 1 week in your school, and the students will only be asked to miss 4 classes, none of which can be their Physical Education class. (1 double class on the first visit, which must take place in the gym, and 2 further single classes on two separate visits).

Deirdre Harrington, BSc in Sport and Exercise Sciences and Kieran Dowd, BSc in Physical Education, both Ph.D. students in the University of Limerick, will assist in administering the study. They will require some assistance from a class teacher in the administration of the study. Additionally, I will closely monitor the study. If you require more in-depth information about the study and the procedures I would be happy to speak with you or forward you on this information.

This study promises to be of great benefit for the participants. I will be in touch with you shortly to enquire whether you are interested in your schools involvement in this study. Should you agree to your schools participation, or if you have any queries, please contact me via the telephone, fax or e-mail information listed at the bottom of this letter. Thank you very much for taking the time to read this information, and I hope to speak to you soon.

Yours sincerely,

Phone: 061-20 XXXX E-mail: <u>Alan.Donnelly@ul.ie</u> Fax: + 353 61 20 XXXX

Prof. Alan Donnelly

PESS Department

University of Limerick



# OLLSCOIL LUIMNIGH

# Parental Information Sheet

Study Title: The Limerick Physical Activity and Fitness Study of Adolescent Girls

## What is the project about?

Regular physical activity during childhood and adolescents is associated with improvements in many physical and psychological health benefits, such as reducing risk of obesity, cardiovascular disease, non-insulin-dependant diabetes, osteoporosis, certain cancers and depression. Regular physical activity is charachterised by taking part in moderate intensity activity for 30 minutes a day 5 times a week. Although these target appear very achieveable, studies show that many Irish adolescents, particularly females, fall short of the prescribed amount of physical activity to maintain a healthy lifestyle.

This study aims to assess participation levels in physical activity in a broad group of female adolescents in the Limerick area. The study will examine these participation levels, and will compare results with prescribed physical activity levels and with other Irish studies findings.

## What will your child have to do?

The first set of tests will be physiological testing. These will take place in an appropriately private space with another female observer present at all times to ensure the safety and protection of both the participants and researchers involved.

- Height and weight will be measured in stocking feet, light shorts and t-shirt by a female researcher to calculate BMI (Body Mass Index).
- Measurement of the waist and hip of your daughter will then be taken.

Appendix B: Information Sheet and Consent form: Study 2

• Skinfolds at four sites will be measured. These are mid-bicep (front of upper arm), mid-triceps (back of the upper arm), sub-scapular (under the shoulder blade) and the iliac crest (side of the body, near the hip bone). All measurements will be taken using a skinfold calliper, and will be performed by an experienced female researcher.

Blood pressure will be measured using a standard procedure. The appropriately sized cuff will be placed around the upper arm and inflated automatically and both systolic and diastolic blood pressures will be recorded.

A 20 meter shuttle run test (Bleep test) will be administered. This is a submaximal test which will involve your daughter running back and forth over a 20 meter distance at specific paces for as long as they can. Your daughter will have the option of dropping out of this test whenever they feel they cannot go any further.

- Physical Activity Levels will be monitored using accelerometry, and is carried out using an accelerometer called an activPAL. The activPAL is a tiny, lightweight box that measures the amount and direction of all the activity that one does. It can tell how long one spends sitting, standing, lying and stepping, and how fast one walks/jogs/runs. The accelerometer is worn on the mid part of the front of the right thigh, and is held onto the skin with a hypoallergenic *stickie*. This *stickie* is both hair and skin friendly. This form of testing is neither invasive, nor will it impede your daughters everyday life. The activPAL only needs to be removed before bathing and any water based activities. The activPAL will be worn for 7 days so we can get any overall picture of her physical activity patterns. Each participant is free to not wear the accelerometer if they feel uncomfortable in any way and can return it before the 7 days are up. This will not affect their future treatment or participation.
- On return of the accelerometers, your daughter will complete a Physical Activity questionnaire. This questionnaire will inform us of your daughters **perceived** activity rates during the same week in which they wore the accelerometer.

#### What are the benefits to your child?

Your daughter will receive a copy of her results, which can inform them of their body composition and cardiovascular fitness. You or your daughter will not be directly informed of the results of this research study. However, it is hoped that the results of this research will inform us of the physical activity levels and physical fitness levels of a broad, random group of female adolescent girls in the Limerick region. Your daughter will also receive a letter, which provides her with a 20% discount off purchases at any Limerick based *Lifestyle Sports* shop, once the activPAL accelerometer has been returned and all questionnaires and physiological tests have been complete.

#### What are the risks?

The UL research Ethics Committee has assessed all risks associated with each testing procedure. All researchers will be appropriately qualified for all testing in the study.

### What if my child does not want to take part?

We are very appreciative of everyone's time, including your time in reading this consent form. However if your child does not wish to take part in the study, it is entirely their decision. They may withdraw from the study at any time and we will thank them for their time and help.

#### Who else is taking part in this Research Project?

It is hoped that 15 other participants from your child's school will take part in this study, and 90 other participants from other schools around Limerick will also be tested.

#### What happens to the information that is gathered?

The information is completely confidential and will only be used for the purpose of this project. Only the people involved in this project will have access to the questionnaires and results of the testing. Once the project is complete, all personal information will be destroyed. Remaining data will be retained for the recommended period of 7-10 years.

Appendix B: Information Sheet and Consent form: Study 2

### Who can I contact for additional information?

If you would like to know more about this study, please do not hesitate to contact any of the investigators listed below.

Name	Title	Contact No.	E-mail
Ms. Deirdre Harrington	Postgraduate Student	086-319XXXX	deirdre.harrington@ul.ie
Mr. Kieran Dowd	Postgraduate Student	087-91XXXX	kieran.dowd@ul.ie
Prof. Alan Donnelly	Senior Lecturer	061-20XXXX	alan.donnelly@ul.ie

Also if you have concerns about this study and wish to contact someone independent, you may contact:

The Chairman of the University of Limerick Research Ethics Committee. C/o Vice President Academic and Registrar's Office University of Limerick Limerick Tel: (061) 20xxxx


## OLLSCOIL LUIMNIGH Parental Consent Form

I have read and understand the parental information sheet and have had due time to consider it. I now fully understand the purposes of the project: *The Limerick Physical Activity and Fitness Study of Adolescent Girls*, and what the data will be used for. I am fully aware of the procedures that will involve my child, and I am aware that she can withdraw from this project at any time, and that her participation is completely voluntary. She will not have to give any reasons as to why she leaves the study if she so chooses to. I am aware that her results will be completely confidential.

I\_\_\_\_\_, understand all that is involved in this study. I hereby agree to allow my child to take part in this research project.

Signed: \_\_\_\_\_

Relationship with the participant:

Date:



## OLLSCOIL LUIMNIGH Participant Information Sheet

Study Title: The Limerick Physical Activity and Fitness Study of Adolescent Girls

What is the project about? Regular physical activity during childhood and adolescents is associated with improvements in many physical and psychological health benefits, such as reducing risk of obesity, cardiovascular disease, non-insulin-dependant diabetes, osteoporosis, colon cancer and depression. Regular physical activity is charachterised by taking part in moderate intensity activity for 30 minutes a day 5 times a week. This study aims to assess participation levels in physical activity in a broad group of female adolescents in the Limerick area. The study will examine these participation levels, and will compare results with prescribed physical activity levels and with other Irish studies findings.

What will you have to do? The first type of testing you will take part in is the physiological testing. This will take place in a private space with a female tester and another female observer will be present at all times. At no point will you be on your own with the people carrying out the tests.

- Height and weight will be measured in stocking feet, light shorts and t-shirt by the female researcher and this will be used to calculate BMI (Body Mass Index).
- Measurement of your waist and hip will then be taken.
- 4 site skinfold test will be administered. The four sites measured are mid-bicep (front of upper arm), mid-triceps (back of the upper arm), sub-scapular (under the shoulder blade) and the iliac crest (side of the body, near the hip bone). All

measurements will be taken using a skinfold calliper, and will be performed by an experienced female researcher.

- Blood pressure will be measured using a standard procedure. The appropriately sized cuff will be placed around the upper arm and inflated automatically. The pressure will be slowly reduced, and both systolic and diastolic blood pressures will be recorded.
- You will participate in a 20 meter shuttle run test (Bleep test). This is a standardised test to measure cardio respiratory fitness in large groups. This is a submaximal test which involves running back and forth a 20 meter distance at specific paces for as long as you can (Many P.E teachers use this as a test in schools, and you may have already completed a bleep test in your current school). You will have the option of dropping out of this test whenever you feel that you cannot go any further..
- Physical Activity levels will be calculated for a week using accelerometry, and is carried out using an accelerometer called an activPAL. The activPAL is a tiny, lightweight box that measures the amount and direction of all the activity that one does. It can tell how long one spends sitting, standing, lying and walking, and how fast one walks. The accelerometer is worn on the mid part of the front of the right thigh, and is held onto the skin with a hypoallergenic stickie. This stickie is both hair and skin friendly. This form of testing is neither invasive, nor will it impede your everyday life. The activPAL only needs to be removed before bathing and any water based activities. The activPAL will be worn for 7 days, including 2 weekend days so we can get any overall picture of your physical activity patterns. Each participant is free to not wear the accelerometer if they feel uncomfortable in any way and can return it before the 7 days are up. This will not affect your future treatment or participation.
- On return of the accelerometers, you will complete a self-report questionnaire. This questionnaire will inform us of your **<u>perceived</u>** activity rates during the same week in which you wore the accelerometer.

#### What are the benefits to you?

You will receive a copy of your results, which can inform you of your body composition and cardiovascular fitness. You will not be directly informed of the results of this research study. However, it is hoped that the results of this research will inform us of the physical activity levels and physical fitness levels of a broad, random group of female adolescent girls in the Limerick region. You will also receive a letter, which provides you with a 20% discount off purchases at any Limerick based *Lifestyle Sports* shop, once the activPAL accelerometer has been returned and all questionnaires and physiological tests have been complete.

#### What are the risks?

The UL Research Ethics Committee has assessed all risks associated with each testing procedure.

### What if I do not want to take part?

We will be delighted with you taking part in the study. However if you do not wish to take part it is entirely up to you. You can withdraw from the study at any time and we will thank you for your time and help.

### Who else is taking part in this Research Project?

It is hoped that 15 other participants from your school will take part in this study, and that 90 other transition year students from schools around Limerick will also participate.

### What happens to the information that is gathered?

The information is completely confidential and will only be used for the purpose of this project. Only the people involved in this project will have access to the questionnaires and results of the testing. Once the project is complete, all personal information will be destroyed. Remaining data will be retained for the recommended period of 7-10 years.

### What if something goes wrong?

In the event of an emergency we have protocol in place to deal with such an emergency.

Appendix B: Information Sheet and Consent form: Study 2

### Who can I contact for additional information?

If you would like to know more about this study, please do not hesitate to contact any of the investigators listed below.

Name	Title	Contact No.	E-mail
Ms. Deirdre Harrington	Postgraduate Student	086-31XXXX	Deirdre.harrington@ul.ie
Mr. Kieran Dowd	Postgraduate Student	087-91XXXX	Kieran.dowd@ul.ie
Mr. Alan Donnelly	Senior Lecturer	061-20XXXX	Alan.donnelly@ul.ie

Also if you have concerns about this study and wish to contact someone independent, you may contact:

The Chairman of the University of Limerick Research Ethics Committee.

C/o Vice President Academic and Registrar's Office

University of Limerick

Limerick

Tel: (061) 20XXXX



# OLLSCOIL LUIMNIGH Participant Consent Form

I have read and understand the subject information sheet and have had due time to consider it. I now fully understand what this project: *The Limerick Physical Activity and Fitness Study of Adolescent Girls* is all about and what the data will be used for. I am fully aware of the procedures involving myself that are involved in this project. I am aware that I can pull out of this project at any time, and that my participation is completely voluntary. I do not have to give any reasons as to why I leave the study if I so choose to. I am aware that my results will be completely confidential.

I\_\_\_\_\_, understand all that is involved in this study. I hereby agree to take part in this research project.

Signed:

Date:

WITNESS SINATURE:

Signed:

Date:



## OLLSCOIL LUIMNIGH Parental Information Sheet

Study Title: Post Primary Girls Physical Activity Research Programme

#### What is the project about?

Regular physical activity during childhood and adolescents is associated with improvements in health benefits, such as reducing risk of obesity, improving the fitness of the heart and lungs, along with developing fitness and strength. Regular physical activity is described as accumulating moderate intensity activity, such as brisk walking, for 30 minutes a day on 5 days of the week.

Female adolescents are recognised as being particularly at risk of being inactive, with large numbers not participating in enough activity to improve health and fitness. Previous studies have identified that non-team, non-competative activities are a more attractive and successful method of improving activity levels. This study will introduce a physical activity intervention lasting 6-8 weeks to a group of female adolescents in 4 schools across the Limerick county region. The school your daughter attends has been selected as a control school. She will have all the same measurements taken, but will not be involved in the intervention. Her involvement in the study is extremely important, as results from her school will be used as a comparison against results from the intervention schools. This study will be very beneficial to your daughter, as she will have the opportunity to get her level of Physical Activity accurately measured using the accelerometers. Your daughter will also learn about her fitness and body composition levels in a private and confidential manner.

#### What will my daughter have to do?

All testing will take place in the Physical Education and Sports Science (PESS) building in the University of Limerick. The first type of testing she will take part in is the physiological testing. This will take place in a private space with a female research student, and another female observer will be present at all times. All measurements obtained will be private and confidential, and at <u>no time</u> will your daughter's information be available to anyone, <u>except the researcher's</u> names on the last page of this document.

- Height and weight will be measured in stocking feet, light shorts and t-shirt by a female researcher and this will be used to calculate BMI (Body Mass Index).
- Measurement of waist and hip circumference will then be taken.
- 4 site skinfold test will be administered. The four sites measured are mid-bicep (front of upper arm), mid-triceps (back of the upper arm), sub-scapular (under the shoulder blade) and the iliac crest (side of the body, near the hip bone). All measurements will be taken using a skinfold calliper, and will be performed by an experienced female researcher.
- Blood pressure will be measured using a standard procedure. The appropriately sized cuff will be placed around the upper arm and inflated automatically. The pressure will be slowly reduced, and both systolic and diastolic blood pressures will be recorded.
- Your daughter will participate in a treadmill test. This measures the difference in breathing as someone exercises, and examines how fit your daughters heart and lungs are. This will involve walking and jogging on a treadmill while breathing through a piece of equipment called a gas analyser. She will breathe through a snorkel type mouthpiece while wearing a nose clip, to ensure all breathing is collected by the analyser. Your daughter will be given a familiarization period, to get used to walking and jogging on the treadmill. She will also be asked to wear a heart rate monitor, which will record her heart rate throughout the exercise. This monitor is attached around the chest under clothing. It emits a signal which is read from a watch worn on the wrist. Once the test begins, your daughter will complete several stages or speeds lasting 3 minutes. The test begins at a slow walk (5 kilometres per hour). After three minutes, heart rate will be recorded and your daughter will be asked to point at a scale identifying how hard she believes she is working. At the end of every three minutes, the speed will be increased by 1 kilometre per hour. The treadmill test will finish when your daughter reaches 85%

of her maximum heart rate, when she points to 17 on the Borg RPE (rate of perceived exertion) scale, or when she decides to stop herself.

- Physical Activity levels will be calculated for a week using accelerometry, which is carried out using an accelerometer called an activPAL. The activPAL is a tiny, lightweight box that measures the amount and direction of all the activity that one does. It can tell how long one spends sitting, standing, lying and walking, and how fast one walks. The accelerometer is worn on the mid part of the front of the right thigh, and is held onto the skin with a hypoallergenic stickie. This stickie is both hair and skin friendly. This form of testing is neither invasive, nor will it impede your daughters everyday life. The activPAL only needs to be removed before bathing and any water based activities. The activPAL will be worn for 7 days, including 2 weekend days so we can get an overall picture of your daughter's physical activity patterns. Each participant is free to not wear the accelerometer if they feel uncomfortable in any way and can return it before the 7 days are up. This will not affect their future treatment or participation.
- On return of the accelerometers, your daughter will complete a physical activity and lifestyle questionnaire. This questionnaire will inform us of why she does/does not choose to take part in physical activity, and of her own perceived level of physical activity

Eight weeks after your daughter completes the accelerometer data, she, along with 19 of her classmates, will be asked to return to the University of Limerick to complete the tests listed above for a second time. All transport to and from the university will be provided, and all groups will be supervised at all times.

#### What are the benefits for my daughter?

Your daughter will receive a copy of your results, which may inform her of her body composition and cardiovascular fitness. She will not be directly informed of the results of this research study. However, it is hoped that the results of this study will inform us of the impact an individual based intervention has on adolescent females in the Limerick region.

#### What are the risks?

The UL Research Ethics Committee has assessed all risks associated with each testing procedure. The treadmill test carries all risks associated with participating in any exercise. An example of these risks may be soreness, stiffness or possible injury to muscles. All researchers will be appropriately qualified and have First Aid certificates in case of any emergencies.

#### What if my daughter does not want to take part?

We will be delighted with her taking part in the study. However if she does not wish to take part it is entirely up to her. Your daughter can withdraw from the study at any point, and we will thank her for her time and help.

#### Who else is taking part in this Research Project?

It is hoped that 19 other participants from your daughter's school will take part in this study, and that a total of 79 other students from schools around Limerick County will also participate.

#### What happens to the information that is gathered?

The information is completely confidential and will only be used for the purpose of this project. Only the people involved in this project will have access to the questionnaires and results of the testing. Once the project is complete, all personal information will be destroyed. Remaining data will be retained for the recommended period of 7-10 years.

#### What if something goes wrong?

In the event of an accident, the department first aid officer will be available to assist in any way possible. In the case of an emergency, the department protocol, in place to deal with such incidents, will be set in motion.

### Who can I contact for additional information?

If you would like to know more about this study, please do not hesitate to contact any of the investigators listed below.

Name	Title	Contact No.	E-mail
Mr. Kieran Dowd	Postgraduate Student	087-91XXXX	Kieran.dowd@ul.ie
Miss. Sarah Breen	Postgraduate Student	061-23XXXX	sarahbreen@ul.ie
Mr. Phelim Macken.	Coordinator, County Limerick Sports Partnership	087-20XXXX	pmacken@limerickcoco.ie
Prof. Alan Donnelly	Professor	061-20XXXX	Alan.donnelly@ul.ie

Also if you have concerns about this study and wish to contact someone independent, you may contact:

The Chairman of the University of Limerick Research Ethics Committee.

C/o Vice President Academic and Registrar's Office University of Limerick Limerick Tel: (061) 20xxxx



## OLLSCOIL LUIMNIGH Parent/Carer Informed Consent

I have read and understand the parental information sheet and have had due time to consider it. I now fully understand the purposes of the project: *Post Primary Girls Physical Activity Research Programme*, and what the data will be used for. I am fully aware of the procedures that will involve my child, and I am aware that she can withdraw from this project at any time, and that her participation is completely voluntary. She will not have to give any reasons as to why she leaves the study if she so chooses to. I am aware that her results will be completely confidential.

I\_\_\_\_\_, understand all that is involved in this study. I hereby agree to allow my child to take part in this research project.

Signed:

Relationship with the participant:

Date:



## OLLSCOIL LUIMNIGH Participant Information Sheet

Study Title: Post Primary Girls Physical Activity Research Programme

### What is the project about?

Regular physical activity during childhood and adolescents is associated with improvements in health benefits, such as reducing risk of obesity, improving the fitness of the heart and lungs, along with developing fitness and strengyh. Regular physical activity is described as accumulating moderate intensity activity, such as brisk walking, for 30 minutes a day on 5 days of the week.

Female adolescents are recognised as being particularly at risk of being inactive, with large numbers not participating in enough activity to improve health and fitness. Previous studies have identified that non-team, non-competative activities are a more attractive and successful method of improving activity levels. This study will introduce a physical activity intervention lasting 6-8 weeks to a group of female adolescents in 4 schools across the Limerick county region. Your school has been selected as a control school. You will have all the same measurements taken, but will not be involved in the intervention. Your involvement in the study is extremely important, as results from your school will be used as a comparison against results from the intervention schools. This study will be very beneficial to you, as you will have the opportunity to get your level of Physical Activity accurately measured using the accelerometers. You will also learn about you own fitness and body composition levels in a private and confidential manner.

#### What will you have to do?

All testing will take place in the Physical Education and Sports Science building in the University of Limerick. The first type of testing you will take part in is the physiological testing. This will take place in a private space with a female research student, and another female observer will be present at all times. All measurements obtained will be private and confidential, and at <u>no time</u> will your information be available to anyone, <u>except the researcher's</u> names on the last page of this document.

- Height and weight will be measured in stocking feet, light shorts and t-shirt by the female researcher and this will be used to calculate BMI (Body Mass Index).
- Measurement of your waist and hip circumference will then be taken.
- 4 site skinfold test will be administered. The four sites measured are mid-bicep (front of upper arm), mid-triceps (back of the upper arm), sub-scapular (under the shoulder blade) and the iliac crest (side of the body, near the hip bone). All measurements will be taken using a skinfold calliper, and will be performed by an experienced female researcher.
- Blood pressure will be measured using a standard procedure. The appropriately sized cuff will be placed around the upper arm and inflated automatically. The pressure will be slowly reduced, and both systolic and diastolic blood pressures will be recorded.
- You will participate in a treadmill test. This measures the differences in your breathing as you exercise, and will examine how fit your heart and lungs are. This will involve walking and jogging on a treadmill while breathing through a piece of equipment called a gas analyser. You will breathe through a snorkel type mouthpiece while wearing a nose clip, to ensure all breathing is collected by the analyser. You will be given a familiarization period, to get used to walking and jogging on the treadmill. You will also be asked to wear a heart rate monitor, which will record your heart rate throughout the exercise. This monitor is attached around your chest under whatever clothing you wear for exercise. It emits a signal which is read from a watch worn on your wrist. Once the test begins, you will complete several stages or speeds lasting 3 minutes. The test begins at a slow walk (5 kilometres per hour). After three minutes, your heart rate will be recorded and you will be asked to point at a scale identifying how hard you believe you are working. At the end of every three minutes, the speed will be increased by 1 kilometre per hour. The treadmill test will finish when you reach 85% of your maximum heart

rate, when you point to 17 on the Borg RPE (rate of perceived exertion) scale, or when you decide to stop yourself.

- Physical Activity levels will be calculated for a week using accelerometry, which is carried out using an accelerometer called an activPAL. The activPAL is a tiny, lightweight box that measures the amount and direction of all the activity that one does. It can tell how long one spends sitting, standing, lying and walking, and how fast one walks. The accelerometer is worn on the mid part of the front of the right thigh, and is held onto the skin with a hypoallergenic stickie. <u>This stickie is both hair and skin friendly</u>. This form of testing is neither invasive, nor will it impede your everyday life. The activPAL only needs to be removed before bathing and any water based activities. The activPAL will be worn for 7 days, including 2 weekend days so we can get any overall picture of your physical activity patterns. You are free to not wear the accelerometer if you feel uncomfortable in any way and can return it before the 7 days are up. This will not affect your future treatment or participation.
- On return of the accelerometers, you will complete a physical activity and lifestyle questionnaire. This questionnaire will inform us of why you do/do not choose to take part in physical activity, and of your own perceived level of physical activity

Eight weeks after you complete the accelerometer data, you will be asked to return to the University of Limerick to complete the tests listed above for a second time. All transport to and from the university will be provided, and all groups will be supervised.

#### What are the benefits to you?

You will receive a copy of your results, which can inform you of your body composition and cardiovascular fitness. You will not be directly informed of the results of this research study. However, it is hoped that the results of this research will inform us of the impact a physical activity intervention, directed at improving participation in individual activities, has on a group of female adolescents.

#### What are the risks?

The UL Research Ethics Committee has assessed all risks associated with each testing procedure. Normal risk of muscle soreness and stiffness associated with any form of physical activity may be experienced by completing the Vo2 sub-maximal test.

Appendix C: Information Sheet and Consent forms Study 3

#### What if I do not want to take part?

We will be delighted with you taking part in the study. However if you do not wish to take part it is entirely up to you. You can withdraw from the study at any time and we will thank you for your time and help.

#### Who else is taking part in this Research Project?

It is hoped that 19 other participants from your school will take part in this study, and that a total of 79 other students from schools around Limerick County will also participate.

#### What happens to the information that is gathered?

The information is completely confidential and will only be used for the purpose of this project. Only the people involved in this project will have access to the questionnaires and results of the testing. Once the project is complete, all personal information will be destroyed. Remaining data will be retained for the recommended period of 7-10 years.

### What if something goes wrong?

In the event of an accident, the department first aid officer will be available to assist in any way possible. In the case of an emergency, the department protocol, in place to deal with such incidents, will be set in motion. Appendix C: Information Sheet and Consent forms Study 3

### Who can I contact for additional information?

If you would like to know more about this study, please do not hesitate to contact any of the investigators listed below.

Name	Title	Contact No.	E-mail
Mr. Kieran Dowd	Postgraduate Student	087-91XXXX	kieran.dowd@ul.ie
Mr. Phelim Macken.	Coordinator, County Limerick Sports Partnership	087-20XXXX	pmacken@limerickcoco.ie
Miss. Sarah Breen	Postgraduate Student	061-2XXXX	sarahbreen@ul.ie
Prof. Alan Donnelly	Professor	061-20XXXX	alan.donnelly@ul.ie

Also if you have concerns about this study and wish to contact someone independent, you may contact:

The Chairman of the University of Limerick Research Ethics Committee.

C/o Vice President Academic and Registrar's Office University of Limerick Limerick Tel: (061) 20xxxx



## OLLSCOIL LUIMNIGH

## Participant Consent

I have read and understand the subject information sheet and have had due time to consider it. I now fully understand what this project: *Post Primary Girls Physical Activity Research Programme* is all about and what the data will be used for. I am fully aware of the procedures that are involved in this project. I am aware that I can pull out of this project at any time, and that my participation is completely voluntary. I do not have to give any reasons as to why I leave the study if I so choose to. I am aware that my results will be completely confidential.

I	, understand all that is involve	d in this study. I hereby agree
to take part in t	his research project.	
Signed:		
Date:		-
<u>WITNESS SIN</u>	ATURE:	
Signed:		-
Date:		-

Prof. Alan Donnelly, PESS Dept., University of Limerick, Castletroy, Co. Limerick.

Dear Principal,

I am writing to you today on behalf of our research team at the University of Limerick, Dublin City University, the Mid-Western Regional Hospital and both the Limerick City and County Limerick Local Sports Partnerships. We are currently planning a large-scale physical activity research project in the Limerick region. The project title is *"Sedentary behaviours, physical activity and health in adolescent females"*. We have recently selected four schools, including your school, which we would hope to participate in this study.

The aim of this study is to investigate and examine physical activity levels and sedentary behaviours in adolescent females aged between 12 and 18 years. We hope to examine how health indices vary across adolescents, and how these behaviours correspond with the measured health indices.

It is expected that a total of 120 female adolescents will participate in this study, with potentially 30 students from your school participating. I have attached a project plan of this study which will explain the full involvement required of you and your students, and how it will affect your school throughout the next 12 months.

I would ask if you could read the attached information sheet. If you have any questions regarding this research study, please feel free to contact my Ph.D. research student, who is in charge of study planning and preparation, Mr Kieran Dowd on (061) 234781 or via email at <u>kieran.dowd@ul.ie</u>. Mr Kieran Dowd will contact you within the next couple of days to arrange a meeting, where you can discuss and examine the advantages and disadvantages of this research project. We look forward to speaking with you, and hope to assist you with any queries or questions you may have.

Yours sincerely,

Prof. Alan Donnelly, Email: <u>alan.donnelly@ul.ie</u> Phone: (061) 20 XXXX

252



# OLLSCOIL LUIMNIGH Principal's Information Sheet

Study Title: Sedentary behaviours, physical activity and health in adolescent females

**Background:** Regular physical activity during childhood and adolescents is associated with health benefits such as reducing the risk of obesity, improving the fitness of the heart and lungs, along with developing fitness and strength. Regular physical activity for adolescents is described as taking part in moderately intense activity, such as brisk walking, for 60 minutes a day on all days of the week. Female adolescents are recognised as being particularly at risk of being inactive, with large numbers not participating in enough activity to improve health and fitness. This study aims to examine the levels of physical activity in a large group of female adolescents in 4 schools, including your school, across the Limerick region. We also wish to investigate how these levels of physical activity affect different aspects of health, such as cholesterol, blood pressure, body composition and fitness. The study will identify levels in physical activity at three different time points throughout one full year, and will also examine the different aspects of health at each of these time points. This study also hopes to examine why each person participates in different levels of physical activity, and how we may best improve these levels of participation.

What is required of you, the principal: As principal of the school, we ask for your consent for the proposed research to take place in your school. Initially, we ask for your consent to obtain a copy of the names and dates of birth of <u>all female students that are aged between 12 and 18 years</u> in your school. This information is required for one reason only, to randomly select our participants. Once this list of names is obtained, 30 students will be randomly selected to participate in the study. <u>The names and dates of birth information will be destroyed once potential participants have been identified</u>. We also ask

you to volunteer a member of the teaching staff to be our contact in the school. This teacher will be asked to assist in arranging meetings and organise test days with the students. Finally, we ask for your support while completing the study in your school. All testing protocol and procedures involved in the research will be explained in detail below.

**Parental/Participant consent:** We will meet with the selected students at the earliest possible convenience. We will discuss the study outline, and explain what is involved in the research. Students will then be provided with parental and participant information sheets and consent forms, which are required to be completed prior to the study beginning. Both of these forms include PAR-Qs (Physical Activity Readiness Questionnaires). Upon return of the completed forms, we will examine the results of the PAR-Q's. If discrepancies are found between the parental and participant PAR-Q's, we will contact the participants GP, to ensure that it is safe for the selected students to take part in the study. We have requested for parents to grant us permission to make this contact with the GP's in the parental information sheet, and all information will be dealt with privately and confidentially. These precautions are necessary to conform to ethical best practice. We would like to emphasise that this study is not extremely dangerous and does not require additional precautions compared to other research of this kind.

Four test periods are involved in this study, with each test period explained below.

#### Test Period 1 (to take place in September/October 2011):

#### Test Day 1

9:00-2:00 Girls that have volunteered to participate in the research will be called to the school gym in pairs. Each participant will be asked to rest on a physiotherapy plinth for 10-15 minutes. Once rested, her blood pressure, pulse wave velocity and carotid intimamedia thickness will be obtained. Pulse wave velocity is measured by simply attaching a clip onto the index finger. The clip is much less painless than attaching a cloths peg onto the top of your finger. Carotid intima-media thickness measures the thickness of a large artery in the neck. This is measured using an ultrasound machine. None of these measures are invasive, and all measurements can be obtained without causing any pain, discomfort or embarrassment. These measures will take a total of 20 minutes to complete on each participant.

#### Test Day 2 (will take place the day after test day 1)

9:00-11:30 Students will be transported from your school to the Mid-Western Regional Hospital, Limerick, where a qualified medical professional will take a blood sample from the student's arm, as may happen at a regular GP visit. The student's will then be briefly asked some questions by the doctor on their current stage of development. This will involve asking the girls if they have had their first menstrual period, showing them standard female puberty diagrams and asking them to self-select their pubertal status. This will be performed by a Specialist Registrar in Paediatrics, working in the Mid-Western Regional Hospital, Limerick. Permission from their parents will be requested before the girls are shown pictures or asked pubertal questions. All of this testing will be done in private, and will be dealt with the greatest confidentiality.

*11:30-1:00* Students will be provided with breakfast/lunch and will be transported back to your school, where they will complete a physical activity and lifestyle questionnaire.

1:00-4:00 Back in the school sports hall, students will then complete some physiological tests, which will include the measurement of height and weight, waist and hip circumference, skinfold measurements and cardiovascular fitness. All of these tests are easy to complete, are not painful and should not cause any embarrassment.

Finally, the students will be provided with an activPAL accelerometer, which will be used to record physical activity over a one week period. The activPAL is a small, lightweight accelerometer, which is used to examine physical activity levels and sedentary behaviour. The accelerometer is a non-invasive, comfortable device which can be worn for long periods of time unnoticed. The activPAL is shown attached to a leg in the picture below.



#### Test Period 2 (March/April 2012): One Day Only

*9:00-12:00* Participants will complete physiological tests: Height and Weight, Waist and Hip circumference, Skinfold measurements, Pulse Wave Velocity, Blood Pressure and Cardiovascular fitness. Participants will once again be provided with an activPAL accelerometer.

12:00-01:00 Participants will be asked to complete a Physical Activity and Lifestyle Questionnaire, as they did on the first test day.

All testing in test period 2 and 3 will take place in the school, as no trip to the hospital will be required (no blood sampling).

#### Test Period 3 (September/October 2012):

All protocol and procedures in test period 3 will be the exact same as test period 1 (two days testing).

There may be other meetings required with the participants when organising these test days, but disruptions to class time and activities within the school will be kept to a minimum at all times. We also hope to provide the participating students with a voucher (approximately  $\pounds 20$ - $\pounds 40$ ) to a local Limerick sports shop as an incentive for the safe return and care of the activPAL devices. This is consistent with the ethos of the study, and may promote further physical activity participation of the participants.

**Benefits to Participants:** There are several benefits to your students for participating in this study. They will learn of the other types of work which are carried out in the hospital, particularly academic research and education. They will also get the experience of observing this work when they are in the hospital. They will learn about physiological testing which takes place in the Physical Education and Sport Sciences department in the University of Limerick. If your students wish, the results of their test will be returned to their GP, and they will be able to obtained and receive an explanation of the results on their next GP visit. Finally, we will inform your students of their levels of participation in physical activity and sedentary behaviour, which may inform them of whether they receive enough activity in their everyday life, or whether they need to participate in more activity.

**Risks:** The Mid-Western Regional Hospital Research Ethics Committee and the University of Limerick Research Ethics Committee have both assessed all risks associated with each testing procedure. Limited risks are associated with this type of research. All measurements are commonly used in both hospital and school settings, and all procedures will be completed by trained and experienced examiners. During the case of accident or emergency in the hospital, trained medical practitioners will be on hand to assist at all times. During the case of accident or emergency in the school first aid procedure will be implemented, and all precautions possible will be taken.

**Exclusion from participation:** This research will include healthy female adolescents between the ages of 12 and 18 years. No leaving certificate students will participate in this study. The only other exclusion criterion from this research is health. A number of measures have been included in this study plan to ensure that only health participants can take part in this study. These measures are outlined above.

**Confidentiality:** The information is completely confidential and will only be used for the purpose of this project. Only the people involved in this project will have access to the questionnaires and results of the testing. Once the project is complete, all personal information will be destroyed. Remaining data will be retained for the recommended period of 7-10 years.

**Voluntary Participation:** We will be delighted with your students taking part in this study. However if they do not wish to take part it is entirely up to them. If they do not wish to participate, they simply return this form <u>unsigned</u> and inform the cooperating teacher that they do not wish to take part in the study. If your students do return the form, and then realise that they do not wish to continue participating in the study, they can withdraw from the study at any time and we will thank them for their time and help. Who can I contact for additional information? If you would like to know more

about this study, please do not hesitate to contact any of the investigators listed below.

Name	Title	Contact No.	E-mail
Mr. Kieran Dowd	Postgraduate Student	087-91 XXXX	kieran.dowd@ul.ie
Prof. Alan Donnelly	Professor	061-20 XXXX	alan.donnelly@ul.ie

Also if you have concerns about this study and wish to contact someone independent, you may contact:

Chairman Education and Health Sciences Research Ethics Committee,

EHS Faculty office, University of Limerick, Co. Limerick, Tel: (061) 23 XXXX



## OLLSCOIL LUIMNIGH Principal Consent

I have read and understand the principal information sheet and have had due time to consider it. I now fully understand what this project: "Sedentary behaviours, physical activity and health in adolescent females" is all about and what the data will be used for. I am fully aware of the procedures that are involved in this project. I am aware that the students in my school can pull out of this project at any time, and that their participation is completely voluntary. The students do not have to give any reasons as to why they wish to leave the study if they so choose to. I am aware that the student's results will be completely confidential. I agree to provide the research team with the names and ages of all eligible students in my school, to allow them to select their sample population. I also give permission to the research group to use the facilities in the school on their test days, and to transport the students to the Mid-Western Regional Hospital, Limerick, on the first and fourth test period.

I \_\_\_\_\_\_ understand all that is involved in this study. I hereby agree to the students taking part in this research project, pending parental consent, participant consent and relevant GP information.

Signed:

Date: / /



OLLSCOIL LUIMNIGH Parental/Guardian Information Sheet:

Study Title: Sedentary behaviours, physical activity and health in adolescent females.

**Background and aims of the study:** Regular physical activity during childhood and adolescents is associated with health benefits such as reducing the risk of obesity, improving the fitness of the heart and lungs, along with developing fitness and strength. Regular physical activity for adolescents is described as taking part in moderately intense activity, such as brisk walking, for 60 minutes a day on all days of the week.

Female adolescents are recognised as being particularly at risk of being inactive, with large numbers not participating in enough activity to improve health and fitness. This study aims to examine the levels of physical activity achieved in a large group of female adolescents in 4 schools, including your daughters school, across the Limerick region. We wish to investigate how these levels of physical activity affect different aspects of health, such as cholesterol, blood pressure, body composition and fitness. The study will examine participation levels in physical activity at four different time points throughout one full year, and will also examine the different aspects of health at each of these time points. This study also hopes to highlight why young people participate in different levels of activity, and how these levels of physical activity and sedentary behaviour can be best improved.

#### Study outline:

This study will begin in November/December 2010. The study will involve approximately 30 female adolescents from four different schools across the Limerick city region. One of the schools selected to participate in this study was your daughter's school. It is expected that

between 100 and 120 girls in total will take part in the study. The study will involve 4 different test periods. Each of these test periods will be described in detail below.

### Test period 1 (September/October 2011):

#### Day 1:

9:00-2:00 Your daughter, accompanied by another girl who is participating in this research, will be asked to go to the school gym. Both girls will be asked to rest on a physiotherapy plinth for 10-15 minutes. Once rested, your daughter's blood pressure, pulse wave velocity and carotid intima-media thickness will be obtained. Pulse wave velocity is measured by simply attaching a clip onto the index finger. The clip is much less painless than attaching a cloths peg onto the top of your finger. Carotid intima-media thickness measures the thickness of a large artery in the neck. This is measured using an ultrasound machine. None of these measures are invasive, and all measurements can be obtained without causing any pain, discomfort or embarrassment. These measures will take 20 minutes to complete on each participant.

#### Day 2:

9:00-11:30 On the second test day, your daughter will not be allowed to eat on the morning of the testing, as she must be in a fasted state. Your daughter, along with the 29 other students from her school, will be transported to the Mid-Western Regional Hospital, Limerick. She will be brought to the paediatric day ward, where a doctor will take a small blood sample (10ml) from her arm, as might happen at a regular GP visit. Your daughter will then be briefly asked some questions by the doctor on her stage of development. This means that a paediatrician who works in the hospital will ask your daughter some questions about puberty. Firstly, they will ask if her if she has had her first menstrual period. If she has not yet had her first period, the doctor will then show her a picture scale (which is attached to this form), and ask her to choose which of the pubertal categories she believes she is in. There will be a female assistant (final year sport sciences students) in the room with your daughter at all times.

*11:30-1:00* Your daughter will then be provided with breakfast/lunch and transported back to her school, where she will complete a physical activity and lifestyle questionnaire. This is a simple questionnaire that will take approximately 30-40 minutes for her to complete.

1:00-3:30 Back in the school sports hall, she will then complete some physiological tests, which will include the measurement of her height and weight, waist and hip circumference, skinfold measurements and cardiovascular fitness. All of these tests are easy to complete, are not painful and should not cause her any embarrassment. All of these tests will be completed in an appropriately quiet and private area, and will be completed by a female examiner. At no time will she be on her own with any examiner, and all information obtained is private and confidential.

Finally, she will be provided with an ActivPAL accelerometer. The ActivPAL is a small, lightweight accelerometer (20 grams), which is used to assess physical activity levels and sedentary behaviour. It is affixed to your daughter's leg using adhesive strips, which attach the ActivPAL onto the midline of her right thigh (demonstrated in the picture below). A standard demonstration will be completed by one of the investigators on how to fix the ActivPAL to the leg. The ActivPAL is worn at all times for 7 days, except when bathing, swimming or in any contact with water. It is a non-invasive device, which can be comfortably worn throughout the day and night.



activPAL accelerometer

### Test Period 2 (to be held in March/April 2012): One Test Day only:

*9:00-11:30* Your daughter will complete each of the physiological tests that were completed on the first test day, including height and weight, waist and hip circumference, skinfold measurements, pulse wave velocity, blood pressure and cardiovascular fitness. She will once again be provided with an ActivPAL<sup>TM</sup> accelerometer. There will be no visit to the hospital, as blood sampling is not to be measured.

11:45-12:30 Your daughter will be asked to complete a Physical Activity and Lifestyle Questionnaire, as she did during the first period of testing.

#### Test Day 3 (to be held in September/October 2012):

All protocols and procedures on test day 3 will be the exact same as test period 1.

Upon full completion of the study, your daughter will receive a small valued sports voucher for a local Limerick sports store (approx.  $\notin$ 20- $\notin$ 40). This voucher will be provided to show our appreciation for the return of the ActivPAL devices in full working order. This has worked in previous studies of this kind, and promotes taking care and monitoring the devices when they are worn. The sports vouchers are also very much in accordance with the ethos of this study and may promote further participation in physical activity.

**Benefits:** There are several benefits to your daughter for participating in this study. She will learn of the other types of work which are carried out in the hospital, particularly academic research and education. She will also get the experience of observing this work while she is in the hospital. Your daughter will learn about physiological testing which takes place in the Physical Education and Sport Sciences department in the University of Limerick. If your daughter wishes, the results of her tests will be returned to her GP, and she will be able to obtained and receive an explanation of the results on her next GP visit. Finally, we will inform your daughter of her levels of participation in physical activity and sedentary behaviour, which may be able to inform her of whether she receives enough activity already in her everyday life, or whether she needs to participate in more activity.

**Risks:** The Mid-Western Regional Hospital Research Ethics Committee and the University of Limerick Research Ethics Committee have both assessed all risks associated with each testing procedure. Limited risks are associated with this type of research. All measurements are commonly used in both hospital and school settings, and all procedures will be completed by trained and experienced examiners. During the case of accident or emergency in the hospital, trained medical practitioners will be on hand to assist at all times. During the case of accident or emergency in the school first aid procedure will be implemented, and all precautions possible will be taken.

**Exclusion from participation:** This research will include healthy female adolescents between the ages of 12 and 18 years only. Once you and your daughter have consented to participate in the research, and have both consistently completed the Physical Activity Readiness Questionnaire (PAR-Q), your daughter will be able to participate in the study.

However, if there are discrepancies between the PAR-Q that you complete, and the one your daughter completes, then a letter must be sent to your daughters GP. This letter will inform your daughters GP that she will be participating in this study, and will ask him/her to contact the principal investigator if he/she feels your daughter should not take part in the study. With this, we will require the <u>name and address of your daughters GP</u> if you consent to participation in this study. Once each of these precautions has been completed, we can proceed with the research.

**Confidentiality:** The information is completely confidential and will only be used for the purpose of this project. Only the investigators involved in this project will have access to the questionnaires and results of the testing. Once the project is complete, all personal information will be destroyed. Remaining data will be retained for the recommended period of 7-10 years.

**Voluntary Participation:** We will be delighted with your daughter taking part in the study. However if she do not wish to take part in the project, it is entirely up to her. If your daughter does not wish to participate, she simply needs to return this form <u>unsigned</u> and inform the cooperating teacher that she does not wish to take part in the study. If your daughter does return the form, and then realises that she does not wish to continue participating in the study, she can withdraw from the study at any time and we will thank her for her time and help.

Who can I contact for additional information? If you would like to know more about this study, please do not hesitate to contact any of the investigators listed below.

Name	Title	Contact No.	E-mail
Mr Kieran Dowd	Postgraduate Student	087-91 XXXX	kieran.dowd@ul.ie
Prof. Alan Donnelly	Professor	061-20 XXXX	alan.donnelly@ul.ie

Also if you have concerns about this study and wish to contact someone independent, you may contact:

Chairman Education and Health Sciences Research Ethics Committee,

EHS Faculty office, University of Limerick,

Tel: (061) 23 XXXX



# OLLSCOIL LUIMNIGH Parent/Guardian Consent Form

I have read and understand the parental/carer information sheet and have had due time to consider it. I now fully understand what this project: "*Sedentary behaviours, physical activity and health in adolescent females*" is all about and what the data will be used for. I am fully aware of the procedures that are involved in this project. I am aware that my daughter can pull out of this project at any time, and that her participation is completely voluntary. I am aware that my daughter's results will be completely confidential. I am also providing the name and address of my daughters GP in accordance with this research. I am giving permission to the investigators involved to contact my daughters GP. I am also giving permission to my daughters GP to provide the research team with any reason why my daughter should not participate in this study.

I \_\_\_\_\_\_ understand all that is involved in this study. I hereby agree to my daughter taking part in this research project, pending relevant GP information.

Appendix D: Information Sheets and Consent Forms Study 4

Name of GP:			-
Address of GP:			-
			-
			-
Parent/ Carer Signatur	e:		
Relationship with part	icipant:	 	
Date: / /			



## OLLSCOIL LUIMNIGH

As your daughter is to be a subject in this project, would you please complete the following questionnaire about your daughter? Your cooperation in this is greatly appreciated.

## **PRE-TEST QUESTIONNAIRE**

Study Title: Sedentary behaviours, physical activity and health in adolescent females

Name: \_\_\_\_\_

Age: \_\_\_\_\_

School:

Date of Birth:	
Date of Diffin.	

	PAR-Q FORM Please mark YES or No to the following:	YES	NO
1)	Has your doctor ever said that your daughter has a heart condition and that		
	She should only do physical activity recommended by a doctor?		
2)	Does your daughter feel pain in her chest when she does physical activity?		
3)	In the past month, has your daughter had chest pain when she was not doing		
	physical activity?		
4)	Does your daughter lose her balance because of dizziness or does she ever		
	lose consciousness?		

5)	Does your daughter have a bone, joint or any other health problem that could be made worse by a change in her physical activity?
6)	Is your doctor currently prescribing drugs for your daughters blood pressure or heart condition?
7)	Do you know of any other reason why your daughter should not undergo physical activity? This might include severe asthma, diabetes, a recent sports
8)	Is your daughter pregnant now or has she given birth within the last 6 months?
9)	Has your daughter recently had surgery?
10)	Has your daughter any blood disorders or infectious diseases that may prevent her from providing blood for experimental procedures?

If you have marked YES to any of the above, please elaborate below:

- ✓ If you answered no honestly to <u>all</u> questions then you can be reasonably sure that you can take part in the physical activity requirement of the test procedures involved in this study.
- ✓ If you have answered yes to <u>any</u> of the questions above, please ensure that you have provided additional information in the area provided above.

I \_\_\_\_\_\_ declare that the above information for my daughter is correct at the time of completing this questionnaire

Date:\_\_/\_\_/\_\_\_\_

### **Scale of Pubertal Development:**

This scale will be shown to your daughter by a specialist registrar Paediatrician during her visit to the hospital. This information is important, as it is very likely that hormonal differences at different developmental stages can affect our examination of the blood sample.




# UNIVERSITY of LIMERICK

# OLLSCOIL LUIMNIGH

Subject Information Sheet

Study Title: Sedentary behaviours, physical activity and health in adolescent females.

**Background and aims of the study:** Regular physical activity during childhood and adolescents is associated with health benefits such as reducing the risk of obesity, improving the fitness of the heart and lungs, along with developing fitness and strength. Regular physical activity for adolescents is described as taking part in moderately intense activity, such as brisk walking, for 60 minutes a day on all days of the week.

Female adolescents are recognised as being particularly at risk of being inactive, with large numbers not participating in enough activity to improve health and fitness. This study aims to examine the levels of physical activity achieved in a large group of female adolescents in 4 schools, including your school, across the Limerick region. We wish to investigate how these levels of physical activity affect different aspects of health, such as cholesterol, blood pressure, body composition and fitness. The study will examine participation levels in physical activity at four different time points throughout one full year, and will also examine the different aspects of health at each of these time points. This study also hopes to highlight why each person participation.

#### Study outline:

This study will begin in November/December 2010. The study will involve approximately 30 female adolescents from four different schools across the Limerick city region. It is expected that between 100 and 120 participants in total will take part in the study. The study will involve 4 different test days. Each of these test days will be described in detail below.

#### Test period 1 (September/October 2011): Day 1:

9:00-2:00 In pairs, each of you will be asked to make their way to the school gym. You will be asked to rest on a physiotherapy plinth for 10-15 minutes. Once rested, your blood pressure, pulse wave velocity and carotid intima-media thickness will be obtained. Pulse wave velocity is measured by simply attaching a clip onto the index finger. The clip is much less painless than attaching a cloths peg onto the top of your finger. Carotid intima-media thickness measures the thickness of a large artery in the neck. This is measured by using an ultrasound machine. None of these measures are invasive, and all measurements can be obtained without causing any pain, discomfort or embarrassment.

#### Day 2:

9:00-11:30 You will not be allowed to eat on the morning of the testing, as you must be in a fasted state. You will be transported from your school to the Mid-Western Regional Hospital, Limerick. You will be brought to the paediatric day ward, where a doctor will take a small blood sample (10ml) from her arm. You will then be briefly asked some questions by the doctor on your stage of development. This means that a paediatrician who works in the hospital will ask you some questions about puberty. Firstly, they will ask you if you have had your first menstrual period. If you have not yet had your first period, the doctor will then show you a picture scale, and ask you to choose which of the pubertal categories you believe you are in. There will be a female assistant present at all times. However, your privacy will be maintained, and no person will have knowledge of anything you say during this testing period.

*11:30-1:00* You will then be provided with breakfast/lunch and transported back to the school, where you will complete a physical activity and lifestyle questionnaire. This is a simple questionnaire that will take approximately 30-40 minutes to complete.

1:00-3:30 Back in the school sports hall, you will then complete some physiological tests, which will include the measurement of height and weight, waist and hip circumference, skinfold measurements and cardiovascular fitness. All of these tests are easy to complete, are not painful and should not cause you any pain or embarrassment. All of these tests will be completed in an appropriately quiet and private area, and will be completed by a female examiner.

Finally, you will be provided with an ActivPAL accelerometer, which will be used to record physical activity over a 7 day period. The ActivPAL is a small, lightweight

accelerometer (20 grams), which is used to examine physical activity levels and sedentary behaviour. The activPAL is worn at all times for 7 days, except when bathing, swimming or in any contact with water. It is a non-invasive device, which can be comfortably worn throughout the day and night.



activPAL accelerometer

## Test Period 2 (to be held in March/April 2012): One Test Day only:

*9:00-11:30* You will complete each of the physiological tests that were completed on the first test day, including height and weight, waist and hip circumference, skinfold measurements, pulse wave velocity, blood pressure and cardiovascular fitness. You will once again be provided with an ActivPAL<sup>TM</sup> accelerometer. There will be no visit to the hospital as blood sampling will not be measured.

11:45-12:30 You will be asked to complete a Physical Activity and Lifestyle Questionnaire, as she did during the first period of testing.

#### Test Period 4 (to be held in September/October 2012):

All protocols and procedures on test day 3 will be the exact same as test period 1.

Upon full completion of the study, you will receive a small valued sports voucher for a local Limerick sports store. This voucher will be provided to show our appreciation for the return of the ActivPAL devices in full working order.

**Benefits:** There are several benefits to you for participating in this study. You will learn of the other types of work which are carried out in the hospital, particularly academic research and education. You will also get the experience of observing this work when you are in the hospital. You will also learn about all other physiological testing which takes place in the

Appendix D: Information Sheets and Consent Forms Study 4

Physical Education and Sport Sciences department in the University of Limerick. If you wish, the results of your testing will be returned to your general practitioner (GP) and you will be able to obtained and receive an explanation of the results on your next GP visit. Finally, we will inform you of your levels of participation in physical activity and sedentary behaviour.

**Risks:** The Mid-Western Regional Hospital Research Ethics Committee and the University of Limerick Research Ethics Committee have assessed risks associated with each testing procedure. Limited risks are associated with this type of research. All measurements are commonly used in both hospital and school settings, and all procedures will be completed by trained and experienced examiners. During the case of any accident or emergency in the hospital, trained medical practitioners will be on hand to assist at all times. During the case of accident or emergency in the school, the school first aid procedure will be implemented, and all precautions possible will be taken.

**Exclusion from participation:** This research will include healthy female adolescents between the ages of 12 and 18 years. Both you and your parent will be asked to consent to participation in the research, and asked to complete a short Physical Activity Readiness Questionnaire (PAR-Q). Once each of these precautions has been completed, we can proceed with the research.

**Confidentiality:** The information is completely confidential and will only be used for the purpose of this project. Only the people involved in this project will have access to the questionnaires and results of the testing. Once the project is complete, all personal information will be destroyed. Remaining data will be retained for the recommended period of 7-10 years.

**Voluntary Participation:** We will be delighted with you taking part in the study. However if you do not wish to take part it is entirely up to you. If you do not wish to participate, simply return this form <u>unsigned</u> and inform your teacher that you do not wish to take part in the study. If you do return the form, and then realise that you do not wish to continue participating in the study, you can withdraw from the study at any time and we will thank you for your time and help.

Who can I contact for additional information? If you would like to know more

Name	Title	Contact No.	E-mail
Mr. Kieran Dowd	Postgraduate Student	087-91 XXXX	kieran.dowd@ul.ie
Prof. Alan Donnelly	Professor	061-20 XXXX	alan.donnelly@ul.ie

about this study, please do not hesitate to contact any of the investigators listed below.

Also if you have concerns about this study and wish to contact someone independent, you may contact:

Chairman Education and Health Sciences Research Ethics Committee,

EHS Faculty office, University of Limerick,

Co. Limerick,

Tel: (061) 23 XXXX



# UNIVERSITY of LIMERICK

# OLLSCOIL LUIMNIGH

Subject Consent Form

I have read and understand the subject information sheet and have had due time to consider it. I now fully understand what this project: *Sedentary behaviours, physical activity and health in adolescent females* is all about and what the data will be used for. I am fully aware of the procedures that are involved in this project. I am aware that I can pull out of this project at any time, and that my participation is completely voluntary. I do not have to give any reasons as to why I leave the study if I so choose to. I am aware that my results will be completely confidential.

I \_\_\_\_\_\_ understand all that is involved in this study. I hereby agree to take part in this research project.

Signed:

Date:

WITNESSES SIGNATURE:

Signed:

Date: \_/\_/\_\_\_

Appendix E: Information Sheet on the activPAL and wear protocol

# activPAL Accelerometer

# Instructions



# Information sheets and Protocol

### What Do I Do?

When you get the device

- 1. You will be given a full demonstration on how to use the activPAL and you will be given help with putting it on for the first time.
- 2. The accelerometer will be attached to your thigh using a stickie and a standard tube support bandage. Be sure that the little man on the front of the device is facing the right way up.
- 3. **Applying the Stickie:** Remove the plastic layer from the stickie and attach the open side firmly to the back of the activPAL. Then, remove the butterfly tabs and apply straight onto the skin, pressing firmly on the activPAL. Try not to use any moisturiser or other creams on the area where the activPAL will be attached, as this can make the *stickie* less sticky!



- 1. **Applying the Tube grip:** Pull the bandage on over your leg and up over your thigh. Envelope the activPAL so it is covered by the bandage at all sides.
- 2. When wearing the activPAL, you can partake in all activities (except waterbased activities) as normal, as it is secure and should not impede your normal activity. Please live your life as you always do.
- 3. Wear the activPAL all day and night, even when you take a nap and go to bed. Remove the activPAL when you are bathing, swimming or doing any other water-based activities. <u>Remember</u>; do not let the device get wet.
- 4. When bathing, remove the fabric, ActivPAL and stickie and leave in a flat, dry, safe place. Once you are dried off, reattach the ActivPAL using the same stickie. It is skin and hair friendly.
- 5. Record on the ActivPAL diary sheet the time and day when you took off and put back on the ActivPAL. Otherwise when you are not wearing the ActivPAL, we will think you were lying down!!

#### **Every Morning and Night:**

Leave the device on all day and all night except when bathing. Please reattach the device as soon as you can afterwards once you have dried off.

#### FAQs

#### Where should I apply the activPAL?

It is most comfortable to wear the ActivPAL on the mid line of the thigh, about one third of the way down between the hip and the knee. However, it will work correctly anywhere on the front of the thigh as indicated on the front panel of the device.

#### What if it is not secure?

Ensure that you have the correct side sticking to your leg. The butterfly tab side will stick to your leg while the clear plastic side will stick to the activPAL. The *stickie* will not work as well if any lotions (moisturiser or fake tan for example) are on the leg.

Often the *stickie* on its own is not secure enough especially during some more vigorous activities. Add the tube grip bandage over the activPAL making sure to create an envelope with the fabric all around the activPAL.

#### When should I remove the activPAL?

The activPAL can be worn comfortably all day and all night and should not impede your everyday activities. It must be removed before bathing and swimming or before any other activities which may result in the activPAL coming in contact with water.

#### How do I know the activPAL is working?

The activPAL is a continuous recorder so will never stop recording. But you can be sure by checking the little light in the front panel-it will flash green every six seconds. If it is not flashing green please call us on the number below!

#### Are there any safety issues or warnings?

Ensure to dry off thoroughly after bathing before reapplying the activPAL. If it drops or falls from the leg don't worry-just check the front panel to see if the green light flashes every 6 seconds. Please do not leave the activPAL lying around in school or public toilets or showers in case it gets lost or stolen. While the activPAL is of no value to you, it is precious to us so please take care of the device at all times!

#### What if I'm in trouble or have a question?

At any stage, day or night if you are having a problem with the device or have a question please give me a call. No matter how small or silly you think the question is, it could turn out to be very important later on.

Name: Kieran Dowd

Email Address: <u>kieran.dowd@ul.ie</u> Home Phone Number: 087-91XXXXX Office Phone Number: 061-23XXXX Deirdre Harrington was the principal author of the paper appended below. I, Kieran Dowd, contributed to study design, aided in data collection, contributed to the writing of each section and proof read the final manuscript for submission. Full text included on additional CD.

#### RESEARCH

Open Access

# Cross-Sectional analysis of levels and patterns of objectively measured sedentary time in adolescent females

Deirdre M Harrington<sup>12\*</sup>, Kieran P Dowd<sup>2</sup>, Alan K Bourke<sup>3</sup> and Alan E Donnelly<sup>2</sup>

#### Abstract

Background: Adolescent females have been highlighted as a particularly sedentary population and the possible negative effects of a sedentary lifestyle are being uncovered. However, much of the past sedentary research is based on self-report or uses indirect methods to quantity sedentary time. Total time spent sedentary and the possible intricate sedentary patterns of adolescent females have not been described using objective and direct measure of body inclination. The objectives of this article are to examine the sedentary levels and patterns of a group of adolescent females using the ActivPAL<sup>TM</sup> and to highlight possible differences in sedentary levels and patterns across the week and within the school day. A full methodological description of how the data was analyzed is also presented.

**Methods:** One hundred and eleven adolescent females, age 15-18 yrs, were recruited from urban and rural areas in the Republic of Ireland. Participants wore an ActivPAL physical activity monitor for a 7.5 day period. The ActivPAL directly reports total time spent sitting/lying every 15 seconds and accumulation (frequency and duration) of sedentary activity was examined using a customized MATLAB<sup>40</sup> computer software programme.

**Results:** While no significant difference was found in the total time spent sitting/lying over the full 24 hour day between weekday and weekend day (18.8 vs. 18.9 hours; p = .911), significantly more sedentary bouts of 1 to 5 minutes and 21 to 40 minutes in duration were accumulated on weekdays compared to weekend days (p < .001). The mean length of each sedentary bout was also longer (9.8 vs. 8.8 minutes; p < .001). When school hours (9 am-3 pm) and after school hours (4 pm-10 pm) were compared, there was no difference in total time spent sedentary (3.9 hours; p = .796) but the pattern of accumulation of the sedentary time differed. There were a greater number of bouts of > 20 minutes duration during school hours than after school hours (4.7 vs. 3.5 bouts; p < .001) while after school time consisted of shorter bouts < 20 minutes.

**Conclusions:** School is highlighted as a particularly sedentary setting for adolescent females. Interventions to decrease sedentary time at school and the use of wearable devices which distinguish posture should be encouraged when examining sedentary patterns and behaviors in this population.

Keywords: Sedentary, adolescent females, ActivPAL, sitting, school, methodology

Deirdre Harrington was the principal author of the paper appended below. I, Kieran Dowd, contributed to study design, aided in data collection, contributed to the writing of each section and proof read the final manuscript for submission. Full text included on additional CD.

# A Steps/Minute Value for Moderate Intensity Physical Activity in Adolescent Females

Deirdre M. Harrington Pennington Biomedical Research Center

> Kieran P. Dowd University of Limerick

#### Catrine Tudor-Locke Pennington Biomedical Research Center

Alan E. Donnelly University of Limerick

The number of steps/minute (i.e., cadence) that equates to moderate intensity in adolescents is not known. To that end, 31 adolescent females walked on a treadmill at 5 different speeds while wearing an ActivPAL accelerometer and oxygen uptake was recorded by indirect calorimetry. The relationship between metabolic equivalents (METs) and cadence was explored using 3 different analytical approaches. Cadence was a significant predictor of METs (r = .70; p < .001). Moderate intensity (3 METs) corresponded to 94 or 114 steps/minute based on the mixed model and ROC analysis, respectively. These two values, and a practical value of 100 steps/minute, were cross-validated on an independent sample of 33 adolescent females during over-ground walking at 3 speeds. The sensitivity and specificity of each value correctly identifying 3 METs were 98.5% and 87.2% for 94 steps/minute, 2.9% and 98.8 for 114 steps/minute and 96.5% and 95.7% for 100 steps/minute. Compromising on a single cadence of 100 steps/minute would be a practical value that approximates moderate intensity in adolescent females and can be used for physical activity interpretation and promotion.

The objective assessment of physical activity using step counts is gaining momentum in both descriptive (27,28) and interventional (4) research. Total steps/day can be measured by pedometer or accelerometer and serve as a practical approximation of daily ambulatory physical activity volume. Physical activity Abstract submitted and presented at the American College of Sports Medicine, Baltimore 2010. Poster included on additional CD.

**Abstract Title:** The Examination of Accelerometer Output: Age/Non-age specific Thresholding and Bout Duration.

Kieran P. Dowd, Deirdre M. Harrington, Alan E. Donnelly.

#### University of Limerick, Limerick, Ireland.

The ability to relate physical activity (PA) to health improvements relies on accurate and precise measurements; often made using accelerometers. The most common method of analysing accelerometer output is classifying activity counts into physical activity levels (PAL's) or energy expenditure (using count-activity thresholds) and summing the results as total minutes per day at each level. Significant differences in PAL's have been noted when analysing accelerometry data using different thresholds. Purpose: To present differences in physical activity levels (PAL's) of female adolescents, using two different methods: total accumulated moderate to vigorous physical activity (MVPA) and bouts of MVPA  $\geq$  10 minutes in duration. Data were analysed using age specific (AS) and non-age specific (NAS) thresholds developed using the same methodologies. Results were then compared with the American College of Sports Medicine (ACSM 2007) PA guidelines (60 minutes of MVPA on at least 5 days of the week). Methods: Participants were randomly selected from 6 schools across the mid-western region of Ireland (N=51). Four to six days of PA was recorded using accelerometry (ActivPALTM, PAL technologies Ltd, Glasgow, Scotland) placed on the mid-thigh of the volunteer's right leg. Data was examined using both AS and NAS thresholds. The resulting data was then processed to identify total daily MVPA and bouts of MVPA  $\geq$  10 minutes in duration. **Results:** Mean AS and NAS total MVPA differed significantly (p < 0.001, t-test). Mean AS  $(20.2 \pm 14.5 \text{ min } d^{-1})$  and NAS  $(18.5 \pm 13.8 \text{ min } d^{-1})$  bouts of MVPA also differed significantly (p < 0.01, t-test). Participants did not achieve the recommended levels of PA using either AS (45.3  $\pm$  19.2 min d<sup>-1</sup>; mean  $\pm$  SD) or NAS (37.9  $\pm$  18.1 min d<sup>-1</sup>) thresholds. When the output was analysed using more stringent guidelines (bouts of MVPA  $\geq$  10 min), recommended PAL's were not achieved. Conclusion: These results illustrate that care must be taken when 1) selecting thresholds, as the lack of threshold specificity may significantly under/overestimate PAL's and 2) analysing duration using either total or bouts of MVPA, as significant differences of estimated PAL's can be observed.

Abstract submitted and presented at the American College of Sports Medicine, San Francisco 2012. Poster included on additional CD.

Abstract Title: Relation between Physical Activity, Sedentary Behaviour and Selected Cardiovascular Disease Risk Factors in Adolescent Females

<sup>1</sup>Kieran P. Dowd, <sup>2, 3</sup> Sarah M. Hughes, <sup>4</sup>Steven Gastinger, <sup>1</sup>Grainne Hayes, <sup>5</sup>Michael Harrison, <sup>6, 8</sup> Alan P. Macken, <sup>7, 8</sup>Clodagh S. O'Gorman, <sup>2, 3</sup>Niall M. Moyna FACSM, <sup>1</sup>Alan E. Donnelly.

<sup>1</sup>Centre of Physical Activity and Health Research, University of Limerick, Limerick, Ireland,<sup>2</sup>CLARITY: Centre for Sensor Web Technology and <sup>3</sup>The Centre for Preventative Medicine, Dublin City University, Dublin, Ireland, <sup>4</sup>Movement, Sport and Health Sciences, University of Rennes 2, France, <sup>5</sup>Centre for Health Behaviour Research, Waterford Institute of Technology, Waterford, Ireland, <sup>6</sup>National Children's Research Centre, Dublin, Ireland, <sup>7</sup>4i The Children's Ark, University Hospital, Limerick, Ireland, <sup>8</sup>4i Centre, Graduate Entry Medical School, University of Limerick, Limerick, Ireland.

Background: Cardiovascular disease risk factor (CVRF) levels for an individual tend to track over time in a given rank within the distribution of the population. Long term exposure to multiple CVRF and lifestyle behaviours confers a lifelong burden of CVD risk resulting in the development of early systemic atherosclerosis. Measurement of carotid intima-media thickness (CIMT) and large artery stiffness (LAS) are commonly used to determine the extensiveness and severity of asymptomatic disease and potential future risk. Purpose: To examine the relation between both physical activity (PA) and sedentary behaviour (SB) and i) selected CVRF and ii) subclinical disease in adolescent females. Methods: Data are presented as mean  $\pm$  standard deviation, except where otherwise indicated. Participants (n=35; age =  $14.4 \pm 1.1$  yrs.; BMI =  $22.6 \pm 2.9$  kg/m<sup>2</sup>; percentage body fat  $31.2 \pm 3.0$ ; mean arterial pressure (MAP) =  $85.0 \pm 8.3$  mmHg) were randomly selected from 4 high-schools in Ireland. Height, weight, percentage body fat, blood pressure, CIMT, and LAS data was obtained for each participant. SB and PA were recorded for 4-6 days using an ActivPAL<sup>TM</sup> physical activity monitor. The amount of time spent sedentary, standing, and in both light and moderate to vigorous PA (MVPA) was calculated. Sedentary data was also examined to identify sedentary bouts of specific duration. Results: The mean left and right-sided CIMT score was 0.47  $\pm$  0.0 cm and the mean LAS index was 5.20  $\pm$  0.58. Participants spent 19.42  $\pm$  0.94 h/d in sedentary activities (9.93  $\pm$  1.03 waking sedentary h/d), 3.08  $\pm$  0.74 h/d standing, 0.73  $\pm$  0.17 h/d in light activity and 0.78  $\pm$  0.26 h/d in MVPA. There was a significant inverse relation between light physical activity levels (independant of standing time) and mean arterial pressure

(MAP), (r = -0.5; p<0.05), BMI (r = -0.5; p<0.01) and percentage body fat (r = -0.4; p<0.05). There was no relation between other PA variables and CIMT or LAS. There was no significant relation between SB and CVRF, CIMT or LAS. **Conclusions:** Objectively measured light PA (independent of standing time) is correlated with MAP, BMI and percentage body fat, but not CIMT or LAS in this population of adolescent females. Although a large amount of time was spent sedentary (9.93 hr), SB were not related to CVRF, CIMT or LAS in this population of adolescent females.

Abstract submitted and presented at the American College of Sports Medicine, Denver 2011. Poster included on additional CD.

Abstract Title: Cadence (steps/minute) Patterns in High and Low Active Adoelscent Females.

<sup>1,2</sup> Deirdre M. Harrington, <sup>2</sup>Kieran P. Dowd, <sup>1</sup>Catrine Tudor-Locke, <sup>2</sup>Alan E. Donnelly

<sup>1</sup>Pennington Biomedical Research Unit, Baton Rouge, Louisiana, USA; <sup>2</sup>Centre of Physical Activity and Health Research, University of Limerick, Limerick, Ireland.

Steps/day has been interpreted as a measure of total daily physical activity, but it does not give information on physical activity intensity. However, minute-by-minute records of stepping data represent cadence (steps/minute), which in turn can be used to infer intensity. Adolescent females have been highlighted as a group who do not engage in adequate physical activity. Little is known about the cadence patterns of those who achieve high steps/day (and are identified as being more active), and those who take lower steps/day. Purpose To investigate differences in cadence patterns between adolescent females achieving high and low mean steps/day. Methods The activPAL was used to collect 7 days of free-living behavior from 52 adolescent girls who were part of a randomized sample. This analysis is based on 8 participants with the highest steps/day values (HIGH group) and the 8 with the lowest steps/day values (LOW group). No significant differences (p<.05) were found between groups' age, height, weight or BMI (mean age 15y 7mo, 1.7m, 67.8kg, 23.5 kg/m<sup>2</sup>). Data was categorized into incrementally increasing (by 10 steps/minute) cadence bands between 1-10 steps/minute up to 170-180 steps/minute. Independent sample t-tests with Bonferroni adjustment were used to compare the number of steps in each cadence band between groups. **Results** The LOW group took  $6696 \pm 344$  steps/day compared with the HIGH group who took 12959 ±1957 steps/day (p<.001). The two groups differed in the number of steps taken at cadences of 100-110 (p<.01) and 110-120 steps/minute (p<.001), with the HIGH group taking more steps in both of these cadence bands. There were no significant differences in the number of steps participants took at any cadence below 100 steps/minute (p=.059 to .939) or above 120 steps/minute (p=.122 to .427). No steps were recorded at between 1 and 20 steps/minute. **Conclusion** Although there are no adolescent-specific studies to assist with interpretation of these results, adult studies have consistently demonstrated that 100 steps/minute demarks the threshold for moderate intensity walking. This cadence represents an achievable target for low active adolescent females to aim for to increase physical activity intensity to match their peers. The fact that no steps were recorded between 1-20 steps/minute may be an instrument artifact.

Abstract submitted and presented at the International Conference on Ambulatory Monitoring and Physical Activity Monitoring, Glasgow 2011. Slides from oral presentation included on additional CD.

**Abstract Title**: Physical Activity and Sedentary Behaviours immediately before and after school.

<sup>1</sup> Kieran P. Dowd, <sup>2</sup>Deirdre M. Harrington, <sup>1</sup>Alan E. Donnelly

<sup>1</sup>Centre of Physical Activity and Health Research, University of Limerick, Limerick, Ireland; <sup>2</sup>Pennington Biomedical Research Unit, Baton Rouge, Louisiana, USA.

Introduction: It is widely acknowledged that the consequences of physical inactivity are a major public health concern, particularly in children and adolescents. Unfortunately, the contribution of commuting to and from school and extracurricular physical activity before and after school is often overlooked when examining levels of daily PA. The purpose of this abstract is to examine the contribution of objectively measured PA to total daily PA in a group of adolescent females during the hour before school (8-9am) and the hour after school (3-4pm). This abstract also describes the objectively measured sedentary pattern of this group of adolescent during this time period. Methods: Participants were randomly selected from 5 schools across the mid-western region of Ireland (N=44). Four to six days of PA was recorded using accelerometry (ActivPAL™, PAL technologies Ltd, Glasgow, Scotland). Data obtained on school days was examined from 8-9am and from 3-4pm. Data was examined for sedentary, light and moderate to vigorous PA (MVPA). Data was also examined for duration of sedentary bouts using a computer program. Results: Participants accumulated 25% of their daily recommended MVPA between 8-9 am and between 3-4 pm (all means ± SD's; 14.9 ± 8.6 mins). No significant differences existed between levels of MVPA achieved between 8-9am (7.72  $\pm$  7.97) and 3-4pm (7.19  $\pm$  6.99; p >.05). The analysis also showed that participants spent 38.2  $\pm$  13.3 mins in light activity, while 66.9  $\pm$  14.8 mins were spent in sedentary activities. Significantly more sedentary time was accumulated in the hour after school  $(36.47 \pm 12.82 \text{ mins})$  than the hour before school  $(30.06 \pm 14.5 \text{ mins}; p < .001, \text{Wilcoxon Sign})$ . Significantly more sedentary bouts less than 10 minutes in duration were also identified in the hour before school (2.4  $\pm$  2.0) compared to the hour after school (1.94  $\pm$  2.0; p=.034, Wilcoxon Sign). Discussion and Conclusion: It is indicated that participants achieved 25% of their daily MVPA immediately before and after school. However, it is also highlighted that considerable periods of time were also spent sedentary, with more time being spent in sedentary activities after school than before school. The time period immediately before and after school provides students with the

opportunity to be involved in active transport or in extracurricular PA. In particular, after school activities have the potential to significantly contribute to daily MVPA.

Presented in this appendix is an example of the mixed model analysis employed in chapter 3. Also included is the method used to examine the concordance correlation coefficient and the standard error of the estimate employed in Chapter 3.

### **Mixed Model Analysis**

	M	odel Dimen	sion <sup>b</sup>		
		Number of	Covariance	Number of	Subject
		Levels	Structure	Parameters	Variables
Fixed Effects	Intercept	1		1	
	ActivPALCounts15	1		1	
Random Effects	ActivPALCounts15 <sup>a</sup>	1	Variance	1	ID
			Components		
Residual				1	
Total		3		4	

a. As of version 11.5, the syntax rules for the RANDOM subcommand have changed. Your command syntax may yield results that differ from those produced by prior versions. If you are using version 11 syntax, please consult the current syntax reference guide for more information.b. Dependent Variable: Cosmed EE.

#### Information Criteria a

-2 Restricted Log Likelihood	300.023
Akaike's Information Criterion	304.023
(AIC)	
Hurvich and Tsai's Criterion	304.098
(AICC)	
Bozdogan's Criterion (CAIC)	312.211
Schwarz's Bayesian Criterion	310.211
(BIC)	

The information criteria are displayed in smaller-is-better forms.

a. Dependent Variable: Cosmed EE.

### Fixed Effects Type III Tests of Fixed Effects <sup>a</sup>

Source	Numerator df	Denominator df	F	Sig.
Intercept	1	149.093	351.917	.000
ActivPALCounts15	1	14.470	201.926	.000

a. Dependent Variable: Cosmed EE.

Parameter						95% Confi	dence Interval
						Lower	
	Estimate	Std. Error	df	t	Sig.	Bound	Upper Bound
Intercept	.971011	.051761	149.093	18.759	.000	.868731	1.073291
ActivPAL	.000677	4.765139E-5	14.470	14.210	.000	.000575	.000779
Counts15							

Estimates of Fixed Effects <sup>a</sup>

a. Dependent Variable: Cosmed EE.

#### **Covariance Parameters**

Estimates of Covariance Parameters <sup>a</sup>

Parameter	Estimate	Std. Error
Residual	.218037	.025261
ActivPALCounts15 [subject = Variance	3.300588E-8	1.266205E-8
ID]		

a. Dependent Variable: Cosmed EE.

#### **Estimated Marginal Means**

#### Grand Mean <sup>b</sup>

			95% Confidence Interval	
Mean	Std. Error	df	Lower Bound	Upper Bound
4.019ª	.214	14.416	3.561	4.477

a. Covariates appearing in the model are evaluated at the following values: ActivPAL Counts per 15 sec = 4501.4742.

b. Dependent Variable: Cosmed EE.

Appendix G: Example of Statistical Analysis Chapter 3

# Calculation of Concordance Correlation Coefficient (CCC)

Observed = Cosmed EE (Cosmed Energy Expenditure)

Fitted = FXPRED Values (Fixed Predicted Values)

- **1.**  $\Sigma$  (Observed Fitted)<sup>2</sup>
- **2.**  $\Sigma$  (Observed Observed Mean)<sup>2</sup>
- **3.**  $\Sigma$  (Fitted Fitted Mean)<sup>2</sup>
- **4.** N(Observed Mean Fitted Mean)<sup>2</sup>

 $CCC = 1 - \frac{\Sigma (Observed - Fitted)^2}{(\Sigma (Observed - Observed Mean)^2 + \Sigma (Fitted - Fitted Mean)^2 + N(Observed Mean - Fitted Mean)^2)}$ 

### Calculation of the Standard Error of the Estimate SE E

$$SEE = \sqrt{\frac{\Sigma \ (Observed - Fitted)^2}{N}}$$

# Table 3.2 Nonparametric Correlations Information

		Correlations		
			Cosmed EE	Predicted_V
Spearman's rho	Cosmed EE	Correlation Coefficient	1.000	.932**
		Sig. (2-tailed)		.000
		Ν	165	165
	Predicted_V	Correlation Coefficient	.932**	1.000
		Sig. (2-tailed)	.000	
		Ν	165	165

\*\*. Correlation is significant at the 0.01 level (2-tailed).

		Correlations		
			Cosmed EE	Predicted_V
Spearman's rho	Cosmed EE	Correlation Coefficient	1.000	.250
		Sig. (2-tailed)		.054
		Ν	60	60
	Predicted_V	Correlation Coefficient	.250	1.000
		Sig. (2-tailed)	.054	
		Ν	60	60

		Correlations		
			Cosmed EE	Predicted_V
Spearman's rho	Cosmed EE	Correlation Coefficient	1.000	.873**
		Sig. (2-tailed)		.000
		Ν	105	105
	Predicted_V	Correlation Coefficient	.873**	1.000
		Sig. (2-tailed)	.000	
		N	105	105

\*\*. Correlation is significant at the 0.01 level (2-tailed).

Appendix G: Example of Statistical Analysis Chapter 5 and Chapter 6

Example of the linear regression statistical analysis included in Chapter 5 and Chapter 6. All analysis is included with the attached CD.

	· ····································					
Model	Variables Entered	Variables Removed	Method			
1	PercLight_Log, Age <sup>b</sup>		Enter			
2	PercMVPA_Lo g <sup>b</sup>		Enter			

# Light (excl. Standing Time) $/\Sigma$ Skinfolds

Variables Entered/Removed<sup>a</sup>

a. Dependent Variable: SumSkinfold\_Log

b. All requested variables entered.

Model Summary						
Model	R	R Square	Adjusted R	Std. Error of the		
			Square	Estimate		
1	.234ª	.055	.045	.17123		
2	.243 <sup>b</sup>	.059	.044	.17129		

### Model Summary

a. Predictors: (Constant), PercLight\_Log, Age

b. Predictors: (Constant), PercLight\_Log, Age, PercMVPA\_Log

ANOVAª											
Model		Sum of Squares	df	Mean Square	F	Sig.					
	Regression	.324	2	.162	5.521	.005 <sup>b</sup>					
1	Residual	5.571	190	.029							
	Total	5.895	192								
	Regression	.349	3	.116	3.966	.009c					
2	Residual	5.546	189	.029							
	Total	5.895	192								

a. Dependent Variable: SumSkinfold\_Log

b. Predictors: (Constant), PercLight\_Log, Age

c. Predictors: (Constant), PercLight\_Log, Age, PercMVPA\_Log

Coefficients <sup>a</sup>										
Model		Unstandardize	ed Coefficients	Standardized Coefficients	t	Sig.				
		В	Std. Error	Beta						
	(Constant)	1.962	.229		8.561	.000				
1	Age	.001	.014	.007	.105	.917				
	PercLight_Log	356	.108	234	-3.297	.001				
	(Constant)	1.964	.229		8.565	.000				
2	Age	.003	.014	.014	.203	.839				
	PercLight_Log	331	.111	217	-2.980	.003				
	PercMVPA_Log	067	.072	068	930	.354				

a. Dependent Variable: SumSkinfold\_Log

Excluded Variables<sup>a</sup>

Model		Beta In	t	Sig.	Partial	Collinearity
					Correlation	Statistics
						Tolerance
1	PercMVPA_Log	068 <sup>b</sup>	930	.354	067	.937

a. Dependent Variable: SumSkinfold\_Log

b. Predictors in the Model: (Constant), PercLight\_Log, Age

Appendix G: Example of Statistical Analysis Chapter 5 and Chapter 6

Example of the binary logistic regression analysis and univariate analysis of variance included in Chapter 5. All additional analysis is included with the attached CD.

# Odds ratios for High Sum of Skinfolds across Light (excl. Standing Time) Quartiles:

			Score	df	Sig.
	-	Age	.250	1	.617
		LightQuart	7.851	3	.049
S: 0	Variables	LightQuart(1)	3.487	1	.062
Step 0		LightQuart(2)	1.714	1	.190
		LightQuart(3)	1.431	1	.232
Overall Statistics			7.981	4	.092

Variables not in the Equation

#### Block 1: Method = Enter

#### Variables in the Equation

		В	S.E.	Wald	df	Sig.	Exp(B)	95% (	C.I.for
								EXI	P(B)
								Lower	Upper
	Age	.068	.189	.130	1	.718	1.070	.740	1.549
	LightQuart			7.440	3	.059			
Step	LightQuart(1)	1.151	.507	5.145	1	.023	3.160	1.169	8.543
1 <sup>a</sup>	LightQuart(2)	1.026	.515	3.972	1	.046	2.791	1.017	7.656
	LightQuart(3)	.294	.551	.285	1	.594	1.342	.456	3.948
	Constant	-2.855	2.976	.921	1	.337	.058		

a. Variable(s) entered on step 1: Age, LightQuart.

	Variables in the Equation									
		В	S.E.	Wald	df	Sig.	Exp(B)	95% (	C.I.for	
			1 1					EX	P(B)	
								Lower	Upper	
	Age	.094	.191	.242	1	.623	1.099	.755	1.599	
	LightQuart	'	1 1	6.634	3	.085				
	LightQuart(1)	1.056	.517	4.170	1	.041	2.875	1.043	7.920	
	LightQuart(2)	.968	.519	3.474	1	.062	2.633	.951	7.287	
Stor	LightQuart(3)	.203	.566	.129	1	.720	1.225	.404	3.713	
Step 1 <sup>a</sup>	MVPAQuart	'	1	.903	3	.825				
I	MVPAQuart(1)	.443	.507	.764	1	.382	1.557	.577	4.204	
	MVPAQuart(2)	.413	.515	.643	1	.423	1.511	.551	4.149	
	MVPAQuart(3)	.346	.522	.440	1	.507	1.414	.508	3.935	
	Constant	- 3.511	3.07 5	1.304	1	.254	.030			

## Block 2: Method = Enter

a. Variable(s) entered on step 1: MVPAQuart.

# Univariate Analysis of Variance: Light Quartiles and Sum of Skinfolds

Detween bubjeets I actors							
-		Value Label	Ν				
	1.00	1st Quartile	49				
L'altroat	2.00	2nd Quartile	47				
LightQuart	3.00	3rd Quartile	49				
	4.00	4th Quartile	49				

**Between-Subjects Factors** 

### **Descriptive Statistics**

Dependent Variable: SumSkinfoldrisk								
LightQuart	Mean	Std. Deviation	Ν					
1st Quartile	1.3469	.48093	49					
2nd Quartile	1.3191	.47119	47					
3rd Quartile	1.1837	.39123	49					
4th Quartile	1.1429	.35355	49					
Total	1.2474	.43263	194					

### Tests of Between-Subjects Effects

Dependent Variable: SumSkinfoldrisk

Source	Type III Sum	df	Mean Square	F	Sig.
	of Squares				
Corrected Model	1.584 <sup>a</sup>	4	.396	2.166	.074
Intercept	52.135	1	52.135	285.277	.000
MVPAQuart	.122	1	.122	.665	.416
LightQuart	1.281	3	.427	2.336	.075
Error	34.540	189	.183		
Total	338.000	194			
Corrected Total	36.124	193			

a. R Squared = .044 (Adjusted R Squared = .024)

### Estimated Marginal Means Light Quart:

Estimates										
Dependent Variable: SumSkinfoldrisk										
LightQuart Mean Std. Error 95% Confidence Interval										
			Lower Bound Upper Boun							
1st Quartile	1.339 <sup>a</sup>	.062	1.218	1.461						
2nd Quartile	1.319 <sup>a</sup>	.062	1.196	1.442						
3rd Quartile	1.184 <sup>a</sup>	.061	1.064	1.305						
4th Quartile	$1.150^{a}$	.062	1.028	1.272						

a. Covariates appearing in the model are evaluated at the following values: MVPAQuart = 2.5103.

#### **Pairwise Comparisons**

Dependent Variable: SumSkinfoldrisk

(I)	(J) LightQuart	Mean	Std.	Sig. <sup>b</sup>	95% Co	onfidence
LightQuart		Difference	Error		Inter	val for
		(I-J)			Diffe	erence <sup>b</sup>
					Lower	Upper
					Bound	Bound
	2nd Quartile	.020	.088	.818	153	.193
1st Quartile	3rd Quartile	.155	.087	.076	016	.327
	4th Quartile	$.189^{*}$	.088	.033	.016	.363
	1st Quartile	020	.088	.818	193	.153
2nd Quartile	3rd Quartile	.135	.087	.124	037	.307
	4th Quartile	.169	.088	.055	004	.342
	1st Quartile	155	.087	.076	327	.016
3rd Quartile	2nd Quartile	135	.087	.124	307	.037
	4th Quartile	.034	.087	.694	137	.205
	1st Quartile	189*	.088	.033	363	016
4th Quartile	2nd Quartile	169	.088	.055	342	.004
	3rd Quartile	034	.087	.694	205	.137

Based on estimated marginal means

\*. The mean difference is significant at the .05 level.

b. Adjustment for multiple comparisons: Least Significant Difference (equivalent to no adjustments).

Dependent Variable: SumSkinfoldrisk									
	Sum of	df	Mean Square	F	Sig.				
	Squares								
Contrast	1.281	3	.427	2.336	.075				
Error	34.540	189	.183						

Univariate Tests pendent Variable: SumSkinfoldrisk

The F tests the effect of LightQuart. This test is based on the linearly independent pairwise comparisons among the estimated marginal means.

#### **Profile Plots**



Covariates appearing in the model are evaluated at the following values: MVPAQuart = 2.5103

	Model 1				Model 2			Model 3		
	β	р	Adj. R <sup>2</sup>	β	р	Adj. R <sup>2</sup>	β	р	Adj. R <sup>2</sup>	
BMI	20	.13	.03	21	.12	.01	-	-	-	
% BF	19	.14	.05	19	.15	.03	-	-	-	
ΣSF	19	.14	.04	19	.16	.02	-	-	-	
HDL-C	.09	.53	.00	.10	.48	01	.05	.72	.01	
LDL-C	16	.24	02	16	.27	04	10	.49	.04	
T Chol.	10	.46	.01	10	.52	01	07	52	02	
TG	13	.33	02	14	.32	03	08	.56	.05	
Insulin	24	.07	.01	26	.06	00	19	.15	.12	
Glucose	16	.23	.10	13	.33	.09	11	.43	.09	
HMW Apo	.05	.71	05	.04	.79	06	.04	.78	08	
LAS	02	.91	06	.05	.76	01	.07	.70	02	
MAP	.01	.97	05	02	.89	06	.00	.99	06	
CIMT	17	.20	.01	17	.23	01	18	.22	03	
CMRS	18	.28	04	17	.32	07	-	-	-	
<b>CMRS-Obs</b>	14	.38	05	16	.36	08	09	.56	.14	

Associations between standing time and cardiovascular risk factors in adolescent females.

 $\overline{BMI} = Body Mass Index; % BF= Percentage Body Fat; \Sigma SF = Sum of Skinfold Thickness; HDL-C = High Density Lipoprotein cholesterol; LDL-C = Low Density Lipoprotein cholesterol; T Chol. = Total Cholesterol; TG = Triglycerides; HMW Apo = High Molecular Weight Adiponectin; LAS = Large Artery Stiffness; MAP = Mean Arterial Pressure; CIMT = Carotid Intima-Media Thickness; CMRS = cardio-metabolic risk score; CMRS-Obs = cardio-metabolic risk score excluding the measure of adiposity. Model 1 adjusted for age and stage of pubertal development; Model 2 adjusted for age, stage of pubertal development and objectively measured MVPA; Model 3 adjusted for age, stage of pubertal development, objectively measured MVPA and adiposity (sum of four skinfolds).$ 

	Model 1			Model 2			Model 3		
	β	р	Adj. R <sup>2</sup>	β	р	Adj. R <sup>2</sup>	β	р	Adj. R <sup>2</sup>
BMI	24	.07	.04	26	.06	.03	-	-	-
% Body Fat	22	.10	.06	23	.10	.04	-	-	-
Σ Skinfolds	22	.10	.05	23	.10	.03	-	-	-
HDL-C	.10	.45	.01	.11	.42	01	.06	.69	.02
LDL-C	18	.19	02	17	.23	03	10	.47	.05
Total Chol.	11	.41	.01	10	.46	01	08	.57	02
Triglycerides	15	.26	01	16	.26	03	09	.53	.06
Insulin	26	.10	00	23	.11	02	14	.31	.13
Glucose	11	.39	.09	09	.53	.09	05	.70	.09
Adiponectin	.08	.56	04	.08	.60	06	.09	.57	08
LAS	01	.97	05	.05	.77	02	.07	.70	04
MAP	03	.83	05	05	.77	07	02	.91	07
CIMT	19	.16	.01	19	.18	01	21	.16	02
CMRS	19	.25	04	17	.32	06	-	-	-
CMRS-Obs	16	.34	05	15	.40	07	06	.68	.19

Associations between light intensity physical activity (including standing time) and cardiovascular risk factors in adolescent females.

BMI = Body Mass Index; % BF= Percentage Body Fat;  $\Sigma$  SF = Sum of Skinfold Thickness; HDL-C = High Density Lipoprotein cholesterol; LDL-C = Low Density Lipoprotein cholesterol; T Chol. = Total Cholesterol; TG = Triglycerides; HMW Apo = High Molecular Weight Adiponectin; LAS = Large Artery Stiffness; MAP = Mean Arterial Pressure; CIMT = Carotid Intima-Media Thickness; CMRS = cardio-metabolic risk score; CMRS-Obs = cardio-metabolic risk score excluding the measure of adiposity. Model 1 adjusted for age and stage of pubertal development; Model 2 adjusted for age, stage of pubertal development and objectively measured MVPA; Model 3 adjusted for age, stage of pubertal development, objectively measured MVPA and adiposity (sum of four skinfolds).

	Model 1				Model 2			Model 3		
	β	р	Adj. R <sup>2</sup>	β	р	Adj. R <sup>2</sup>	β	р	Adj. R <sup>2</sup>	
BMI	.16	.23	.02	.17	.23	.00	-	-	-	
% Body Fat	.09	.49	.03	.09	.54	.01	-	-	-	
$\Sigma$ Skinfolds	.13	.33	.02	.12	.39	.00	_	_	-	
HDL-C	.11	.41	.01	.11	.43	01	.15	.30	.04	
LDL-C	08	.57	04	09	.53	05	13	.36	.03	
Total Chol.	05	.74	.01	06	.70	01	07	.65	02	
Triglycerides	02	.86	03	03	.86	05	07	.63	.05	
Insulin	.07	.60	05	.08	.60	06	.03	.83	.09	
Glucose	.05	.70	.07	.01	.93	.07	01	.96	.08	
Adiponectin	19	.18	01	18	.23	03	18	.23	05	
LAS	.19	.28	03	.12	.52	01	.11	.54	02	
MAP	.09	.57	04	.14	.41	04	.13	.45	04	
CIMT	.22	.11	.02	.21	.14	.01	.22	.14	01	
CMRS	.11	.49	06	.09	.60	09	-	-	-	
<b>CMRS-Obs</b>	.09	.60	06	.09	.61	09	.06	.70	.15	

Associations between prolonged sedentary periods (sedentary periods lasting more than 30 minutes in duration) and cardiovascular risk factors in adolescent females.

BMI = Body Mass Index; % BF= Percentage Body Fat;  $\Sigma$  SF = Sum of Skinfold Thickness; HDL-C = High Density Lipoprotein cholesterol; LDL-C = Low Density Lipoprotein cholesterol; T Chol. = Total Cholesterol; TG = Triglycerides; HMW Apo = High Molecular Weight Adiponectin; LAS = Large Artery Stiffness; MAP = Mean Arterial Pressure; CIMT = Carotid Intima-Media Thickness; CMRS = cardio-metabolic risk score; CMRS-Obs = cardio-metabolic risk score excluding the measure of adiposity. Model 1 adjusted for age and stage of pubertal development; Model 2 adjusted for age, stage of pubertal development and objectively measured MVPA; Model 3 adjusted for age, stage of pubertal development, objectively measured MVPA and adiposity (sum of four skinfolds).